

PRE-ENTRY ACADEMIC AND NON-ACADEMIC FACTORS INFLUENCING TEACHER EDUCATION STUDENTS' FIRST-YEAR EXPERIENCE AND ACADEMIC PERFORMANCE

by

SUBETHRA PATHER

Thesis submitted in fulfilment of the requirements for the degree

Doctor of Education

in the Faculty of Education

at the Cape Peninsula University of Technology

Supervisor: Prof Rajendra Chetty

Mowbray Campus April 2015

CPUT copyright information

The thesis may not be published either in part (in scholarly, scientific or technical journals), or as a whole (as a monograph), unless permission has been obtained from the University.

DECLARATION

I, Subethra Pather, declare that the contents of this thesis represent my own unaided work, and that the thesis has not previously been submitted for academic examination towards any qualification. Furthermore, it represents my own opinions and not necessarily those of the Cape Peninsula University of Technology.

Signed

Date

ACKNOWLEDGEMENTS

What counts in life is not the mere fact that we have lived. It is what difference we have made to the lives of others that will determine the significance of the life we lead.

- Nelson Mandela

The successful accomplishment of this thesis would not have been possible without the support and guidance from a number of people. I should like to express my heartfelt gratitude to all who enriched this journey of intellectual pursuit. In particular I wish to thank:

- My family: my husband Shaun, for his patience, love and devotion; and my sons Hushav and Nishen for their understanding and endurance during these last five years.
- My parents, my late dad Bobby and mum Radha Padayachee: thank you for being my number one supporter. Your spiritual guidance and unconditional love are forever appreciated.
- My father-in-law, Dr Juggie Pather: thank you for your assistance and constant encouragement.
- To family and friends: thank you for your ongoing support and always keeping my sprits high. My brothers Shalen and Samish, my sister-in-laws: Sagree, Preba, Romy and Mayendree, and all my wonderful cousins and dear friends.
- Colleagues at CPUT and other institutions: thank you for sharing your academic thoughts with me and keeping me focused. A special thank you to my friend and colleague, Prof. Niri Dorasamy; I appreciate the late night calls and the constant support.
- Participants in my research: I should like to acknowledge all the participants in this research project. Your valuable input and time are greatly appreciated.
- Most importantly I am indebted to my supervisor, Prof Rajendra Chetty, without whose expertise, guidance and mentorship I would not been able to achieve such success. Thank you for sharing your wisdom with me.

I also wish to acknowledge the support from the following organisations:

- The National Research Foundation (NRF) of South Africa. The completion of this study would not have been achievable within the required time frame without receiving the NRF Sabbatical Grant. Opinions expressed in this thesis and the conclusions arrived at, are those of the author, and are not necessarily to be attributed to the NRF.
- The Centre for Critical Research on Race and Identity (CCRRI) at UKZN: I am grateful for being awarded the DHET Doctoral Scholarship from CCRRI and participating in a collaborative project titled: Education and Emancipation: A Critical, Intervention-Oriented Investigation of Obstacles and Opportunities Within the Higher Education and Training Sector in South Africa.

The financial assistance of the NRF and CCRRI's DHET Doctoral Scholarship award towards this research is acknowledged.

DEDICATION

For my parents: Amma and Papa

Matha, Pitha, Guru, Deivam (Mother, Father, Teacher & then God)

ABSTRACT

The research question that guided this doctoral study is: How do pre-entry academic and non-academic factors influence teacher education students' first-year experience and academic performance? The study was designed within the qualitative research paradigm and employed a case study strategy to collect both quantitative and qualitative data. The quantitative approach included a questionnaire that was completed by 195 respondents. The qualitative data was obtained from one-on-one and focus-group interviews with eight participants that were purposively selected. The conceptual framework developed for this enquiry took into consideration the significance of student diversity in understanding first-year experience and thus employed concepts from two sociological models, Tinto's (1975; 1993) integration model (social and academic integration) and Bourdieu's (1984; 1990) theoretical tools of capital, habitus and field. Six key themes emerged from the data: determination, self-reliance, fitting-in, out-of-habitus experience, positioning oneself to succeed and challenges.

The unequal distribution of economic, social and cultural capital created disparities between students' habitus and schooling experiences which influenced the way they integrated into their first year at university. The study revealed that more mature students than school-leavers and gap-students are entering higher education. Further, the majority of first-year students are unable to fund their studies and source external funding or engage in part-time employment. Students pursued financial aid before focusing on academic activities. Engagement in the social domain remained marginal. Students' determination to change their economic circumstances was the primary factor that influenced their attitudes and actions at university.

Higher education needs to consider student diversity, financial constraints of disadvantaged students, first-year curriculum planning and delivery, and the high cost of university studies. It needs to move away from viewing entering students from a deficit model, to capitalise on their qualities of determination, optimism, enthusiasm and openness to learning, thereby creating an inclusive first-year experience that could encourage retention and student success.

TABLE OF CONTENTS

	DEC	LARATION	i
	ACK	NOWLEDGEMENTS	ii
	DED	DICATION	iv
		TRACT	
		LE OF CONTENTS	
		of Tables	
		of Figures SSARY: ABBREVIATIONS AND ACRONYMS	
	GLU	SSART. ADDREVIATIONS AND ACRONTINS	XIII
CHAP	TER 1		1
1.1	Сна	LLENGES FACING HIGHER EDUCATION IN SOUTH AFRICA	1
1.2	BAC	KGROUND TO RESEARCH PROBLEM	3
	1.2.1	Changing HE landscape: from elite to mass-based system	3
	1.2.2	Student attrition and persistence	5
	1.2.3	Dealing with the problem: focus on first-year students	5
1.3	Brid	DGING THE GAP IN THE LITERATURE?	6
1.4	STA	TEMENT OF PROBLEM	8
1.5	Тне	RESEARCH CONTEXT: TEACHER EDUCATION IN SOUTH AFRICA	9
	1.5.1	Current teacher profile in South Africa	9
	1.5.2	High teacher attrition rate: a national concern	
	1.5.3	Teacher training in South Africa	
	1.5.4	Teacher education students	14
1.6	Pur	POSE OF STUDY	15
	1.6.1	Research questions	
	1.6.2	Research aims and objectives	
1.7	Res	EARCH DESIGN AND METHODOLOGY	16
1.8	CLA	RIFICATION OF BASIC CONCEPTS AND TERMS	19
1.9	ORG	GANISATION OF THE STUDY	21
1.1	0 C	CONCLUSION	22
СНАР	TFR 2	2 CONCEPTUAL FRAMEWORK	23
2.1		RODUCTION	
2.2		NIFICANCE OF A CONCEPTUAL FRAMEWORK	
2.3	Τιντ	O'S INTEGRATION MODEL	
_	2.3.1	Empirical studies on Tinto's student integration model	
	2.3.2	Critique of Tinto's model	
2.4	Βου	IRDIEU'S THEORETICAL TOOLS	
2	2.4.2	Empirical studies using Bourdieu's theoretical tools	
2	2.4.3	Critique of Bourdieu's conceptual tools	45
2.5	Acc	DNCEIVED CONCEPTUAL FRAMEWORK FOR THIS STUDY	47

2.6	CON	CLUSION	52
СНА	PTER 3	LITERATURE REVIEW	54
3.1	INTR	ODUCTION	54
3.2	Wha	T CONSTITUTES THE FIRST-YEAR EXPERIENCE?	54
3.3	SIGN	IIFICANCE OF THE FIRST-YEAR EXPERIENCE	55
3.4	Емр	IRICAL STUDIES ON THE FIRST-YEAR EXPERIENCE	57
	3.4.1	First-year retention	
	3.4.2	First-year transition	
	3.4.3	First-year student preparedness	
3.5		ENING ACCESS AND DIVERSITY	66
3.6	FAC1	FORS INFLUENCING THE FIRST-YEAR EXPERIENCE	70
	3.6.1	Academic factors	71
	3.6.2	Non-academic factors	73
	3.6.3	Institutional factors	
3.7	CON	CLUSION	77
СНАР	PTER 4	RESEARCH METHODOLOGY	79
4.1	INTR	ODUCTION	79
4.2	Ουτι	LINE OF THIS CHAPTER	
4.3	B PHIL	OSOPHICAL PERSPECTIVES	81
	4.3.1	Constructivist/interpretivist viewpoint	
	4.3.2	Pragmatic viewpoint	
4.4	RES	EARCH STRATEGY OF ENQUIRY: CASE STUDY	85
4.5	RES	EARCH DESIGN	
	4.5.1	Selection of research participants	
	4.5.2	Ethical considerations	
	4.5.3	Data collection methods	
	4.5.4	Mailed questionnaire	
	4.5.5	One-on-one interviews	
	4.5.6	Focus-group interview	
	4.5.7	Analysis of evidence	
	4.5.8	Quantitative data analysis	
	4.5.9	Qualitative data analysis	
	4.5.10	Trustworthiness of the study	
4.6	CON	CLUDING REMARKS	124
СНА	PTER 5	QUESTIONNAIRE FINDINGS	125
5.1	INTR	ODUCTION	
5.2	PRES	SENTATION OF DATA	
5.3	DIVE	RSITY OF FIRST-YEAR STUDENTS IN THE BED PROGRAMME	

	5.3.1	Demographic information	127
	5.3.2	Socio-economic factors	133
	5.3.3	Pre-entry academic factors	139
	5.3.4	Students' pathway to university	145
5.	4 Stu	DENTS' EXPECTATIONS OF THEIR FIRST-YEAR EXPERIENCE	147
	5.4.1	Expectations of academic integration	147
	5.4.2	Expectations of social integration	152
	5.4.3	Students' perceptions	157
5.	5 SUM	MARY OF FINDINGS	160
	5.5.1	Who are the first-year students entering teacher education programmes?	160
	5.5.2	What are first-year students' expectations of university life?	163
5.	6 Con	CLUDING REMARKS	165
СНА	PTER 6	QUALITATIVE FINDINGS	166
6.	1 Intr	ODUCTION	166
6.	2 Pres	SENTATION OF QUALITATIVE DATA	167
6.	3 Des	CRIPTION OF PARTICIPANTS	167
	6.3.1	Joyce	168
	6.3.2	Thabo	170
	6.3.3	Letti	173
	6.3.4	Nadia	175
	6.3.5	Alley	178
	6.3.6	Katlyn	180
	6.3.7	Lisa	183
	6.3.8	Tasha	186
	6.3.9	Synopsis of participants	187
6.	4 SOR	T-CARD ACTIVITY: FACTORS THAT POSITIVELY CONTRIBUTED TO FIRST-YEAR EXPERIENCE	188
	6.4.1	Synopsis of sort-card activity	195
6.	5 Thei	MES AND SUB-THEMES THAT EMERGED FROM THE INTERVIEWS	196
	6.5.1	Main theme 1: Determination	197
	6.5.2	Main theme 2: Resilience	201
	6.5.3	Main theme 3: Fitting in	205
	6.5.4	Main theme 4: out-of-habitus experience	209
	6.5.5	Main theme 5: Positioning oneself to succeed	214
	6.5.6	Main theme 6: Challenges	218
6.	6 CON	CLUSION	224
СНА	PTER 7	DISCUSSION OF FINDINGS	226
7.	1 Intr	ODUCTION	226
7.	2 Pre-	ENTRY ACADEMIC AND NON-ACADEMIC FACTORS INFLUENCING FIRST-YEAR EXPERIENCE A	٧D
AC		PERFORMANCE	228

7	.2.1	Diversity in student population	.228
7	.2.2	Prior school experience	.232
7	.2.3	Economic status	.234
7	.2.4	Family support	.236
7	.2.5	Life circumstances	.238
7.3	Adju	ISTING TO FIRST-YEAR EXPERIENCE AND ACADEMIC PERFORMANCE	.240
7	.3.1	Academic integration	.241
7	.3.2	Social integration	.244
7	.3.3	A sense of belonging	.246
7.4	CON	CLUDING REMARKS	.248
7	.4.1	Pre-entry academic and non-academic factors influencing first-year experience and	
а	caden	nic performance	.248
7	.4.2	Adjusting to the first-year experience and academic performance	.250
СНАР	TER 8	CONCLUDING REMARKS	252
8.1		ODUCTION	
8.2		NOPSIS OF THE RESEARCH JOURNEY	
8.3	Rese	EARCH OUTCOMES	
8	.3.1	Theoretical contribution	
8	.3.2	Contribution to practice	.257
8.4	CRIT	IQUE OF THE RESEARCH AND FUTURE RESEARCH OPPORTUNITIES	
8	.4.1	Future research	.260
8.5	A pe	RSONAL REFLECTION OF WHAT THIS JOURNEY MEANT TO ME AS THE RESEARCHER	.260
8.6	CON	CLUSION	.261
	Refe	rences	263
	APPI	ENDIX A: Pre-entry questionnaire	290
	APPI	ENDIX B: Interview Schedule	301
	APPI	ENDIX C: Consent form	303
	APPENDIX D: Ethical clearance		304
	APPI	ENDIX E: Sort Cards	307
	APPI	ENDIX F: Focus-group interview	309
	APPI	ENDIX G: Job category grouping	313

List of Tables

TABLE 4:1 PARADIGMS, METHODS AND TOOLS	82
TABLE 4:2 FIRST YEAR ENROLMENT FIGURES FOR BED FOUNDATION PHASE STUDENTS: 2008–2012	92
TABLE 4:3 PROFILES OF PARTICIPANTS	94
TABLE 4:4 OVERALL RELIABILITY OF INSTRUMENT	118
TABLE 4:5 STRATEGIES USED TO ENHANCE THE QUALITY OF THE RESEARCH OUTCOMES OF THIS STUDY	124
TABLE 5:1 AGE AND GENDER DISTRIBUTION OF FIRST-YEAR BED STUDENT RESPONDENTS	129
TABLE 5:2 HOME LANGUAGE OF RESPONDENTS IN THE STUDY	132
TABLE 5:3 PARENTS' ACADEMIC QUALIFICATIONS	133
TABLE 5:4 PARENTS' JOB CLASSIFICATION OF RESPONDENTS IN THE STUDY	135
TABLE 5:5 NATURE OF SUPPORT FROM FAMILY	136
TABLE 5:6 FIRST-GENERATION UNIVERSITY STUDENTS	145
TABLE 5:7 YEAR PRIOR TO ENTERING UNIVERSITY	146
TABLE 5:8 ACADEMIC PREPAREDNESS	147
TABLE 5:9 ACADEMIC EXPERIENCES	149
TABLE 5:10 ACADEMIC SUPPORT	151
TABLE 5:11 ACADEMIC COLLABORATION	152
TABLE 5:12 EXPECTATION ON SOCIALISING	154
TABLE 5:13 OUT-OF-CLASS DISCUSSIONS	155
TABLE 5:14 SELF-PERCEIVED EXPECTATIONS OF SOCIAL INTEGRATION	157
TABLE 5:15 PERCEPTIONS OF LECTURER EXPECTATIONS	158
TABLE 5:16 EXPECTATIONS OF LECTURERS	160
TABLE 6:1 SYNOPSIS OF PARTICIPANTS WITH EMERGENT THEMES AND SUB-THEMES	188
TABLE 6:2 PARTICIPANTS RANKING OF FACTORS INFLUENCING FIRST-YEAR EXPERIENCE – SORT-CARD	
ACTIVITY	190
TABLE 6:3 IDENTIFYING THE TOP THREE FACTORS INFLUENCING FIRST-YEAR EXPERIENCE FROM THE SOR	.т-
CARD ACTIVITY	193
TABLE 6:4 FACTOR RANKED AS THE NO. 1 SORT CARD BY PARTICIPANTS	195
TABLE 6:5 KEY THEMES AND SUB-THEMES	197
TABLE 7:1 STRUCTURE FOR DISCUSSION OF FINDINGS	227

List of Figures

FIGURE 1:1 CHAPTER 1 – OVERVIEW	2
FIGURE 2:1 CONCEPTUAL MAP OF CHAPTER 2	
FIGURE 2:2 A CONCEPTUAL FRAMEWORK TO EXPLORE INFLUENCES ON FIRST-YEAR STUDENTS' EX	PERIENCE
AND ACADEMIC PERFORMANCE	50
FIGURE 4:1 RESEARCH DESIGN	
FIGURE 4:2 VISUAL MODEL OF THIS STUDY'S SEQUENTIAL COLLECTION OF MIXED DATA	
FIGURE 4:3 COMPONENTS OF DATA ANALYSIS AS APPLIED TO THIS STUDY	
FIGURE 5:1 AGE AND GENDER DISTRIBUTION OF RESPONDENTS	
FIGURE 5:2 RACIAL COMPOSITION OF THE RESPONDENTS IN THE STUDY	131
FIGURE 5:3 FINANCIAL SUPPORT FOR FIRST-YEAR UNIVERSITY FEES	
FIGURE 5:4 MODE OF TRANSPORT OF RESPONDENTS IN THE STUDY	
FIGURE 5:5 TYPE OF SCHOOL ATTENDED BY RESPONDENTS IN THE STUDY	
FIGURE 5:6 PERIOD IN WHICH RESPONDENTS COMPLETED GRADE 12	
FIGURE 5:7 GRADE 12 RESULTS FOR ENGLISH HOME LANGUAGE AND FIRST ADDITIONAL LANGUA	GE 142
FIGURE 5:8 GRADE 12 MATHEMATICS RESULTS	
FIGURE 5:9 GRADE 12 MATHEMATICS LITERACY RESULTS	
FIGURE 5:10 EXPECTATIONS OF FIRST-YEAR ACADEMIC EXPERIENCES	
FIGURE 5:11 EXPECTATIONS OF FIRST-YEAR ACADEMIC SUPPORT	
FIGURE 5:12 EXPECTATIONS OF FIRST-YEAR ACADEMIC COLLABORATION	
FIGURE 5:13 FIRST-YEAR STUDENTS' EXPECTATIONS OF SOCIALISING	
FIGURE 5:14 EXPECTATIONS OF OUT-OF-CLASS ACADEMIC DISCUSSIONS	
FIGURE 5:15 SELF-PERCEIVED EXPECTATIONS OF SOCIAL INTEGRATION	
FIGURE 5:16 FIRST-YEAR STUDENTS' PERCEPTIONS OF LECTURER EXPECTATIONS	
FIGURE 5:17 FIRST-YEAR STUDENTS' SELF-PERCEIVED EXPECTATIONS OF LECTURERS	

The legacy of apartheid education places an enormous responsibility on universities. The sins of our fathers visit us in the underprepared generation of learners we have to guide and shape to be the creative and productive future of our country. The challenge is to be caring, responsible and innovative; and to be truly South African in our practice and approach, being neither patronising nor claiming of entitlement.

President Nelson Mandela (1998),

(Address by President Nelson Mandela on the installation of Professor Bundy as Vice-Chancellor and Principal of the University of the Witwatersrand)

GLOSSARY: ABBREVIATIONS AND ACRONYMS

ACT American College Test CHE **Council on Higher Education** CPUT Cape Peninsula University of Technology DBE Department of Basic Education DHET Department of Higher Education & Training DoE Department of Education ECD Early Childhood Development FET Further Education and Training FP **Foundation Phase** FYE **First-Year Experience** GET General Education and Training HEI Higher Education Institution ISP Intermediate and Senior Phase NAO National Audit Office NBT National Benchmark Test NPC National Planning Commission NSC National Senior Certificate OECD Organisation for Economic Co-operation and Development SES Socio-Economic Status STD Self-Determination Theory TE **Teacher Education**

CHAPTER 1 INTRODUCTION

1.1 Challenges facing higher education in South Africa

Since the advent of democracy, the student population at South African universities has become racially and culturally diverse. In accordance with the rich diversity of a mixed student population, students' needs and expectations have become increasingly heterogeneous (Nelson et al., 2011). The need for Higher Education Institutions (HEIs) to appreciate, and adjust to, such diversity is particularly urgent among first-year students. The first-year experience has been highlighted in recent research as a critical factor with regard to persistence, academic performance, retention, assessment and success (Harvey et al., 2006). McInnis (2001) identified the first year as a year in which the highest degree of academic failure and attrition occurs. Students' first-year experience and perceptions have a major impact on their study options and participation in higher education. The first-year experience may be a decisive factor in the success or failure of students' university studies (Kantanis, 2000; Hillman, 2005). McInnis (2001), together with Nelson et al. (2011) demonstrate that the significance of religious, racial, economic and cultural diversity on students' first-year experience has been underestimated: further research is essential on how diversity can add value to the first-year experience of all students and consequently the overall pass rate if the relevant tertiary institution understands, appreciates and makes adjustments for such diversity. Consequently, many HEIs have begun to shift their attention to analysing who their first-year students are and the levels and scope of their first-year experience. Nelson et al. (2011) note that this trend has, to date, been depicted loosely and imperfectly as 'student-centredness'.

Given the foregoing, this study aims to investigate South African higher education students' first-year experience and academic performance. The selected problem parameter for the study is a cohort of first-year teacher education students in a university of technology. The selected setting is used as a vehicle to explore certain key aspects of students' first-year experience: students' pre-entry academic and non-academic backgrounds and the influence of such factors on their social and academic integration in their initial year of study. McKenzie and Schweitzer (2001) identified social and academic integration as strong predictors of university life satisfaction. For a first-year student to be successfully integrated into university life,

integration must take place in both the social and academic domains (Tinto, 1975; Hillman, 2005). First-year students' level of integration can be influenced by internal and external factors. Tinto (1975, 1993) considers students' initial commitment to the university and their goal to graduate as directly influential in their level of social and academic integration which could determine whether they decide to stay and persist or drop out of university.

The structure of this chapter is as follows: background to the research problem, neglected areas in the literature with regard to this study, teacher education in South Africa as the context for the study, the research questions, aims and objectives of the study, research design and methodology, definition of concepts used, and finally a conclusion. Figure 1:1 below provides a graphical overview of Chapter 1.



Figure 1:1 Chapter 1 – overview

1.2 Background to research problem

Since 1994, South Africa's new democratic government has committed itself to transforming higher education and its inherited apartheid social and economic structure (Badat, 2010). The South African Constitution of 1996 (South Africa, 1996) and the White Paper of 1997 contain the primary policies which underpin the transformation of higher education. The underlying objective of the policies was to eradicate all forms of unfair discrimination and advance the process of redress of past inequalities (DoE, 1997). One of the key imperatives of transformation was that student enrolment should be expanded and access broadened to reach a wider distribution of social groups and classes (DoE, 1997).

1.2.1 Changing HE landscape: from elite to mass-based system

The Ministry of Education had a specific role to play in advancing transformation goals: the creation of a transformed, democratic, non-racial and non-sexist system of higher education (DoE, 1997). Two specific outcomes that the Ministry of Education sought to achieve were:

- increasing and broadening participation to reach a wider distribution of social groups and classes, including adult learners, thereby accommodating a larger and more diverse student population (DoE, 1997:1.13); and
- the promotion of equity of access and fair chance of success to all (DoE, 1997:1.14).

During the era of white nationalist government control of education, the University of South Africa (Unisa) used distance learning to reach students from all racial groups in South Africa. This was the only institution, in the pre-apartheid era, which provided access to a non-racial mass education at a tertiary level. At traditional universities, however, there was racial segregation, which still has an effect on academic staff and students. Freedom in 1994 brought with it a drastic shift in focus in South African higher education at traditional universities which are the focus of this present discussion. The pool of students entering higher education expanded and became richly diverse (Jarvis, et al., 2004; McKenzie & Schweitzer, 2001). Student numbers drastically increased, doubling enrolment figures from 473 000 in 1993 to 893 024 in

2010 (DHET, 2013). Female students outnumbered male students by an increasing margin of 55,5% (CHE, 2009). The new enrolment trends also showed a different racial imbalance with more students from historically under-represented groups entering higher education and resulting in at least 50% representation of African¹ students in traditional universities (CHE, 2009).

These changes in the country's student demographics indicated a shift in higher education institutions from being elite and racist to inclusive and democratic. The elite system of the past favoured and cossetted a single race group of students who generally possessed the fundamental skills necessary for higher education (Leathwood & O'Connell, 2003; Trow, 2006). The change to mass education has successfully produced a student body that is more representative of the South African population, one which reflects diversity in language, socio-economic and academic backgrounds, culture, religion and race (Asmar, 2004). However, the broadening of university access brought with it a host of new challenges for South Africa's HEIs. Foremost was resistance to such change from traditionally white elitist tertiary institutions. Little attention has been paid to the difficulties faced by underprivileged students from underachieving schools, economically depressed environments and homes offering minimal learning support. Many traditional universities have been slow to adapt their staff component to reflect the national profile. Moreover, such institutions have not been prompt to make substantial adjustments or sacrifices to ensure their students' social or academic wellbeing. Such tardiness prevents the management at universities from acknowledging the potential advantages and benefits that may be drawn from a diverse and socially representative student body whose democratic and open interaction is possibly the most reliable guarantor of understanding and future equity in the country as a whole.

It seems fatuous to point out that such problems are not unique to South Africa. Many international countries faced with student increases at HEIs were challenged with addressing problems that included under-preparedness of first-year students (Engstrom, 2008; Kift et al., 2010), concerns with retention and persistence rates

¹ This study makes use of the four population groupings used by Statistics South Africa: African, coloured, Indian and white. Black refers to the African, Coloured and Indian population.

(Pascarella & Terenzini, 2005; Tinto 2007) and the need to institute first-year support programmes (Morsi et al., 2007; Tinto, 2007).

1.2.2 Student attrition and persistence

Within the context of massification, a number of problems presented themselves, especially with regard to student attrition and persistence. The National Plan for Higher Education expressed concern that South Africa's HE throughput rates were too low and that the graduation rate of less than 22% for a three-year generic bachelor's degree was one of the lowest in the world (DoE, 2005). Interestingly, this is a problem faced internationally as well, but at a lower level. For example, in the United States of America an average four-year university degree suffers the loss of 29% of its new students in the first year of study (ACT, 2004) while South Africa loses 30% of its students in the first year and an additional 20% in the second and third years of study (DoE, 2005). Tinto (1993) identified the first-year student cohort as the group at greatest risk of attrition from university. In Schneider's (2010) study on first-year student attrition the following areas of concern were highlighted: prior academic performance, academic and social integration, and academic and psychological readiness. According to Terenzini et al. (2001) and Terezini and Reason (2005), attrition rates are even greater among low-income, historically under-represented students and first-generation students.

1.2.3 Dealing with the problem: focus on first-year students

The Stakeholder Summit on Higher Education Transformation in South Africa acknowledged the need to maximise students' chances of success (DHET, 2010). The summit highlighted the following areas that needed attention: importance of, and challenges around, understanding students' first-year academic and social experience; the influence of the First-Year Experience (FYE) to high dropout and low throughput rates; and finally the importance of providing first-year student support for academic success (DHET, 2010). The focus on first-year intervention has been identified in other national and international studies. These studies all emphasise the importance of the first year of study with regard to providing foundation knowledge; frontloading academic support; and enhancing academic success and persistence. For example, Reason et al. (2006) identified the first year as the most critical year for

laying the foundation on which students' subsequent academic success and persistence rest. Other studies on the FYE reveal the first year of study to be the most significant phase of the student experience in relation to academic success and persistence (Tinto, 1995; McInnis, 2003; Yorke & Longden, 2008). According to Krause (2006), exploration of students' first-year experience assists in determining student attitudes and approaches to learning and, as a result, makes it possible to improve retention and student success.

1.3 Bridging the gap in the literature?

This study bridges a gap in the literature by contributing empirical research to the first-year experience field by means of examining what, how and why certain preentry academic and non-academic factors influence students' first-year experience and academic performance. The understanding of this phenomenon was pivotal to the increase in the diversity of the higher education student body that has added complexity to the nature of students' first-year experience. Traditionally, much of the research and literature on higher education student experience and retention has overlooked the influence of students' backgrounds and characteristics (Reay, 2012). Tinto, who for the last 35 years has been acknowledged as the leading expert on student retention, makes reference to students' backgrounds and characteristics in passing and places greater emphasis on students' social and academic integration within the campus climate (Tinto, 2007). However many of the studies on student retention consider the students' pre-entry characteristics, such as socio-economic status, academic experience or age, as control variables rather than considering each of these key factors 'investigated as variables whose effects are important to understand' (Walpole, 2007:8). In addition, Wilcox et al. (2005) note that many studies on the first-year experience that employ Tinto's concept of social and academic integration rarely discuss the concept of integration in detail. These studies lack analysis on how and why such social and academic integration takes place. Harvey et al. (2006) concur with this formulation and acknowledge that although there has been a large amount of data collected on students' first-year experience at the institutional level, a relatively insignificant amount has been reported with a view to explicitly exploring students' personal experiences in their first year of study.

In South Africa there has been limited research on university students' first-year experience. The majority of the literature reviewed for this study focused on research questions related to the academic and social issues within the university environment, such as student learning, academic support, social networks, student retention, and dropout rates. Having reviewed international studies, for example, Dumais and Ward (2010), Lane and Taber (2012) and Gaddis (2013), I noticed a shift in focus toward the role played by social class, capital and habitus vis-à-vis students' educational experiences and academic success. I realised that the issues external to the university environment cannot be ignored when investigating the firstyear experience. I recognised this area as a gap in the literature concerning the firstyear university experience. There need to be more studies in South Africa that investigate students' first-year university experience by considering the influences of factors outside of the university environment, such as students' habitus, different forms of capital resources that students bring with them into the university environment, and their prior schooling experiences. Given the context in which higher education institutions in South Africa have rapidly increased in student complexity, I believe that there is a growing urgency for a critical exploration and understanding of students' higher education experiences through the voice and individual experiences of first-year students.

This study thus aimed to bridge this gap in the literature by identifying and exploring the influences of pre-entry academic and non-academic factors on students' first-year experience and academic performance. The focus of this study is on investigating and identifying how certain pre-entry factors influence students' social and academic integration in their first year of study. By employing the pragmatic viewpoint together with concepts from Bourdieu's (1984) and Tinto's (1975, 1993) theories, I sought to bridge this gap in the literature. Tinto's (1975, 1993) model on integration and Bourdieu's (1984) conceptual tools – capital, habitus and field – allowed this study to provide an in-depth understanding of the complexities of this phenomenon. This study makes use of a conceptual framework that integrates the concepts from these two theorists to consider the significance of pre-entry factors such as prior schooling experience, religious, racial, economic and cultural diversity in influencing the way students negotiate their first-year university experiences.

Tinto's model of integration provided the scope to be descriptive and provide explanations at a practical level. Bourdieu's concepts of capital, habitus and field were used as both conceptual and analytical tools. Finally, the justification for employing concepts from Bourdieu's (1984) and Tinto's (1975, 1993) theories in the conceptual framework model was to address the lacunae in the literature on the first-year experience. The use of concepts from these two theorists provided an explanation of how the social, academic and cultural environment in which students were raised shaped their attitudes and interpretation of their first-year experience, as well as their capacity to engage with academic and social integration in their first year of study (Bourdieu & Passeron, 1977).

1.4 Statement of problem

The high attrition rate at undergraduate level, and more specifically at first-year level, has raised concerns about the low throughput rates, graduation rates, and the quality of student experience within the context of expanding student numbers and resource constraints (Wilcox et al., 2005; CHE, 2013). These concerns have prompted an urgent need to profile more accurately the students entering higher education so that appropriate measures could be implemented to ensure a quality university experience which could result in improved throughput and graduation outputs. Presently, South African's higher education student population is very diverse. Students entering higher education come from differing social, economic and cultural backgrounds, life experiences, and levels of education which influence their university expectations, needs and academic potential. Limited empirical research on these issues within the South African context has resulted in a challenge, specifically when trying to address issues on low throughput and graduation rates in higher education.

If South Africa is to offer quality education, it is imperative that HEIs understand who their students are, where they come from, and the qualities and attributes these students bring with them to higher education. In addition, HEIs need to move away from viewing today's entering students from a deficit model and try to capitalise on their qualities of determination, resilience, enthusiasm and openness to learning.

1.5 The research context: teacher education in South Africa

One of South Africa's greatest challenges is providing quality education to sustain the country's human resources. In this regard teacher education plays a pivotal role as Wolhuter (2006:124) states: 'Any education system stands or falls by the guality of its teaching profession, and therefore, by implication, the quality of its teacher education programmes.' Good quality teachers imply the need for good quality teaching programmes. Owing to the poor literacy and numeracy levels of school students in South Africa judged against international standards and the low percentage of Grade 12 students qualifying with a university entrance certificate, the country's schooling system is facing several serious quality problems at the tertiary level (Wolhuter et al., 2012). This is a cause for concern because education is viewed as key to effecting an economic, social, moral, political and value reconstruction of society (Wolhuter, 2010:6-7). The Council on Higher Education (CHE, 2013) adds that South Africa's social stability greatly depends on the establishment of an educated, trained and employable workforce. In addition to the country's national concerns about the quality of school education, contributory fears relate to the high teacher attrition rate and the imminent shortage of teachers in the country.

1.5.1 Current teacher profile in South Africa

Latest statistics on teachers in South Africa (DBE, 2012) indicate that there are approximately 440 000 teachers employed. The *Integrated Strategic Planning Framework for Teacher Education and Development in South Africa: 2011-2025* reveals that although 89% of these teachers have a professional teaching qualification, only 18% of them are graduates, that is, have a four-year degree or equivalent (DBE & DHET, 2011). In addition to the worrying statistics on teacher qualifications, the age and experience of South African teachers indicate a concern for future supply and demand in the teaching profession. A large percentage (42%) of teachers is aged between 45 and over, while only 5% are under the age of 30 (DoE, 2009). This implies that not many young students are choosing teacher education. Teacher experience statistics reveal that many newly qualified teachers are not entering the profession in South Africa. For example, in 2004 there was a total of 36% of teachers with fewer than five years' experience compared with 9% in

2009. The loss of young teachers will certainly create a drain on the vitality of the teaching force (Shisana et al., 2005).

The demographics of teachers in South Africa in some significant ways represent the demographics of the country. The Department of Education (DoE, 2009) reveals that the percentage of teachers according to the four racial groupings in South Africa is as follows: 79% of the teachers are African, 10% white, 9% coloured and 1% Indian. Close to two-thirds (67%) of the teachers in South Africa are women.

1.5.2 High teacher attrition rate: a national concern

Presently the teacher attrition rate in the country has stabilised around 5%–6% per annum (DoE, 2007:7; DoE, 2009:22). According to Crouch and Perry (2003), the high attrition rate peaked with the redeployment and rationalisation of teaching around 1997/1998. Teachers were found to be in excess at their schools in accordance with the teacher–pupil ratio set by the South African government: this led to many teachers being redeployed to areas of teacher shortage. This caused many teachers either to resign, opt for early severance packages, or be dismissed. In addition to the redeployment of the teachers at schools, the closure of the colleges of education that were spread across the country, including some in remote rural areas, resulted in a limited number of students gaining access to teacher education providers were concentrated mainly in the richest provinces, making accessibility a major problem (Welch & Gultig, 2002).

Rationalisation and redeployment escalated teacher attrition and contributed to a shortage of teachers in the country. Welch and Gultig (2002:17) highlight the following consequences on teacher education:

- The teaching profession was perceived by the public to have limited career prospects.
- It created the impression that South Africa had too many teachers.
- Many newly qualified teachers could not find employment.
- There was a decline in the prestige, predictability and attractiveness of teaching as a profession.

In Gordon's (2009) study it was noted that a high rate of teachers leaving the profession was experienced close to the end of their careers, for example, through early retirement, medical incapacity, or death. The effect of HIV-related illnesses increased attrition across all phases in teachers' careers. By contrast, the Organisation for Economic Co-operation and Development (OECD, 2008) notes that the highest rate of attrition actually occurs early in the teacher's career owing to job changes, further education and childbirth. Other factors of attrition were ascribed to the school environment: overcrowding of schools and classrooms; inadequate incentives and job rewards; poor parent participation; policy overload; conflicting role demands; lack of administrative and collegial support; insufficient involvement in decision making; and favouritism and nepotism (OECD, 2008; Pitsoe, 2013).

Currently, a major concern is South Africa's teacher attrition rate, which is much higher than the replenishing rate. According to Steyn (2006), if the country's attrition rate of teachers remains at 5% and teacher training institutions experience an attrition rate of 10%, which is not unreasonable, this implies that there should be approximately 84 900 students in training which will break even with the attrition rate. Steyn (2006) highlights that teacher education institutions in South Africa have an enrolment figure of around 25 600. This low replenishing rate is supported by Gordon (2009), who warns that South Africa would need between 20 000 and 30 000 new teachers for the next decade. Gordon (2009) voices a concern that the rate of graduates from teacher education institutions in South Africa is far lower than required in the profession. Nevertheless, South Africa is not alone with regard to the teacher shortage crisis. Pitsoe (2013) indicates that this trend seems to be global: two-thirds of the world's countries experience severe teaching shortages. Pitsoe quotes Unesco's 2009 statistics that divulge a global total of 10.3 million teachers needs to be recruited between 2007 and 2015 to address this shortfall.

In Pitsoe's (2013) article on teacher attrition in South Africa, he critically analyses the country's teacher retention policy and proposes that by rethinking attrition and retention policies through the lens of the incentive theory of motivation, teacher attrition can be reduced and its impact on the country avoided. Pitsoe (2013:317): identifies four most promising options that can assist South Africa to reduce teacher attrition through motivation:

- Higher salaries: An across-the-board salary increase for all teachers will make the teaching profession more attractive.
- Differentiated salaries: Additional compensation should be provided to teachers teaching in designated shortage areas, for example, in specific subject areas, certain geographical regions such as rural areas, or within student populations: low-achieving or high poverty. This gesture would improve the attractiveness of particular teaching positions to alleviate chronic shortages.
- Smaller class sizes: Reduction in class sizes as an alternative would be favoured by many teachers especially in overcrowded schools. This would improve working conditions and make teaching more attractive.
- Mentoring: Mentoring programmes provide support, advice and encouragement to new and inexperienced teachers which would make their employment more attractive. This would improve working conditions of teachers.

High teacher attrition rates escalate the demand for more teachers in South Africa. To add to this dilemma of urgent teacher demand is the immediate recruitment and migration of South African teachers to other countries such as the United Kingdom, Australia, New Zealand, Dubai and countries in Asia. In a news article captioned 'Teacher exodus' (Samodien, 2008), she contends that the poaching of South African teachers and principals by overseas institutions has become a worrying factor for the Department of Education. In Samodien's interview with a teacher migration specialist, he estimated that between 25% and 30% of all graduating teachers leave annually to teach abroad. The migration specialist added that between July 1997 and July 2006, more than 10 000 teachers from South Africa had migrated to the UK, a number much higher than in any other country. Although it is not clear as to how many South Africans are presently teaching abroad or how long they remain abroad (Bertram et al., 2006) it is still a concern that needs to be addressed.

The second pressing concern with regard to the dilemma of high teacher demand is highlighted by the increasing population of those under the age of 18 in South Africa. Statistics South Africa's 2011 census reveals that the country's total population is

estimated at 50 million people, of whom 18.5 million are children under the age of 18. School-going children therefore constitute 37% of the total population, a rapid growth by about 6% (one million) over the 10-year period from 2002 to 2011 (Statistics South Africa, 2012). For the country to provide good quality education to the increasing school population, the teacher supply–demand crisis needs to be addressed. Thus the focus of attention has now turned to teacher training institutions as a way of helping to alleviate the problem.

1.5.3 Teacher training in South Africa

Under apartheid, higher education in South Africa was skewed in ways designed to entrench the power and privilege of the ruling white minority (Bunting, 2006:52). Higher Education institutions were established along the racial divide. From 1990– 1994 there were 21 universities and 15 technikons (polytechnics) that constituted higher education and 93 teacher training colleges catering for the African, coloured, Indian and white population groups of South Africa. The highest number of institutions from the three distinct types catered exclusively to the white population of the country. It was in this context that the new higher education policies of the new democratic government sought to reshape the system into one that met the goals of equity, democratisation, responsiveness and efficiency for all South Africans (Bunting, 2006).

In the new democratic South Africa, teacher training colleges underwent a radical transformation. Owing to the restructuring of higher education, many teacher training colleges were forced to close their doors; some colleges were forced to merge with other teacher training colleges which were later incorporated into higher education in the form of universities and technikons. Essop (2008) acknowledges that by incorporating the colleges of education into higher education it laid the basis for elevating the role, status and quality of teacher education in the country. However, the restructuring process brought with it a number of negative consequences, such as the following:

 The reduction in the number of teacher training institutions offering teacher education programmes resulted in a significantly lower number of individuals enrolling for such programmes.

- The fees that individuals had to pay at the higher education institutions were significantly higher than those at teacher training colleges.
- As many of these higher education institutions were located in urban areas, this limited the access of many potential students from rural and out-of-town locations (DBE & DHET, 2011).

Although initially these negative consequences drastically reduced enrolment numbers in teacher training programmes from 1995–2005, the past few years have shown a change in this trend. In 2005 there were 23 610 student teachers enrolled fulltime at contact universities in the BEd programme. By 2007 there were 33 546 student teachers enrolled at the 22 universities across the country. The 2009 enrolment numbers showed an increase of 25% from 2008 (DBE & DHET, 2011). In the same year, 2009, there were 5 942 new teachers who graduated (DHET, 2010: 8), but this figure was still far below the suggested requirement of between 20 000 and 30 000 new teachers annually (Gordon, 2009).

1.5.4 Teacher education students

There have been few studies that focus on the essence of the character or nature of being a teacher education student in South Africa. Wolhuter et al. (2012) maintain that not much research is available on what exactly inspires, motivates or drives student teachers to enrol in teacher education programmes. In addition, there is a gap in the literature with regard to understanding who our teacher education students are, what qualities they bring with them into the profession, and what attributes of the student teachers could be enhanced to ensure that students remain and succeed as newly qualified educators. Studies undertaken on career paths of education graduates reveal that many newly qualified teachers drop out of teaching within the first five years in their careers (Geyser & Wolhuter, 2001). A significant percentage of final-year teacher education students indicate that they have no plans to enter the teaching profession (Bertram et al., 2006).

This study focuses on pre-entry factors influencing teacher education students' firstyear experience and academic performance. The examination of such factors is intended to contribute to knowledge concerning who the students entering teacher education in South Africa are. If teacher education institutions wish to address the issue of teacher supply, it is imperative that they understand the nature of the students entering the teaching profession. Wolhuter et al. (2012) assert that many students enter teacher education, but opt for the profession only as a choice of last resort. Pitsoe (2013) agrees with Wolhuter et al. and adds that teaching has become a 'stopgap' profession or a profession of 'last resort'. This mind-set of teacher education students has resulted in many prospective teachers not being intrinsically committed to the profession (Wolhuter et al., 2012). This has a negative impact on the quality of the teacher who exits from the teacher training institution and subsequently enters the teaching profession as an uninspired and uninspiring teacher.

1.6 Purpose of study

The purpose of this study is to identify and explore pre-entry factors that influence teacher education students' first-year experience and academic performance. A cohort of first-year BEd undergraduate students at a university of technology was selected as a case study. A pivotal process was the investigation of students' preentry academic and non-academic factors and their influence on how the students integrate into the academic and social environment during their first year of study. Several questions emerged during the review of literature on the first-year experience, academic performance and retention, but the following key research questions outlined in the next section guided this research.

1.6.1 Research questions

Main RQ: How do pre-entry academic and non-academic factors influence teacher education students' first-year experience and academic performance?

The following sub-questions were identified:

- Sub-RQ1: Who are our first-year teacher education students?
- Sub-RQ2: What are the pre-entry academic and non-academic factors that firstyear teacher education students bring with them to higher education?

- Sub-RQ3: To what extent do pre-entry factors that students report influence their social and academic interaction in their first year of study?
- Sub-RQ4: To what extent do these factors contribute to the students' first-year experience and academic performance?

This study was located within a university of technology in Cape Town, South Africa.

1.6.2 Research aims and objectives

The over-arching aim of the study is twofold: first, to identify and explore factors influencing the first-year experience and academic performance of teacher education students and second, to identify strategies for first-year student support.

The objectives of the research are to:

- gain a better understanding of who the first-year students entering teacher education programmes are;
- gain familiarity with specific pre-entry factors that influence teacher education students' first-year experience;
- ascertain how these factors influence students' first-year academic and social integration and academic performance; and
- propose strategies to enhance the quality of students' first-year experience in higher education.

1.7 Research design and methodology

This research study is located predominantly within the pragmatic worldview, which is unique in key worldview elements. Pragmatism, according to Tashakkori and Teddlie (2010), is an example of a single paradigmatic stance. According to Wheeldon and Åhlberg (2012), the pragmatic approach to understanding reality and developing knowledge occurs through the use of both quantitative and qualitative data collection and analysis. Using this process provided this study with an enhanced understanding of enquiry from which the fundamental nature of students' first-year experience and academic performance was elicited. This study is designed within the qualitative research paradigm and uses the case study approach as the main research strategy. According to Harrison (2002), a case study is more aptly described as a strategy than a method: it sets out to address the understanding of a phenomenon within its operating context. Hancock and Algozzine (2006) describe a phenomenon to be a particular event, situation, programme, or activity which is researched in its natural context and is bounded by space and time. This study was focused principally on students' first-year experience at a higher education institution as its situation. It was bounded by a particular group of students, specifically first-year teacher education students, while the bounded time period was for one academic year, namely their first year of study.

The study employed both quantitative and qualitative data-collection methods which consisted of two distinct stages. The first phase of the study collected quantitative data through questionnaires mailed to all first-year students accepted into the BEd undergraduate programme. The objective of the quantitative data was to provide an initial description of first-year students entering the BEd programme and purposefully select participants for the qualitative phase of the study. Purposeful sampling provided a source of information-rich cases that were studied in depth.

Qualitative evidence was collected from eight first-year BEd students using semistructured interviews and focus-group interviews. Priority in this study was given to the qualitative method because qualitative research represented the major aspect of data collection and analysis. Qualitative research refined the study and defined the experiences and perceptions of participants in much greater detail. According to Babbie and Mouton (2001), qualitative research pursues a deeper understanding and description of human experiences than explanation and prediction of human behaviour. Participants selected for the study reflected the gender and racial balance of the BEd first-year cohort of 2012 of the case study. The purpose was to ensure that the conclusions adequately represented the range of variation.

Data was collected over a period of one year. The collection of data was sequential: the quantitative data was collected first. The rationale for collecting quantitative data was that it provided a general picture of the research problem and provided a backdrop for the first two research questions: who our first-year teacher education students are and what pre-entry academic and non-academic factors they bring with them to higher education. Quantitative data was integrated with the qualitative data. Quantitative data contributed to the selection of participants for the case study and development of questions for semi-structured interviews. Semi-structured interviews with selected participants were conducted at the end of the first term, three months after commencement of the course. At the end of the academic year, the same set of participants were invited to a focus-group interview. The intention was to identify preentry factors that teacher education students considered influential either in assisting or hindering their first-year experience. Participants' final academic scores were used as secondary data to determine academic performance. The results of the two phases were integrated during the discussion of the outcomes of the entire study.

The analysis of data relied mainly on thematic analysis as a means of organising data. According to Braun and Clarke (2006), thematic analysis is a qualitative analytic method for identifying, analysing and reporting patterns or themes within data. For the purpose of this study, for the researcher to be fully engaged with and immersed in the corpus of data, thematic analysis provided the most effective means of organising and describing how participants made meaning of their first-year experience in rich detail. For example, student teachers' perceptions, attitudes, understanding, knowledge, values, feelings and experiences were analysed in an attempt to approximate their reality.

For the development, support and management of quantitative analysis, the Statistical Package for the Social Sciences (SPSS, since 2015 officially named IBM SPSS Statistics) was used. SPSS is capable of handling large amounts of data and was useful in electronically capturing and storing data from all the questionnaires received from respondents. The use of SPSS in this study facilitated examination of relationships between different questions, generating descriptive data and creating graphical illustrations of the data.

For the development, support and management of qualitative analysis, the Atlas ti computer program assisted; the Computer-Aided Qualitative Data Analysis Software (CAQDAS) was used. CAQDAS has become a basic tool for qualitative researchers and is designed not to undertake the analysis for the researcher, but rather as a tool to aid the researcher in the processing of analysis. In this study, qualitative data collected from one-on-one interviews and focus-group interviews was uploaded to

Atlas ti. The CAQDAS package facilitated the processes of segmenting, categorising, annotating, retrieving, and searching within and across individual transcripts captured and categories identified. More importantly, Atlas ti allowed analysis of qualitative data in this study to be handled in an efficient and accessible manner.

1.8 Clarification of basic concepts and terms

Pre-entry academic factors: For this study, pre-entry academic factors refer to students' prior schooling experience: the school they attended, their academic performance at school, and their interactions with school staff and peers.

Pre-entry non-academic factors: For this study, pre-entry non-academic factors refer to demographic, socio-economic, family, and community influences.

First-year student (FY): Refers to an undergraduate student in his/her first year of study at an HEI.

First-Year Experience (FYE): FYE has a broad definition and encompasses all aspects of student life on campus. In this study, the identification by Krause et al. (2005) of key areas that make up the FYE is used to understand phenomena such as transition to university; aspirations of students; balancing work, study and social commitments; social and academic integration on campus; student satisfaction; and response to courses in the programme of study. The term 'student experience' is used interchangeably with student engagement, which is often applied in higher education (Harvey, 2008).

Student Expectations: In this study, student expectations refer to students' anticipation of a particular occurrence, FYE. According to James (2002a), student expectations are highly individualised and linked to their diverse nature.

Student Attrition: Refers to first-year attrition and is defined as the number of students who do not return to their chosen programme after the first year of enrolment. It refers to students who drop out of the programme.

Student Retention: Refers to students who complete the first year of study and have enrolled in the second year of study in the same programme. In this study student retention and first-year persistence are used interchangeably.

First-Generation Student: Refers to a student whose grandparents, parents or siblings have not earned a tertiary degree (Engle, 2007).

Academic Integration: Is described in this study as the students' academic associated perceptions and experiences with regard to their learning and the activities necessary for their learning: programme orientation; commitment and affiliation; students' formal interaction with lecturers and peers; use of the library; and attending out-of-class academic support activities.

Social Integration: Is described as the students' socially associated perceptions and experiences that arise from their social involvement with other students and lecturers in or out of university: informal interaction with lecturers; peer relations; participation in campus activities; use of campus facilities; and sense of belonging in the campus community.

Field: Bourdieu's concept of field is described as a socially structured space in which individuals play out their engagements with one another. Bourdieu regards every field as a situation of struggle, competition or conflict: the objective for each individual is to optimise his or her accumulation or retention of 'capital' (Swartz, 1997). This study is contextualised within the higher education field. The focus of this study is on what factors first-year students bring into the field of higher education that supports or constraints their first-year experience.

Capital: Bourdieu defines capital as any resource that holds symbolic value within a field and therefore acts as a currency that individuals take with them to the field. Bourdieu identifies the following three types of capital: economic capital, which he regards as capital that can be immediately and directly be converted into money; cultural capital that is mainly a product of education; and social capital that indicates an individual's social connections or networks (Bourdieu, 1986:245). Each of these types of capital has a social value and can be 'inherited', through the circumstances of early upbringing, or accumulated, exchanged and leveraged, like economic capital (Swartz, 1997).

Habitus: Bourdieu's concept of habitus emphasises the importance of history: it consists of 'a set of historical relations "deposited" within individual bodies in the form of mental and corporeal schemata of perception, appreciation and action' (Bourdieu & Wacquant, 1992:16). Thus habitus creates dispositions to act, interpret experiences, and think in a certain way (Lehmann, 2007). In this study first-year students' habitus is investigated to understand integration into the higher education field.

1.9 Organisation of the study

This thesis consists of eight chapters. Chapter 1 provides a brief introduction to this study. Chapter 2, titled 'Conceptual Framework', outlines the two principal theories: Tinto's (1975, 1993) student integration model, focusing mainly on academic and social integration, and Bourdieu's (1984) theory of practice, focusing mainly on his theoretical tools: capital, habitus and field. These two fundamental theories are used to develop the conceptual framework for exploring how certain pre-entry factors were entrenched in a student's habitus and capital that influenced his/her social and academic integration into the higher education field.

Chapter 3 provides a detailed review of literature that pertains to the first-year experience in the higher education field. First-year experience literature examined in this section includes increasing student access, student diversity, student retention, first-year transition, and academic success.

Chapter 4 details the research design and methodology used in this study. It provides the rationale and justification for the study's design within the qualitative research paradigm. It employs a case study strategy to collect both quantitative and qualitative data to address the research questions of this study.

Chapter 5 presents quantitative data derived from the mailed questionnaires received from the 195 respondents prior to their entering the BEd programme. This section makes use of descriptive statistical analysis and inferential techniques that include the use of correlations and *t*-test values to discuss the following characteristics of the first-year BEd student cohort entering higher education: demographics, socio-economic circumstances, prior academic experience, and students' expectations of their first year of study.

Chapter 6 presents the qualitative empirical evidence derived from in-depth one-onone interviews and the focus-group interview. The five concepts within the conceptual framework, habitus, capital, field, social integration, and academic integration, provided a suitable framework to present and discuss pre-entry academic and non-academic factors that participants identified as either supporting or hindering their first-year experience and academic performance. The qualitative evidence was presented under the following six themes: determination, self-reliance, fitting-in, out-of-habitus experience, positioning oneself to succeed, and challenges.

In Chapter 7, empirical findings from Chapter 5 and 6 of this study are interpreted and discussed. Owing to the wide range of themes and sub-themes that emerged from Chapter 5 and 6, and in order to offer clarity in the discussion chapter, the themes and sub-themes were further categorised, grouped and reduced to two data sets for discussion: pre-entry academic and non-academic factors influencing the first-year experience and academic performance; and adjusting to first-year experience and academic performance.

Chapter 8 concludes the study with a synopsis of the research journey, knowledge developed and recommendations for future action and research.

1.10 Conclusion

This introductory chapter provided a comprehensive introduction to the discussion and research on the first-year experience in higher education. The background and rationale for such a study was identified, and the influence of this study in contributing to the research body of knowledge on the first-year experience was highlighted. The context of the study, teacher education, was underscored. The research design, methodology and key concepts were emphasised in this chapter. The next chapter outlines the conceptual framework that informed this study.
CHAPTER 2 CONCEPTUAL FRAMEWORK

2.1 Introduction

The key purpose of this study is to identify and explore how first-year teacher education students' pre-entry academic and non-academic factors influence their first-year experience and academic performance. I review a range of theories that focus on the first-year experience at HEIs. What was evident from the literature reviewed was that many of the theories highlighted the first-year experience from different perspectives, such as student retention, student attrition, academic performance, student engagement, and student success (Astin, 1985, 1993; Bean, 1990; Pascarella & Terenzini, 1991; Tinto, 1993; Berger & Milem, 2000; Terenzini & Reason, 2005). My research concern is complex and based in the South African context: many of the theories I reviewed partially covered aspects of the concern. These theories add to the understanding of studies on students' involvement in academic and social systems of the university, but the focus of these theories remains fairly limited (Reason, 2009). From the considerable list of literature that I reviewed, it was evident that many of the theories did not significantly address my research questions. The theories did not explicitly incorporate or emphasise the role and/or influence that students' habitus and cultural capital (Bourdieu, 1984) may have on the way they experience university life and academic performance. Many of the theories on student persistence, retention and success focused on issues within the institutions, that is, the intra-university environment, but neglected to take into account students' prior dispositions.

The widening and increased participation in South Africa's higher education milieu over the past 20 years has resulted in diverse students entering universities. Many students are from disadvantaged backgrounds and educational experiences (CHE, 2013). In alignment with this study's research concern, exploring students' pre-entry dispositions becomes particularly important in understanding and interpreting students' first-year experience and academic success in the South African higher education context. In this regard, I focus on specific concepts from two theories to inform this study: Tinto's (1975, 1993) student integration model, concentrating mainly on academic and social integration, and Bourdieu's (1984) theory of practice, focusing mainly on the theoretical concepts he defines: capital, habitus, and field.

A sound and well-considered theoretical matrix is essential when conducting research. Imenda (2014) states that the importance of theory to inform research is mainly because theory and research reinforce what is learned through practice to create knowledge within a given discipline. Kerlinger (1973:9) defines theory as 'a set of interrelated constructs or concepts, definitions, and propositions that present a systematic view of phenomena by specifying relations among variables, with a purpose of explaining and predicting the phenomena'. When a researcher is conducting research, he/she needs to select a theoretical platform or specific concepts within a theory to inform his/her research. Nonetheless, Imenda (2014) adds that if a researcher finds that his/her research problem cannot be meaningfully researched with reference to only one theory or those concepts that reside within one theory, then the researcher can bring together a number of related concepts to explain or provide a broader understanding of the phenomena under investigation. This synthesis of concepts can be regarded as a conceptual framework which provides an integrated way of examining a particular research problem (Liehr & Smith, 1999).

In this study, concepts from two theories are used to develop a conceptual framework that provides conceptual tools for exploring the interrelations between first-year students and their social contexts. The conceptual framework developed for this study not only provided a basic structure on which to build this study (Bell, 2005) but provided grounds for my research to gather together all the 'isolated bits' of empirical data into a coherent conceptual framework (Cohen et al., 2001). This framework allowed me to explore the first-year students' use of their personal habitus and capital in supporting or challenging their first-year experience and academic performance in the new higher education field.

A conceptual framework can be a visual or written product that explains, either graphically or in narrative form, the main aspects to be studied, According to Miles and Huberman (1994:440), a conceptual framework lays out the key factors, constructs or variables, and presumes relations among them which are used as a way to understand or explain the phenomena under investigation. The following main research question provided guidance in developing the conceptual framework for this study:

24

How do pre-entry academic and non-academic factors influence teacher education students' first-year experience and academic performance?

The conceptual map, Figure 2:1 on page 26, provides a guide to my thinking process which culminated in a comprehensive conceptual framework being developed for this study. In this chapter, the next section (Section 2.2) highlights the significance of a conceptual framework. It considers various authors' views with regard to constructing, using and interpreting a conceptual framework. Jabareen's (2009) guideline for developing a conceptual framework is used to inform the development of this study's conceptual framework, which is based on the work of two theorists, Tinto (1974, 1993) and Bourdieu (1984).

Section 2.3 below is divided into three parts. The first elaborates on Tinto's (1974, 1993) student integration model. The second part of this section reviews empirical studies undertaken in higher education that are underpinned by Tinto's integration model. The final part of this section concludes by exploring some criticism of Tinto's integration model.

Section 2.4 considers the second theorist's contribution to the conceptual framework of this study. This section is divided into three parts. The first part elaborates on Bourdieu's (1984) conceptual tools: field, habitus, and cultural capital. The second part reviews empirical studies completed on higher education that utilise Bourdieu's concepts of habitus, capital, and field. The final section includes a critique of Bourdieu's three concepts.

Section 2.5 of this chapter concludes with the visual presentation that places this study within a clear and well-constructed framework which, according to Atkins and Wallace (2012), allows the study to be conceptualised with similar studies but not regarded as a stand-alone study. The five concepts: habitus, capital, field, social integration, and academic integration are discussed with reference to this study.



Figure 2:1 Conceptual map of Chapter 2

2.2 Significance of a conceptual framework

Miles and Huberman (1994) refer to a conceptual framework as the researcher's map of the territory being investigated. The practical relevance of a conceptual framework to a study can be achieved by making use of concepts that provide a set of general signposts within the field of that study (Bryman, 1988). Cohen et al. (2001:13) add that 'concepts enable us to impose some sort of meaning of the world; through them reality is given sense, order and coherence. They are the means by which we are able to come to terms with our experience'.

Smyth (2004) regards a framework as a research tool that may assist a researcher to develop an awareness and understanding of the situation under scrutiny; to interrogate, review and reform the research investigation; and finally to be able to communicate the research under scrutiny.

For this study the following main features provide a structure and thinking tool in developing the conceptual framework that guided the investigation.

A conceptual framework:

- is not merely a collection of concepts but, rather, a construct in which each concept plays an integral role;
- does not provide a causal/analytical setting, but rather an interpretative approach to social reality;
- provides understanding and interpretation of intentions rather than offering a theoretical explanation or hard facts on knowledge;
- is indeterminist in nature and therefore does not enable us to predict an outcome; and
- can be developed and constructed through a process of qualitative analysis.

Finally, the sources of data consist of many discipline-oriented theories that become the empirical data of the conceptual framework analysis (Jabareen, 2009).

Smyth (2004) cautions that researchers developing a conceptual framework should be aware of the subjectivity of the process. A conceptual framework is a construction of knowledge bounded by the life-world experiences of the researcher developing it and therefore should not be attributed a power that it does not have. In addition, Miles and Huberman (1994) advise that a conceptual framework could be restrictive because researchers should be aware that not all data analysed fits within the framework which, in turn, limits the results of the study.

With the above cautions in mind, I believe I have remained open to fresh and unexpected incidents in the data and research investigation while using the conceptual framework as a map to guide the focus of this research. The concepts and ideas of Tinto (1974, 1993) and Bourdieu (1984) were used as a way of understanding and explaining the phenomena of the first-year experience. The

framework influenced the literature reviewed in this study; the approach to the study; the methods used; and the analysis (Atkins & Wallace, 2012).

The identification of a suitable conceptual framework for this study was important. It enabled me to locate this study within a particular context in which I could examine students' first-year university experience with more rigour and credibility. Incorporating Tinto's (1974, 1993) model of integration, underpinned by Bourdieu's (1984) conceptual tools, namely, cultural capital, field, and habitus as a conceptual framework for this study, enabled me to undertake research which was 'systematic, credible, verifiable, justifiable, useful, valuable and trustworthy' (Wellington, 2000:14). Concepts from Tinto and Bourdieu supported one another but articulated their respective phenomena, and established what Jabareen (2009) refers to as a framework-specific philosophy.

A framework-specific philosophy of this study's conceptual framework was necessary to understand university students' first-year experience in the South African context. By exploring the first-year experience of students in the South African context through the use of Bourdieu's habitus, capital, and field concepts, brought into focus South Africa's unjust apartheid era, the social and economic injustices, and the concomitant inequality in the education system. The use of this conceptual framework guided this study and provided a more holistic understanding of the complexity of the first-year experience. As stated by Czerniewicz and Brown (2013), Bourdieu's concepts of habitus, capital, and field enabled the unravelling of complex inter-relations between background, identity and the demands of higher education. Understanding first-year students through the use of this framework should enable first-year curriculum planners, lecturers and academic support services to be more mindful and sensitive when planning for a diverse first-year student body.

The use of the conceptual framework developed for this study allowed me to be more explicit in my thinking with regard to my research focus. It heightened my focus on first-year students' use of their habitus and capital when entering the new higher education field. First-year students entering the university bring with them their backgrounds and past dispositions (habitus) which are influenced by different forms of capital: social, economic, and cultural. Consequently, students' habitus and capital

28

produce practices that assist them to navigate the first-year educational experience, which, in turn, have an influence on their social and academic integration at university. Exploration of the varying degrees of students' social and academic integration emphasises the congruency between the university as a field and the students' habitus. The conceptual framework in this study permitted me to understand the complexity of certain characteristics and experiences from the students' habitus and capital: how such factors influenced their academic and social integration and ultimately shaped their first-year experience and academic performance. It assisted me in identifying and explaining why some students felt like 'a fish out of water', while others were like 'a fish in water' (Bourdieu & Wacquant, 1992).

Finally, my incorporation of constructs from Tinto's integration model (1975, 1993) and Bourdieu's (1984) conceptual tools was shaped by the need to understand and interpret the social realities of the students. My research commitment is less to an approach and more to exploring the problem of how students' pre-entry academic and non-academic characteristics influence their first-year university experience and academic performance.

To provide the backdrop to the construction of the conceptual framework for this study, the following sections examine Tinto's (1975, 1993) student integration model and Bourdieu's (Bourdieu, 1984) conceptual tools in more detail.

2.3 Tinto's integration model

A dominant influence of theoretical literature on student retention and first-year experience has been the work of Vincent Tinto from the USA (Tinto, 1975; 1993). He developed a longitudinal student integration model that seeks to explain the process of student departure from higher education. Tinto's model (1975) theorises that interactions between students' characteristics and experiences in the university environment influence students' academic success and first-year transition. According to Tinto, students entering higher education retain attributes such as family background, skills, abilities, and prior education that influence their initial choices of goals and commitments. When a student's goals and commitments don't align with their university experiences, this may cause the student to become

academically and socially disconnected, which most likely, would lead to the student's dropping out of university. Thus Tinto's (1975) model suggests that students assign the following reasons to their dropping out of higher education: personal attributes, educational goals, and/or their commitment to the institution.

Tinto's (1975) interactionalist theory of student departure suggests four distinct features in the process to student departure:

- Student entry characteristics
- Initial commitment to goal and institution
- Social and academic integration
- Subsequent commitment to goal and institution

Tinto states that students enter university with various individual characteristics, such as background characteristics (e.g. family background, parental educational level, socio-economic status); individual attributes (e.g. age, sex, race, ability), and prior academic experience (e.g. schooling experience, grades), that play a role in student departure. Tinto hypothesises that students' entry characteristics directly influence their initial commitment to the institution and goal towards graduation. He adds that students' entry characteristics can also directly affect their decision to either stay or drop out of university.

Tinto's model suggests that students' initial commitment to an institution and goal to graduate consecutively influence their degree of integration into the social and academic domain of the institution. Academic integration, which refers to students' formal interactions with lecturers, students and support staff, can be defined under two dimensions: structural and normative. According to Tinto (1975:104), structural integration involves the students meeting the explicit standards of the university required for academic achievement, while normative integration relates to an individual's identification with the normative structure of the academic system. Tinto (1975:106) also adds that the normative integration takes the form of congruency between the individual's intellectual development and the intellectual environment of the university. Social integration refers to students' non-formal networks and associations with peers, academic staff and extra-curricular activities at university.

Tinto (1975:107) suggests that social integration concerns the degree of congruency between the individual student and the social system of the university.

In Tinto's (1975) model, the formation of students' subsequent commitments to the university and their own goals to succeed is directly affected by their level of social and academic integration. He posits that the greater the level of academic integration, the greater the level of subsequent commitment to the students' academic goals. Similarly, Tinto adds that the greater the level of social integration, the greater the level of the students' subsequent commitment to the university (Tinto, 1975:110). He adds that students' initial commitments also shape their subsequent university and goal commitments which will influence their university persistence. In this regard, Tinto's model asserts that students who engage in formal and informal academic and social integration are less likely to leave their institution. Harvey et al. (2006) further contend that the integrative and positive experiences at university reformulate the students' goals and commitment, thus reinforcing commitment.

Tinto's (1993) theory of student departure highlights academic and social integration as two constructs that represent the process of student adjustment to their new university environment. He adds that these two clusters of behaviour influence students' overall performance and affective responses to the university experience. Tinto (1975, 1993) suggests that compensatory mechanisms can be applied with regard to academic and social integration, and goal and university commitment. For example, Tinto suggests that a high level of academic integration can compensate for a lower level of social integration and a high level of social integration can composite for a lower level of academic integration; however he does caution that an excessive level of social integration can cause an impact on academic performance. This study examines Tinto's two constructs, social and academic integration, to understand how and why certain pre-entry factors influence students' integration into the university environment.

The next two sections (2.3.1 and 2.3.2) highlight the many studies that validate, test and apply Tinto's model to different types of higher education context. Section 2.3.2 provides a critique of Tinto's student integration model.

2.3.1 Empirical studies on Tinto's student integration model

Tinto's (1975, 1993) student integration model provided a conceptual framework which many researchers used to develop their empirical evidence on student persistence, attrition, academic performance, success and engagement at tertiary institutions (Abramson & Jones, 2003; Yorke & Longden, 2004; Horstmanshof & Zimitat, 2007; Kuh et al., 2007). Tinto's integration model was applied to different institutional settings, for example, single institutions, small campuses, multiple institutions and within faculties and subjects (Wilcox et al., 2005; Stuart, 2006; Noyes, 2007-2008; Pather, 2013).

Researchers using Tinto's integration model have contributed a great deal to the understanding of what affects student dropout and persistence. Pascarella and Terenzini (1978, 1979) indicated that the model was appropriate in predicting those students at risk. Researchers have applied Tinto's integration model to a range of different contexts within higher education institutions to identify the factors that influence university experience, student performance and persistence. The following studies highlight the flexibility of Tinto's model within the different higher education contexts.

In Duquette's (2000) study on disabled students, Tinto's integration model was used as a framework for exploring university experience and persistence among university students with disabilities. The variables, student background and characteristics and social and academic integration, were examined to provide an insight into the students' university experience and reasons for persisting or dropping out. The findings of the study revealed that the variables that were more important in predicting disabled students' success and persistence were their background characteristics and academic integration, rather than social integration. Family support, goal commitment and supportive lecturing staff were found to be important elements related to persistence among students with disabilities.

In another context Tinto's model was applied to students with a military background. In this study DiRamio and Jarvis (2011) used Tinto's (1975) model to identify variables influencing veteran students' academic success and persistence. The study focused specifically on students' pre-entry attributes that influenced their

32

academic success and persistence. Finance was cited as one of the main issues influencing the veteran students' academic success and persistence. The findings from this study were useful in initiating introductory discussions on veteran students' access to funding to enhance university experience and persistence. The study created a platform for future research on veteran students and financial aid in higher education.

Longwell-Grice and Longwell-Grice (2008) used Tinto's model on a multiple case study involving working-class first-generation white male students in their first year of study. Their study focused on social and academic integration with specific attention to student-lecturer interactions. Their study revealed that first-generation workingclass white male students felt intimidated by the idea of seeking academic support from lecturing staff. Longwell-Grice and Longwell-Grice asserted that universities need to find proactive ways to develop student-lecturer interactions to improve student success and retention. In a similar study completed on first-generation college students, Woosley and Shepler (2011) used Tinto's model to examine predictors of social and academic integration, institutional satisfaction and homesick-related anxiety, early in the students' integration experiences at college. Their study acknowledges the importance of the institution in creating a campus environment that enables students to feel accepted and promotes academic performance.

According to Tinto (1995), students' decisions to withdraw from campus will depend on the level of integration within the two domains. Therefore much of the literature on university experiences that I reviewed focused on empirical studies relating to students' institutional experiences with a specific focus on their level of academic and social integration. Some studies used Tinto's concepts of academic and social integration to examine the influence of both the social and academic integration of students into the university environment. Other studies focused solely on either academic or social integration to explain academic success, persistence and dropout. For example, in the study of Krause et al. (2005), they focused mainly on academic interaction with staff and students to investigate academic success. Studies by Inman and Pascarella (1998) and Benham (2006) focused on successful classroom and learning experiences to explain the level of academic integration and academic success. However, studies undertaken on students' social integration highlight the following influential factors with regard to persistence and academic success: student and peer interaction (Harris, 2006), out-of-classroom contact with faculty (Pascarella & Terenzini, 2005) and integration into the campus learning community (Jaffee et al., 2008).

Some research findings, however, indicated that social integration is more influential on students' withdrawal decisions than academic integration. Mackie (1998) found that dropping out of campus in the early part of the course often results from a failure in social integration, such as difficulties in making friends, feeling alienated, homesick or lonely. Wilcox et al. (2005) stressed that compatible friends provide direct emotional support, equivalent to family relationships, as well as buffering support in stressful situations, which is essential to retention. In addition Thomas (2002) acknowledged that students' new social networks at university often provide support to overcome retention difficulties. Harvey et al. (2006) maintained that social and academic factors both play a role in withdrawal but it would be hasty to give priority to one over the other.

Consequently, other empirical studies that used Tinto's model (1975, 1993) revealed that the model did not support their research, specifically when applied to minority and under-represented students. It was acknowledged that Tinto's model does not take into account cultural variables (Tierney, 2000; Guiffrida, 2006). In Guiffrida's (2006) study, he attempted to integrate the social and cross-cultural psychological principles into Tinto's integration model. Guiffrida's intention was to strengthen and enhance cultural sensitivity and make Tinto's model more descriptive of minority students' academic achievement and persistence. Guiffrida used key components of theories on self-determination and job involvement to create a framework for explaining cultural norms' and motivational orientation's impact on college students' achievement and persistence. His study suggests that the integration of cultural norms and motivational orientation into Tinto's integration model may allow Tinto's theory to recognise how diverse socialisation experiences affect motivation towards academic achievement and persistence. Guiffrida (2006) adds that the use of his framework can provide a more comprehensive multicultural understanding of students' commitment and subsequent academic performance and persistence.

Other researchers have built on Tinto's integration model to offer a different perspective through which student dropout and persistence could be examined. For example, Bean (1980) proposed an examination of student attrition through organisational perspectives. His research focused on student attrition and those factors influencing non-persisters. Bean applied the theoretical concept of job turnover to tertiary education and reported on similarities between leaving the world of work and leaving university. Bean's model synthesised Spady's (1970) social integration process model with Tinto's (1975) model to create a new model that included attitudinal variables. Bean (1980) suggested that student attrition was affected by the following: student background variables; interaction by students within the institution; the influence of environmental variables (finances, family support); and the presence of attitudinal variables and student intention.

During the same period Pascarella and Terenzini (1980) expanded on the work of Tinto (1975), Spady (1970), Bean (1980) and Astin (1975). In their theory on student intent and persistence, social and academic integration formed the basis. Pascarella and Terenzini (1980) outlined their student involvement theory from the perspective of student interactions with academic staff and peers. They provided a connecting relationship model addressing both direct and indirect effects of student involvement and interaction. A study by Pascarella and Terenzini (1980) focused on interactions and inter-relationships between students and academic staff; it revealed that the amount of time that students spent with academic staff, both in and out of the formal classroom, strongly influenced student intent and persistence.

Accordingly, Tinto (1982) acknowledged that his 1975 model did not focus on characteristics or forces external to the institutional environment. He stated that the limitation of his theory was 'what we took to be self-evident in its development has apparently proven not to be' (Tinto, 1982:688). Tinto added the need to include additional ethnographic information such as background variables and to assess the role of academic and social integration factors into his conceptual model of persistence (Metz, 2002). Thus Tinto's 1987 model posited the following five major theoretical research bases for developing and understanding the evolving nature of student persistence: psychological, societal, economic, organisational, and interactive. In spite of Tinto's revision to his 1975 theory and the fact that it is the

most studied and tested model in the literature (Braxton & Hirschy, 2005), criticism of his model on student integration is ongoing. The next section highlights some of these criticisms that relate to this study.

2.3.2 Critique of Tinto's model

Tinto's theory focused heavily on traditional, white young American first-year students in private residential institutions. Consequently, this has led to his model being criticised for the following reasons: being too homogeneous (Pascarella & Terenzini, 1983; Attinasi, 1989; Brunsden et al., 2000); its inability to explain racial minority student retention (Tierney, 1992; Stage & Anaya, 1996; Rendón et al., 2000); lack of research on the attrition of older students (Bean & Metzner, 1985); and neglecting the 'widened community' of students that augmented participation has brought (Rhodes & Nevill, 2004).

Another critique of Tinto's 1975 model is the use of academic and social integration which is central to his model on student integration. Tierney (1992) noted that these constructs were too broad and did not address specific examples that could be related to non-traditional elements in higher education. Melguizo (2011) adds that Tinto does not provide explicit definitions or clarity to academic and social integration or measuring of such constructs. This has resulted in researchers defining and measuring these constructs to fit their own purpose. In addition, Tinto does not explain how students become socially and academically integrated into the institutions (Bean & Eaton, 2000).

Another significant critique of Tinto's integration model was cultural limitation. Tierney (2000) critiqued Tinto's model for misrepresenting the cultural aspects of transition which placed too much emphasis on withdrawal as an individual matter. Tierney (1992) also criticised Tinto's misinterpretation of Van Gennep's anthropological rites of passage: Tinto's theoretical presupposition that students must undergo a three-stage process of separation, transition and integration to successfully persist until degree completion. This theory may hold potentially harmful consequences for racial and ethnic minority students. For example, Tinto's (1993) assertion that students need to separate from past associations and traditions to become integrated into the social and academic systems in the institution is strongly opposed by Tierney (1992). Tierney argues that Tinto's use of the concept from Van Gennep's (1960) transitional model is not applicable to minority students because the model was intended to describe development progression within a single culture rather than adaptation from one culture to another. Tierney further states that the misinterpretation is potentially harmful as it encourages minority students to separate from their cultural traditions and supportive relationships.

Many researchers felt that Tinto's integration model has limited applicability since it is best suited to institutional analysis of the persistence of traditional undergraduate students (Rendón et al., 2000). Nonetheless, the work of Tinto (1975, 1993) has been particularly important in establishing the role of the institution in promoting an environment for student integration and remains relevant in this regard. Tinto (2007), however, acknowledges the limitations of his model and recognises a range of models, some sociological, psychological and economic in nature, that can be used to explain student retention. This study acknowledges Tinto's limitations for the current study and reviews an alternative model that goes beyond the explanatory and descriptive, creating a deeper understanding of factors influencing students' first-year experience and integration into the university environment.

2.4 Bourdieu's theoretical tools

A complementary approach to Tinto's integration model draws on Bourdieu's (1984) theoretical tools, capital, field, and habitus, which Bourdieu termed his three 'thinking tools' (Wacquant, 1989:50). Bourdieu uses these three concepts to explain how the environment in which people are raised and their conditions of cultural and material existence shape their attitudes, their means of interpreting the world and their capacity to engage in academic dialogue (Bourdieu & Passeron, 1977). Bourdieu's theoretical tools are predominantly geared to understanding the social world. This, according to Maton (2008), is not simply the result of one's habitus, but rather of relations between one's habitus and one's current circumstances. For Bourdieu, habitus, capital, and field are unavoidably interrelated, both conceptually and empirically (Bourdieu & Wacquant, 1992:96-97). To talk of habitus without field, or claim to analyse 'habitus' without analysing 'field', extracts habitus from the very contexts which give it meaning and in which it operates (Maton, 2008).

Bourdieu encourages researchers to adopt a relational mode of thinking that goes beyond surface empirical practices to explore more deeply their underlying structuring principles. James (2011) elaborates that Bourdieu's theoretical tools enable researchers to be more specific about their unit of analysis. He adds that Bourdieu helps the researcher to choose a 'social space', or a 'field', or a set of relationships defined by differences in some form of capital and/or power, while not losing sight of the individuals. Seemingly, Bourdieu's tools aim to help transform our ways of seeing the social world. Using Bourdieu's conceptual tools to explore and understand students' per-entry factors influencing their first-year experience is essential to this study.

(a) The concept of field

In examining Bourdieu's theoretical tools, Warde (2004) expresses that the concept of field, although central and essential to the theoretical foundation of the analysis, does not play a substantive role as habitus and cultural capital, which accomplishes all the interpretive work. Bourdieu describes field as a socially structured space in which individuals play out their engagements with one another. Warde (2004) elaborates on this description by conceptualising a field as a relatively autonomous structured space, which has been socially instituted, thus having a definable but contingent history of development which agents recognise and refer to its history. Thus Warde characterises the field as possessing the following features:

- Having more autonomy than others, with some parts of fields more autonomous than other parts.
- As an arena of constant struggle for 'stakes', particular types of fieldspecific and generic capitals.
- The dynamics of a field arise from the positions, dispositions and positiontaking of agents.
- It could be altered in accordance with the experience within the field itself.

In Bourdieu's (1984) explanation of 'field', he frequently employs the analogy of a game when conveying the sense of activity or activities within a field. Hillier and Rooksby (2002) elaborate on Bourdieu's analogy of a game and state that to be successful in a game requires participants not only to understand and follow the rules of the game, but also to have a sense of the game. Playing in the game

requires constant awareness of, and responsiveness to, all the participants involved in the game. It requires assessment of one's own team players' resources, strengths and weaknesses and also those of the opposition team. Strategies required to play the game include improvisation, flexibility and intuition. Few players come into the game with the abilities described. Insights and a sense of the game, a habitus, develop with experience. Players learn from experience what is possible and not possible; how to work effectively within existing practices in the field and how the rules might be modified. The players' activities are constructed as much by the external limits of the rules and regulations as their own internalisation of what they can or cannot do and want to do in particular circumstances. Thus entering a new field could be supportive of some participants while posing challenges for others.

A field could be an organisation or system: government, university, business, or education in which actors play out their engagement with one another. The actors' behaviours will be related to their position in the field and also the resources available to them (Hillier & Rooksby, 2002). Like a game, each field encompasses the 'rules of the game' and the taken-for-granted practices that could be explicitly or implicitly imposed on those who seek to enter it. This in turn structures practice by defining the range of possible and acceptable actions or behaviour available to those operating within the field (Grenfell, 2004). According to Jackson (2008), a field encompasses its own rules and regulations, and its own objectives, values and strategies on how to play the game, concluding that the concept of field is restrictive regarding the social context in which the actors are embedded and through which their actions are aimed. Bourdieu (1984) discusses the field as a competitive arena where individuals compete and struggle to accumulate and monopolise different kinds of capital which DiMaggio (1979) states can protect or enhance individuals' existing resources.

From the above description of the field and the practices within it, it is evident that within a specific field, an individual's capital and habitus come into play. Berger (2000) asserts that each field has its specific logic that dictates in what ways different amounts and types of capital can be used for competitive advantage. However, it is habitus that offers an insightful way of understanding social interactions within a field.

39

(b) The concept of capital

In order to 'play the game' in the field, Bourdieu (1984:224) states that individuals need to have some existing stock or capital that is relevant to the new field. Bourdieu defines capital as any resource that holds symbolic value within a field and acts as a currency that actors take with them to the field. Bourdieu identifies the following three types of capital: economic capital, which he regards as 'immediately and directly convertible into money' (Bourdieu,1986:245); cultural capital refers mainly to the products of education, and whether these are visible in individuals: accent, vocabulary, behaviour etc., connected to objects such as qualifications, or connected to institutions, such as schools and universities (James, 2011); and social capital, which denotes individuals' social connections or networks of lasting relations that have been established and continue to expand (Grenfell & James, 1998).

According to Bourdieu (1971), individuals with access to similar types and amounts of capital share a common habitus. Berger (2000) concurs and elaborates that individuals' shared level of access to capital develops a shared worldview among these individuals as their perceptions, experiences, preferences and tendencies become standardised. This, however, serves as a mechanism to marginalise others who have access to different amounts and types of capital within a field. DiMaggio and Mohr (1985) regard capital as having a cumulative effect; the greater, the earlier the accumulation, the easier it becomes to expand one's personal assets.

The concept of the 'field' is closely linked to that of capital. As mentioned by Bourdieu and Wacquant (1992:101), 'capital does not exist and function except in relation to a field'. Therefore within a field, individuals hold unequal positions and experience unequal trajectories based upon volume and composition of their portfolio of capital (Wacquant, 1998). Bourdieu (1986) attributes success in higher education to cultural capital, both in the quantity and type inherited from a person's family background. He describes cultural capital to be cumulative and explains that students with higher initial levels of cultural capital, which is acquired from primary socialisation in the family, tend to be able to use this investment of cultural capital to gain further cultural wealth through the secondary socialisation process in schools and/or higher education (Bourdieu, 1973). Bourdieu sees cultural capital as a

resource which, based on its distribution among students of different social classes, enables some to profit in the new field while others fail.

It is the students' cultural capital that determines their individual experiences and consequently the congruence or fit between their habitus and field in which they enter. Bourdieu and Wacquant (1992:127) describe this as follows:

... social reality exists, so to speak, twice, in things and in minds, in fields and in habitus, outside and inside social agents. And when habitus encounters a social world of which it is the product, it is like a 'fish in water': it does not feel the weight of the water and it takes the world about itself for granted.

(c) The concept of habitus

Habitus appears to be an elusive, indeterminate concept, yet it is central to Bourdieu's' distinctive sociological approach, 'field' theory, and philosophy of practice. It is crucial to Bourdieu's originality and contribution to the field of social science (Maton, 2008). Habitus can be described as a set of values, practices and norms which people assimilate as part of who they are and how they operate. Habitus is 'a past which survives in the present and tends to perpetuate itself into the future' (Bourdieu, 1977). It represents how individuals make use of their past and present experiences to address a current or future situation. Bourdieu explains habitus as 'a system of lasting, transposable dispositions which, integrating past experiences, functions at every moment as a matrix of perceptions, appreciations, and actions' (Bourdieu, 1971). Bourdieu defines habitus as a property of actors, whether individuals, groups or institutions, that comprises a 'structured and structuring structure' (Bourdieu, 1998:170). It is 'structured' by one's past and present circumstances, such as family upbringing and educational experiences. It is with 'structuring' that one's habitus helps to shape one's present and future practices. It is a 'structure' in that it is systematically ordered rather than random or haphazard. This 'structure' comprises a system of dispositions which generate perceptions, appreciations and practices (Bourdieu, 1990:53).

Bourdieu used habitus to explain how structures are limited in their power to control individuals, as individuals bring dispositions to what structures offer to produce practice (Maton, 2008). For Bourdieu, individuals' habitus or dispositions and their expressions of power operate within fields which are structured arenas. The concepts of field and habitus prove useful when examining the interaction between

students as individuals and the institutions they attend (Bourdieu & Passeron, 1977). For Bourdieu and Passeron, the individuals and the institution exist in structural relations to each other. The individuals use their experiences and practice which are influenced by their personal habitus to navigate within the higher education field. The following quotation encapsulates Bourdieu's theoretical tools in a few condensed lines:

A field consists of a set of objective, historical relations between positions anchored in certain forms of power (or capital), while habitus consists of a set of historical relations 'deposited' within individual bodies in the form of mental and corporeal schemata of perception, appreciation, and action (Bourdieu & Wacquant, 1992:6).

It is therefore evident that the field and habitus are both manifestations of power, where field is a set of power relations objectified within institutions, and habitus a set of power relations embedded within individuals. As Bourdieu highlights, if a field is the game, then habitus is the 'sense of the game' (Bourdieu, 1990:61; Bourdieu & Wacquant, 1992:98).

Webb et al. (2002) highlight four important aspects that Bourdieu associates with habitus:

- First, our knowledge, the way we understand the world, our beliefs and values, are always constructed through the habitus.
- Second, our cultural trajectories exert an influence toward certain attitudes, values or the way we behave.
- Third, the habitus can be understood as 'the feel for the game'; it is always constituted in moments of practice and brought out when a set of dispositions meet a particular problem, choice or context.
- Finally, habitus operates at a level that is partly unconscious because habitus is arbitrary; there is nothing natural or essential about the values we hold, the desires we pursue or the practices in which we engage.

In summary, habitus operates at various levels: in one's thoughts, actions, use of language, and in how one embodies experiences of structures and relations (field). Although habitus is a whole body experience, it can't be solely attributed to an individual, since dispositions are created and recreated through social interaction and tradition. Thus James (2011) describes Bourdieu's approach as 'relational': the

study of the social world is not only about the relations between things or people but also about the substance of the things or individuals themselves. In this regard Bourdieu's (1984) concepts of capital, habitus and field can be represented in his theory of practice relational equation: Capital + habitus + field = practice. Practices are actions undertaken by individuals or groups that represent what they do. That is, the values of cultural capital, social capital, economic capital and habitus in a society can be best understood in relation to where individuals work (fields). James (2011) acknowledges that these concepts have more practical importance and can help a researcher choose a social space or field, or a set of relationships defined by differences in some form of capital and/or power, while not losing sight of the individuals.

2.4.2 Empirical studies using Bourdieu's theoretical tools

Many researchers have applied Bourdieu's theoretical tools to explore differential higher education experiences by gender, race and class of the varying student population, such as minority students (Ovink & Veazey, 2010); black students (Jones, 2001); first-generation and non-traditional students (Watson et al., 2009); low-income and working-class students (Reay et al., 2010; Devas, 2011); and gender (Dumais, 2002).

A considerable number of studies were reviewed and focused specifically on Bourdieu's concept of habitus to explain, interpret and understand institutional habitus, individual habitus, collective student habitus, and habitus of academic staff (Nash, 2002; Thomas, 2002; Reay, 2004; Jawitz, 2009; Kloot, 2009). Other studies reviewed used cultural capital as a tool to explore its influence on student behaviour and experiences. For example, in the study by Ball et al. (2002), they explored the notion of cultural capital and the role it plays in individuals making the choice to attend university and the choice of programme to study at university. In this study they found that students with strong cultural capital are likely to make more informed choices and therefore have more commitment and clearer goals than students with a lower cultural capital. Martin and Spenner's (2009) study, conducted at an elite university, examined how students' habitus is associated with types of economic, social, cultural and human capital. In addition, their study investigates how these types of capital influence academic achievement, college activities and postgraduation plans. Martin and Spenner's findings revealed similar results to those of Ball et al. (2002), which suggests that the different forms of capital, especially cultural capital which more privileged students simply inherit, play an important role in educational attainment. Bourdieu (1986) concurs that schools and higher education institutions are sites for cultural reproduction where particular forms of cultural capital are promoted, which according to Lane and Taber (2012) has implications for social mobility and economic success. Lane and Taber add that if a student's cultural capital aligns with that which is privileged in education institutions and by teachers/lecturers, then that student is more likely to be academically successful.

In Longden's (2004) study, which influenced the development of this study's conceptual framework, he explores Bourdieu's cultural capital and social reproduction theory as an alternative lens to understanding, explaining and interpreting student retention within the context of higher education in the United Kingdom. His study questions the appropriateness of Bourdieu's lens within the different social and cultural settings for interpreting data relating to students' early departure. Longden argues that although Bourdieu's (1984) lens has not been widely used in the UK, his theory has been helpful in viewing and interpreting data where influences of socio-economic factors are important. For Longden's (2004) study, Boudieu's lens provided a different perspective on the importance of previous experience of education and its influence on a student's habitus and forms of capital prior to engaging with higher education. His study is based on the view that all institutions have cultural capital formed on the wider perception of the status of the institution. Thus by the institution's promoting particular forms of culture, the education system has the ability to influence how well a student's habitus and levels of cultural capital available to them to access resources can promote or inhibit their educational activities in the higher education environment. Similarly, Lane and Taber (2012), agree that social status or social class position, habitus and capital, are influenced by the organising effects of education and supported by how well a student's cultural capital aligns with the cultural capital promoted in these institutions.

Longden (2004) concludes that viewing early student departure through Bourdieu's lens emphasises the role played by schools in influencing early cultural experiences

which provide the cultural capital that is used by students within the higher education arena. In addition, students' experiences are influenced by the higher education institution's own perceived cultural capital. The findings from Longden's (2004) study suggest that a disjunction between the relative amounts of cultural capital that the institution and the individual has can either promote and support persistence in higher education or undermine persistence, resulting in early departure. He concludes by advising researchers to view their research problem under investigation from the perspective of a theoretical framework or through a lens, as he did in his study, to bring about a deeper understanding of student departure from higher education. He adds that only when a phenomenon is interpreted against a theoretical framework or through a lens, can action be tested and evaluated with increased understanding of the underlying factors which affect the problem to be identified. This study takes heed of Longden's recommendation and uses a conceptual framework to guide the investigation of the study.

2.4.3 Critique of Bourdieu's conceptual tools

Bourdieu's theoretical tools have been applied and reviewed by many researchers (Thomas, 2002; Longden, 2004; Reay, 2004; Reay et al., 2007). There have been several researchers who have criticised his concepts (Kingston, 2001; Sullivan, 2001, 2002; Reay, 2004). In this section, criticism of Bourdieu's conceptual tools is discussed and in particular the relative vagueness of his concepts of habitus and cultural capital.

Kingston (2001) and other authors have criticised Bourdieu for the vagueness of his concepts and state that the lack of a more explicit description of, for example, his cultural capital concept, has resulted in a wide variety of variables used to study this concept. To compound the vagueness, Bourdieu in his own work identifies the following as some elements of cultural capital: education credentials, linguistic capabilities, and school systems. In his work on educational credentials, Bourdieu's (1973) view is that cultural capital is inculcated in high-class homes and thus enables high-class students to gain higher educational credentials than other students. Bourdieu has been criticised by Sullivan (2002) for not being precise enough about exactly which of the resources associated with high-class homes contribute to cultural capital and how these resources are converted into educational credentials.

However, Bourdieu admits that his cultural capital concept is not as precise in definition as economic capital and points out that cultural capital must be fluid to be reflective of the society which is being studied. Sullivan (2002), however, acknowledges that the concept of cultural capital, although not constructed concisely by Bourdieu, is substantive enough to be potentially useful to empirical researchers.

Another concern is with Bourdieu's concept of habitus, which is regarded as central to his conceptual tools, but Sullivan (2002) argues that this concept is theoretically incoherent and has no clear use for empirical researchers. Habitus holds the same criticism as cultural capital, of being too vague and nebulous, to be operationalised and interpreted (Kingston 2001; Lareau & Weininger, 2003; Wildhagen, 2010). Although habitus has been reported as the most misunderstood, misused, and contested of Bourdieu's concepts, it is still Bourdieu's most widely cited concept (Maton, 2008). Many researchers in the field of education have referred to habitus in a broad range of theoretical tasks (Nash, 1999; Dumais, 2002, 2006; Reay, 2004; Wacquant, 2005; Davey, 2009). Habitus has been conceived by Bourdieu as a mediating device between structure and agency, yet his concept of habitus has been frequently criticised for being overly deterministic and unable to account for individual change. DiMaggio (1979) observed that the notion of habitus is completely deterministic, thereby leaving no place for individual agency or individual consciousness. What is problematic, according to Warde (2004), is that Bourdieu developed habitus in part as a means of countering the undue emphasis on consciousness in social science. Bourdieu states that habitus is neither a result of free will, nor determined by structures, but created by dispositions which are created and reproduced unconsciously, 'without any deliberate pursuit of coherence ... without any conscious concentration' (Bourdieu, 1984:170). This sparked criticism by many authors. Firstly Bourdieu's lack of a clear explanation of 'a set of dispositions' caused many authors to condemn the concept of habitus as 'ambiguous and overloaded' (Nash, 1990:446). Secondly, his unconscious aspect of habitus is critiqued for neglecting the mundane everyday reflectivity, what Sayer terms 'our inner conversations' (Sayer, 2005:29), thus leaving researchers with little choice but to innovate and invent new, seemingly consistent notions to make sense of Bourdieu's concepts (Atkinson, 2011:332).

46

However, Bourdieu (1990:116) denies the charge of determinism on the grounds that the same habitus will produce different practices in different social fields and the habitus can be changed by changed circumstances. Reay (2010) points out that without a confined definition of habitus, this ironically contributes to a potential strength of the concept, specifically when conducting empirical work, as 'it makes possible adaptation rather than the more constricting straightforward adoption' of the concept within the applied study (Reay, 1995:358).

In sum, Bourdieu's three conceptual tools should be understood as an interrelated system and not considered individually. As Frank (2002) explains, cultural capital provides the tools for success within a particular field, which includes understanding how and why to comply with the rules of the field. Habitus gives an individual the confidence, feeling of comfort and sense of entitlement to use these tools in the field. Thus Bourdieu's concept of habitus enables us to understand individuals as a complex mix of their past and present, which is always in the process of completion but never complete, which Bourdieu describes as 'a power of adaptation' (Bourdieu, 1993:88).

2.5 A conceived conceptual framework for this study

After reviewing Tinto and Bourdieu's theories and considering the criticism of their work, it was evident that the strengths of both these models and the lacunae and ambiguity of certain concepts should form part of this study's conceptual framework. In this regard the framework developed provided a more suitable and expanded conceptual one for exploring the interactions and influences between students and their particular contexts. The views of Tinto (1975, 1993) and Bourdieu (1984) that contribute to this study are:

Tinto's integration model:

- 1. Students entering university have various individual characteristics that influence their first-year experience and academic performance:
 - Family background (socio-economic status, parents' education levels and expectations)
 - Individual students' attributes (ability, age, gender, race)

- Pre-academic experience (characteristics of school attended and academic achievement)
- 2. Students' prior dispositions and experiences influence their initial commitment to the university and goal to be successful.
- 3. This commitment or lack of commitment would in turn influence the way students' academically and socially integrate at university.
- 4. The extent to which students socially and academically integrate at university would subsequently affect commitment to the university and the goal to be academically successful.

Bourdieu's conceptual tools:

- Social, economic and cultural capital which is used by individuals during their own ongoing struggle to maintain and improve their own position in the university environment. Their families are the primary agents responsible for transfer of capital.
- 2. Habitus is described as a set of values, practices and norms which individuals assimilate as part of who they are and how they operate. It determines individuals' actions, aspirations and expectations.
- 3. Field is the social space where actions take place and individuals compete to protect or increase their capital.
- 4. Capital, habitus and field are relational and interact to produce social outcomes for individuals, groups and societies.

The justification for employing concepts from Bourdieu's (1984) and Tinto's (1975, 1993) theories in the conceptual framework model was to address a gap in the literature on the first-year experience. I sought to fill this gap by contributing empirical research to the first-year experience field by means of examining and understanding what, how and why certain pre-entry academic and non-academic factors, such as religious, racial, socio-economic and cultural diversity, influence students' first-year experience and academic performance. The gaps, clarity and significance of pre-entry factors that needed further investigation formed the focus of this study's conceptual framework. The framework provided a more holistic approach to fully explore pre-entry factors influencing the way students negotiate their first-year experience.

Taking into account Jabareen's (2009) main features mentioned earlier in Section 2.2 page 28, provided a structure and thinking tool for this study. It assisted with the development of an appropriate conceptual framework to investigate the phenomenon in the study. The conceptual framework provided a tool to guide this study to consider the significance of the range of pre-entry factors influencing students' first-year experience and academic performance. It provided sufficient opportunities to understand, analyse and interpret the factors contributing to the respective complexities identified. In addition, the conceptual framework with its integrative use of concepts from Bourdieu (1984) and Tinto's (1975, 1993) theories, formed an analytical and theoretical tool to investigate and answer the following research questions of this study:

Main RQ: How do pre-entry academic and non-academic factors influence teacher education students' first-year experience and academic performance?

The following sub-questions were identified:

- Sub-RQ1: Who are our first-year teacher education students?
- Sub-RQ1: What are the pre-entry academic and non-academic factors that firstyear teacher education students bring with them to higher education?
- Sub-RQ2: To what extent do pre-entry factors that students report influence their social and academic interaction in their first year of study?
- Sub-RQ3: To what extent do these factors contribute to the students' first-year experience and academic performance?

Figure 2:2 on page 50, which graphically represents the conceptual framework developed, identifies four sets of constructs that provide a more comprehensive array of influences on students' first-year experience and academic performance: students' habitus; students' different forms of capital; institutional context; and individual students' first-year experience.



Figure 2:2 A conceptual framework to explore influences on first-year students' experience and academic performance

The methodological approach that was used to understand the first two constructs, student habitus and capital, located under the pre-entry factors in Figure 2:2, was initially directed by a quantitative analysis to obtain an overview of the first-year students; this was followed by a qualitative analysis to elicit a more in-depth understanding of students' individual contexts. This type of analysis assisted in identifying patterns, trends and expectations of first-year students entering university. The next two constructs, the institutional context and the individual students' first-year experience, represented in the middle block of Figure 2:2 on page 50, was guided primarily by a qualitative analysis. The use of one-on-one interviews and the focus-group interview allowed me to understand how the identified pre-entry factors influenced students' first-year experience as it was located within an individualistic gaze.

This study's conceptual framework posits that students come to university with a range of academic and non-academic (social and personal) background characteristics and experiences, which in Figure 2:2 are referred to as student habitus and forms of capital. Students make use of varying levels of their different capitals' resources and habitus to negotiate their first-year university experience. Different experiences and characteristics from their habitus and capital resources available to them may positively and/or negatively influence their integration into higher education. These dispositions both prepare and dispose the students in varying degrees to integrate socially and academically in their new field of higher education. The more satisfying the interaction at university, the better the student would experience a sense of integration. When this 'fit' between the university community and the student experience is lacking, the student senses incongruence between his/her institution or academic community, and how he/she perceives him or herself. This incongruency Tinto (1987:53) refers to as isolation the 'absence of sufficient interactions whereby integration can be achieved'. Congruency and isolation will influence students' overall university experience and academic performance.

Therefore taking into account students' habitus and capital available to them will ensure a better understanding of their academic and social integration, which is necessary for understanding their sense of belonging or fit. Without this sense of congruence between students and their educational pursuits, it could be presumed that they are less likely to navigate successfully through the first year of university. A student's sense of belonging is linked to that sense of congruence. Thus, pre-entry factors, such as the environment in which individual students were raised, their forms of capital, and habitus, will influence their attitude, interpretation and ability to integrate into the higher education field. This in turn shapes students' first-year experience and academic performance and could determine whether the student decides to persist into the second year or drop out of university.

Finally, the conceptual framework which was developed, assisted in understanding how new students who bring with them their unique habitus and forms of capital that they have accumulated as part of their personal histories, including those associated with their family, schooling and socio-economic status, have influenced their experiences in the higher education field. The way students navigate their first year of university could either be in tension or congruence with their sense of self, their habitus and the capital that they bring to the higher education field. This study, with the use of the conceptual framework developed, investigates this issue.

2.6 Conclusion

Establishing a conceptual framework for this study enabled me to locate the study within a particular context and 'make sense' (Cohen et al., 2001:1), in a meaningful and manageable way, of teacher education students' first-year university experience and the pre-entry factors influencing their participation at university. The dual-theory approach which utilised Bourdieu's (1984) thinking tools and concepts from Tinto's (1975, 1993) integration model equipped me to identify gaps in our knowledge and postulate the existence of previously unknown phenomena in this particular study. It was through this integrated approach that I was able to explore:

- the relations between the field of HE: the habitus and forms of capital that the first-year students bring to this field and its influence in their integration into the field; and
- how the interactions within this field influence student behaviour, aspirations and the congruence of their first-year university experience and academic performance.

Bourdieu and Tinto's concepts formed this study's conceptual framework and the basis of a methodology which enabled me to undertake a more rigorous investigation of first-year students' actions and interactions in the education field.

The next chapter will present a review of the literature on the transition to university and the nature of the first-year experience.

CHAPTER 3 LITERATURE REVIEW

3.1 Introduction

This research study investigated teacher education students' first-year university experience and academic performance. The research question that was probed investigated how pre-entry academic and non-academic factors influence teacher education students' first-year experience and academic performance. In an effort to identify these factors and gain a clearer understanding of first-year experience in higher education, this chapter reviewed relevant empirical research and theories under the following headings to provide a holistic view of studies on the first-year experience globally:

- What constitutes the first-year experience?
- Significance of the first-year experience.
- Empirical studies on the first-year experience: retention, transition and academic success.
- Widening access and student diversity in first-year higher education.
- Factors influencing the first-year experience: academic factors, nonacademic factors and institutional factors.

3.2 What constitutes the first-year experience?

In trying to address the above question: What constitutes the first-year experience? Scott's (2012) traditional image of the first-year university experience provides a good synopsis. He describes it to be 'one of exciting intellectual and personal discoveries, independence in thought and behaviours, widening horizons, and growth in confidence' (Scott, 2012:17). Such an experience, Scott acknowledges, is a reality for some first-year students, but for the majority of first-year students, especially in the South African context, this experience is marred by failure, loss of confidence, and perhaps disillusionment.

Researching and addressing the phenomenon of the first-year experience has become an important focus in many countries primarily as the first year of study at university has been identified as a period which has a major impact on future academic success. Numerous national and international studies that focused on the first-year experience at higher education institutions indicate that there is no single first-year experience (Tinto, 1993, 1995, 2007; Krause et al., 2005; Pascarella & Terenzini, 2005; Harvey et al., 2006; Kuh et al., 2007; Yorke & Longden, 2008; Leibowitz et al., 2012; Levy & Earl, 2012; Scott, 2012). It is multi-dimensional and multi-layered, therefore making it impossible to cite one specific action, event or definition that can best describe what constitutes a first-year experience. However, what emerged from many of the studies on this phenomenon is that students' response to the first-year experience is based on a combination of academic and non-academic factors, internal and external to the university environment. Empirical studies reporting on students' responses to tertiary studies indicate that the term 'student experience' has a broader meaning which encompasses all aspects of student life (Harvey, 2004). In a ten-year national study of the first-year experience in Australian universities (Krause et al., 2005), the following key areas of the first-year experience are listed: aspirations of students; first-year transition to university life; balancing work, study and social commitments; engaging with groups on campus; career prospects; ease of access to resources; responses to courses students are studying; and satisfaction with teaching and delivery of materials.

3.3 Significance of the first-year experience

The rapidly increasing student population from a diverse range of backgrounds has heightened research into the experiences of first-year students both nationally and internationally (Krause et al., 2005). Why is the first-year experience at university more important than that of any other year? One possible response is that the first year at university is regarded as transitional. Some studies have identified this year as a period where cognitive, personal and social development takes place (Scott, 2012); a make-or-break for learning (Terenzini & Reason, 2005); playing a critical role in determining the likelihood of graduation (Upcraft et al., 2005); and a period that may affect the development of attitudes towards continuing learning at university and beyond (Hillman, 2005). Krause et al. (2005:9) designate the first year of university as the 'most critical time for engaging students with their learning community and equipping them with the requisite skills not only to persist, but to be successful and independent in their learning throughout their undergraduate years and beyond'.

The first year of study is a significant time to evaluate how prepared students are for academic studies and whether they need to be supported. Research on first-year student support indicates that academic support programmes and initiatives have contributed to an increase in student engagement and academic performance that has positively impacted on student retention (Kuh, 2001; Kuh et al., 2005; Smith, 2011). Research on student support indicates that orientation of first-year students is critical for integration, retention and determining an opportune time to address student support initiatives (Shupp, 2005). Mullendore and Banahan (2005:391) consider orientation to be 'about new beginnings for students, their families, and the institution'. These authors add that orientation provides a platform for the institutions 'to introduce a strong learning environment, build the foundations for academic success, welcome students and families to the campus community, promote student interactions with faculty and staff, and convey the values and traditions of the institution' (Mullendore & Banahan, 2005:391). During this period students connect with peers, staff and resources that play a key role in their academic success.

The first-year experience has been identified as a year in which the greatest amount of failure and attrition occurs (McInnis, 2001). In response to this first-year challenge, the student needs to be successfully integrated into the university environment. As Tinto (1995) asserts, being both socially and academically integrated into the university domain can increase academic success and persistence. According to McKenzie and Schweitzer (2001), lack of social integration and poor academic performance are strong predictors of attrition. Astin (1993) adds that the more active and involved students are, the more academically and socially integrated they become, leading to increased student satisfaction and greater willingness to remain in the institution and persist with their studies. Studies of student success and retention indicate that students who felt more connected to peers, the academic community or activities in the institution were more likely to stay and succeed (Tinto, 1988; Pascarella & Terenzini, 1991; Astin, 1993).

Gathering information on students' perceptions of higher education and their firstyear experiences is important as it provides vital input for first-year curriculum planners, administrators and lecturers that could be utilised to improve student retention and academic success (McInnis, James & Hartley, 2000). Hillman (2005) stresses that this information could be used to investigate course attrition and lifelong learning. Scott (2012:19) concurs that exploring the first-year experience could contribute to improving the quantity and quality of the country's outcomes. This has major implications for social, political and economic development, particularly in a context of scarcity of skills.

3.4 Empirical studies on the first-year experience

'First-year students can and will do better when placed in intentional, intellectual and social campus environments that challenge and support their efforts to succeed' (Upcraft et al., 2005:524).

Many of the key research initiatives on the first-year experience over the past 25 years have been based on the core belief expressed by Upcraft et al. (2005) above. The central importance of the first-year experience in higher education became an international concern when countries began to recognise the value of focusing on the first year of study to enhance academic success and retention (McKenzie & Schweitzer, 2001; Yorke & Longden, 2004; Kuh et al., 2005; Upcraft et al., 2005; Harvey et al., 2006). Examples of initiatives are:

- In 1986, the National Resource Center (NRC) in the USA launched its first international meeting for the first-year experience.
- In 1995, Queensland University of Technology in Australia hosted the first Pacific Rim Conference on the First-Year Experience in Higher Education (FYHE);
- In 2005 the European First-year Experience Conference was launched.
- In 2008 the first South African Conference on the First-Year Experience was inaugurated (Nutt & Calderon, 2009).

Much of the research on the first-year experience has been conducted in the context of retention, transition, preparedness and academic success (Harvey et al., 2006). In reviewing the literature on the first-year experience, it is evident that extensive research surrounding this phenomenon has been undertaken in the USA, UK and in Australia. Most of the American studies on the first-year experience have focused on academic and social integration to explore retention and related issues (Tinto, 1975,

1993, 2007; Kuh, 2001; Pascarella & Terenzini, 2005; Reason et al., 2006; Kuh et al., 2007). UK studies have focused mainly on first-year preparedness and student satisfaction to explain retention and withdrawal (Thomas, 2002; Yorke & Longden, 2004; Harvey et al., 2006; Noyes, 2007-2008; Russell, 2008). In comparison, the Australian focus on the first-year experience was expressed in efforts to enhance student success and reduce attrition (James et al., 1999; McInnis, 2003; Hillman, 2005; Kift & Nelson, 2005; Krause, 2006). McInnis, James and Hartley's (2000) study provided an exploration of first-year trends across universities in Australia which built a picture of the overall character and quality of the first-year experience. A later study by Krause et al. (2005) provided a more detailed insight into the firstyear experience of Australian undergraduate students. They noted that there was a significant increase in the diversity of the university student population and an awareness of its importance to academic achievement and retention. In South Africa, the focus on extended degree programmes and student support led many researchers to associate the first-year experience with 'struggling students' (Leibowitz et al., 2012); therefore much of the research on the first-year experience in South Africa focused on student support and academic development programmes (Fraser & Killen, 2003; Bitzer, 2005; Scott et al., 2007; Steenkamp et al., 2009; Levy & Earl, 2012; Scott, 2012).

Empirical research on the first-year experience completed both nationally and internationally, ranged from small samples of students in a single course, department or faculty to substantial samples that involved the entire institution. Some studies ranged from a single institution to longitudinal multi-institutional studies (Harvey et al., 2006). In the South African context much of the research on the first-year experience has been quantitative and from the perspective of the institution (Scott et al., 2007; CHE, 2013). There has been a wide range of studies on the first-year experience. The focus of these studies varies from small-scale studies focusing on students' learning outcomes and development, to broader, large-scale studies focusing on student success, transition, retention and support; however, there do seem to be recurring themes that emerge. In this regard, the next sub-section of the study will focus on literature reviewed under three broad areas of the first-year experience: retention, transition, and preparedness.
Nutt and Calderon (2009) note that studies of the first-year experience evince a real concern for students and a desire to see them succeed. They add that a considerable number of these studies seem to have a limited focus on the first-year experience from the students' perspective. In addition, fewer studies explore the students' prior background influences on their first-year experience in higher education. This is evident in many of the retention and persistence studies that use models to explain higher education student outcomes. Bean (1980), Pascarella and Terenzini (1983) and Tinto's (1988), models view background characteristics as an external factor that influences student outcomes within institutions and ultimately students' decisions to persist or drop out. However, Stage and Hossler (2000) voice their concern that background characteristics have not been fully explored and still remain unexplained. This study attempts to contribute to this area of contestation by foregrounding student perspectives on the pre-entry academic and non-academic factors influencing their first-year experience and academic performance.

3.4.1 First-year retention

Research evidence from across the globe indicates unequivocally that first-year retention is the key to successful degree completion (Tinto, 1993; Kuh, 2001; Krause et al., 2005; Pascarella & Terenzini, 2005; Upcraft et al., 2005; Yorke & Longden, 2008; Van der Meer et al., 2010). There have been many studies that set out to investigate either one single factor or multiple factors influencing the first-year experience and retention, but it is clear that the elements contributing to the first-year experience and retention are complex.

Of considerable influence on retention studies was the work of Tinto (1993). He focuses on explaining student withdrawal by taking into account student levels of commitment to their goals and the institution. This led to Tinto's conceptualisation of first-year engagement. The understanding of student engagement in learning outcomes and persistence was explored by Pascarella and Terenzini (1991) and Kuh (2001). Other retention studies explored the role of individual student characteristics in understanding persistence (Dey & Astin, 1989). Yorke's (1999) study highlights the complexity of withdrawal patterns among students and emphasises the role of institutions in understanding and responding to student withdrawal. Yorke and Longden's (2004) project that focused on first-year fulltime

undergraduate students' experiences across institutions in Wales and England, evinced a better understanding of student withdrawal and highlighted four reasons for students discontinuing their studies:

- Unsatisfactory university experience
- Inability to cope with academic demands
- Wrong choice of course
- Personal factors such as family circumstances, finance, illness, loneliness, etc.

Empirical studies on the relations between the institutional environment and first-year retention highlight the key role that the campus environment plays in students' decisions to remain or withdraw. Many of these studies focus on student satisfaction and social/academic integration in relation to student retention. For example, studies on student satisfaction reveal that students' decisions to withdraw from campus are often linked to their dissatisfaction, lack of motivation, or adjustment to the campus environment (Nes et al., 2009). This could be due to poor service, quality of experience, unfulfilled expectations or incorrect course choice (McInnis, Hartley, Polesol & Teese, 2000; Yorke, 2000). In Fraser and Killen's (2003) study at a university in South Africa, first-year students identified the following five factors for their failure: inadequate preparation for academic work; boring presentation by lecturers; irregular attendance at classes or tutorials; lack of self-discipline; and poor examination techniques. Assistance and support given to students by the institutions in their first year of study could help increase their chances of success.

3.4.2 First-year transition

Wahr et al. (2009:434) define student transition as 'the process students go through to become ready to engage in learning as they move into new educational contexts, for example, school-leaver to first-year university student; vocational program to higher education degree; education in one cultural and language setting to another'. Thus first-year transition could be regarded as a period of adjustment and change.

Much of what is known about student transition in higher education has been gathered over the past 50 years by international scholars (Harvey et al., 2006).

Hooper (2005:91) highlights that many of the earlier studies on student transition experiences were underpinned by an institutional agenda to minimise casualties and costs in a bid to maximise outcomes and efficiency. Levy and Earl (2012) acknowledge that in recent studies there has been a significant change in focus: students, rather than academic staff, have been repositioned at the centre of the university experience and purpose. Levy and Earl (2012:xiii) add that in so doing, considerations and representations of the university experience from the students' perspective have become a vital part of gaining valuable information on how more students, institutions and societies are qualitatively and quantitatively beneficial. This focus is evident in many of the recent studies on student transition (McInnis, James & Hartley, 2000; Perry & Allard, 2003; Kift & Nelson, 2005; Krause, 2006; Wahr et al., 2009; Smith, 2011; Levy & Earl, 2012).

Studies undertaken on the first-year experience and transition indicate that many students encounter difficulties in moving from school or work to university (Terenzini, 1993; McInnis, James & Hartley, 2000; Scanlon et al., 2007). Some of the challenges identified were: lack of connectedness and involvement (Perry & Allard, 2003); difficulties in developing a student identity (Scanlon et al., 2007; Briggs et al., 2012); unhappiness and dissatisfaction (Yorke, 2000); and finally, feeling lonely and isolated (Pargetter, 2000; Lawrence, 2003). A factor creating a challenge for firstyear transition is the change in the identity of a 'typical university student'. This was evident in McInnis and James's (1995) study, which revealed that a university student's identity developed at a slow pace. In the study of Scanlon et al. (2007), the slow development of a student identity was largely due to the increase in diversity of the student population: mature-aged students, foreign students, first-generation students and also students with part-time employment. They add that this diversity has influenced students' feelings of connectedness to the university which have impacted on their identity development and ultimately their decision to stay at or leave university.

Regardless of the many difficulties experienced by students in their first-year transition, there are many studies that reveal that student transition is not all negative. For example, national studies in Australia revealed that the majority of first-

year students enjoyed being university students; had positive experiences most of the time; and felt happy at university (McInnis & James, 1995; McInnis, James & Hartley, 2000; Hillman, 2005).

Many researchers have used a range of theoretical perspectives to understand or explain student transition to university better. Two frequently used and quoted theoretical perspectives are Tinto's (1988; 1993) application of Van Gennep's transitional model on 'rites of passage' and Bourdieu's (1973) concept of cultural capital (Scanlon et al., 2007). Tinto (1988) adapts Van Gennep's three kinds of rituals, separation, transition, and incorporation, to explain that for students to be successfully integrated into university life they need first to go through a period of separation from their past associations, which includes their families. He argues that only those students fully incorporated socially and academically into the university environment feel connected and persist in their studies.

Bourdieu's (1973) concept of cultural capital has been used extensively in student transition studies. These studies highlight students' cultural capital as crucial in shaping their decisions in respect of the school to university transition. According to Pitzalis et al. (2008), cultural capital, which is linked to social origins, is considered the variable that mainly influences the process of transition from high school to university and is used as a predictor of educational achievement in different theoretical frames. This concept is applied to educational institutions that produce the values, attitudes and dispositions of the dominant group in society. Bourdieu argued that schools whose students traditionally make the transition to university would be expected to transmit cultural capital to students as a necessary prerequisite for transition to university (Scanlon et al., 2007). Owing to the diversity in students and their educational backgrounds, not all students enter university with the dominant cultural capital, thereby creating a relative disadvantage for some students to fit into the university environment.

Much of the literature on transition studies reveals that for students to perform a successful transition, they need to develop a sense of belonging and connection with new peer groups and the wider academic community. Perry and Allard (2003) further claim that students need to make connections between the experiences they bring with them to the new environment and their new knowledge obtained at university.

Students are not alone in this process; Wilson asserts that what has become apparent at universities is that supporting and facilitating effective transitions into and throughout higher education has turned out to be increasingly student orientated, professional and accepted as institutional core business (Wilson, 2009: 1 in Levy & Earl, 2012). Universities now recognise their core role in supporting the successful transition of first-year students to improve graduation rates.

Pym et al. (2011) suggest that when providing support to first-year students, universities need to move away from a deficit model to a more value-added model that focuses on harnessing students' agency and actively nurturing their social connectedness and sense of community to enhance first-year academic performance. They add that this type of intervention, especially in the South African higher education context, takes into account students' range of diverse histories, experiences and context to develop a supportive community that attempts to promote a sense of belonging. The aim of the study of Pym et al. (2011) in implementing a value-added model to enhance academic performance was to reduce alienation and demoralisation among students who negotiate their transition into university.

3.4.3 First-year student preparedness

The preparedness of first-year students for higher education can be regarded as an important factor in determining first-year retention and academic success. There is a growing concern about the under-preparedness of students and the effectiveness of the skills they bring with them to university (Nutt & Calderon, 2009). Studies over the last 30 years indicate that student under-preparedness is a complex phenomenon which manifests itself in a range of ways, for example, from struggling with the formal curriculum to difficulties in adjusting to independent study and a new academic environment (CHE, 2013).

Taylor and Bedford's (2004) study notes that university academics regard students' level of preparedness, motivation and academic ability to be significant factors for university completion or non-completion. Entering students' level of preparedness may be influenced by factors such as high school final exit results (Grade 12 results); quality of primary and secondary education; language barriers; and socio-economic

status; however many national and international higher education institutions rely solely on students' final Grade 12/matriculation results. The validity of South African students' Grade 12 results has raised major concerns about university success rates, especially among first-year students. Leibowitz et al. (2012) observes that many first-year students who arrive at higher education institutions with high expectations have their hopes for academic success ruined before their first academic year is over. Students' Grade 12 results seem to provide unrealistic expectations with regard to higher education studies.

In this regard there has been a trend in South African higher education institutions to explore alternative ways of assessing students' preparedness for university studies. Several assessment tests such as the National Benchmark Test (NBT) have been implemented. For example, the NBT reports have shown that high school graduates are not sufficiently prepared for the challenges of higher education. Case et al. (2013) report that only a small minority of prospective students obtain NBT results which indicate that they are sufficiently prepared for mainstream university study. In addition, Pym et al. (2011) contend there is a clear relation between students' high school matriculation results and their university entrance tests when students' academic success is scrutinised systematically.

Fraser and Killen (2003) add that universities in South Africa make use of the entering students' Grade 12 results to identify at-risk students and offer appropriate academic support in their first year of study. The NBT results are used to place students in extended curriculum programmes at universities.

Further considerations in student preparedness are students' socio-economic circumstances and educational background. These factors play a major role in determining students' subsequent decisions to further their studies and their choice of institution. Students' personal circumstances and the area in which they live influence the quality of school education they receive and the knowledge and skills they bring with them into higher education. In the South African context, the disadvantaged educational background due to the apartheid past has contributed negatively to the achievement of poor students in higher education. These students, labelled disadvantaged students, according to Huysamen (2000), attended historically 'black' schools that experienced unequal education systems, resources,

64

quality of teachers and infrastructure. Fraser and Killen (2003) add that the quality of school attended has the power to determine the type of tertiary institution that students attend, assuming they were given the opportunity to attend university. A second distinctive characteristic of disadvantaged students as identified by Woollacott and Henning (2004) is that most of these students are likely to have English as a second or third language and are generally under-prepared for universities in South Africa, where most have English as the medium of instruction.

Many of these key factors identified above influence the way students experience their university education. These are factors which many university academics believe are beyond their control, which leads to Jansen and Van der Meer's (2012:10) relevant question, 'Who prepares students for academic studies: high school or university?' Scott (2012:23) asks a similar question: 'Where does responsibility for improving student performance lie?' In Yeld's (2013:5) article, 'Time to stop blaming schools', she cautions that higher education institutions' response to high failure and dropout rates is that under-preparedness is the responsibility of schools. Yeld claims this apportioning of blame is misguided. The responsibility, she avers, should be higher education's focus, and urges that higher education needs to start where schooling has left off as confirmed in Jansen and Van der Meer's (2012) study. They examine high school students' perceived preparedness for university study. Jansen and Van der Meer's study was conducted between two secondary education systems in two countries. Students from the Netherlands' school system that specifically trains students for university entrance were compared with students in the New Zealand school system who follow a broad-based general curriculum. Their results reveal no significant difference between learners in the two different school systems with regard to their self-perceived readiness for university. The authors suggest that, rather than reforming secondary education, the first year in higher education should be reformed to prepare students for university.

Studies in South Africa (Slonimsky & Shalem, 2005; Scott et al., 2007) indicate that the South African school system, with its continuing educational inequalities and worst knowledge practices due to apartheid education and embedded socioeconomic conditions, is the primary cause of students' under-preparedness for higher education. Scott (2012), acknowledges that, in spite of the above statement and universities commonly disclaiming responsibility for academic performance problems in higher education, 'it is necessary on pragmatic grounds for the higher education sector to identify factors affecting student performance that are within its control, and to act on these to the best of its ability' (Scott, 2012:24). This strand of thinking was highlighted by other authors of retention studies (Braxton et al., 2004; Tinto, 2007; Kuh et al., 2008) when they suggest that universities should play a more active role in reducing attrition. A suggestion offered by Jansen and Van der Meer (2012) is to utilise a pedagogical practice at a practical level and a scaffolding approach to develop the academic skills needed by first-year students so that they can gain early opportunities to academic success.

3.5 Widening access and diversity

In South Africa, the transformation goals relating to higher education are to achieve a transformed, democratic, non-racial and non-sexist system (DoE, 1997). Specific outcomes sought include:

- increasing and broadening participation to reach a wider distribution of social groups and classes, including adult learners, thereby accommodating a larger and more diverse student population (DoE, 1997: 1.13); and
- the promotion of equity of access and fair chances of success to all (DoE, 1997: 1.14).

The widening of access has resulted in an even greater number of students, from a broad range of diverse socio-economic and academic backgrounds, entering higher education.

The South African government's objective for increased and more diversified higher education participation is expressed in both the Green Paper for Post-School Education and Training (DHET, 2012a) and the National Development Plan 2030 (NPC, 2012). The Council on Higher Education (CHE) in South Africa stresses that representation among higher education graduates, in terms of race, gender and social class, is essential for equity, developing talent across the population, setting a balanced development agenda, and maintaining the public's respect and faith in higher education (CHE, 2013:32). In addition, the objective for more diverse higher

education participation by the South African government will ensure that no individuals are excluded from society based on race, gender and social class; and more importantly that the dominant hegemonic power structure that developed during the apartheid era would be dismantled and democratised.

Although female representation at higher education institutions has increased to 58% and black (African, coloured and Asian/Indian) enrolment has increased to 78% (DHET, 2013), Scott et al. (2007) highlight that participation rates in higher education still remain highly disproportionate. For example, the proportion of African students in higher education has increased from 49% in 1995 to 66% in 2011 (CHE, 2013; DHET, 2013), but if one takes into consideration South Africa's racial composition (African, coloured, Asian/Indian and white) 79% of the population constitutes Africans; the participation rate of only 12% for African students in higher education is considered therefore as being under-represented, while white students seem to be over-represented with a participation rate of 54% (CHE, 2009). This racial imbalance is reflected in undergraduate graduation rates which have major implications for correcting employment equity, income and development inequalities.

The Council on Higher Education, the Green Paper on Higher Education and other educational policies stress the need for graduates to be representative of the country's population to meet the changing demands of the South African society and economy (DHET, 2012a; CHE, 2013:32). Concerns of access and diversity in higher education seem to structure representation around the issues of race, gender and social class; many reports and studies on higher education mainly focus on race and gender. The notion of social class which may be more difficult to define is largely omitted or, as noted in many reports and studies, is included in race and gender. There is an increasing debate on the notion of social class as a structure other than race. For example, a University of Cape Town research paper, states that social class in higher education has largely been neglected and social scientists have been silent with regard to the issue of whether or not class is important in the 'new' South Africa (Seekings, 2003:6). Seekings (2003) states that in the 'new' South Africa, class inequalities are highly visible. However, emphasis on race, especially in official statistics, serves to obscure even the possibility of collecting data on other criteria, such as class. Chetty (2012) argues that, unlike race, social class is not constitutionally protected. He adds 'there is no guarantee of equal opportunity on the basis of poverty and no political mobilisation on the part of poor, in contrast to mobilisation activities of race pre-1994' (Chetty, 2012:14). Therefore, in examining the objective for increased access and diversity and to meet the policy goals of a justifiable representation in higher education, all three issues, race, gender and social class, must be considered in an integrated way.

Importantly, such an objective is not unique to South Africa. There seems to be a global trend to increase the education of knowledge workers for 21st century economies (Levy & Earl, 2012). The Australian Government has committed itself to increase access to higher education by increasing participation by 20%, specifically from low socio-economic status (SES) groups by 2020. The Australian Government has targeted an additional 217 000 graduates by the year 2025 who would more accurately reflect Australia's SES demographics (Australian Government, 2009:12-13). The South African government has set a similar target, to increase the participation rate to 23% by 2030 (DHET, 2013). Since South Africa's participation rate has been steadily increasing from 761 090 in 2007 to 900 000 in 2011, the Green Paper calls for total university enrolment to increase to 1.5 million by 2030 (DHET, 2012a:x; CHE, 2013:41).

Consequently, the widening of access and diversity in higher education both nationally and internationally has brought with it a host of new challenges for these institutions (Wagner, 2000; Fraser & Killen, 2003; McInnis, 2003). McInnis and James (1995) warn that many universities and academics are not sufficiently aware of the increasing diversity of the student population or how this may impact on their institution and/or staff. They caution that institutions may have to deal with higher dropout rates and low throughput and graduation rates which will have financial implications. This has an impact on academic staff, as they will have to accommodate students who need additional academic support. This implies that lecturers will have to become more creative and innovative in their teaching and planning of programmes for a more diverse student population. This results in financial and planning constraints as more time and resources are required for staff to prepare adequately for the diverse student population.

The greatest academic challenge will be experienced in the first-year programmes. McInnis (2003) cautions that the increasingly diverse student population makes it difficult to identify the profile of a typical first-year experience. This challenge highlights the need for higher education institutions to profile the first-year students accurately, in order to estimate and respond to the diversity experienced within the rapidly changing mass higher education system (James et al., 1999). Smit (2012) suggests that to respond to the diversity in student body requires a change in thinking:

We need to thoughtfully consider the readiness of higher education institutions to respond to students, and cultivate the will to learn in our students. We need to find ways to research the full texture of the student experience and to value the prehigher education context of students (Smit, 2012:369).

Many studies internationally have attempted to understand higher education diversity and the changing profiles of students focused on the following cohort of students: first in family (first-generation students); mature students (over 20 years) entering higher education; fulltime students employed while studying; and experiences of students from low socio-economic backgrounds (Tinto, 1993; McInnis & James, 1995; Yorke, 1999; McInnis, James & Hartley, 2000; James, 2002b; Terezini & Reason, 2005). Many of these studies acknowledge the importance of understanding student diversity and the influence it has on academic outcomes.

Harvey et al. (2006) attribute first-year persistence and success to the diverse nature of students' prior experience and expectations. They conclude that a composite from an intricate mixture of students' characteristics, external and institutional factors, contribute to persistence. Fraser and Killen (2003) add that this diverse nature could have either a negative or positive influence on students' first-year experience which would ultimately contribute to students' decisions to persist or leave the programme of study. Although Gregory (2012) contends there is limited African research on the issues of student persistence, there are many Western persistence models that could be used to understand the phenomenon in African universities (Tinto, 1975; Pascarella & Terenzini, 2005). In Tinto's (1975) model of student integration, he advocates that the fit between the student and the institution plays a key role in the likelihood of persistence. Tinto adds that the degree of institutional commitment a student feels and the subsequent persistence is shaped by congruency between the

student's motivation and ability, and the institution's academic and social characteristics. Yorke and Longden (2008) reveal that understanding persistence is multifaceted. They caution that there are more complex factors that influence student persistence, some of which are external to the university environment.

3.6 Factors influencing the first-year experience

There is a large body of research on factors influencing the first-year experience in higher education (Thomas, 2002; Fraser & Killen, 2003; ACT, 2007; Yorke & Longden, 2008; Jansen & Suhre, 2011; Cleyle & Philpott, 2012). In some studies these factors are referred to as cognitive and non-cognitive (Sedlacek, 2004, 2011), while other literature refers to them as academic or non-academic (Tinto, 1993; Nora et al., 1996; Lotkowski et al., 2004; Pascarella & Terenzini, 2005; Miller, 2013). Regardless of the distinction, the majority of the variables researched under factors influencing first-year experience include levels of social and academic integration; pre-college ability; finances; institutional support; psychological and attitudinal factors; level of parents' education; socio-economic status and the level of family support (Tinto, 1993; Pascarella & Terenzini, 2005). Some studies focus on a single factor influencing students' first-year experience, others pay attention to a multitude of factors, thus highlighting that it is difficult to single out one particular factor that may influence student experience and academic success.

Mansfield et al. (2009:502-503) suggest that gaining a better understanding of factors contributing to the first-year experience and academic success will result in a greater ability to offer positive and timely interventions to improve first-year students' grades and their successful completion of higher education.

There have been many theorists who have developed models to understand a wide range of factors presumed to influence the first-year experience. Many of these models acknowledge pre-entry variables in understanding students' activities and outcomes at university: for example, Tinto's (1975) student integration model; Astin's (1985, 1993) Input-Environment-Output (I-O-E) model; Bean's (1980) attrition model; Weidman's (1989) student socialisation model and Berger and Milem's (2000) college impact model. These models make reference to background characteristics that could be divided into three categories: psychological, academic and

demographic. Astin's (1993) I-O-E model foregrounds the relation between the environment (activities that students engage in at university) and outcomes (the desired aim and objective of the programme enrolled in). Astin's model cautions that the first-year experience cannot be understood without taking into account the students' own input: the personal qualities that they bring with them to the university. Tinto's (1975) student integration model highlights the interrelated effect of students' family background, individual attributes and pre-entry schooling and how these influence students' goals, together with the role of institutional commitment. What is clear from the empirical studies using these models is that there is no single factor that can be identified to have influenced students' first-year experience above others. Influential factors can have a different outcome at different institutions depending on the institutional climate, course of study, changing diversity of student population and other internal and external variables (Harvey et al., 2006; Yanto et al., 2011). Harvey et al. (2006) maintain that the first-year experience is a shifting target that evolves and changes with time and culture. It requires research and practice, and continuous re-evaluation is required for relevance and enhancement. They caution that researchers should not be complacent that once the first year has been attended to, later experiences can be ignored.

The sub-section below considers factors influencing the first-year experience under three broad headings: academic factors, non-academic factors, and institutional factors. These factors are considered separately; however it is important to stress that these factors are inter-related, complex and often overlap as noted earlier in this chapter.

3.6.1 Academic factors

Academic and non-academic factors play an important role in helping students to achieve academic success. McKenzie et al. (2004) note there has been extensive research that supports the relations between prior academic performance and university performance. ACT (2007) reveals that the strongest predictors of university persistence and degree attainment are prior academic achievement and course selection. Other studies have shown that prior academic performance is similarly a good predictor of first-year students persisting to their second year of study (Nora et al., 1996; Pascarella & Terenzini, 2005). In McKenzie and

Schweitzer's (2001) investigation into the factors influencing academic success, they found that while integration into university, self-efficacy and employment responsibilities were predictive of university grades, it was the students' previous pre-tertiary academic performance that was the most significant and reliable predictor of university performance. There have been mixed responses to what is a more accurate predictor of academic performance and academic success at university. Sedlacek (2011) notes in recent literature that grades are becoming less useful as indicators of student achievement or predictors of future student success in higher education. He maintains that high school grades no longer appear as useful in differentiating student academic achievement as they once were. Sedlacek (2011:88) attributes this to what he terms 'grade inflation'. In South Africa there is limited evidence of the predictive validity of the National Senior Certificate (NSC) examination results. These are not necessarily the best predictor for academic success at tertiary institutions, although they are the primary determinant for university entrance (Fraser & Killen, 2003). Jansen (2012) cautions that the levels set for passing the NSC examination are so low (30% in some subjects and 40% in others), that students who receive below 70% in the NSC examination should seriously consider whether to enrol in higher education studies. He adds that many South African universities have thus opted to administer alternative admission examinations to gain a more realistic measure of the potential of students entering their institutions.

For Terenzini and Pascarella (1978), prior academic performance, as a predictor of university performance and attrition, contributes only a small percentage: for them a more significant predictor is the students' academic and social integration at university. Malefo's (2000) study of black students at a predominately white university in South Africa indicates that variation in students' academic performance could be attributed more to the students' background variables than to cognitive factors. Sedlacek (2011) differs and demonstrates that these predictions could be improved by adding non-cognitive variables, such as students' perceptions, adjustment and motivation. He adds that cognitive and non-cognitive variables would give a more complete picture of applicants' abilities and would increase the diversity of entering students to include students from different races and those with less traditional backgrounds.

A second, related issue to prior academic performance is understanding entering students' educational experiences, since these play an important role in university achievement. In South Africa, the inequalities in the public school system, mainly due to inadequate resources, infrastructure, quality of teachers and socio-economic conditions of students, have contributed to the 'generally poor and racially skewed performance in higher education' (CHE, 2013). Thus Berger (2000) stresses that the key to understanding student retention better and shaping future chances of persistence would require universities to understand the role played by students' prior educational experiences from a social reproductive perspective. He states that prior school experiences contribute to the individual accumulation of cultural capital of students which has an 'effect on whether or not students are likely to attend university and if so, an effect on where students choose to go and how likely they are to persist' (Berger, 2000:102).

3.6.2 Non-academic factors

It is clear that predicting academic performance and student retention in the first year of study is not simply a matter of investigating academic factors. There is ample evidence that factors such as motivation, attitudes, goal attainment, family education, soci-economic status, income, student identity and student demographics are likely to influence student success at university (Tinto, 1975; Stage & Hossler, 2000; McKenzie & Schweitzer, 2001; Thomas, 2002; Harvey et al., 2006). In the South African higher education context there have been few studies of socio-cultural and psychological factors impacting on first-year academic performance (Pym et al., 2011).

Research on personal factors such as motivation, personal interest, confidence, efficacy and self-esteem provides evidence that suggests that relationships exist between such variables and university academic performance (Kim et al., 2010). Goals and commitment are important factors concerning student retention.

Cleyle and Philpott (2012) indicate that numerous studies have demonstrated a connection between pre-enrolment educational goals and persistence. This view is evident in Tinto's (1975) theory where he acknowledges the importance of pre-entry educational goals and states that the level of one's commitment to one's educational

goals is directly related to persistence. Bean and Metzner (1985) discuss educational goals as a background variable in their model of non-traditional student attrition.

Students' life goals influence their commitment to their studies. Toni and Olivier's (2004) study on first-year black female students in a predominately white university in South Africa notes that these students' identities at university are shaped by their life goals, which are ignited by specific feelings and characteristics using their past experiences. Toni and Olivier (2004:194) suggest that lecturers need to understand that 'past experiences' form the root of a student's beliefs, sense of security, confidence, self-esteem and identity: such factors contribute to students' educational goal achievement and academic success.

Another important factor that generates contrasting views is financial assistance. The National Audit Office in the United Kingdom (NAO, 2007) identifies financial assistance is not a major factor in student persistence: it is not the sole reason why students withdraw from university. There are many studies that reveal that students are involved in paid work, but little evidence shows that paid work reduces students' levels of academic achievement (Fjortoft, 1995; Light, 2001; NAO, 2007). Zimitat (2003) explored the impact of employment and family commitments on the first-year experiences of three different groups of fulltime students:

- Group 1: fulltime students not in paid employment.
- Group 2: fulltime students in fulltime paid employment, with few family responsibilities.
- Group 3: fulltime students in fulltime paid employment who are primary income earners and primary carers in their household.

The findings revealed that there were significant differences in the experiences and behaviours of Group 2 students compared with Group 1 and 3 peers. Group 2 students spent less time on campus, found lecturers less stimulating and valuable, experienced more difficulty with motivation to study, and spent less time preparing for and attending scheduled teaching activities. These students were therefore significantly more likely to consider dropping out of campus. Zimitat concludes that the support of family and university friends appears to be a critical factor in their persistence at university. Cabrera et al. (1992) found that financial assistance

created an equal playing field among those students who were financially advantaged and those who were not.

Another important factor influencing academic performance is parental support and commitment. Studies on the relations between parental support and university adjustment have been limited, with the majority of studies quantitative in nature (Mounts et al., 2006; Yazedjian et al., 2007). Much of the research on student parental support and commitment while at university has been inconsistent. Some studies found positive relations between parental support and university adjustment (Martin, et al., 1999), while other studies found no significant relations between the two variables (Dennis, et al., 2005). For example, in Nora and Cabrera's (1996) study on minority students, although minority students had perceptions of discrimination and prejudice at university, this negative influence was negated by the that their families students' perceptions were supportive and provided encouragement while they were at university. Several studies found that parental education and encouragement to attend college or university were strongly associated with students' intention to attend such institutions (Stage & Hossler, 2000). In Stage and Rushin's (1993) study, parental encouragement was regarded as a strong predictor of persistence and it further influenced other factors such as educational goals and institutional commitment which had a positive influence on students' studies.

Perna and Titus's (2005) study highlights a different focus on parental influence and university adjustment. They show that students whose parents did not attend university tend to arrive at university less prepared than their peers whose parents attended university. The main contributory factor to this underpreparedness was that their parents might not be familiar enough with the university's expectations to prepare children adequately for first-year university studies. Student persistence literature reveals that parents' higher education levels and incomes are strongly related to successful involvement in university and indirectly to persistence (Astin, 1975; Tinto, 1987).

3.6.3 Institutional factors

Institutional factors can contribute to the type of first-year experience encountered by students which can ultimately impact on their decision to either stay at or exit from university. Yorke (1999) draws attention to students' prior expectations of an institution. He adds that if there is a mismatch between a student's expectations and the reality of an institution, this would contribute to dissatisfaction with the student's first-year experience and hence his/her decision to withdraw. In addition, a student's perception of the institution's commitment and support, and the ease in which the student can navigate the internal operations of the university, would have a direct impact on their willingness to persist in their studies (Thomas & Yorke, 2003). A similar finding was found in Fraser and Killen's study on post-enrolment factors that students viewed 'a better understanding of the mechanisms and functions of the institution' would contribute to their success in the institutions and programmes of choice (Fraser & Killen, 2003:262).

Cleyle and Philpott (2012) acknowledge that there are many factors within the institution that could be instrumental in influencing students' success and retention: access to information and advice; connectivity to campus services; availability of courses; the quality of the offering of these courses; quality of programmes; and the quality of instruction and guidance offered. Kuh (2001) adds that the culture of the institution undoubtedly also influences student persistence. Thomas reveals that if students feel that they do not fit in, that their social and cultural practices are inappropriate and that their tacit knowledge is undervalued, they may be more inclined to withdraw early in the first year of study (Thomas, 2002:431). Thomas argues that if an institutional habitus is inclusive and accepting of differences, and does not prioritise one set of characteristics, but rather celebrates diversity and differences, this would allow students from diverse backgrounds to find greater acceptance of and respect for their own knowledge, which will ultimately promote persistence in higher education (Thomas, 2002:432).

3.7 Conclusion

This chapter revealed that the body of literature pertaining to the first-year experience is extensive. The nature of the research questions outlined in Chapter 1 necessitated a wide coverage of literature on the first-year experience in higher education. In addition, the complexity of what constitutes the first-year experience dictated that the literature be surveyed as widely as possible. Pertinent issues, such as the significance of the first-year experience, empirical studies and theories on the first-year experience, widening access and diversity in higher education, and factors influencing the first-year experience were reviewed and discussed in this chapter.

The literature reviewed in this chapter acknowledges that the first-year experience is multi-dimensional and multi-layered, making it difficult to cite one specific action to describe what constitutes a first-year experience. The literature reveals that the first-year experience is significant. It represents a year in which a student's cognitive, social and personal development crystallises; it is a 'make or break' year for academic learning; a year to front-load academic support; a year in which developing attitudes towards life-long learning are established, and finally a year in which successful social and academic integration can impact positively on academic success and retention.

The review of literature in this chapter covered empirical studies of the first-year experience: most of the studies have been conducted in the context of first-year retention, transition, preparedness, and academic success. The literature reviewed discloses many of the empirical studies have been completed extensively in the USA and UK and in Australia. Empirical studies in South Africa have been limited primarily to first-year student support. A prominent feature identified when reviewing literature on the first-year experience was that many of the theories used by researchers to underpin their studies on the first-year experience were those of Tinto's (1975) student integration model; Pascarella and Terenzini's (1991) student engagement; Tinto's (1988) adaptation of Van Gennep's (1960) transition model, and Bourdieu's (1973) concept of cultural capital.

With regard to literature reviewed on factors influencing the first-year experience, it is evident that there is a large body of research identifying a multitude of factors which define students' first-year experience. There is no one factor that could be identified as influencing students' first-year experience: the influential factors could have different outcomes at different institutions depending on the institutional climate, course of study, diversity of students, and internal and external variables.

Finally the first-year experience is identified as a shifting target that evolves and changes with time and culture and thus has to be continuously researched and evaluated, while practices need to be updated to be relevant and enhanced. The next chapter, Chapter 4, outlines the appropriate research design adapted for this study.

CHAPTER 4 RESEARCH METHODOLOGY

4.1 Introduction

Sahu (2013:2) explains that research is 're-search', in which 're' means again and again, and 'search' means a voyage of knowledge, thus emphasising that research engenders an original contribution to an existing stock of knowledge. Kothari (2006) notes that research could be defined as a scientific and systematic search for pertinent information on a specific topic. Thus the research undertaken in this study investigates pre-entry academic and non-academic factors influencing the first-year experience and academic performance of BEd undergraduate students at a university of technology.

The over-arching aim of this study was twofold. First, to identify and explore preentry factors influencing the first-year experience and academic performance of teacher education students and, second, to identify and propose strategies for firstyear student support.

The objectives of the research as explained in Chapter 1 are to:

- gain a better understanding of who the first-year students are who enter teacher education programmes;
- gain familiarity with specific pre-entry factors that influence teacher education students' first-year experience;
- ascertain how these factors influence students' first-year academic and social integration and academic performance; and
- propose strategies to enhance the quality of students' first-year experience in higher education.

The research questions that guided this study's methodology were:

Main RQ: How do pre-entry academic and non-academic factors influence teacher education students' first-year experience and academic performance?

The following sub-questions were identified:

Sub-RQ1: Who are our first-year teacher education students?

- Sub-RQ2: What are the pre-entry academic and non-academic factors that firstyear teacher education students bring with them to higher education?
- Sub-RQ3: To what extent do pre-entry factors that students report influence their social and academic interaction in their first year of study?
- Sub-RQ4: To what extent do these factors contribute to the students' first-year experience and academic performance?

The next sections describe the methodology or logical sequence that was followed to investigate the research phenomenon. Kothari (2006) explains that the scope of a study's research methodology is much wider than that of the research methods. A research methodology includes the theoretical perspective (Bogdan & Biklen, 2007) and considers the general logic behind the specific techniques, collection and analysis methods used in the context of the study. Thus the alignment of the research methodology with this study's research aims, objectives and the associated research questions was a fundamental consideration when designing this study. I therefore reflected carefully on the purpose of the study before making the various methodological decisions.

4.2 Outline of this chapter

This chapter follows Creswell's (2003) line of thinking which emphasises that research methodology includes a researcher's plans, procedures and decisions that begin from broad philosophical assumptions to strategies of enquiry to detailed methods of data collection and analysis. In order to describe the research methodology employed in this study to investigate and explain the phenomenon of the first-year experience and academic performance, this chapter is divided into three main sections, namely, the philosophical perspectives; case study as a research strategy of enquiry; and the research design.

The first section presents this study's philosophical perspectives. It presents pragmatism as the general philosophical perspective, including some elements of

constructivism/interpretivism that inform this study. The constructivist/interpretivist perspective employs the assumption that meanings in the social world are constructed by individuals as they engage with the world they are interpreting (Creswell, 2003). Thus constructivism offers a descriptive, reflective, interpretive and engaging mode of enquiry from which the fundamental nature of students' first-year experience and academic performance may be elicited.

Pragmatism offers a practical assumption of 'what works' as the truth regarding the research questions under investigation (Tashakkori & Teddlie, 2003:52). The pragmatic worldview offers the use of those methods most appropriate for studying phenomena where the essential emphasis is on actual behaviour and the beliefs that stand behind such behaviour (Morgan, 2007:67). The pragmatic philosophy of common sense uses human enquiry as its focal point. It was used as a mode of enquiry from which the students' habitus and cultural capital could be examined to understand their influence on the first-year experience and academic performance.

Section 2 discusses the case study approach as the main research strategy of enquiry. The use of a case study strategy allowed this study to explore teacher education students' first-year experience in greater depth and collect detailed information with the use of a variety of data collection procedures over a period of one year (Stake, 1995).

The third section highlights the research design by including the following: the sample design; the research method for the study which included the sequential collection of quantitative and qualitative data and the data collection tools used; the analysis and interpretation methods employed; and the trustworthiness of the data. In addition, this chapter briefly discusses the conceptual framework outlined in Chapter 2 as a theoretical lens to analyse and interpret the data.

4.3 Philosophical perspectives

All research undertaken needs a foundation for its enquiry and researchers need to be aware of the implicit worldview/knowledge/paradigm they bring to their studies. Yin (2011:285) contends that 'a worldview consists of a set of beliefs about the acceptable qualities of research and how it should be done'. For Kuhn (1962), a paradigm is an agreed worldview that is embodied in the beliefs, practices, and products of a group of scientists (cited in Tashakkori & Teddlie, 2010:85). Creswell (2003) highlights four worldviews used in research (refer to Table 4.1) and argues that they all have common philosophical elements. He adds that that there are no set standards for the different worldviews, but they continue to evolve. Tashakkori and Teddlie (2010) state that the differing worldviews can and should coexist as they offer healthy contrasts and dialogues. Table 4:1 below presents the different worldviews and describes the characteristics under each view.

Paradigm	Methods (primarily)	Data collection tools (examples)
Positivist/ Postpositivist	Quantitative. 'Although qualitative methods can be used within this paradigm, quantitative methods tend to be predominant' (Mertens, 2005:12).	 Experiments Quasi-experiments Tests Scales
Interpretivist/ Constructivist	Qualitative methods predominate although quantitative methods may also be utilised.	 Interviews Observations Questionnaires Document reviews Visual data analysis
Transformative	Qualitative methods with quantitative and mixed methods. Contextual and historical factors described, especially as they relate to oppression (Mertens, 2005:9).	 Diverse range of tools – particular needs to avoid discrimination, e.g., sexism, racism, and homophobia.
Pragmatic	Qualitative and/or quantitative methods may be employed. Methods are matched to the specific questions and purpose of the research.	 May include tools from both positivist and interpretivist paradigms, e.g., interviews, observations, testing and experiments.

Table 4:1 Paradigms, methods and tools

Source: (Mackenzie & Knipe, 2006:201).

4.3.1 Constructivist/interpretivist viewpoint

In summary, constructivism holds that meaning is constructed by individuals as they engage with the world: their interpretations and the way they make sense of this world are based on their historical, cultural and social perspectives (Mason, 2002). In addition, the constructivists prefer qualitative methods of collecting data and the creation of meanings from data collected in the field is always social (Creswell, 2009). The constructivist view proclaims that there is no single reality, but multiple realities which are subjective and influenced by the context of the situation.

Two of the research questions of this study, RQ3 and RQ4 (outlined above in Section 4.1) required participants to construct meanings of the first-year experience with regard to their social and academic interactions in their first-year of studies. According to Crotty (1998), meanings are both constrained and constructed by one's culture, and historical and social perspectives; the social environment is seen as an extension of the experiences of the individual. In this view, the ontological orientation of the interpretive view resonates with the objective of this study which focused on exploring the internal reality of teacher education students' pre-entry characteristics and their influence on the first-year experience and academic performance. The main epistemological assumption of knowing reality was through exploring the experiences of others regarding a specific phenomenon. In this study, students' voices and their interpretations of their own experience were the medium explored to comprehend realities embedded in their first-year experience. Finally, the case study strategy was deemed appropriate for qualitative data collection, especially since I needed to conduct an in-depth study of a small number of participants that addressed RQ 3 and RQ4.

4.3.2 Pragmatic viewpoint

This research study is situated largely within the pragmatic worldview, which stands alone in key worldview elements. Pragmatism, according to Tashakkori and Teddlie (2010), is referred to as an example of a single paradigmatic stance. The uniqueness of this paradigm is that its focus, unlike that of other worldviews, is on the consequences of the research and the importance of the research questions over the methods used. Creswell and Plano Clark (2007) describe the pragmatic worldview as:

... typically associated with mixed methods research. The focus is on the consequences of the research; on the primary importance of the question asked rather than the methods; and multiple methods of data collection inform the problems under study. Thus it is pluralistic and oriented toward 'what works' and practice (Cresswell & Plano Clark, 2007:23).

According to Wheeldon and Åhlberg (2012), the pragmatic approach to understanding reality and developing knowledge is through the use of both

quantitative and qualitative data collection and analysis. This process provides an enhanced understanding of the phenomenon under investigation. For Cameron (2011), pragmatism in its simplest sense is viewed as a practical approach to a problem. Thus epistemologically, the key to pragmatism is practicality, using the 'what works' approach to addressing research problems.

From a philosophical and methodological stance, Johnson and Onwuegbuzie (2004) assert the following:

Pragmatism will be productive because it offers an immediate and useful middle position philosophically and methodologically; it offers a practical and outcomeorientated method of inquiry that is based on action and leads, iteratively, to further action and the elimination of doubt; and it offers a method for selecting methodological mixes that can help researchers better answer many of their research questions (Johnson & Onwuegbuzie, 2004:17).

Tashakkori and Teddlie (2003) identify the following characteristics as a philosophical worldview of pragmatism:

- Both quantitative and qualitative research methods may be used in a single study.
- The research question should be of primary importance, with a much greater emphasis than either the method or the philosophical worldview that underlies the method.
- The forced-choice contrast between postpositivism and constructivism should be abandoned. Pragmatism opens the door to multiple methods, different worldviews and different assumptions.
- The use of metaphysical concepts such as truth and reality should be avoided. Pragmatists sees truth as what works at the time to provide the best understanding of a research problem.
- A practical and applied philosophy should guide methodological choices.

Morgan (2007) describes pragmatism as a philosophy. It has strayed somewhat from the original focus according to Mertens (2010), but still resembles the ideas of the earlier pragmatists. The focus on 'lines of action', 'warranted assertions', and 'workability', is described by Morgan (2007:67) as:

... on actual behaviour ('lines of action'), the beliefs that stand behind those behaviours ('warranted assertions'), and the consequences that are likely to follow from different behaviours ('workability').

The pragmatists' goal is to search for useful points of connection. Shields (2003) avers that pragmatism in its simplest definition is viewed as a philosophy of common sense, which uses purposeful human enquiry as its focal point. The uniqueness of pragmatism identified by Tashakkori and Teddlie (2003) is that it aims to unlock the process of enquiry to all possibilities in the most practical way.

In a pragmatic view on ontology the emphasis is on creating knowledge through lines of action or joint actions (Morgan, 2007). The value of the research is judged on its effectiveness and establishing what works. For this study I draw on multiple knowledge claims including those of the constructivist viewpoint (Section 4.3.1 above) to gain an in-depth understanding of the nature of reality, taking into account the participants' personal experiences and their cultural, social and historical perspectives. The pragmatic view on how we gain knowledge of what we know (epistemology) is deemed most appropriate to understand the phenomena under investigation. This study makes use of a case study strategy to best understand such phenomena. First-year students within the Faculty of Education were chosen as the case. This allowed me, from a pragmatist view, to position myself as an insider and interrogate what 'interests' me and is of 'value' to me and to collect and utilise the results in ways that can bring about positive consequences within my own value system (Tashakkori & Teddlie, 1998:30).

4.4 Research strategy of enquiry: case study

A case study was my choice of research strategy for exploring pre-entry factors influencing the first-year experience and academic performance. According to Harrison (2002), a case study is more aptly described as a strategy than a method: it sets out to understand phenomena within its operating context. Yin defines a case study as an empirical enquiry which 'investigates a contemporary phenomenon within its real-life context, especially when the boundaries between phenomenon and context are not clearly evident' (Yin, 2003:13). Hancock and Algozzine (2006) describe a phenomenon as a particular event, situation, programme, or activity which is researched in its natural context and is bounded by space and time. A case study

could be regarded as a suitable strategy when the 'how' or 'why' questions are being posed (Yin, 1984:1). Atkins and Wallace (2012) state that the case study strategy has the ability to go beyond the 'how' and 'why' questions: it offers a way of investigating connections, patterns and context, and of reflecting on the details and the bigger picture under investigation. With the above explanation in mind, the case study strategy was an acceptable approach to capture an in-depth understanding of issues and the complexities of first-year students within their natural context.

Yin (2003:5) identifies six kinds of case studies based on a 2 x 3 matrix. He states that a case study could either be single, focusing on one case only, or multiple, which focuses on two or more cases within the same study. He then classifies three types of case studies: descriptive, exploratory or explanatory. A descriptive case study design presents an in-depth description of a phenomenon within its context. Exploratory and descriptive case study designs can involve theory generation however, only explanatory case study design is appropriate for theory testing (Yin, 1994). However, Stake (1995) includes three other types: intrinsic, instrumental and collective. Stake expounds that an *intrinsic* case study is when the researcher has an interest in the case; when a group of cases are studied together it is referred to as a *collective* case study; and an *instrumental* case study is when the case is used to understand more than what is obvious to the observer.

This study can be classified as an instrumental single-case design (Stake, 1995). A defining characteristic of this case study was its intensive investigation of a single unit (Yin, 1994; 2009). Its primary concern was with particularistic, descriptive and heuristic analysis of a single unit within a bounded system (Merriam, 2009). Merriam (1998:30) defines *particularistic* as focusing on a particular situation, event, programme or phenomenon; *descriptive* as producing a rich, thick description of the phenomena under investigation; and *heuristic* as enlightening the researcher's understanding of the phenomena and bringing about the discovery of new meaning to the phenomenon being studied. To capture a deep understanding of pre-entry factors influencing the first-year experience of teacher education students, I sought to embrace the single case study approach. An undergraduate BEd programme at a single campus in the Faculty of Education at a university of technology became the medium for exploring the first-year experience and academic performance of these

students. The descriptive case study design was considered most appropriate for this study, although according to Yin's (2003) six kinds of case studies, this study could be regarded as an explanatory case study as it does engage with Tinto's (1975, 1993) student integration theory and Bourdieu's (1984) conceptual tools. In this case study, the concepts from the two theories were not used as theory testing, but more to support the description of the phenomenon (Yin, 1984). The use of a descriptive case study design was a more appropriate choice for this study: it allowed me to investigate the students' first-year experience with much rigour and gain a deeper understanding of the pre-entry academic and non-academic factors influencing their experiences.

According to Stake's (1995) classification of case studies, this study is firmly established as instrumental: it focused on teacher education students as a means of understanding their first-year experience in higher education. As a lecturer employed in the Faculty of Education at a university of technology, and with a vested interest in academic support for at-risk first-year students, a pragmatic decision, based on easy access, was taken to focus the case within the teacher education context. This case could be regarded as intrinsic (Stake, 1995). But it can be argued that this study goes beyond self-interest to incorporate the insights gained to understand the phenomenon at a level that would be of significance to all stakeholders involved with first-year students and should therefore be understood as an instrumental case study.

Hancock and Algozzine (2006) describe a case study phenomenon as a particular situation, event, activity or programme bounded by space and time. This study's situation was principally focused on students' first-year experience at a higher education institution. It was bounded by a specific group of students - first-year teacher education students; the bounded time period was for one academic year - the first year of study only. Merriam (2009) suggests that insights gleaned from case studies can directly influence policy, procedures, and future research. The case studied in this research was aimed at gaining a better understanding of who our first-year students are and what pre-entry attributes they bring with them to higher education. The information gathered from this study may influence first-year curriculum planners and academic support programmes to plan inclusive first-year

programmes that could positively influence students' first-year experience. This could lead to an improved retention and success rate among the first-year student intake.

Case study research has many advantages. Reasons for utilising the case study strategy for this study were:

- Case studies often build upon tacit knowledge and provide a thick description of the case under investigation (Merriam, 1998:12).
- They have the ability to accommodate both quantitative and qualitative research techniques (Yin, 2003).
- They have the ability to deal with a variety of evidence, for example, documents, artefacts, interviews and observation (Merriam, 1998:12).
- Besides being able to explore or describe qualitative data in the real environment, case studies can assist in exploring the complexities of reallife situations (Zainal, 2007).
- They allow the researcher to explore the phenomena without restricting the context: the examination of the data is conducted within the context of its use (Yin, 1984).
- They are compatible with different philosophical viewpoints. Rezgui and Miles (2010) acknowledge that case studies are often associated with interpretivist and pragmatic viewpoints.

The use of case studies as a research design has been criticised by many researchers. For example Yin (2009:14) and Atkins and Wallace (2012:111-112) identify three types of arguments against case study research: the danger of generalising from the particular; accusations of lack of rigour; and the size of documents and unclear management structure.

The danger of generalising from the particular: The main concern with regard to case study research is its use of a small sample; therefore the results are often seen as not generalisable. Atkins and Wallace (2012) suggest that researchers should acknowledge such limitations of the research, in terms of scope and transferability, and should not offer any elaborate claims made on the basis of one case.

Accusations of lack of rigour: Yin (2009:14) acknowledges a second concern as a lack of rigour in the investigation. He admits that case study researchers have been accused of being 'sloppy', not following systematic procedures, and allowing biased pre-determined views to influence the direction of the findings and conclusions.

The size of documents and unclear management structure: A third concern in respect of case studies is that this type of research can be too long, difficult to conduct and can produce a massive amount of documentation which can result in the data's not being systematically managed and organised (Yin, 1984).

Although there have been criticisms of the use of case study research, the validity and reliability of a case study can be established by following certain requirements. Yin (2003:34) highlights four tests commonly used to establish the quality of any empirical social research undertaken. Case studies form part of social research, so the following four tests are relevant to case studies: construct validity, internal validity, external validity, and reliability. Validity and reliability and their relevance to this study are discussed in detail in Section 4.5 below.

4.5 Research design

According to Yin (2011), all research studies have an implicit blueprint or design, whether planned by the researcher or not. He defines research designs as 'logical' plans where logic involves the link between the research questions, the data to be collected, and strategies for analysing the data so that the findings can address the intended research questions (Yin, 2011:76). This study's research design is highlighted in Figure 4:1 on the following page (page 90). This section includes a discussion on participant selection, ethical considerations, data collection methods, data analysis, and trustworthiness of the study.



Figure 4:1 Research design

4.5.1 Selection of research participants

It was appropriate to invite all in-coming first-year students accepted for the BEd programme to participate in this research. The research site was the Faculty of Education, where English is the medium of instruction. The total fulltime enrolment figure at this campus for undergraduate teacher education programmes is 2 378. Approximately 500 students are enrolled in the first year of study across the three undergraduate programmes (CPUT, 2014). In 2012, at the commencement of this study, students from the three undergraduate programmes: Foundation Phase (FP); Intermediate and Senior Phase (ISP) and Further Education and Training (FET), which constituted a total intake of approximately 300 first-year students, became the participants for this study.

According to Mouton (1996), sampling in social research refers to procedures which produce a representative selection of population elements. For this study, a purposive sampling procedure was used. A key characteristic of purposeful sampling, particularly suitable for this study, was that not all members of the population are equivalent data sources. Participants selected for the qualitative sample were believed to be information-rich cases: selection was based on prior identified criteria for inclusion (Wiersma & Jurs, 2005). Participants chosen for inclusion in this study were 'new' first-year teacher education students accepted for the BEd programme. They were prospective students entering their first year at university, and therefore able to provide information on their prior expectations of university student.

To address the first two research questions, it was essential to include the total firstyear cohort accepted in the BEd programmes. The first stage of data collection was quantitative in nature. The total first-year cohort of 300 students accepted into one of the three BEd programmes (FP, ISP and FET) formed part of the quantitative sample. Pre-entry questionnaires were mailed to all students: 195 of the 300 responded. The 195 respondents became participants for the quantitative part of the data collection. The response rate of the mailed questionnaires was 65%, which according to Bryman (2008) is an acceptable response rate. The acceptable sample size guaranteed a minimal risk of bias in the findings as well as a greater opportunity

91

for the validity of the findings. For the qualitative data-collection phase, a purposive sample was selected from respondents of the mailed questionnaires. There are many variations of purposeful sampling, Wiersma and Jurs's (2005) maximum variation sampling offered the most suitable strategy for the selection of participants for the qualitative data collection. Maximum variation sampling is a strategy by which individuals are selected for the sample: they provide the greatest differences in certain characteristics. Participants who were selected for this phase were chosen from one specific course from the BEd programme, the foundation phase course. Students registered in the programme specialise in teaching Grade R to Grade 3 learners. Characteristics (Wiersma & Jurs, 2005) that prompted my choice of sample from the foundation phase cohort were as follows: the immediate shortage of foundation phase teachers in the country, specifically in rural areas; the low participation rate of black and male students in the programme; and the high firstyear attrition rate in this programme. Table 4:2 below indicates the profile of student enrolment in the foundation phase programme over a six-year period, from 2008 to 2012.

Year	African	Coloured	Indian	White	Total Female	African	Coloured	Total Male	Total
2008	8	57	2	130	197				197
2009	11	75	3	116	205		2	2	207
2010	9	68		120	197				197
2011	10	77		118	205				205
2012	22	61	2	121	206	1		1	207

Table 4:2 First year enrolment figures for BEd foundation phase students:2008–2012

These figures represent enrolments according to gender and race on the two satellite campuses offering teacher education programmes at the selected university of technology (CPUT, 2014). It is the only course in the programme that does not reflect the demographics of the country. Another reason for choosing participants from the foundation phase programme is that the selected university of technology is

the only university of technology in South Africa that offers the BEd foundation phase programme. Thus the exclusivity of the foundation phase programme and students intrigued me to focus on first-year students in this particular programme.

Participants for the qualitative data were chosen from the foundation phase cohort of the mailed questionnaire who indicated an interest in participating in the qualitative research. Foundation phase volunteers who provided their contact details formed the initial research pool. From the initial pool of 75 volunteers, a manageable sample of 12 participants was selected. I employed purposive selection to ensure that the sample cohort was representative of the foundation phase group and that information-rich data was collected. Gender and race were used initially to purposively select participants to be representative of the total foundation phase group. The second criterion used to narrow the selection to 12 participants was to ensure that the sample reflected the diversity of the total BEd foundation phase student group with regard to demographics, prior academic performance, and socio-economic characteristics, which were obtained from the completed mailed questionnaires (refer to Table 4:3 on page 94).

I felt that 12 participants would be a manageable number of students with whom to conduct in-depth interviews before the end of the first term of university. However, at the time of the interviews, which was eight weeks into the academic year, two of the selected participants had dropped out of the course, citing teaching as a wrong career choice. An additional two participants were unavailable for interviews owing to their heavy workloads and personal responsibilities; thus only eight of the selected participants became part of the final sample and were involved in the interview process. Table 4:3 on the next page highlights the profile of the participants.

Table 4:3 Profiles of participants

Participants	Age	Gender	Race	Profile
Joyce	22	Female	African	Single, one child, orphan, lives in Langa, Western Cape, received a scholarship to attend a private school, received a bursary for BEd, first-generation student, part-time job, uses public transport (taxi), does not own a computer.
Thabo	24	Male	African	Single, one child, orphan, mother was a domestic worker (single parent), lives in Dunoon, Eastern Cape, attended public school in rural area (no-fees school), moved away from home, applied for student loan, presently funding his own fees, has a part-time job, lives in student residence, uses public transport (taxi), does not own a computer.
Tasha	19	Female	White	Single, parents have professional jobs, lives away from home, with grandparents in Constantia, Western Cape. From Hermanus, attended public school (ex Model C fee paying), parents paying for Bed, not first-generation student, uses own car to campus and also owns a laptop.
Nadia	20	Female	Coloured	Single, parents divorced, mother remarried, mum professional employment (teacher), lives at home in Ottery, Western Cape, attended public school (ex Model C fee paying), parents paying for BEd, first-generation student, uses public transport (train), owns a computer.
Letti	27	Female	African	Single, father farm worker (semi-skilled labour), mother housewife, lives in Stellenbosch, Western Cape, attended public school (ex Model C fee paying), BEd fees part individual & part funded, first-generation student, has a part-time job, uses public transport (taxi), does not own a computer.
Alley	21	Female	Coloured	Single, parents divorced, mother remarried, parents have clerical jobs, lives at home, Milnerton, Western Cape, attended public school (fee paying), parents paying for studies, first-generation student, uses public transport (taxi), owns a computer.
Katlyn	19	Female	White	Single, parents have skilled jobs, divorced. Mother remarried, lives in Table View, Western Cape, attended public school (ex Model C, fee paying), first-generation student, she and her parents contribute to her fees, has a part-time job, has her own car but also travels with lift club, owns a laptop.
Lisa	19	Female	White	Single, parents divorced, lives with mum – remarried, both parents have professional jobs, lives in Wynberg, Western Cape, not first-generation student, parents paying for BEd studies, attended public school (Ex Model C fee paying), uses private transport (own car), owns a laptop.
The total intake of foundation phase students registered for the 2012 BEd programme consisted of 98% women and 2% men. For the first time since the inception of the programme at this particular campus has the programme experienced such a diverse student population, with 48% being coloured female, 26% African female, 24% white female, and 2% Indian female. The programme has also for the first time enrolled two male African students. The drastic change in the racial diversity of students entering the foundation phase programme was due to the Government of South Africa offering teacher education bursaries to address the shortage of foundation phase teachers in the country. More African foundation phase teachers are needed in rural schools since foundation phase learners at primary school need to be taught in their mother tongue. As reflected in Table 4:2 on page 92, the faculty still has a majority white female student population of 58% in the foundation phase programme, a total from both campuses.

4.5.2 Ethical considerations

Silverman (2011:417) states that ethical considerations have become one of the contemporary tendencies in qualitative research, diverting attention from knowledge production to social justice. Ethical considerations should be sustained during the entire research process (Kvale, 1996), from the start of the investigation to the final writing up of the study. This section uses Silverman's (2011) three main issues identified as ethical guidelines to guide the discussion: codes and consent, confidentiality, and trust.

Codes and consent: Ethical codes guarantee that identities and records of participants are confidential. When findings are published, the participants are not identifiable (Bryman, 2008). Kvale (1996) cautions that ethical codes do not provide definite answers to the normative choices made during a research investigation but can provide guidelines that must be judged according to their relevance to specific situations. Siverman (2011) contends that codes and consent refer to 'informed consent'. In this study there were two layers of informed consent. First, approval was sought from the institution's Faculty Ethics Committee. A formal proposal was submitted to the committee, providing full disclosure of the aim, objectives and procedures of the research project and the participants that would be selected for the project. A research ethical clearance compliance form (see Appendix D), issued by

the Faculty Ethics Committee, had to be completed. Once approval had been obtained from the Faculty Ethics Committee, I ensured that throughout my research process ethical decisions and procedures were followed. The second consent was obtained from the participants. For the mailed questionnaire, prior written or verbal consent could not be obtained. All first-year prospective students who received the mailed questionnaire were informed of the following: the questionnaire formed part of a DEd research project; filling in the questionnaire was optional; no costs would be incurred since a self-addressed envelope was provided; and finally, confidentially would be preserved with regard to participants' information and identity (see Appendix A: Pre-entry questionnaire).

Participants selected for the one-on-one and focus-group interviews were given a written consent form to sign. Prior to the prospective participants signing the consent form, the following explanation and description, adapted from Cohen et al. (2008) and Bryman (2008), was used to inform prospective participants of the procedures of the research study so that they could make an informed choice as to whether they wanted to participate in the study or not:

- The purpose, aims and procedures of the research were outlined to the prospective participants.
- They were assured that no risk, discomfort or negative outcomes would result from their participation.
- Their participation would be voluntary.
- They were free to refuse to answer any questions.
- They could withdraw from the interview at any time.
- Benefits that might derive from the research to the individual, the researcher and the faculty.
- They were assured of confidentiality in respect of their identity and information.

Once the above had been outlined to the prospective participants, they were given an opportunity to ask questions with regard to any aspect of the research. When the prospective participant was fully satisfied and understood what was expected of his or her role in the research study, the written consent form was signed. **Confidentiality**: Confidentiality is linked to the code of ethics and informed consent. Confidentiality means that the researcher is obliged to protect the participants' right to privacy (Silverman, 2011). This entails protecting the participants' identity and protecting the information given as a 'shared secret' (Cohen et al., 2008). Bryman (2008:123) warns researchers that 'the right to privacy is a tenet that many of us hold dear, and transgressions of that right in the name of research are not regarded as acceptable'. Cohen et al. (2008) add that researchers need to keep faith with those who have helped them. In order to ensure confidentiality and keep faith with the participants, I undertook the following actions: verbally before the interview I stated explicitly to the participants that their right to privacy would be upheld at all times; the confidentiality of the participants was once again acknowledged in the written consent form; and finally, the protection of the participants' privacy was sustained by using pseudonyms and changing their identifiable features when I reported on their interviews.

Trust: The third important concern associated with ethics is trust. This refers to the relationship between the researcher and the participants. To build trust between the researcher and the participant, the research procedures should be transparent (Yin, 2011) at all levels. Special care should be taken at the time of recording and reporting on participants' responses. Before the commencement of the interviews, permission from participants was obtained for the use of a tape recorder during interviews. Participants were assured that they could have access to their information at any time requested. Finally, participants were given transcripts of their interviews to revise, add or delete any of their comments.

The three ethical issues of consent, confidentiality and trust are closely linked. Therefore the role of the researcher when undertaking research should be to design, review and undertake the investigation in a way that ensures research integrity and quality.

4.5.3 Data collection methods

Data collection methods refer to a variety of techniques that could be used for gathering information. According to Cohen et al. (2008), there is no single prescription for specific data collection instruments to be used. The choice should be

'fitness for purpose'. This study makes use of both quantitative and qualitative data collection methods. Creswell and Plano Clark (2007) explain that the mixing of methods focuses on quantitative and qualitative data being collected, analysed and interpreted in a single study. Qualitative data is conveyed in words and quantitative data is presented numerically (Merriam, 2009). Creswell (2009) advises that researchers can make interpretations and identify patterns from statistical data and themes and patterns can be presented numerically.

Calls for the integration of a quantitative and qualitative research approach have been advanced by many disciplines. Molina Azorin and Cameron (2010) highlight the growing popularity of mixed methods research across all fields or disciplines. This admixture had been recommended mainly because mixing the two data sets provides the researcher with a better understanding of the problem. The study is able to achieve a more superior research outcome than one obtained from exclusively qualitative or quantitative data. The rationale for using both qualitative and quantitative data collection methods in this study was that neither quantitative nor qualitative methods on their own could capture the complex issue of the students' pre-entry factors and their influence on their first-year experience. When used in combination, both methods complement each other and provide a more complete picture of the research problem (Tashakkori & Teddlie, 1998).

The use of quantitative and qualitative methods in this study provided the following strengths to the phenomenon under investigation:

- The use of both methods provided strengths that offset the weaknesses of using either quantitative or qualitative data on their own.
- The mixing of quantitative and qualitative data collection tools provided more comprehensive evidence for studying the research problem.
- Using both methods provided satisfactory answers to the research questions under investigation.
- Using mixed methods encouraged me to collaborate across the quantitative and qualitative divide which occasionally can serve to limit the approaches to enquiry.
- Mixed methods research is 'practical' and allowed me the freedom to use all methods possible to address the research problem.

 The use of mixed methods encourages the use of multiple worldviews or paradigms rather than the typical association of certain paradigms for quantitative research and others for qualitative research. In this study I used both the pragmatic and interpretivist worldviews to understand and interpret the phenomena under investigation.

The strengths of the mixed-methods approach assisted me in obtaining a better understanding of factors influencing teacher education students' first-year experience and academic performance. Both the numerical and interpretive methods of data collection assisted in my understanding the way students perceived and experienced the world. Using the mixed-methods approach has value but it also has its challenges. Driscoll et al. (2007) caution that collecting and analysing quantitative and qualitative data can be time-consuming and expensive, and may lead researchers working under tight budgetary or time constraints to reduce sample sizes or limit the time spent interviewing. Another challenge that could arise and retard the research process is that researchers may have expertise in one datacollection method and not the other. This could adversely affect collection and interpretation of data. These issues are important: Creswell and Plano Clark (2007) state that these challenges are not too great to overcome, and that there are strategies that could be used to address these issues.

This study used the sequential method to collect data. Quantitative data was collected first and regarded as phase one of the data collection process. The collection of qualitative data followed and was regarded as phase two. As represented in Figure 4:2 on the following page (page 100), several data collection methods were used to answer the research questions. In Phase 1, questionnaires were used that addressed Research Question 1 and 2. In Phase 2, one-on-one interviews and focus-group interviews collected the qualitative data and addressed Research Questions 3 and 4.



Figure 4:2 Visual model of this study's sequential collection of mixed data

4.5.4 Mailed questionnaire

According to Bryman (2008), the term 'questionnaire' is reserved for contexts in which a series of usually closed questions are completed by the respondents. Questionnaires that are answered by respondents independent of an interviewer are referred to as self-administered questionnaires. Curtis and Curtis (2011) note that increasingly more self-administered questionnaires are completed without any interaction between participant and interviewer, for example, participants complete an electronic survey via a link in an Email or a mailed questionnaire. This study used the traditional mail questionnaire format to gain information from pre-entry first-year teacher education students. Three hundred questionnaires (see Appendix A) were sent by post to all students accepted for entry into the first year of the BEd undergraduate programme in the Faculty of Education at the university of technology which was chosen as the research site. I was cognisant of the fact that the sample population selected comprised new first-year students entering higher education and that some might not be familiar with self-administered questionnaires. Respondents had to read and complete the questionnaire without the interviewer's instructions or guidance. I had to ensure the following design in my questionnaire:

- Few open questions.
- Instructions that were easy to follow and not complicated.
- Unambiguous wording.
- Relatively brief and quick to answer to avoid 'respondent fatigue' (Bryman, 2008:261).

It was hoped that the information received from the questionnaires would provide an overview of the students entering the programmes and allow for more in-depth probing for the follow-up qualitative data collection. The self-administered questionnaire was divided into four sections: Section A, titled *Background Information*, requested mainly demographic and socio-economic information. Section B, titled *Pre-University Academic Background*, enquired about the students' prior school experience, subjects and computer skills. Section C focused on their career aspirations as teachers, and finally section D consisted of 40 statements on students' expectations of their first-year university experience (refer to Appendix A). The first three sections made use of the dichotomous question type which asked

closed questions requiring, for example, a yes/no response. Cohen et al. (2008) acknowledge that dichotomous questions are useful: they compel respondents to make a choice. Such questions funnel subsequent questions. For example, if 'yes', proceed to question X. It is also possible to code responses quickly.

Section D made use of the rating scale question type. According to Cohen et al. (2008:325), this type of question is a reliable device for researchers. The use of rating scales builds in a degree of intensity, sensitivity and differentiation of responses while still generating numbers. It provides the opportunity to determine frequency and correlations, and fuses measurements with opinions from analysing data quantitatively and/or qualitatively (Cohen et al., 2008:327). Section D consisted of 40 statements which sought to gain knowledge on the students' prior expectations of their first year of study. To give respondents a range of responses to the statements, the Likert scale was used which consisted of four categories ranging from 'strongly agree', 'agree', 'disagree', and 'strongly disagree'. The reason for using a four-point scale is that participants are required to make a decision on the statement. As Cohen et al. (2008) note, there often is a tendency for respondents to opt for the mid-point. The rating scale is a powerful tool that offers many opportunities to be flexible, but some of its cautionary factors include the inference of the intensity of feelings between two categories; for example, 'strongly agree' and 'agree' match that of 'strongly disagree' and 'disagree'; there are limited checks on whether the respondents are telling the truth; and finally, there is no way of knowing if respondents wanted to provide additional comments on what was asked. To give respondents an opportunity to voice their opinions/concerns, an open-ended question was provided at the end of the questionnaire that requested students to respond to any concerns/expectations with regard to their first-year experience.

Data collected from the questionnaire survey was primarily used to complement the qualitative part of this study. Much of the quantitative data collected provided a descriptive analysis of the first-year student cohort entering the teacher education programme at the selected campus. It provided a list of pre-entry academic and non-academic factors that were used to follow up in the in-depth interview process. To gain a greater sense of confidence and the maximum benefit from using questionnaires, it was crucial that I piloted the instrument. Bryman (2008) advises

that it is always desirable to conduct a pilot study before administering a selfcompletion questionnaire: it will ensure that the survey questions operate well (e.g. for clarity, elimination of ambiguities, target audience readability levels). It can play a role to ensure that the research instrument as a whole functions well. The questionnaire used in this study was first examined by my supervisor to check for the appropriateness of the items on the questionnaire for the research study. The modified version was then examined by other colleagues and experts in the field to ensure it was suitable for the study. Finally, the revised questionnaire was piloted on a small group of 30 teacher education students who had completed their first year of study. The students provided feedback on the clarity of the questionnaire items, instructions and layout. They provided feedback on the appropriateness of certain groups of questions/themes and use of words and readability levels of the target group. Once the adjusted questionnaire had been completed, it was mailed to all first-year students accepted for the BEd programme in the next year (2012). To avoid any additional costs to the participants, a self-addressed pre-paid envelope was included with the questionnaire.

4.5.5 One-on-one interviews

Kvale (1996) regards the qualitative research interview as a construction site for knowledge. He adds that 'an interview is literally an *inter view*, an inter-change of views between two people conversing about a theme of mutual interest' (Kvale,1996:14). Thus Kvale regards the interdependence of human interaction and knowledge production as central to the social context that provides the multiple perspectives needed for understanding a particular phenomenon. Cohen et al. (2008) concur with Kvale (1996), and add that although an interview enables participants to discuss their interpretations of the world they live in and also express situations from their own point of view, the researcher concerned with collecting data about life is inseparable from human embeddedness: both parties are part of life itself. Thus, in considering my research questions and research design, I opted to conduct in-depth interviews to collect rich data from the participants to better understand the phenomenon under investigation.

Interviewers could apply different approaches to interviewing participants. Interviews could range from being highly structured, questionnaire-driven interviews at one

extreme to open-ended conversational formats on the other (Merriam, 2009). In highly-structured interviews, questions and the order of the questions are determined ahead of time. Semi-structured interviews are guided by a list of questions or issues to be explored, but the exact wording and order are not pre-determined. Unstructured interviews involve no pre-determined criteria and are essentially exploratory.

For the purpose of this study, I decided to implement the standardised semistructured interview technique. In this type of interview the sequence of usually open-ended questions is determined in advance. Respondents are asked the same basic questions in the same order, which provides consistency. An interview schedule (refer to Appendix B) as well as picture cards (refer to Appendix E) was used to conduct the interviews. Patton (1987:116) offers the following strengths and weaknesses of using the semi-structured interview technique:

The strengths of this type of interview are:

- Respondents answer the same questions, so increasing comparability of responses.
- A complete set of data for each respondent on the topic addressed in the interview is produced.
- Interviewer bias is reduced through common questions.
- The instrumentation used in the evaluation is available for review.
- Evidence organisation and analysis are facilitated.

The weakness of this type of interview is that there is limited scope for flexibility should the respondents want to broaden their responses. To overcome the latter, I did not institute any constraints or limits to the way the participants discussed their views and experiences in response to the interview questions.

Formulating interview questions: The interviews were guided by the interview schedule consisting of questions that were divided according to the seven themes related to pre-entry factors and the first-year experience (see Appendix B). The themes included were: family background; pre-entry schooling experience; first-year university expectations; first-year experiences; academic integration; social

integration; and support structures. In devising the questions for the interview schedule, I took into account my main research question of this study. My aim was to draw responses from the participants with regard to the phenomenon under investigation. However, in formulating the questions I did not draw attention to the seven themes, but worded my questions in a general sense to probe the themes. The formulation of my questions was guided by the conceptual framework discussed in Chapter 2 and outlined briefly in the latter part of this chapter. The main objective was that the questions associated with the seven themes should aid in eliciting responses from the participants that would reveal their habitus, forms of capital and prior educational experiences which would facilitate a deeper understanding of how certain pre-entry factors identified by the teacher education students influenced their first-year experience and academic performance.

The use of picture sort cards: For the final question on the interview schedule, participants were shown ten picture cards (see Appendix E). These cards had to be sorted, ranging from the card that had the most significant influence, to the card that had the least influence on their university experience. The use of sorting techniques involves participants sorting such items as physical objects, pictures or cards containing words into different groups, so allowing their item categorisations to be elicited (Whaley & Longoria, 2009). According to Nurmuliani et al. (2004), card sorting is a knowledge-elicitation method used to capture information about different ways of representing domain knowledge. It provides valuable insight into participants' understanding of and perceptions on a particular issue. Österåker (2001) states that picture cards allow participants to be more expressive and verbal. This increases the amount of information given by participants. He elaborates that picture cards allow participants to become informants rather than respondents and to tell stories rather than answer structured questions. Thus, using the sort-card activity concurrently with the in-depth interview provided rich data on the students' perceptions and influences on their first-year university experiences. Saunders and Thornhill (2004) point out that an advantage of using a card sort activity in a concurrent in-depth interview allows the interviewer greater exploration of reasoning on particular issues raised by the participants.

The ten picuture themes outlined in the sort-card activity in this study were: family support; financial support; lecturer support; community support; academic student support; new campus friends; school experience; need to succeed; self-confidence; and an open card to discuss any other influence not mentioned. Rugg and McGeorge (2005) recommend that there should be no fewer than eight cards and a maximum of between 20 and 30 cards per sort activity under investigation. The sort-card activity was a crucial component in this study: it was related to the teacher education students' thought processes on important factors either supporting or not supporting their university careers. This activity encouraged participants to reflect critically on situations and sometimes even change their arrangement of the cards.

Participants were free to discuss any issue or link the picture on the card to any of their life experiences. My intention to gather as much knowledge on the students' prior university experience and factors influencing the first-year experience prompted me to implement the picture card technique.

Conducting the interview: The interview was conducted using the following guidelines, adapted from Struwig and Stead (2001:98-99):

- Participants were allowed to go into more detail on those issues they considered to be important in contributing to their first-year experience.
- At no time did I impose my viewpoint on the participants. Instead I assumed a passive role but probed when I thought a certain issue needed clarity.

The length of the interview was dependent on the depth of response from the participants. On average, substantive discussions in the interview, after preliminary introductions had been completed, were between 40 to 45 minutes.

At the commencement of the interview, I thanked participants for being part of the first-year experience study. I offered the participants some introductory remarks on the research study and then requested permission to use the audio tape-recorder.

In addition to the tape-recording, I made notes while the interview was taking place. According to Babbie and Mouton (2001), notes taken during the interview are important aspects of enhancing the credibility of research undertaken within the qualitative paradigm. The review of both the notes taken during the interviews, as well as actual transcripts of the interviews, served to provide a more in-depth and accurate understanding of the interactions with the participants. For example, a transcript will not show that a student's facial expression was negative when speaking about a particular issue.

All interviews were conducted on the campus of the university of technology selected. This was a suitable venue: it was where the first-year teacher education students were located, and it was convenient for the participants. No additional travel time or cost was incurred. The participants were comfortable being interviewed in an environment with which they were familiar.

The one-on-one interviews were conducted in the first term, over a period of one month, in March 2012.

4.5.6 Focus-group interview

The focus-group interview is similar to a one-on-one interview since both are casecentric, meaning there are a few cases and more variables, and both are led by the researcher in semi-conversational format. Curtis and Curtis (2011) acknowledge that there are significant differences. A focus-group interview presents a more natural environment than that of the one-on-one interview because participants in focus groups are influencing, and influenced by, the other participants in the group, just as they are in real life (Krueger & Casey, 2000). Travers (2006) describes a focus group as the term suggests: *focus* having a limited (focused) area of interest and *group* pointing to the number of participants occupied in the interview. Morgan (1988) further explains that a focus group is a form of group interview that depends on the interaction within the group as group members discuss a topic supplied by the interviewer.

In this study, the focus-group technique was used as a follow-up from the mailed questionnaires and the one-on-one interviews. The focus-group interview was conducted with the same participants involved in the one-on-one interviews. The aim of selecting the same group of participants for the focus-group activity was to generate additional opinions and insights on their first-year experience. Since this activity took place at the end of the participants' first year of study, I felt that they

would be able to reflect on the last ten months and provide more in-depth responses. A second aim in using the same sample was to gather the participants as a collective and get them to reflect on and clarify their perspectives on their first-year experience (Liamputtong & Ezzy, 2005). The use of this technique with the same participants not only allowed me to gather rich contextualised data, but also was useful for:

- gathering data on attitudes, values and opinions of the participants;
- empowering participants to speak out and voice their own opinions in their own words;
- encouraging groups rather than individuals to voice their opinions;
- providing greater coverage of issues that would not have been possible in the one-on-one interviews and questionnaires; and finally
- triangulating the one-on-one interviews and questionnaires (Morgan, 1988).

Thus a focus-group interview can be regarded as a type of in-depth interview accomplished in a group. Cohen et al. (2008) draw attention to the fact that when conducting group interviews, the unit of analysis becomes the view of the collective and ceases to be that of the individual participants. The focus is on the interaction inside the group.

Advantages and disadvantages of focus groups: The use of focus groups in relation to other research methods has grown in popularity. Early researchers (Watts & Ebbutt, 1987; Bogdan & Biklen, 1992; Arksey & Knight, 1999), cited in Cohen et al. (2008), have acknowledged the following advantages and disadvantages of using focus groups:

Advantages of focus groups:

- A greater potential for discussion to develop, thus generating a wider range of responses.
- Useful for gaining insights from participants for either a follow-up individual interview or from an individual interview already conducted.
- Quicker than conducting individual interviews and therefore less time consuming.

• Allows the researcher the opportunity to generate data quickly and at a low cost.

Disadvantages of focus groups:

- One participant may dominate the interview.
- Topics that are of a sensitive nature may cause participants to become silent or antagonistic to others in the group.
- There is no scope for individual probing.
- The researcher has less control over the data that is generated.
- The researcher needs to be an expert or well trained to conduct focusgroup interviews.

In spite of the disadvantages enumerated above, the use of the focus group method facilitates the collection of rich and interesting data. Patton (2002:386) explains: 'The objective is to get high-quality data in a social context where people can consider their own views in the context of the views of others.'

In this regard Merriam (2009) advises that the advantages of focus-group interviews need to be weighed against the disadvantages. In addition, focus-group data-collection methods should be used when the researcher regards them as the best way to obtain the best data that addresses the research questions of the study under investigation.

Participants in the focus group: All eight participants were invited to the focus group interview which was conducted at the end of the students' first year of study. Only six of the participants were available to be part of the focus group. Although there is no hard-and-fast rule with regard to the number of participants to be included in a focus group, Merriam (2009) suggests that between six to ten participants is an acceptable number. Morgan (1988) suggests between four to twelve people to a group. The six participants used in this study's focus group served the purpose of eliciting rich data on the students' first-year experiences and their attitudes, values and opinions on everyday life. According to Macnaghten and Myers (2004:65), 'focus groups work best for topics people could talk about to each other in their everyday lives – but don't'. A key feature of focus groups is the interaction within the

group which can be affected if the participants are not comfortable with one another (Curtis & Curtis, 2011). Because of the homogeneity of the group selected for this study's focus group (all the participants were first-year students on the BEd foundation phase programme), it ensured that the discussions were focused on their experiences in their first year of study. The homogeneity of the group ensured that participants had something to say and felt comfortable saying it (Cohen et al., 2008). As participants knew one another from attending the same classes, this ensured good group dynamics. This eased the process of building sound relations which usually occurs at the beginning of a focus-group meeting. This encouraged a frank and open discussion in which the participants were able to add to others' views and opinions and also agree and disagree with one another (Curtis & Curtis, 2011). The use of a homogeneous group of first-year students ensured that the aim of a focus group was achieved:

Unlike a series of one-on-one interviews, in a focus group participants get to hear each other's responses and to make additional comments beyond their own original responses as they hear what others have to say. However participants need not agree with each other or reach any kind of consensus. Nor is it necessary for people to disagree (Patton, 2002:386).

The role of the researcher: According to Bloor et al. (2001:57), focus groups are naturalistic rather than natural events and therefore cannot be left to chance and circumstance. Their naturalism has to be contrived carefully by the researcher. The role of the researcher, referred to as the moderator or facilitator, is regarded as the core of the focus group (Curtis & Curtis, 2011). The moderator's key role is to encourage all the participants in a focus group to express their opinions with as little direction as possible given by the moderator (Yin, 2011). The questions posed by the moderator are crucial: they will determine interaction among the group. The moderator needs to be able to balance the chairing of the focus group meeting so that the focus group dynamics are not compromised. The moderator should be able to address the following issues that may arise with some participants: being too talkative, too quiet, too noisy, aggressive, and dominating or monopolising the conversation. The moderator's role is to make sure that no participant is either unnecessarily marginalised, subject to blame or ostracised for holding a different opinion (Cohen et al., 2008). Thus the success of the focus group will require the moderator to be vigilant, cautious and polite when managing group dynamics.

However, in this study, I limited my role as researcher in the focus group to primarily directing the questions and ensuring that key topics were discussed (Flick, 2006). For example, I would pose a question and then step aside and observe so that the discussion could take on a 'natural' feel, allowing the participants to discuss the topics highlighted in an everyday, relaxed way. I kept my interruptions to a minimum to create a space in which the discussions on the students' first-year experience could unfold through an autonomous exchange of opinions from the participants (Curtis & Curtis, 2011).

Formulating focus-group questions: The focus-group activity followed from oneon-one interviews so I needed to obtain depth of coverage with certain key issues identified from the earlier interviews. Thus obtaining good data from the focus group meant asking good questions in the most appropriate and effective way. Merriam (2009) advises that the way in which questions are worded is a crucial consideration in extracting the type of information desired. Merriam offers the following guidelines that researchers could adhere to when drafting focus group questions:

- Make certain that what is being asked is clear to the participants.
- Questions need to be understood in familiar language.
- Use words that reflect the participant's worldview.
- Avoid technical jargon or terms and concepts from your particular discipline.

For the focus group activity I made use of the semi-structured interview technique. I wanted to examine students' habitus, capital, and field. I wanted to ascertain how these three concepts of Bourdieu (1984) influenced the way students perceived or interacted/negotiated their first year on campus. However, I could not directly ask the participants to reflect on their habitus, capital, or field: these are theoretical concepts so I had to think of a way of turning the theory into practical questions that the participants would be able to understand and relate to. My best option was to pose the questions in themes. The three themes identified for the focus group questions that addressed Bourdieu's three thinking tools were: background information, students' first-year experience, and student support. In order to stimulate an in-depth response from the participants, I formulated my questions by making use of five of Patton's (2002) suggested six types of questions to achieve a successful interview:

- Background/demographic questions: these refer to questions that seek to elicit particular demographic information such as age, education, income, job, etc., from participants.
- Experience and behaviour questions: these types of questions elicit responses that highlight participants' behaviour, actions and activities. For example, asking participants in this study to talk about their first day of orientation, such as they first thing they did.
- 3. Opinion and value questions: in these types of questions the researcher is interested in participants' beliefs or opinions on a particular issue. For example, following from the above question, a subsequent question would be: what is your opinion of orientation programmes which provide support to first-year students?
- 4. Feeling questions: these questions draw on the affective dimension of human life. Researchers look for adjectival responses such as: anxious, sad, happy, scared, etc. For example, in this study, a follow-up question would be: How did you feel on your first day on campus?
- 5. Sensory questions: these are similar to experience questions, but in this type of questioning the researcher tries to elicit more specific data about what the participants saw, heard, touched, etc. For example, when participants responded to a question about group work activity, the information retrieved from the data was what they observed with regard to the group dynamic, the type of conversations they heard from the group or from students in other groups.

However, when questioning, I was cautious in asking the 'why' questions, as Patton (2002) advises that although this type of question can sometimes uncover specific insights or lead to new questions, there is still that danger of the 'why' question leading to speculation about causal relationships and also to dead-end responses.

Conducting focus-group interviews: The focus group interviews took place at the end of the academic year after the students had written their last examination paper. Students were more relaxed at this time. The physical space chosen to conduct the

interview was one of the classrooms that had movable furniture and was secluded and noise free, thus allowing for limited distractions. Taking into consideration that the success of a focus-group interview relies on the interaction between the group participants, I arranged the desk and chairs so that all participants were visible to one another. I had coffee and tea available for the participants.

In order to have a successful focus group discussion I followed Finch and Lewis's (2004) five stages in conducting focus group interviews. These stages are:

- Stage 1: Setting the scene and ground rules
- Stage 2: Individual introductions
- Stage 3: Opening topic
- Stage 4: Discussions
- Stage 5: Concluding remarks

The focus group commencement time was 10:00. As students walked in, I welcomed them and thanked them for making the time to be present for the focus group interview. Stage 1: setting the scene and ground rules commenced once all six participants were in the room. I officially welcomed the participants once again as a group and then provided them with a brief introduction of what the study was about, the background information, the purpose of the study and what their contributions would be to this study. I had a Microsoft PowerPoint presentation with nine slides. The first two welcomed the students and set the rules for the focus group activity (refer to Appendix F: Focus-group interview) which I went through with the students. Confidentiality was stressed and the need to respect each other's views and allow all participants a voice. I stressed that there were no right or wrong answers and that all views expressed in the focus group would be valued and confidential and would not be discussed outside this venue. I once again informed the participants that this session would be recorded, transcribed and given to them to edit and comment on before using the information. Stage 1 was short and not technical or intimidating in any way. I wanted participants to feel relaxed and comfortable.

Stage 2 involved individual introductions: it was short and slide 3 in Appendix F covers this procedure. Because the main part of this focus group activity was to investigate how habitus, capital, and field played a role in students' perceptions,

actions and interactions in their first year at university, understanding the students' background information was particularly important in gaining the information required. In the first activity, students had to provide a general introduction of themselves to the group.

In Stage 3, the opening topic, the activity chosen, ensured that students felt confident about the topic and comfortable to share with the group. Each student had to reflect on his or her life history and produce a timeline with all the important milestones leading to the day they arrived at university (see slide 4 in Appendix F). Students were given 20 minutes to complete this task and then 5 to 10 minutes to present their timelines to the group. The group could then interact with the participant. This activity provided me with information on the students' capital, habitus, and field. It allowed participants to establish a degree of familiarity; gave each participant an opportunity to speak and listen; and allowed participants to feel valued and an expert in their field.

Stage 4, which Finch and Lewis (2004) label 'discussions', encompassed eliciting conversations with regard to students' prior expectations and present experiences in their first year of university. This activity created robust discussion that sometimes strayed outside the ambit of the topics but produced rich information on how students' cultural beliefs and perceptions influenced their first-year experience.

In concluding the focus-group interview, I thanked participants for their candour and generosity in sharing their personal stories. I assured them once again that the information provided by them would be confidential and that they would have an opportunity to read the transcripts of the focus-group interview and comment or correct any misrecording of the discussions.

4.5.7 Analysis of evidence

This section describes the data processing and analysis used to understand the quantitative and qualitative data collected. As Henning et al. (2004) affirm, data analysis is a process that requires analytical craftsmanship in order to have the ability to analytically capture understanding of the data. I made use of qualitative and quantitative data analysis to process the interpretation of the data about students' first-year experience and pre-entry factors influencing these experiences. However, it

is important to state that the data analysis activity was not undertaken solely at the end of the data collection period, but was an ongoing process throughout the study of the first-year student experience (Miles & Huberman, 1994).

In this study, quantitative data was collected first to provide an overview of the profile of students entering the teacher education programme and to capture their expectations of their first-year university study. The quantitative data collected addressed two of the research questions: 'Who are our first-year teacher education students?' and 'What are the pre-entry academic and non-academic factors that firstyear teacher education students bring with them to higher education?' The quantitative data collected informed the qualitative data process to answer the study's main research question: 'How do pre-entry academic and non-academic factors influence teacher education students' first-year experience and academic performance?' More specifically, it contributed to addressing the two sub-questions investigated in the qualitative part of the study: 'To what extent do pre-entry factors that students report influence their social and academic interaction in their first year of study?' and 'To what extent do these factors contribute to the students' first-year experience and academic performance?' The following section discusses the quantitative and qualitative data analysis separately to highlight the importance of both data sets to the study. The quantitative and qualitative data collection methods were supportive of each other and thus intrinsically linked with the success of this study.

4.5.8 Quantitative data analysis

The quantitative statistical analysis was conducted by using a computer software package (Statistical Package for the Social Sciences: SPSS version 22.0). The data analysed was collected from the survey instrument: pre-entry questionnaires that were sent to students prior to their entering the first-year BEd undergraduate programme. A variety of data analysis was undertaken to examine the profile of the first-year BEd students entering higher education; explain the reliability of the instrument used; and to answer the research questions framing the study. The following statistical procedures were followed:

- Descriptive statistical analysis
- Statistical inference
- Reliability analysis (Cronbach's alpha)

Descriptive statistics: Descriptive statistics describe a situation by summarising information in a way that highlights the important numerical features of the data. In this study, summary statistics were generated for each of the 138 items included in the instrument. These descriptive statistics included the mean scores, standard deviation and the frequency distribution for each response. The statistics were compiled and expressed in numbers and percentages for all demographic, independent and dependent variables, for example, age, gender, school attended, grades obtained, parents' highest educational level, parents' occupations and students' expectations of their first-year experience. The completed statistical analysis reported on the total sample and the total responses from the three individual BEd programmes. Finally, the results from the descriptive statistical analysis were summarised in the form of text and presented in the form of graphs, cross-tabulations and other figures. The intention of the quantitative data analysis was to describe the phenomena in the study rather than prove or test a hypothesis or research question. The descriptive statistics provided the backdrop to this study as it offered valuable information on the profile of first-year students entering the BEd programme: their pre-entry characteristics, and expectations of their first-year university experience. In addition it allowed for certain matters obtained by the quantitative data to be further probed and explored in the qualitative process of the study.

Statistical inference: The analysis of the quantitative data made use of inferential statistical methods: Fisher's exact test and *t*-test. Fisher's exact test is a statistical significance test used in the analysis of cross-tabulation tables. In this study, Fisher's exact test was used to test significance in relations between demographic data, socio-economic data, prior schooling experience and students' prior expectations of their first-year university experience. Much of the analysis of the quantitative data relied heavy on *p*-value significance testing which Fisher regarded as self-explanatory measures of evidence that did not need further support. Wetzels et al. (2011) explain that Fisher interpreted these *p*-values as evidence against the null

hypothesis which implies that the smaller the *p*-value, the more evidence there is against the null hypothesis. Wetzels et al. add that most researchers adopt a 0.05 cut-off which indicates a *p*-value less than 0.05 constitutes evidence for an effect: those greater than 0.05 do not. In this study, various categorical relations were determined and the Fisher test value was used to determine significance of relations between the variables. For example, the Fisher test value was used to determine whether the first-year BEd cohort in the three programmes played a role in terms of respondents being the first in their immediate family to attend university. The Fisher *p*-value of 0.638 implied that the cohort in the three programmes did not have an effect on their being the first to attend university.

The *t*-test, which is designed to test if a difference between two means is significant, was applied to the data in this study to make statistical inferences. One example of the use of *t*-tests to ascertain significant differences between two means was between the mean expectation value and the mean perception value. Analysis of variance (ANOVA) was used to determine whether the means between the BEd first-year groups (FP; ISP and FET) were comparable. Statistical testing in this study was undertaken at 0.05 level of significance.

Reliability analysis: Reliability analysis is used to determine the stability of the instruments used to measure the variables in a study. Polit and Beck (2004) add that reliability refers to the consistence of an instrument over time and conditions. In quantitative data analysis, reliability is usually reported as a coefficient with a theoretical range of 0.00 and 1.00, with the 1.00 representing perfect reliability. However, researchers usually strive for reliability coefficients greater than 0.70, but occasionally will settle for coefficients as low as 0.60. Watson and Flamez (2014) highlight that there is no clear-cut indication of what amount of reliability is considered sufficient. However, they provide the following guideline: an estimate of 0.90 is preferred and regarded as a high degree of reliability; a 0.80 can be regarded as highly acceptable, and from 0.7 to as low as 0.60 as acceptable. But a reliability estimate below 0.60 is usually judged to be unacceptably low. As the estimate gets closer to 0.50, it is more likely that the instrument scores are a product of chance rather than a true score of the variance. Finally, a measurement of below 0.50 may

be regarded as being heavily influenced by measurement errors (Mitrushina et al., 2005; Watson & Flamez, 2014).

To determine the reliability of the instrument used in this study, I made use of Cronbach's alpha (α) to determine the internal consistency and average correlation. The reliability calculation made use of 35 instrument items, excluding 5 items which were stated bi-directionally. The Cronbach's alpha for the 35 items ranged from 0.564 to 0.776, with an average measurement of reliability of 0.668 as shown on below in Table 4:4.

Case Processing Summary						
		Ν	%			
Cases	Valid	178	91.3			
	Excluded	17	8.7			
	Total	195	100.0			
a. Listwise deletion based on all variables in the procedure.						

 Table 4:4 Overall reliability of instrument

Reliability Statistics				
Cronbach's Alpha	No. of Items			
0.668	35			

The resultant coefficient score of 0.668 obtained for this study therefore can be regarded as an acceptable score (Watson and Flamez, 2014), bearing in mind that this was a newly developed instrument. Inferences from the quantitative study that had a low reliability were probed and sustained in the qualitative study. This process clarified, verified and enriched the data from the survey questionnaire and positively contributed to the validity, reliability and triangulation of the study.

4.5.9 Qualitative data analysis

At the onset I should like to outline that my engagement with the data proved that the qualitative data analysis activity was not a neat and tidy linear process. I discovered that this procedure was quite challenging and found myself going back and forth, in a cyclical pattern, in order to gain a deeper and more in-depth understanding of the

information retrieved. To be fully engaged with and immersed in the corpus of data of this study, I utilised thematic analysis to guide this process (Miles & Huberman, 1994; Holliday, 2007). Thematic analysis provided the most effective means to establish how participants made meaning of their first-year experience. The first-year teacher education students' perceptions, attitudes, understanding, knowledge, values, feelings and experiences were analysed in an attempt to approximate their reality.

According to Braun and Clarke (2006), thematic analysis is a method suitable for identifying, analysing and reporting patterns (themes) within a set of data. They acknowledge that a theme captures something important about the data in relation to the research question and represents some level of patterned response or meaning within the data set. Thematic analysis was deemed to be an appropriate analytical tool for the qualitative data of this study (Alhojailan, 2012):

- Its ability to detect and identify factors that influence any issue generated by the participants.
- Its flexibility allowed the use of both inductive and deductive methodologies.
- It provided the opportunity to code and categorise data into themes, thus noting patterns and different levels of themes.

In this study the thematic analysis was underpinned by the conceptual framework developed. Bourdieu's conceptual tools of habitus, field, and capital and Tinto's social and academic integration produce codes that consequentially converge to represent themes that emerged from shared experiences among participants. Each participant's data set was explored and positioned in relation to the new field of higher education that they entered. Bourdieu makes explicit that his conceptual 'thinking' tools are not simply theoretical tools but are working tools for empirical application (Bourdieu, 1998:9). New themes and sub-themes that emerged were acknowledged and reported on.

For the development, support and management of the qualitative analysis, the ATLAS.ti computer program assisted, and the Computer-Aided Qualitative Data Analysis Software (CAQDAS) package was used. CAQDAS is designed not to

undertake the analysis for the researcher, but rather to facilitate the process. The use of ATLAS.ti for this study freed me from the manual methods of coding, comparing and categorising, providing more time for reflecting on the data and identifying or relating concepts pertaining to the study.

The thematic analysis process adopted for this study followed the model described by Miles and Huberman (1994:12) which consisted of three linked stages: data





reduction, data display and data conclusion - drawing/verifying. The illustration in Figure 4:3 on page 120 highlights the components of the data analysis of this study.

The qualitative data included one-on-one interviews with the eight participants and the focus-group interview. In preparation for the data reduction process, the first step entailed transcribing of interviews, providing pseudonyms to all participants, and entering all data onto ATLAS.ti. The first phase of the data-reduction process involved reading through all the transcribed data to obtain a general sense of the information and reflect on the overall meaning (Creswell, 2009). While reading the data for a second time, I started recording general thoughts about the data, made brief notes in the margins, and underlined passages that interested me. During this phase I also went back to the recorded interviews to listen to the tone of the conversations and make connections between the participants' thoughts and ideas. This allowed me to appreciate the complete picture of the data collected.

Phase 2 of the data reduction involved highlighting sentences from each participant's transcript relevant to the research. While going through the text, I was mindful of the study's research questions and conceptual framework. In this regard I highlighted excerpts from the respondents' full text that could contribute to this study.

Phase 3 of the data reduction involved my going through all the highlighted sentences/paragraphs and breaking them down into smaller segments. These segments became my first set of themes from the data. As suggested by Ryan and Bernard (2003), I read through the full text of each participant once again to compare, contrast and search for information that appeared missing in the first level of themes. I considered the following questions in this process: How is this text different from the next? What kinds of things are mentioned in both? Was there a deeper meaning in one response compared with another? This allowed the data, under the first level of themes, to develop further. Once this process was complete, the data was ready for identifying and classifying according to the second level of themes or codes. But before moving onto coding, I had to ensure that the first set of themes represented the whole text. Thus validity of the themes was the next important step.

Phase 4 involved the validating of the text, as stressed by Miles and Huberman (1994). Validating themes in the early and late stages of data analysis is essential. Outside reviewers, namely colleagues, fellow researchers and experts in the field, were involved in evaluating and confirming that the first set of themes was compatible with the whole text. The main purpose of this process was to build reliability in the themes analysis coding, thus providing accurate and reliable themes for the second level of coding.

Phase 5 involved the data display which is the second main step in Miles and Huberman's model (1994). This step involved organising the data to make sense of the data collected. The following procedures were followed: reviewing the research questions to identify information that relates to similar concepts; utilising the conceptual framework to arrange concepts and thoughts of the data; grouping topics that were related to one another; and finally, using descriptive words to turn these topics into categories. This process allowed me to make sense of the data collected, avoid data overload, as well as view and enhance the data for research more distinctly (Miles & Huberman, 1994). The data-display process of the study involved a variety of techniques (tables, graphs, narratives and direct quotations) which afforded me the opportunity to gain a more in-depth understanding of the data. For example, the direct quotations provided supportive meanings to the interpretation of data (Patton, 2002). The use of tables provided an appropriate and convenient way to draw comparisons between the different themes of the data. By utilising different display techniques, I was able to frame my thoughts and focus, and make the necessary connections from the data to reach justifiable conclusions in this study.

The final phase of this process, titled data drawing and conclusions (Miles & Huberman, 1994), involved extracting the relevant data and once again verification of the data. In order to generate meaning and interpret the data, this phase could not be separated from the data-reduction phase. The data-drawing and conclusion step included identifying relevance of similar or contrasting statements; emergent patterns; interrelations among factors and variables; and, finally, exploring the validity of the findings by building conceptual coherence and consistency (Alhojailan, 2012), so that they fit the conceptual framework of the study.

4.5.10 Trustworthiness of the study

The case study approach has received considerable criticism regarding its possible lack of accuracy and generalisability compared with quantitative methods (Bessoondyal, 2005). According to Lincoln and Guba (1985), trustworthiness is an important aspect of qualitative research; it is equivalent to the concepts of reliability and validity. Reliability is generally understood to concern the replicability of research and the obtaining of similar findings if another study using the same methods was undertaken (Lewis & Ritchie, 2003). Validity, according to McMillan and Schumacher (2001), refers to the degree to which the interpretations and concepts have mutual meanings between the participant and the researcher. According to Lincoln and Guba (1985), there is no validity without reliability and therefore no credibility without dependability. They further state that a demonstration of validity in research is sufficient to establish the reliability of that research. Therefore, according to Lewis and Ritchie (2003), in discussing reliability and validity, qualitative researchers are interested in the confirmation of findings, which can be assessed through examining credibility, transferability and dependability.

Based on the above discussions, Table 4:5 on the next page, page 124 outlines the framework adopted to enhance the reliability and validity of this study.

Table 4:5 Strategies used to enhance the quality of the research outcomes ofthis study

	 In-depth individual interviews followed by a focus-group interview with the same participants. 				
	Referential adequacy: audio-recording evidence.				
	 Verbatim accounts of interviews, tape-recorded interviews were transcribed. 				
Credibility	 Member checks: transcripts were checked and confirmed with participants. 				
	• Outside reviewers used to evaluate and confirm themes and codes of transcribed text.				
	 Direct quotations of participants were used to illustrate participants' meanings in the discussion section. 				
Transferability	• Participants were purposefully selected to maximise the range of specific information for the research.				
	 Audit trail: an external audit trail was applied to establish levels of dependability and confirmability of the study. 				
Dependability	• Audit trail included: audio-tapes of interview, transcripts of interview, field notes and hard copy of all documents coded by qualitative and quantitative software package.				

Finally, this methodology chapter underpinned the issues of reliability and validity by making explicit the research process that was followed in this study. In addition, by describing each stage of the research process in detail, it allowed for the transparency of the information and the process.

4.6 Concluding remarks

In this chapter the research methodology of the study was explained and the research methods detailed. The chapter was divided into three sections: the first section described the philosophical perspectives of the study; the second section outlined the use of the case study as a research strategy of enquiry; and the final section explicated the research design, which included both quantitative and qualitative data collection methods and analysis. Finally, this chapter concluded with a discussion of the trustworthiness of the study by providing a brief outline of the reliability and validity techniques used in the study. In the next chapter the analysis of the quantitative findings is discussed.

CHAPTER 5 QUESTIONNAIRE FINDINGS

5.1 Introduction

This chapter presents quantitative results obtained from questionnaires administered to all first-year BEd students prior to registration. In total, 300 questionnaires were mailed to all students accepted for the BEd programme and 195 questionnaires were returned. This resulted in an acceptable response rate of 65%.

The questionnaire consisted of 138 items, with levels of measurement at a nominal and ordinal level. The self-administered questionnaire was divided into four sections which measured various themes.

- Section A, titled 'background information', requested mainly demographic and socio-economic information.
- Section B, titled 'pre-university academic background' enquired about the students' prior school experience, Grade 12 results and computer skills.
- Section C, focused on their career aspirations as teachers.
- Section D, consisted of 40 Likert scale statements on students' expectations of their first-year university experience (Appendix A).

The first three sections made use of the dichotomous question type which asked closed questions, requiring for example, a yes/no response. Section D made use of the rating scale question type which consisted of four categories ranging from 'strongly agree', 'agree', 'disagree', to strongly disagree.

Results obtained from the questionnaires formed part of Phase 1 of the datacollection procedures. The questionnaire was fundamentally designed to provide an overview of the first-year profile of students entering the BEd programme. Data gathered enabled me to obtain a synopsis of the following research question:

Who are our first-year teacher education students?

The questionnaires led to a general perspective on pre-entry factors, partially to address the following research question:

What are the pre-entry academic and non-academic factors that first-year teacher education students bring with them to higher education?

In addition, the 195 respondents from the questionnaire survey assisted with the purposive selection of participants for the qualitative phase of the study. Finally, the completed questionnaires were used to design the one-on-one interview schedule to address inconsistences and concerns that were identified in the quantitative data collection. This allowed for a more intense and meaningful probe into factors influencing students' first-year experience and academic performance.

The quantitative data collected from the questionnaires was analysed with SPSS version 22.0. The results from the descriptive statistical analysis were illustrated in the form of graphs, cross-tabulations and other figures. Inferential techniques were used in the analysis of this study's quantitative data. The data is presented as a total sample and as individual samples among the three BEd programmes: FP, ISP, and FET. To determine whether there was any relation between cohorts of the three programmes and each item analysed, Fisher's exact test was conducted. All *p*-values lower than 0.05 show a significant relation. The test implies that the differences observed are significant, which means that the different groups do not share the same ideas or have similar trends. The presentation of the quantitative data obtained is presented in the next section.

5.2 Presentation of data

This chapter presents the data under two main categories: diversity of first-year students entering the BEd programme and students' expectations of their first-year experience. In the first category, first-year student profiles are examined as a total sample. The diversity between the three first-year BEd programmes is described. In the second category, students' expectations are described as a total sample and also within the three programmes.

The data is presented under the following categories and sub-categories:

- Diversity of first-year students in the BEd programme:
 - Demographic factors: age, gender, race, language.

- Socio-economic factors: academic qualifications of parents, job classification of parents, family support, financial support, mode of transport to university.
- Pre-entry academic factors: type of school attended, year of high school completion, Grade 12 results, first-generation university students.
- Student pathways to higher education.
- Students' expectations of their first-year experience:
 - Academic integration: academic preparedness, academic support, and collaborative interaction.
 - Social integration: students' expectations of socialising at university, involvement in academic discussions outside of formal class, students' self-perceived expectations of social integration.
 - Student perceptions: perceptions of lecturers' expectations of firstyear students, students' self-perceived expectations of academic lecturers.

5.3 Diversity of first-year students in the BEd programme

This section reports on the descriptive analysis of the demographics of the respondents entering the BEd programme, namely age, gender, race and home language. The demographic data analysed is presented first as a total sample and then, according to the three BEd programmes offered at the university of technology, that is, FET, FP and ISP.

5.3.1 Demographic information

5.3.1.1 Age and gender

From the total sample (N=195), 23% (n=45) respondents were male and 77% (n=150) were female. The age groups ranged from: younger than 20 years; 20–29 years; 30–39 years and 40–50 years. The findings showed that a majority of respondents, i.e., 53.3% (n=104) were in the less than 20 years age category; 40%

(n=78) within the 20–29 years age category; 6.2% (n=12) in the 30–39 years age category and 0.5% (n=1) were in the 40–50 age category.

In examining the age and gender results among the three first-year BEd programmes, overall it was noted that a similar trend to the one revealed by the total sample of the study was identified: there were slight differences within the three programmes. Table 5:1 on page 129 indicates all three programmes, FET, FP and ISP had the highest number of respondents in the lower than 20 age category: 60.4%, 48% and 53.7% respectively. The total age mean for the entire sample was 20.91 (sd= 4.478, N=195). A similar trend in age group was observed in the three programmes. The mean and standard deviation for each group was: FET, 21.02 (sd= 5.740, N=53); FP, 21.20 (sd= 4.496, N=75) and ISP, 20.49 (sd= 3.169, N=67).

Table 5:1 on the next page shows the figures and percentage for the younger than 20 years age category and the total of the three programmes. The results indicate that the highest number of male and female respondents was found to be in the younger than 20 years of age category for the FET and ISP programmes. Foundation phase had no male respondents in this age group. The FP group had two male respondents only in the programme. Differences in gender can also be observed from Table 5:1 which indicates the following male to female ratio: FET programme is approximately 1:1 (49.1%: 50.9%); FP is 1:36 (2.7%: 97.3%) and ISP 1:3 (25.4%: 74.6%).

Eurther Education 9 Testsing (EMC)					Gender	
Further Education & Training (EMS)				Male	Female	Total
FET	Age (Grouped)	< 20 years	Count	13	19	32
			% within Age (Grouped)	40.6%	59.4%	100.0%
			% within Gender	50.0%	70.4%	60.4%
			% of Total	24.5%	35.8%	60.4%
	Total		Count	26	27	53
			% within Age (Grouped)	49.1%	50.9%	100.0%
			% within Gender	100.0%	100.0%	100.0%
			% of Total	49.1%	50.9%	100.0%
Foundation Phase			haaa	Gender		Total
			nase	Male	Female	Total
		< 20 years	Count	0	36	36
FP	Age (Grouped)		% within Age (Grouped)	0.0%	100.0%	100.0%
			% within Gender	0.0%	49.3%	48.0%
			% of Total	0.0%	48.0%	48.0%
	Total		Count	2	73	75
			% within Age (Grouped)	2.7%	97.3%	100.0%
			% within Gender	100.0%	100.0%	100.0%
			% of Total	2.7%	97.3%	100.0%
Intermediate & Sen			ior Phase	Gender		Total
				Male	Female	Total
ISP -	Age (Grouped)	< 20 years	Count	10	26	36
			% within Age (Grouped)	27.8%	72.2%	100.0%
			% within Gender	58.8%	52.0%	53.7%
			% of Total	14.9%	38.8%	53.7%
	Total		Count	17	50	67
			% within Age (Grouped)	25.4%	74.6%	100.0%
			% within Gender	100.0%	100.0%	100.0%
			% of Total	25.4%	74.6%	100.0%

Table 5:1 Age and gender distribution of first-year BEd student respondents

In this regard there was no significant difference in the number of respondents in the three BEd programmes and age (Fisher's exact test, *p*-value = 0.524), but gender did show significant differences (p = 0.000).



Figure 5:1 below illustrates the results of age and gender distribution for the total study sample.

Figure 5:1 Age and gender distribution of respondents

Male Female

Figure 5:1 above illustrates there is a decreasing ratio of each gender with increasing age. Also evident is that for the first three age groupings, there are approximately three times as many women as men in the BEd programme. The mean age for the total sample is 20.9 and the male to female ratio is 1:3.33 (45:150). This data implies that with the first-year student population having an average age of 20.9 years and the age category for the younger than 20 years being the highest (i.e. 53%), the majority of the first-year student population are either school leavers or gap-year students. The balance of the student population (i.e. >21 years and older), can be regarded as mature students.
5.3.1.2 Race



Figure 5:2 below indicates the racial composition of the respondents in this study.

Figure 5:2 Racial composition of the respondents in the study

From the total sample N=195, the distribution of the first-year student population according to race for the total sample was: 54.4% (106) coloured; 26.7% (n=52) African; 17.4% (n=34) white and 1.5% (n=3) Indian. However, the data indicates a significant difference (p = 0.000) in the three programmes and race. Coloured respondents comprised the largest grouping in each of the three programmes with 58.5% registered in FET, 58.2% in ISP and 48% in FP. More African students (i.e. 36.5%) from the total sample registered for FET than for any of the other programmes. In a similar but contrasting trend, the majority of the total sample of white participants (i.e. 73.5%) registered for ISP. The Indian respondents comprised the smallest grouping in all three programmes.

5.3.1.3 Home language

From the entire sample of first-year students, the data revealed a total of 52.8% of the respondents were English home-language speakers, while 23.6% (n=46) spoke isiXhosa; 21.5% (n=42) spoke Afrikaans and only a small percentage of 2.1% (n=4) spoke other home languages. Language between the three programmes showed a significant difference (p = 0.002). Among the FET group of respondents, English and isiXhosa were the two common home languages spoken: 39.6% and 32.1% respectively. Table 5:2 below summarises the home languages used by the respondents.

	Further Education & Training (EMS)						
				FP	ISP	Total	
	English	Count	21	54	28	103	
	English	% within BEd Programme	39.6%	72.0%	41.8%	52.8%	
	Afrikaans	Count	14	9	19	42	
-anguage		% within BEd Programme	26.4%	12.0%	28.4%	21.5%	
Lang	Lang	Count	17	11	18	46	
	isiXhosa	% within BEd Programme	32.1%	14.7%	26.9%	23.6%	
	Other	Count	1	1	2	4	
	Other	% within BEd Programme	1.9%	1.3%	3.0%	2.1%	
Total		Count	53	75	67	195	
Total		% within BEd Programme	100.0%	100.0%	100.0%	100.0%	

Table 5:2 Home language of respondents in the study

English was the dominant home language among respondents in FP and ISP. For example, FP had the highest number of respondents, 72% (n=54) with English as home language and ISP had 41.8% (n=28). The ISP respondents had a fairly even distribution of respondents with Afrikaans 28.4% (n=19) and isiXhosa 26.9% (n=18)

as home language. The FP group had the lowest respondents with Afrikaans 12% (n=9) and isiXhosa 14.7% (n=11) as home language.

5.3.2 Socio-economic factors

Socio-economic factors in this study refer to the social and economic experiences and realities that shape the respondents' lifestyles and personalities. This section reports on the following factors: academic qualifications of parents, job classification of parents, family support, financial support, and respondents' mode of transport to university.

5.3.2.1 Academic qualifications of parents

Table 5:3 below illustrates the parents' academic qualifications for the total sample in this study.

			Academi	ic qualific	ations o	f parents	
		FE	ĒT	F	Р	IS	P
		Count	%	Count	%	Count	%
Did your father complete a degree?		7	13.2%	14	18.7%	11	16.4%
		46	86.8%	61	81.3%	56	83.6%
Did your mother complete a degree?		3	5.7%	14	18.7%	7	10.4%
		50	94.3%	61	81.3%	60	89.6%

Table 5:3 Parents' academic qualifications

The overall trend of parents' academic qualifications indicates low completion rates by both parents. From the total sample of participants, 83.6% (N=163) of their male parents and 87.7% (N=171) of their female parents did not obtain a university qualification. The percentages of the male parents ranged from 81.3% (n=61) in FP to 86.8% (n=46) in FET and for the female parents, 81.3% (n=61) in FP to 94.3% (n=50) in FET.

Respondents' male parents in the FET and ISP programmes show a higher rate of completion of a university degree than their female parents. For example, in the

category 'yes' to degree completion, the male parent completion for FET was 13.2% (n=7) and for ISP, 16.4% (n=11), while the female parent completion for FET was 5.7% (n=3) and for ISP, 10.4% (n=7). The male and female parents of the FP respondents showed the highest level of education with an equal number of completion rates among both parents, i.e. 18.7% (n=14). The lowest number of parents who had completed university comprised the parents of the FET respondents with only 13.3% (n=7) male parents and 5.7% (n=3) female parents.

Fisher's exact test shows no significant differences across all three programmes with regard to the male parents' level of education (p = 0.758) while the female parents' level of education across the three programmes showed a significance difference (p = 0.079).

5.3.2.2 Job classification of respondents' parents

The guidelines for the job classifications (Appendix G) were extracted from the South African skills structure of the Organising Framework for Occupations (OFO) 2012 (DHET, 2012b). Table 5:4 on page 135 makes use of this classification and categorises the respondents' parents' occupations into the three categories: advanced, intermediate, and entry-level jobs.

From the total sample, N=195, similar numbers of male and female parents were classified with jobs in the advanced category: male parents, 24.4% (n=42) and 22.2% (n=41) female parents. However more men, 30.2% (n=52) were classified as having intermediate skills jobs while more women, 28.6% (n=53) were classified with entry-level jobs.

Table 5:4 on the next page (page 135) indicates that parents of respondents from FP had the highest number of advanced and intermediate skills jobs: male parents involved in advanced skills jobs comprised 31% (n=22) and intermediate skills jobs, 35.6% (n=26). Female parents involved in advanced skills jobs were 33.8 (n=25) and intermediate skills jobs, 25.7% (n=19). By contrast, parents of respondents from FET had the highest number of parents that were unemployed (male parents, 13% and female parents, 19.6%).

			Occupation of parents			
			FET	FP	ISP	Total
	Advanced	Count	5	22	15	42
	Advanced	% within BEd Programme	10.9%	31.0%	27.3%	24.4%
~	Intermediate	Count	14	26	12	52
Father Occupation by Category	Intermediate	% within BEd Programme	30.4%	36.6%	21.8%	30.2%
Cate	Entry (Count	14	7	11	32
o O	Entry	% within BEd Programme	30.4%	9.9%	20.0%	18.6%
ion	Deeeeed	Count	4	2	3	9
Ipati	Deceased	% within BEd Programme	8.7%	2.8%	5.5%	5.2%
	Not Applicable	Count	0	4	4	8
er C	Not Applicable	% within BEd Programme	0.0%	5.6%	7.3%	Total 15 42 % 24.4% 12 52 % 30.2% 11 32 % 18.6% 3 9 % 5.2% 4 8 % 4.7% 3 7 % 4.1% % 4.2% % 12.8% % 12.8% % 12.8% % 12.8% % 12.8% % 12.8% % 12.8% % 13.0% % 22.2% % 13.3 % 17.8% % 28.6% 0 2 % 1.1% 1 3 % 1.6% % 2.7% % 1.6% % 2.7% % 13.0% % 13.0% % 13.0%
ath	Pensioner	Count	3	1	3	7
	Pensionei	% within BEd Programme	6.5%	1.4%	5.5%	4.1%
	Linomployed	Count	6	9	7	22
	Unemployed	% within BEd Programme	13.0%	12.7%	12.7%	12.8%
Total		Count	46	71	55	172
TOLAI		% within BEd Programme	100.0%	100.0%	100.0%	100.0%
	Advanced	Count	5	25	11	41
	Auvanceu	% within BEd Programme	9.8%	33.8%	12.7%12.8%55172100.0%100.0%114118.3%22.2%83313.3%17.8%2053	
	Intermediate	Count	6	19	8	33
>	Internediate	% within BEd Programme	11.8%	25.7%	13.3%	17.8%
lobe	Entry	Count	16	17	20	53
Category	Litty	% within BEd Programme	31.4%	23.0%	33.3%	28.6%
\sim	Deceased	Count	1	1	0	2
tion	Deceased	% within BEd Programme	2.0%	1.4%	154227.3%24.4%125221.8%30.2%113220.0%18.6%395.5%5.2%4.1%87.3%4.7%5.5%4.1%72212.7%12.8%12.7%12.8%114118.3%22.2%13.3%17.8%205333.3%28.6%020.0%1.1%13.3%17.8%205333.3%28.6%020.0%1.1%1.6%31.7%1.6%0.0%2.7%112418.3%13.0%92415.0%13.0%	
Mother Occupation b	Not Applicable	Count	0	2	1	3
Occi		% within BEd Programme	0.0%	2.7%	1.7%	1.6%
ier (Pensioner	Count	3	2	0	5
Aoth		% within BEd Programme	5.9%	2.7%	0.0%	2.7%
2	Unemployed	Count	10	3	11	24
	Onemployed	% within BEd Programme	19.6%	4.1%	18.3%	13.0%
	Housewife	Count	10	5	9	24
	TIOUSEWIIE	% within BEd Programme	19.6%	6.8%	15.0%	13.0%
Total		Count	51	74	60	185
iotai		% within BEd Programme	100.0%	100.0%	100.0%	100.0%

Table 5:4 Parents' job classification of respondents in the study

In addition, Table 5:4 highligh that the male parents of the FET respondents also had the highest number of entry-level jobs, 30.4% (n=14), while female parents of respondents from the ISP programme had the highest entry-level jobs: 33.3% (n=20).

Fisher's exact test shows a significant difference across the three programmes with regard to parents' occupations, with p=0.004 for the female parents and p=0.005 for the male parents.

5.3.2.3 Family support

From the total sample of respondents, 99% (N=192) acknowledged that they had received full support from their families with regard to tertiary studies. All FP respondents indicated they received full support: 100% (n=75), while a high percentage of ISP and FET respondents indicated a positive response to family support: 98.5% (n=65) and 98.1% (n=52) respectively. Table 5:5 below provides information on the nature of family support received by the respondents.

	Number	Percentage
Tangible	62	39%
Financial: paid partial fees; full fees; registration fees	24	15%
Informational : career advice; university selection; context shaping; finding information	25	16%
Assistance: filling in application forms; registration process	13	8%
Intangible	96	61%
Emotional: nurture	21	13%
Personal advice: spiritual; empathy; goal setting	15	9%
Companionship: sense of belonging	6	4%
Encouragement	28	18%
Positive reinforcement	26	16%
TOTAL	158	

Table 5:5 Nature of support from family

Responses from the total sample of first-year students were categorised either as being of a tangible or intangible nature. It is important to note that not all respondents

provided an explanation of the type of support received from their families. From the replies received (N=158), it was evident that most of the respondents received a more intangible (61%) nature of support than a tangible one (39%). The three highest responses with regard to family support were encouragement, 18%; positive reinforcement, 16%; and informational support, 16%.

5.3.2.4 Financial support

From the total sample, the majority of respondents, 93.9% (N=168) were unable to fund their first year of university study. Over 50% of respondents sourced outside funding, with the highest number of respondents receiving bursaries: 43% (N=77) and 10.1% (N=18) securing student loans. The second highest source of payment was from parents: 40.8% (N=73). A small percentage of students, 6.1% (N=11) were able to fund their own first-year university studies. Figure 5:3 below shows the different paths sourced by the respondents to fund their first year of studies.

From the three programmes, the majority of the FP parents, 58% (n=40), paid for their children's first year of study with only a small number, approximately a third,



Figure 5:3 Financial support for first-year university fees

34.7% (n=24) of FP respondents relying on external sources to fund their first year of study. By contrast, close to two- thirds, 71.1% (n=32) of the FET respondents relied more on external sources to fund their first-year study, with a total of 64.4% (n=29) obtaining bursaries and 6.7% (n=3) securing student loans.

Fisher's exact test, p=0.002, shows significant differences between the three programmes and the type of financial support received for university fees.

5.3.2.5 Mode of transport of respondents to university

From the total sample, the majority of the respondents, 71.2% (N=136), relied on public transport to travel to university: 30.9% (N=59) utilised the public taxi service; 22.5% (N=43) used the public bus service; while 17.8% (N=34) used the train. However 18.8% (N=36) respondents used their own cars to commute to university. Figure 5:4 below illustrates this trend.



Figure 5:4 Mode of transport of respondents in the study

Figure 5:4 shows a clear pattern among respondents in the three programmes and the mode of transport used. It is evident from the data and visible in the illustration

above that the majority of FET and ISP respondents made use of public transport, some 35.8% and 34.4% respectively, used a taxi to commute to university, while the majority of FP respondents, 32.4%, used their own cars to commute to university. A small percentage of respondents, 9.9% (N=19), walked to university. The majority of these respondents were from FET, 18.9 (N=10) with 12.5% (N=8) ISP and only one respondent, 1.4%, from FP. These students could be students living close to the university or living in university residences.

5.3.3 Pre-entry academic factors

This section refers to the respondents' prior schooling experience and includes the following sub-categories: type of school attended; year of high school completion; Grade 12 results; first-generation university students; and students' pathway to university.

5.3.3.1 Type of school attended

The South African Department of Education has classified public schools into twocategory systems: either fee-paying or no-fee schools. The No-Fee School Policy (NFSP) was introduced in 2007 with the Department of Basic Education identifying the no-fee schools and providing them with funding (DBE, 2011). Many of the schools that were classified as fee-paying schools were ex-Model C schools: schools classified as white schools during the apartheid era in South Africa and regarded as well resourced in terms of infrastructure, quality of teachers, teaching and learning resources, and government funding.

From the sample, the majority of respondents, 79% (N=154), attended schools in urban areas. The highest number of respondents were from FP, 83% (N=64); second highest, ISP, 76.1% (N=51); and third FET, 73.6% (N=39). With regard to the two-category school fee system, the fee-paying schools in Figure 5:5 on the following page refer to the 'yes' column and no-fee schools to the 'no' column. The 'not applicable' column refers to respondents who have either attended private schools or Further Education and Training (FET) colleges to complete high school.



Figure 5:5 Type of school attended by respondents in the study

From the total sample (N=195), 75.9% of the participants attended no-fee schools; 15.9% attended fee-paying schools and 8.2% attended other schools (private and FET colleges). The majority of respondents who attended fee-paying schools were from FP, 78.8% (n=59), while the highest number of respondents who attended no-fee schools were from ISP, 79.1% (n=53). The largest number of respondents in the 'not applicable' category (private school and FET college) were from FP, 13.3% (n=10).

Fisher's exact test, showed that there was no significant difference between the three programmes and location of schools, p=0.208. However a significant difference was evident between the three programmes and fee school type attended by respondents, p=0.024.

5.3.3.2 Year of high school completion

Figure 5:6 on the next page (page 141) illustrates the period in which respondents in this study completed Grade 12. It is important to note that data was collected from respondents in the year prior to the commencement of their first-year 2012 academic programme.

Results show that the majority of respondents, 57.9% (N=113) completed Grade 12 in the year 2010 and 2011: 31.8% (N=62) and 26.2% (N=51) respectively. Between the periods 2001 to 2009, a total of 37.4% participants completed Grade 12. A total of 26.2% (N=51) of respondents entered first-year university directly from school and can be regarded as school leavers. The gap-year students represent participants who finished Grade 12 in 2010, 31.8%. The FET cohort had the highest number of respondents entering university directly from school, which represented almost half of the FET cohort, 43.3% (n=23). FP had 25.3% (n=19) and ISP had 13.4% (n=9) respondents.



Figure 5:6 Period in which respondents completed Grade 12

5.3.3.3 Grade 12 results

Taking into account the low literacy and numeracy rates among pupils in South African schools, this section focuses mainly on respondents' Grade 12 performance in the following subjects: English as a Home Language (HL) and First Additional Language (FAL); and Mathematics and Mathematics Literacy. A subsequent intention of reviewing these two subjects is the fact that they are mandatory courses for all first-year students in the BEd programme. Students have a choice either to

register for English HL or FAL; and Mathematics, either as a major or an elective. These courses are compulsory courses in the first two years of study. Figure 5:7 below illustrates participants' performance in Grade 12 English HL and FAL within the three programmes.

The majority of the respondents from the total sample obtained a D symbol for both language categories: 50% (N=41) in HL and 33.7% (N=28) in FAL. The second highest scores in all three programmes obtained a C symbol for both language groupings: 28% HL and 30.1% FAL.



Figure 5:7 Grade 12 results for English Home Language and First Additional Language

Fisher's exact test reveals no significant difference between the three programmes and the results obtained from English Home Language (p=0.342) and English First Additional Language (p=0.803).

Mathematics and Mathematics Literacy results: From the study sample a total of 70.2% (N=137) respondents had either taken Mathematics or Mathematics Literacy at Grade 12 level: 41.6% (N=57) respondents completed Grade 12 Mathematics and 58.4% (N=80) respondents completed Grade 12 Mathematics Literacy.

The following two figures, Figure 5:8 below and Figure 5:9 on page 144 illustrate the respondents' Grade 12 results for Mathematics and Mathematics Literacy respectively.

Figure 5:8 illustrates the Grade 12 mathematics results of the 57 respondents. As a total sample, the highest score obtained by the majority of respondents was a C symbol, 28.1% (N=16); the second highest score was an F symbol, 21.1% (N=12)



Figure 5:8 Grade 12 Mathematics results

and the third highest was a D symbol, 19.3% (N=11).

The high frequency of scores obtained for Grade 12 mathematics was located in the lower symbols. As individual programmes, the highest numbers of scores for each programme cohort were: FP, 47.4% (n=9) with a C symbol; ISP, 22.2% (n=4) with a C symbol and FET, 35% (n=7) with a D symbol. Fisher's exact test, p=0.164 indicates no significant difference was identified between respondents in the three programmes and their Mathematics results.

With regard to the Mathematics Literacy results, from the total sample of 80 respondents who completed Mathematics Literacy in Grade 12, the scores were



Figure 5:9 Grade 12 Mathematics Literacy results

much higher than those of the respondents who completed Grade 12 Mathematics. The frequency of the Mathematics Literacy scores was located in the higher symbols, an equal number of respondents, 22.2% (N=18) scored either a B or a C symbol and 15% (N=12) scored A symbols. Within the individual programmes, FET and FP had the highest number of respondents who obtained A, B and C symbols: 81% (n=17) and 76.6% (n=23) respectively. By contrast, ISP had the highest number of respondents, 62% (n=18) who obtained D, E and F symbols. Fisher's exact test showed a significant difference, p=0.001, among the respondents in the three programmes and the Mathematics Literacy scores.

5.3.3.4 First-generation university students

For this study, 'first-generation university student' refers to a student who is the first member of his/her immediate family to attend university. The status of respondents

in terms of being the first in their immediate family to attend university is illustrated in Table 5:6 below.

			Which BE you be	d Program registerin		Total
			FET	FP	ISP	
		Count	36	48	48	132
Are you the first in your	Yes	% within BEd programme	67.9%	64.0%	71.6%	67.7%
immediate family to attend university?	No	Count	17	27	19	63
		% within BEd programme	32.1%	36.0%	28.4%	32.3%
		Count	53	75	67	195
Total		% within BEd programme	100.0%	100.0%	100.0%	100.0%

Table 5:6 First-generation university students

From the total study sample, a high majority of 67.7% (N=132) indicated they were first-generation university students. As evident in Table 5.6 above, FET respondents showed a similar trend to the total study sample: 67.9% (n=36) were first-generation students and 32.1% (n=17) non-first-generation students. Comparing first-generation students among the three programmes reveals the ISP programme had the highest number of first-generation students: 71.6% (n=48). By contrast, FP had a high majority of non-first-generation students: 36% (n=27). Fisher's exact test, p=0.638 indicates no significant difference in the respondents across all three programmes.

5.3.4 Students' pathway to university

Results show that the majority of the respondents, 39% (N=75) were neither at school nor engaged in paid employment a year before commencing university. Only 26.6% (N=51) of the total sample were school leavers who entered university directly after completing high school. Of the 34.4% (N=66) respondents that were working, data confirms they were involved in entry-level paid employment. Table 5:7 on the following page (page 146) illustrates the results.

Year prior to entering the BEd programme at university		FET	FP	ISP	Total
Student	Count	21	20	10	51
Student	% within BEd programme	39.6%	27.0%	15.4%	26.6%
Marking	Count	8	35	23	66
Working	% within BEd programme	15.1%	47.3%	35.4%	34.4%
	Count	24	19	32	75
Unemployed	% within BEd programme	45.3%	25.7%	49.2%	39%
	Count	53	74	65	192
Total	% within BEd programme registering for.	100.0%	100.0%	100.0%	100.0%

Table 5:7 Year prior to entering university

Comparing students' pathways to university among the three programmes, the data displays the FET programme to have the highest number of respondents, 39.6% (n=21) entering university immediately after completing high school. The FP group had the highest number of respondents, 47.3% (n=35) entered university from paid employment, while both the FET and ISP groups indicated a high number of unemployed respondents entering university: 45.3% (n=24) and 49.2% (n=32) respectively. Fisher's p-value=0.000 shows a significant difference in respondents' pathways to university among the three groups.

In spite of the different pathways to university, a high number of respondents, 96.9% (N=189), as illustrated in Table 5:8 on page 147, indicated a positive response to preparedness for university study. A similar pattern was confirmed by the respondents in all three programmes, FP, 97.3%; ISP, 97% and FET, 96%.

			FET	FP	ISP	Total
	Yes	Count	51	73	65	189
Do you feel that you are	162	% within BEd Programme	96.2%	97.3%	97.0%	96.9%
prepared for university life?	No	Count	2	2	2	6
	INO	% within BEd programme	3.8%	2.7%	3.0%	3.1%
Total		Count	53	75	67	195
Total		% within BEd programme	100.0%	100.0%	100.0%	100.0%

Table 5:8 Academic preparedness

5.4 Students' expectations of their first-year experience

This section analyses the scoring patterns of respondents obtained from the expectation survey which formed part of the pre-entry questionnaire. Respondents had to rank each statement from: 'strongly agree', 'agree', 'disagree', to 'strongly disagree'. In this study, the levels of disagreement were collapsed to show a single category of 'disagree' and a similar procedure was followed for the levels of agreement to show a single category of 'agree'. This is allowed because of the acceptable levels of reliability.

In this section, the students' expectation responses are divided into three categories: academic integration, social integration, and student perceptions. All figures below are a summary of the variables that constitute the sub-sets of the three categories. The figures below represent only the level of agreement percentages. To determine whether there was any relation between the three programme cohorts and each statement, Fisher's exact test was used in this section. The test results are shown in the tables below. All *p*-values lower than 0.05 show a significant relation, which implies that differences observed are significant and therefore suggests that different groups do not share the same ideas for a particular statement.

5.4.1 Expectations of academic integration

Academic integration in this study is described as the students' academic associated perceptions and experiences with regard to their learning and activities necessary for

their learning. This section examines respondents' expectations of academic integration under the following sub-headings: academic experiences; academic support; and academic collaborative interaction.

5.4.1.1 Academic experiences

Results show that the levels of agreement are similar across the statements. Some statements have high levels of agreement, while others are low. However, they are consistent because some statements are negatively worded. Figure 5:10 below illustrates the results.



Figure 5:10 Expectations of first-year academic experiences

Fisher's exact test results for the above statements are displayed in Table 5:9 on page 149, the test reveal five statements have *p*-values lower than 0.05, indicating that first-year students' expectations of academic experiences in the three BEd programmes are dissimilar with regard to academic activities. For example, the

percentages for 'Going to schools for teaching practice will be easy and I do not expect to encounter any difficulties' (p = 0.028) for the different groups range from 52.2% for ISP to 73.1 for FET and the percentages for 'My first-year university work load will be easier than school' were all low, ranging from 8% for FP to 15.7% in FET; the *p*-value 0.017 indicated a significant difference amongst expectations in the three programmes.

	Fisher's Exact <i>p</i> -value
Going to schools for teaching practice will be easy and I do not expect to encounter any difficulties	.028
I expect that group work will be an important feature of lectures	.056
My first-year university work load will be easier than school	.017
I expect to spend a lot of time preparing for lectures	.731
During my first year of studies there will only be a few assignments and projects	.009
I expect my first-year courses to be difficult	.018
I expect to write fewer formal tests in my first year of study compared with school	.025

Table 5:9 Academic experiences

5.4.1.2 Academic support

All the results reveal high percentages with regard to students' expectations of academic support statements. The highest agreement scores were for the following statement: 'I will seek academic help from senior students assigned as peer tutors in the various subjects if I have difficulties with my courses'; the percentages ranged from 93.3% for FP to 98.1% for FET. The lowest scores were from the statement relating to the student's awareness of academic services offered at university: 'I am aware of academic support services, e.g., writing centre, peer tutors and tutorials that are available on campus to support students that need help'. The percentages ranged from 75.8% for ISP to 84% for FP. Figure 5:11 on page 150, illustrates the results.



Figure 5:11 Expectations of first-year academic support

Fisher's exact test for the above statements is illustrated in Table 5:10, on page 151. The table shows that none of the *p*-values are lower than 0.05. This implies that the differences observed are not significant and respondents from all three programmes share similar views with regard to academic support.

Table 5:1	0 Academic	support
-----------	------------	---------

	Fisher's Exact <i>p</i> -value
Librarians will be important people to help me with finding information for my assignments and projects	.212
I will seek academic support from my friends whenever I need help with my academic work	.713
I am aware of academic support services: writing centre, peer tutors and tutorials that are available on campus to support students that need help	.620
I will seek academic help from senior students assigned as peer tutors in the various subjects if I have difficulties with my courses	.134

5.4.1.3 Academic collaboration

Figure 5:12 below illustrates the results for academic collaboration. The results reveal that the level of agreement across all statements was high. The majority of respondents were positive and looked forward to participating in class discussions: 'I am looking forward to contributing to lively classroom discussions and debates'. The percentages ranged from 100% for FET to 94.7% for FP. The negative statement scored the lowest, 'It will be a challenge for me to participate in group work during class lessons', implying that a small percentage of the respondents from all three



Figure 5:12 Expectations of first-year academic collaboration

programmes expected group work during class time to be a challenge: 30.8% in FP; 21.6% in FET and 20.9% in ISP. Fisher's *p*-values in Table 5:11 below indicate there were no significant differences between respondents' responses in the three programmes and their views on academic collaboration.

	Fisher's Exact <i>p</i> -value
I expect that group work will be an important feature of lectures	.347
It will be a challenge for me to participate in group work during class lessons	.862
I am looking forward to contributing to lively classroom discussions and debates	.228

Table 5:11 Academic collaboration

5.4.2 Expectations of social integration

Social integration in this study is described as the students' socially associated perceptions and experiences that arise from their social involvement with other students and lecturers in or out of formal lecture time. This section presents the results under the following sub-headings: student's expectations of socialising at university; involvement in out-of-class academic discussions; and students' self-perceived expectations of social integration.

5.4.2.1 Socialising at university

Figure 5:13 on the next page (page 153) illustrates the results of first-year students' expectations of socialising at university. The results for the total sample revealed that positive statements related to making friends: 'I will make many new friends' and 'I will make friends from different racial groups' scored highly: 91.2% and 99.5% respectively. The statements relating to joining social clubs and attending social functions scored a slightly lower score of 88.9% and 89.5% respectively. The negatively worded statement, 'It will be more difficult to make friends at university than it was at school or work', had the lowest level of agreement, implying respondents expected that making friends on campus would be easier than at school.



Figure 5:13 First-year students' expectations of socialising

For the statements from the above figure, Fisher's exact test In Table 5:12 on the following page (page 154) indicates there were no significant differences between the respondents' answers from all three programmes and the statements on socialising at university, with the exception of one statement, 'I will make friends from different racial groups', which indicated a significant difference of p = 0.015. Although the percentages seem close, the test implies that there is a difference in terms of the ISP group's expectations of making friends from other racial backgrounds compared with those of FET and FP groups.

	Fisher's Exact <i>p</i> -value
I expect to join organisations/clubs on campus, e.g. sports club, student union, etc.	.101
I will make many new friends	.280
I will make friends from different racial groups	.015
I will attend many social functions, e.g. sports day, student braai, fund- raising events, etc. in my 1st-year	.827
I will be in contact with campus friends and peers using one or more of the following social networks: Facebook, Email, SMS, MXit	.237
It will be more difficult to make friends at university than it was at school or work	.486

Table 5:12 Expectation on socialising

5.4.2.2 Out-of-class academic discussions

The levels of agreement for respondents from the three programmes were similar for all statements. Respondents' perceptions of lecturers' availability to assist them outof-class time, 'Lecturers will make themselves available outside of the formal lecture time to assist and advise me', scored the highest in all three programmes: 83.6% for ISP, 81.1% for FP, and 80.8% for FET. In contrast respondents' perceptions of being in contact with their lecturers through social networks scored the lowest out of all three statements. The scores for the statement, 'I expect to be in contact with my lecturers using any of the following social networks: Facebook, Email, SMS, MXit' were 63.1% for ISP, 60.8% for FET, and 54.7% for FP. The negatively worded statement, 'I will not be involved in academic discussions with my peers outside of formal lectures', varied slightly with 89% of FP, 75% of FET, and 70% of ISP respondents indicating that they would be involved in academic discussions with their peers outside of formal lectures. Figure 5:14 on page 155 illustrates this result.





In Table 5:13 below, Fisher's exact test indicates that none of the *p*-values is lower than 0.05. This implies that the differences observed were not significant and that respondents in all three programmes shared similar views with regard to out-of-class discussions

	Fisher's Exact <i>p</i> -value
I will not be involved in academic discussions with my peers outside of formal lectures.	.084
Lecturers will make themselves available outside of the formal lecture time to assist and advise me.	.917

Table 5:13	Out-of-class	discussions
------------	---------------------	-------------

to I expect to be in contact with my lecturers using any of the following social .603 networks: Facebook, Email, SMS, MXit

5.4.2.3 Self-perceived expectations of social integration

The results in Figure 5:15 below indicate that respondents' replies for all the statements were consistent. All respondents perceived themselves as being sociable and having a positive reaction to socialising with peers and lecturers outside formal classes. They responded that they felt more comfortable with engaging in conversation with peers than lecturers (89.4% and 71.8% respectively). Respondents perceived themselves to be focused and not easily distracted by socialising at university, as indicated in the results for the following statement, 'Joining social clubs/organisations at university will distract me from academic work'. The low levels of agreement: FP, 14.7%; FET, 21.2%; and ISP, 25.4% confirm the respondents' level of focus. The results indicated a stronger response with regard to the following statement, 'My friends can easily persuade me to bunk lectures', showing a low level of agreement from all groups: 0% (FET), 2.7% (FP), and 4.5% (ISP). The above result indicates that respondents perceived themselves as not easily influenced by their friends.



Figure 5:15 Self-perceived expectations of social integration

From Table 5:14 below, Fisher's exact test shows no significant difference for all the statements on students' perceptions of social integration with the exception of the following statement, 'Making friends from different racial groups would be difficult for me', which showed a significant difference, p=0.023. The percentages of level of agreement ranged from 2.7% in FP, 7.7% in FET, and 8.9% in ISP.

	Fisher's Exact <i>p</i> -value
I am shy and will not personally introduce myself to my lecturers outside of class.	.942
Having conversations with my class peers outside the classroom is easy for me.	.889
Having a conversation with my lecturer outside of formal class will be easy for me.	.828
Making friends from different racial groups would be difficult for me.	.023
My friends can easily persuade me to bunk lectures.	.376
Joining social clubs/organisations at university will distract me from academic work.	.269

5.4.3 Students' perceptions

The students' perceptions of academic staff are presented under two sub-headings: students' perceptions of what their lecturers expect of them as first-year students, and the students' self-perceived expectations of what is required of them from their lecturers, with regard to academic activities.

5.4.3.1 Perceptions of lecturers' expectations

The results show that the respondents' replies for both the positively worded statements and negatively worded statement, were consistent. All the positive statements scored high percentages, ranging from 100% to 94% and the negative statement: 'My lecturers will not expect me to attend every single lecture', scored low percentages ranging from 1.5% to 5.8%.



Figure 5:16 First-year students' perceptions of lecturer expectations

Figure 5:16 illustrates that in all three programmes, respondents' perceptions of their lecturers' expectations of their academic activities were high. Fisher's exact test indicates the same outcome observed in Table 5:15 below. None of the *p*-values was lower than 0.05, implying that respondents shared similar views with regard to their perceptions of lecturer expectations.

	Fisher's Exact <i>p</i> - value
My lecturers will expect me to write well-structured academic essays	.072
My lecturers will expect me to know how to correctly reference my assignments and projects	.066
My lecturers will not expect me to attend every single lecture	.549
My lecturers will expect me to take part in class discussions and debates	.833
My lecturers will expect me to be able to use the computer for assignments and projects	.940

Table 5:15 Perce	ptions of lecture	^r expectations
		onpoolationo

5.4.3.2 Self-perceived expectations of lecturers

The results in Figure 5:17 below show that all respondents rated the following two statements highly: 'I expect that my lecturers will provide regular feedback to me on class work, assignments and tests' and 'I expect the lecturers to prepare/equip me before I go to schools for my first teaching practice session'. The scores for these statements ranged from 86.6% to 96.2%. However, respondents' reactions to lecturers' knowing them by their first name scored the lowest, with the following responses from the three programmes: 55.8% in FET, 68.7% in ISP, and 72% in FP.



Figure 5:17 First-year students' self-perceived expectations of lecturers

None of the *p*-values was lower than 0.05 in the statements below, implying that there was no significant difference in the way the different groups responded to the statements in Table 5:16 below.

	Fisher's Exact <i>p</i> -value
Lecturers will not know me and other students by their first name	.501
Lecturers will make themselves available outside of the formal lecture time to assist and advise me	.764
I expect the lecturers to prepare/equip me before I go to schools for my first teaching practice session	.633
I expect that my lecturers will provide regular feedback to me on class work, assignments and tests	.081

Table 5:16 Expectations of lecturers

5.5 Summary of findings

The quantitative findings are summarised under the following two headings: Who are the first-year students entering teacher education programmes? and What are firstyear students' expectations of university life?

5.5.1 Who are the first-year students entering these teacher education programmes?

The data for this section is discussed under the following sub-headings: demographic factors, socio-economic factors, and pre-entry academic factors.

Demographics factors: The demographic factors took into account the age, gender, race, and home language of the sample. From the total sample of 195 respondents, the gender ratio for male and female participants was approximately 1:3 with a total of 45 men and 150 women. The findings reveal the majority of male respondents (57.7%) preferred the FET programme, while the majority of female respondents (48.6%) preferred the FP programme. The age of participants from the total sample ranged from 19 to 45 years. Just over 50% of the sample population represented a young student population (53.3%, N=104) which belonged to the lower than 20 years of age category, while the balance represented 20 years and older. These results indicate that there is a growing number of mature students returning to formal

education. The growing diversity in the student population was evident in the racial composition of first-year students. The majority of respondents in the sample were identified as coloured (54%), with Africans representing 26.7%, white participants constituting 17.4%, and Indian only 1.5%. An interesting finding is that certain race groups appear to dominate a particular BEd programme. For example, of the total sample of 52 African participants, 36.5% were registered in the FET programme; of the 34 white students, 73.5% were registered for the FP programme; and from a total of 106 coloured participants, 36.7% were registered in the ISP programme. Finally, from the total sample, the majority of the participants were English first-language speakers (52.8%), with isiXhosa being the second largest and Afrikaans the third largest at 23.6% and 21.5% respectively.

Socio-economic factors: This section summarises the quantitative data from parents' education, parents' employment type, financial support, mode of transport, and family support. The overall trend in parents' education reveals that the majority of the parents of the respondents, 83.6% of the male parents and 87.7% of the female parents, do not possess a tertiary qualification. The findings disclose that parents of FP participants showed the highest level of completion rates among the total sample of participants' male and female parents: 43.7% and 58.3% respectively. In this regard the finding of parents' employment type clearly signposts that parents of FP participants had the highest number of advanced and intermediate skills jobs, while the male parents of FET participants had the highest unemployment numbers.

The participants' parents' qualifications and jobs impinge on financial support. The majority of the participants (93.9%) were unable to fund their first year of university studies. Over 50% of participants sourced outside funding: 43% received bursaries while 10% secured student loans. From the three programmes, the findings indicate that the majority of FP participants (58%) had their studies paid for by their parents with only a small number, approximately around a third (34.7%), relying on external sources of funding. By contrast, close to two-thirds (71.1%) of FET respondents relied more on external sources to fund their first-year studies. The majority of the total sample population (71.2%) relied on public transport to commute to university. A small percentage of 18.8% made use of private transport and 9.9% walked to

university. Of the total sample of 36 participants using their own vehicles to travel to university, the percentages of participants within the three programmes are as follows: 66.6% FP, 19.6% ISP, and 13.8% FET. Finally, the type of support received by the participants from their immediate family reveals that most of the support was intangible (61%). For example, participants acknowledged their families for the following intangible support: emotional, encouragement, positive reinforcement; personal advice, and companionship. The nature of the tangible support (31%) was mainly financial, informational and assisting with registration.

Pre-entry academic factors: This section focuses on type of schooling; year of school completion; Grade 12 results; first-generation students; and student pathways to university. From the total sample, 79% of participants attended urban school while only 21% attended rural schools. Results indicate that the majority of FP respondents attended fee-paying schools (ex-Model C schools), while the highest number of respondents from the ISP cohort attended no-fee schools. The findings disclose that the majority of respondents from the total sample (57.9%) completed Grade 12 in 2010 and 2011. From the 2011 cohort, 26.2% could be referred to as school leavers: they entered university directly after completing Grade 12. From the 2010 cohort, 31.8% could be regarded as gap-year students since they entered university a year after completing matric. Between the periods 2001 to 2009, and the period prior to 2000, a total of 37.4% and 4.6% participants respectively completed Grade 12. These participants could be referred to as mature students.

The data on participants' Grade 12 academic performance reveals average scores for English Language and Mathematics. The majority of participants from the total sample obtained a D symbol for English Home Language and First Additional Language. The results of the participants that completed Grade 12 Mathematics Literacy revealed a high frequency in the upper scores: A, B and C category. Participants who completed Mathematics in Grade 12 showed a high frequency in the lower scores: D, E and F. The findings further disclosed that the majority of the participants in the sample were first-generation students (67.7%). From the total sample, ISP had the highest percentage (36.3%) of first-generation students, while FP had the highest percentage (42.8%) of non-first-generation students. Finally, although the majority of participants were not school leavers (34.4% employed in fulltime jobs and 39% unemployed) prior to entering university, a substantive number of participants (96.9%) responded positively to preparedness for university studies.

5.5.2 What are these first-year students' expectations of university life?

The findings from this section were drawn from the Likert scale survey on student expectations of first-year university experience. The summary of the data for this section includes the following sub-headings: students' expectations of academic integration; students' expectations of social integration; and students' perceptions of academic staff.

Expectations of academic integration: The findings are summarised under the following divisions of expectations: academic experience, academic support, and academic collaboration. The statements referring to first-year students' expectations of academic experience consisted of seven statements. The participants within the three programmes shared similar views on only two statements: importance of group work activity in class and time spent on preparing for class. Students' expectations of five statements showed a significant difference in views: difficulty of teaching practice; difficulty of course; number of formal tests written; number of assignments; and work load. These results indicate that many participants' expectations of academic experience were varied, thereby indicating a discrepancy against their actual experiences.

With regard to participants' response to the four academic support statements, the findings reveal there was no significant difference among the three programmes. This implies that all students shared the same view with regard to their expectations of academic support. Although all students scored highly on the statements referring to academic support, all had equally low scores on the statement of their awareness of the different academic support services offered by the university. The findings bring to light the willingness of the participants to make use of academic support to improve their performance.

There was no significant difference in academic collaboration between the participants in the three programmes. All participants felt positive about class discussions and debates, but some participants were apprehensive and scored a low response for the statement on 'participating in group work will be a challenge'.

Expectations of social integration: participants' expectations are summarised under the following divisions: socialising at university; out-of-class discussions; and self-perceived expectations of social integration.

Overall the responses from participants in all three programmes had a high level of agreement with the statements on socialising at university. There was no significant difference for four of the five statements. One statement that indicated a significant difference of p=0.015 related to making friends from different race groups. The ISP responses varied from the responses of FET and FP.

The out-of-class statements did not score as highly compared with the academic integration statements. Low levels of agreement were found in all of the out-of-class statements: contact with lecturers on social media; lecturers' availability out-of-class time; and out-of-class discussions with lecturers and peers. All the statements showed no significant difference among the three programmes, implying that all participants in the three programmes shared a similar view.

From the six statements on participants' self-perceived expectations of social integration, five statements showed no significant difference. This implies that participants shared similar views on these four statements that were highly rated. The one statement that showed a significant difference of p=0.023 related to 'making friends from different race groups may be difficult for me'. The FP score varied from FET and ISP.

Expectations of academic staff: The summary of the findings of this section is addressed under two areas: students' perceptions of lecturers' expectations; and students' self-perceived expectations of lecturers.

There was no significant difference between the five statements relating to students' perceptions of lecturers' expectations and participants' rating in the three programmes. The statements related to the following perceptions of lecturers' expectations of participants: use of computer; class participation; class attendance; correct referencing; writing well-structured essays. Participants rated all the positive statements with high levels of agreement and the negative statement with low levels of agreement. The results reveal that participants shared similar views with regard to their lecturers' expectations of them as first-year students.

Finally, the results from the statements on students' self-perceived expectations of their lecturers indicated there were no significant differences between these statements and the participants in the three programmes. All participants shared similar views with regard to the following statements: regular feedback from lecturers; lecturers equipping students for first teaching practice; lecturers' availability; lecturers' knowing their names. Although all statements were rated highly, participants did score one statement relatively low compared with the other statements: lecturers knowing the students by their first name. These results disclose that participants had high expectations of their lecturers.

5.6 Concluding remarks: who are the first year students entering these teacher education programmes?

The use of descriptive statistics together with the use of figures, tables, crosstabulations and inference statistics provided a more detailed and realistic overview of the diversity of the respondents entering first-year teacher education programmes. The quantitative data provided a good synopsis of who our first-year teacher education students are, highlighting student demographics, socio-economic status, prior schooling, and expectations of the first-year university experience. Results from this chapter provided a backdrop to this study and allowed for a more in-depth qualitative investigation of first-year students and factors influencing their university experience and academic performance. The qualitative findings are presented in the next chapter, Chapter 6. Discussions from both the quantitative and qualitative data are discussed in Chapter 7.

CHAPTER 6 QUALITATIVE FINDINGS

6.1 Introduction

This chapter presents qualitative findings obtained from one-on-one interviews which included a sort-card activity and data from the focus-group interview. The qualitative process was designed to obtain a more in-depth understanding of pre-entry factors influencing students' first-year experience and academic performance. One-on-one interviews with the eight participants, who were purposefully selected, allowed me to probe more deeply into participants' backgrounds to gain a better understanding of how their habitus and different forms of capital supported and/or challenged their first-year integration.

Chapter 5, to a degree, addressed the first two research questions. The primary purpose of the quantitative data was to gain an overview of the profile of first-year students entering the teacher education programme. In this chapter all four of the sub-questions are addressed to provide a more in-depth understanding of the phenomenon and to answer the main research question with more clarity and focus. Thus the following research questions will be considered when presenting and discussing the qualitative data analysis:

Main RQ: How do pre-entry academic and non-academic factors influence teacher education students' first-year experience and academic performance?

The following sub-questions were addressed:

- Sub-RQ1: Who are our first-year teacher education students?
- Sub-RQ2: What are the pre-entry academic and non-academic factors that firstyear teacher education students bring with them to higher education?
- Sub-RQ3: To what extent do pre-entry factors that students report influence their social and academic interaction in their first year of study?
- Sub-RQ4: To what extent do these factors contribute to the students' first-year experience and academic performance?
6.2 Presentation of qualitative data

The qualitative findings are presented and discussed under three main sections in the remainder of this chapter:

- The first section provides a description of each of the eight participants' backgrounds including their habitus and capital: social, cultural, and economic.
- The second section discusses the results of the sort-card activity and factors influencing participants' first-year experience.
- The final section discusses themes and sub-themes that emerged from the one-on-one and focus-group interviews. The five main concepts from the conceptual framework: habitus, capital, field (institutional context), academic integration, and social integration, provided a lens through which to analyse and understand the data.

6.3 Description of participants

Providing a description of each participant is a fundamental feature of qualitative research. Patton (2002:437) acknowledges that 'a thick, rich description provides the foundation for qualitative analysis and reporting'. In this study, a detailed description of each participant provides a context in which to analyse the influences of their preentry factors, including habitus and forms of capital in relation to the way each participant reported on his or her individual first-year experience. The rich description also assists by putting into context the quotations used in the latter part of this chapter. This section provides a description of all eight participants by highlighting one broad theme for each participant. The broad theme selected is not exclusive to the selected participant only. The broad themes identified give rise to several subthemes that focus on the complexities of the participant's experiences as parts of a whole. This section concludes with a table providing an overview of the participants, together with broad themes and sub-themes that emerged from the data. These themes and sub-themes are expanded on, taking into consideration participants' one-on-one interviews and focus-group interview discussed later in this chapter. Pseudonyms were used to ensure the anonymity of the participants.

6.3.1 Joyce

Theme: Determination – I am here to get a degree.

Joyce is a 22-year-old African, female student. She was born in the Eastern Cape. There are five members of her family: her parents, one sister, her twin brother and herself. Joyce is from a working-class family. Her parents separated when she was five years old and she does not have much recollection of her time with her father. For most of her childhood she was in the care of her grandmother since her mother moved to Johannesburg to increase her chances of finding a job. When Joyce was eleven years old, her grandmother died. She and her siblings had to move to Johannesburg to live with their mother. A year later, Joyce's mother had died. Once again Joyce and her younger siblings had to relocate, and they moved to Cape Town to live with her aunt.

While in Cape Town, Joyce was fortunate enough to gain a scholarship to attend a private high school in Cape Town. She considered this experience to be a lifechanging experience. Everything she learned about life was gained from her experiences at this school. While at high school Joyce fell pregnant and her son was born in 2006. Her aunt was disappointed with her and requested she move out of the house. Joyce moved in with her son's father and his family. In spite of these challenges, Joyce completed Grade 12 in 2008 and enrolled at university the following year. With limited knowledge and advice on what career path to follow, Joyce within a few months into her first year of study dropped out of university. She found herself a job as a salesperson in a departmental store. Joyce's life was not without personal trauma. Her twin brother died in 2007 and her sister in 2010. She felt alone and her only solace was from her boyfriend and son: the only two people she now regards as family.

Joyce embraces her culture and traditions, but she has a conflict with how her culture expects her to behave and think. She stated that in her culture, grandparents and parents 'groomed' them to respect elders and other people. The elders in the family prescribe how to relate to people. Joyce feels such prescription becomes a 'personal thing' as you grow up. She feels that growing up in today's society, being

exposed to different races or cultures and observing how people perceive themselves, encourage you to see people in different ways.

Joyce recalls missing out on her childhood and taking on adult responsibilities at a young age. Now that she has a son, Joyce feels she needs to provide for him. She does not want him to go through the same hardships that she experienced as a child. Her decision to go back to university, to increase her chance of getting a better job and uplifting their lives, was based on this belief. In 2012 Joyce was back at university and enrolled in the BEd programme. She felt confident about her studies and knew exactly what she wanted from life. Her decision to go back to university was an easy one: she wanted a better life for herself and her son. She felt teaching was the best job for her now that she had a son. Joyce feels secure in her choice of career and highlights that the country will always need teachers. Joyce is aware that among her race group there is no respect for the teaching profession, but she is determined to make this her career and successfully complete her studies.

Joyce is the first in her family to attend university. She wishes her mom were alive to witness this event. She is enjoying her campus experience and feels comfortable on campus. Talking to her lecturers and peers comes quite naturally to her: being in a private school has prepared her for university life. Joyce was surprised at the multi-racial composition of the student population in her course. She certainly did not expect to see so many white students at this university studying to become teachers. Her perception was that white students would choose better courses since they had better education and would certainly be attending the more 'prestigious' universities. From Joyce's previous university experience she expected more autonomy as a student. But, she is happy to have her lecturers fully engaged in her studies.

Joyce feels confident that her school experience has equipped her to adapt academically and socially to the university environment. She has experienced schooling at a township school and a private high school and therefore can clearly see the disparities in the quality of education in the two types of school environments. She pities students with township education who are entering university. She believes that they are entering university without a good educational experience and will therefore struggle to fit in. She believes that township school students are not equipped with the 'right attitude towards their work'. She is strategic

169

when it comes to choosing peers for group work and projects. She is determined to succeed in her studies and wants nothing or nobody to hold her back. She emphasises that her choice of peers for group assignments is determined by her peers' prior schooling and their commitment to their studies.

Joyce feels comfortable communicating with diverse students in her course. She feels that having been in a good school which had students from different race groups equipped her to engage with students from different race groups. She feels privileged in this regard. She observed that her African peers do not feel comfortable mixing with students from other race groups. She attributes this situation to their home environment and prior school experience. Joyce is focused on her studies and regards herself as a motivated and determined person. She tries to instil some of these qualities in her African peers. She feels they need to be more focused and know the privileges and opportunities of attending university.

Joyce does not have many friends at university: her main focus is on her studies and her son. This has resulted in her restricting her time for friends or socialising although she does regret missing out on her childhood when she was growing up. Joyce believes if she wants to be successful in her studies she needs to make certain sacrifices. She has no issues with time management and effectively manages to balance her university work, duties at home and responsibilities as a mum. Joyce is grateful that she does not need to work part-time to support her family or pay for her studies. She received a bursary for her studies which helped ease her responsibilities as a parent and student. Joyce's financial stability has afforded her the privilege to dedicate herself to successfully completing her studies and obtaining her teaching degree so that she can take care of her son and herself.

6.3.2 Thabo

Theme: Resilience – The need to be financially stable.

Thabo is a 24-year-old African male student from the Eastern Cape. He is from a working-class family and was raised by his mother who was a domestic worker. His father died when he was young; his mother took care of his older sister and him. His mother died of HIV/Aids when Thabo was ten years old. Thabo and his sister went to live with his aunt. He attended a public school in a small rural town. After the death

of his mother, Thabo and his sister were cared for by his extended family. They did not have a permanent home and moved around between their three aunts and grandmother. Thabo's childhood memories are of economic hardships and social ills. He was surrounded by uncles who came home at weekends and engaged in drinking alcohol, smoking drugs and fighting. He was grateful that they gave him pocket money. Thabo used the money to buy cigarettes, which he sold to pupils at his school to make some money to purchase his school uniform and stationery.

When Thabo turned 18, he moved to Cape Town to look for a job to support his sister, aunts and grandmother. While working, he completed a part-time computer course to procure a better job and send more money home to his extended family. He worked for a company doing data capturing and managed to save some money. When Thabo was retrenched from his job, he decided he needed to obtain a stable job with a good salary. These circumstances motivated him to attend university as a fulltime student. His decision to enrol for a BEd degree was influenced by his aunt. She told him he should be a teacher because she had watched him take care of her children and do homework with them. His passion, however, is to be a psychologist. Thabo is the first in his family to attend university. He applied for a student loan and hopes he will get it: he finds it difficult to balance being a fulltime student and having a part-time job. He works night shifts as a security guard to try and support himself and his family back in the Eastern Cape. He says it does not really affect his concentration because he tries to get a quick nap now and then at work. He finds the academic work somewhat challenging: he regards himself as a mature student who completed Grade 12 in 2006. He believes that his rural school education did not equip him for university life. His teachers at school never motivated them to study further. He recalls his teachers being content if the learners at school reached Grade 12. He has a few peers at university whom he regards as friends. They all work together and help one another, but they don't socialise away from campus. There is little time or money for socialising since he has to juggle his fulltime studies and parttime job commitments. However, Thabo is focused on his goal of completing his degree and getting a permanent, secure job. He feels positive, motivated and selfdriven. He says his tough upbringing with limited resources, money, food, good clothes and a stable home, has made him deal with challenging situations. Thabo's behaviour is governed by his strong cultural and traditional beliefs which influence

the way he interacts with his peers and lecturers. He does not make contact with his lecturers unless they request to see him. He regards them as elders. In his culture you do not have much contact with your elders. He believes his lecturers should be respected and not approached unnecessarily. This behaviour affects his interaction with his lecturers in and out of class; he will not debate or challenge them in any way; neither is he confrontational about issues relating to his academic work.

Thabo's restricted exposure to other race groups due to his rural schooling experience and home environment resulted in him having difficulties in interacting with students and lecturers from other race groups. Initially, when Thabo arrived at university, he found this situation awkward and soon found out that many of his African peers felt the same way. Thabo and his African peers/friends discussed this issue and realised that these students from other race groups were students just like them, with similar academic concerns and goals. Thabo has a few friends from other race groups that he chats to in class, but his friends look at him strangely when he gets too 'friendly' with them. They pass funny remarks, laugh and sometimes call him a 'coconut'. Thabo says this banter does not bother him as it is a new experience for him and his friends. Thabo is passing all his courses, but struggles with English language studies. He is not doing well in academic literacy: he is afraid he is going to fail that course. He says lecturers assume that students know how to write an essay or do referencing. Not much support is given in class and he cannot make use of academic support services because he is involved in part-time employment that demands a lot of his time. Thabo sometimes wishes he did not need to be so selfreliant and had hoped that his extended family would contribute towards his education. He firmly believes that if he was financially stable he would have been able to focus on his studies and obtain better marks instead of worrying about supporting himself, his extended family and his studies. Thabo is hopeful that next year he will get a student loan or bursary if his first-year results are good. This will allow him to complete his studies in the required time of four years and then he can look forward to a well-paid stable job as a teacher. He says his family would be proud of his achievement: this would allow them to brag about his success in their home town, but more importantly he would be able to take care of them financially.

6.3.3 Letti

Theme: Determination – Self-improvement: key to being someone.

Letti is a young 27-year-old African female student. She belongs to a working-class family. Letti's father is a labourer on a wine farm in Stellenbosch on which she and her five siblings were raised. Her mother is a seasonal worker on the wine farm but her main responsibility is to take care of the household duties. Letti presently has two older sisters and two younger brothers. In 1999 her eldest sister was shot dead in a tragic incident which was traumatic for the whole family. Letti does not feel comfortable talking about the incident. In spite of this traumatic incident, Letti recalls her childhood as being fun and exciting. Her parents were happy people who always wanted the best for their children. Letti regards her father as a hard worker, who has never missed a day at work. Letti grew up in an extended family environment. Her parents took on the responsibility of raising two of their daughters' children and Letti's cousin when her uncle died. She proudly acknowledges that her mother is presently taking care of eight boys and three girls, including herself.

Letti's mother tongue is Sesotho, yet she regards Afrikaans as her first language. She attended schools where Afrikaans was the language of instruction. She recalls her primary school experience with mixed emotions. Letti attended a coloured primary school during the time of apartheid in South Africa. She remembers the racial tension and being subjected to racial slurs from other learners at school. Her fondest memories of primary school were taking part in beauty pageants and winning prizes. She ascribes the pageants to contributing to her sense of self-confidence. Letti's recollection of high school is one filled with gratitude and joy. She regards her high school experience as most memorable. She attended a Model C high school, which was regarded as a white high school during apartheid. It was an all-girls school with the majority of students and staff white. She recalls a total of four black students in her school. She is grateful to her parents who had sent her and her sister to this school. Letti is aware of the financial sacrifice her family had to make to send her and a sister to such a well-resourced school with high school fees. She felt guilty about being selected to go to this school. She did not want to waste her parents' money and believed the only way she could show her appreciation to her family was

by working diligently and passing each year. Letti did not regard herself as a bright student – rather as a dedicated and hard-working one.

She appreciated being in a well-resourced school and exposed to new experiences and cultures. She found her teachers to be highly motivated: they encouraged learners to study further. The vice-principal of her school inspired her to be career oriented and she played a major role as Letti's mentor and advisor. Letti completed Grade 12: many of her friends were going to university, yet, she chose to work and contribute financially to her family. She felt guilty about spending her parents' savings. She worked as a cashier and a waitress for two years. Letti soon realised that these types of entry-level jobs would not give her the stability and security that she needed to better her standard of living. She approached the vice-principal of her high school, who advised her to consider furthering her studies. The vice-principal also secured a sponsor for Letti's first year of study. Letti enrolled in a college close to home and after a successful first year, she was given a bursary to complete her diploma in Early Childhood Development (ECD). Letti found a job at a primary school close to home and taught Grade R for two years. She taught in English which was a challenge for her, but one that she was willing to embrace. Letti was earning a good salary and had a stable job. She was a motivated and ambitious person. She wanted more from her life; she felt she needed to better her qualifications and decided to enrol at a university to obtain a teaching degree.

In 2012 Letti arrived in Cape Town to attend university. She felt nervous, anxious and scared. She had only heard about university from her friends, but had never visited one. In addition this was her first experience of being away from her home and family. Although Letti was confident that her high school experience, her training in ECD and her teaching experience as a Grade R teacher would be a sound support in her university studies, she still felt nervous owing to being unfamiliar with university expectations. She was afraid of failing her first year, but to her surprise, she passed all her courses with good grades. Her expectations of university were not quite what she experienced. She expected university work to be much tougher than she had experienced. She did not expect to see so many mature students at university. She easily fitted in with the older students and expressed that she 'liked it as it made her feel comfortable'. In this regard, Letti felt she could identify with the

mature students; they were more focused and determined to succeed, just as she was.

Letti regarded herself as shy and would not ask questions in class. She felt being an Afrikaans first-language speaker also contributed to her non-participation in class. She considered the language barrier confusing at times: it restricted her from answering questions in class since she did not want her answers to come out incorrectly and cause her to look 'stupid'. Letti is aware that non-participation in class may sometimes be misinterpreted by some of her lecturers as her not understanding the academic work.

Letti stays at the university residence and has good support from her peers at the residence with regard to her academic work. They help one another and have formed a study group. They do not socialise out of residence: going out to movies, clubs or dinner. Financially, she and her friends cannot afford to engage in such activities. Letti paid for her first year of studies and residence fees with the money she had saved from working as a Grade R teacher. However, Letti is concerned that her savings will soon be exhausted. She has applied for a bursary and hopes her application will be successful. She urgently needs extra money for food, to purchase stationery and other resources needed for academic projects. Letti goes home on weekends: she has a part-time job as a cashier at the farm stall on the wine farm where her father is employed.

Letti's father is so proud of her and talks highly of her to the people in the community. She is the first in her family to attend university. Letti feels honoured when she is in her community and children refer to her as 'teacher'. Letti wants to be successful in her studies so that she can take care of her parents, buy them a house and make them feel comfortable and secure.

6.3.4 Nadia

Theme: Fitting in – This is where I am supposed to be.

Nadia is a 20-year-old coloured student who completed matric in 2010 and took a gap year before enrolling at university in 2012. She is from a middle-class family. There are five members in her family: her parents, younger brother, sister and she.

Nadia was born in Mitchell's Plain, an area once classified as a coloured residential area during the apartheid era, but soon moved to Ottery which she regards as a good area. It was designated a white residential area before 1994. She recalls her childhood as perfect, attending many family functions and outings. Nadia and her siblings attended a good primary school, stating it was an ex-Model C school. She remembers her first day at primary school, quite vividly, being accompanied by her mother. She remembers the period in which her perfect life was ruined. She was twelve years old when her father walked out on their family. Nadia recalls that moment as being 'sad, disappointing, emotional and crucial', a day she will never forget. Nadia attended Belgravia High school which she regards as a well-resourced school with dedicated teachers who prepared them adequately for university life. She recalls her teachers motivating and encouraging them to work hard. Her teachers informed them that university studies are much tougher than school. In 2010, Nadia delivered the valedictory speech on behalf of the Grade 12 class. She was proud of her achievements and matriculated with good results. It was Nadia's intention to attend university but she decided to take a gap year. In that year she obtained her driver's licence and worked at Edu-care, a childcare centre. It was work experience gained at this centre that inspired her to pursue a career in teacher education.

Nadia is not the first in her family to attend university: her mother has a teaching qualification. Nadia feels comfortable at university: her expectations of campus life were influenced by her mother and friends at university. She feels her school experience prepared her for university study, but more importantly her mother being a teacher and having done the same degree is able to give her sound advice and help her with her course work. Nadia and her mother have daily conversations about her activities on campus. Nadia's mother provides emotional and instructional support. She goes through day-to-day university activities with ease. She does not feel stressed or pressured and regards her academic performance as mediocre. Nadia admits that she is lazy and leaves work till the last minute and thus does not get the high marks that she has the ability to obtain. She acknowledges she does not put in the effort needed to produce good results: if she devoted more time to her studies this would be possible.

Nadia has many friends at university. She regards herself as a social person and therefore making friends on campus was quite easy for her. She feels social interaction is an important part of university life. Nadia believes that having friends at university provides a source of information when it comes to academic work. Nadia regards herself as a sociable person, yet in class she regards herself as shy. She does not feel confident enough to participate in or contribute to class discussions, but adds that if she was directly asked a question she would respond. She asserts that she is not afraid to approach her lecturers if she has any academic issues. Nadia is enjoying her first-year university experience and is determined to be successful even though she is just passing her courses. She admits she needs to create a better balance between her social and academic life: she feels challenged when it comes to managing her time with regard to academic activities.

In her personal life, she is trying to balance her home commitments and academic work. Nadia and her mother are constantly quarrelling about house chores. Nadia refuses to do chores around the house because she feels her studies should take priority over house chores, which she regards as her mother's responsibility. This constant bickering has put a strain on their relationship although Nadia does appreciate that it is difficult for her mother, who is a single parent and the bread-winner of the family. Notwithstanding, Nadia feels her mother needs to be more tolerant of her situation as a student. In addition, Nadia's estranged father has made contact with her on Facebook. Nadia refuses to respond to his request: she believes the first year of study is crucial, it requires her full attention and therefore precludes her father's request.

Financially, Nadia seems to have little anxiety. She is grateful that she does not have to worry about money issues. She is well catered for at home and all her university expenses are covered by her bursary, which she acknowledges gives her financial independence. Nadia is aware that without the bursary it would have been difficult to pay for her university fees since her mother is the only person with an income in her family. Nadia feels she needs to succeed in her studies as she is mindful of her bursary and feels an obligation to do well. She feels the sponsors of the bursary have invested in her and she needs to succeed.

6.3.5 Alley

Theme: Out of habitus experience – Culture shock.

Alley is a 21-year-old coloured, female student from Cape Town. She is from a middle-class family and she lives at home with her mother, stepfather and two younger brothers. Her brothers are from her mother's marriage to her stepfather whom Alley regards as her 'own father': he is the only father she knows. Her parents divorced when she was three years old and her mother married her stepfather when she was four years old. Alley's biological father died when she was seven. Alley is the first in her family to attend university. Neither of her parents has ever been to university and they do not have any formal tertiary qualifications. Alley describes her upbringing as being well structured: she was taught to be disciplined and to respect people. She was brought up in a religious household: all members of her family are actively involved in the church. Alley herself is a group leader at her church. Her upbringing was guided by the teachings from the Bible and she adds that this has given her focus and a purpose in life.

Alley completed Grade 12 in 2009. She attended a Christian school and had nuns as her teachers. Alley enjoyed her schooling. She believes that having been in a Christian school environment, which had a strict work ethic, has not only prepared her for university studies and the real world, but has given her boundaries in which to live her life. Her teachers at school had constantly told them that university was not like school and that they had to work independently at university: 'nobody is going to baby you like we do at school'. Her school experience had taught her to be disciplined and focused. At the end of Grade 12, Alley decided to take a gap year which turned into two years before going to university. Alley was not sure what career path she would pursue and therefore needed the time off. She did a lot of volunteer work in the first gap year. During the second gap year she volunteered as a teacher assistant at a pre-school that her step-brother attended. It was this experience that made her realise her calling in life: she wanted to be a teacher to make a difference to the lives of the next generation. Alley consulted with her mother, her aunt (who was a teacher), her pastor at the church and friends to ensure that she was on the right career path. Her decision was supported by her family and

friends. Her friends also assisted her in choosing a university that was best suited to her choice of course.

Alley entered the first year of university with mixed feelings. Although she was motivated and goal oriented, she felt nervous, scared and anxious: she had been out of the learning environment for two years. Alley was unsure as to whether she would be able to fit into her new environment. Alley's strict upbringing, strong religious beliefs and rigid school experience influenced the way she perceived university life. She anticipated that the university environment would be similar to the military with strict rules and procedures to be followed with regard to academic work and social behaviour. She was taken aback during orientation when issues with regard to HIV/Aids, pregnancy, and health and personal services offered at the university were discussed. She could not believe that she would be interacting with students who, although the same age as herself, had more life experience than she did. She was surprised to learn that many of the students her age were sexually active, some pregnant and others with children. This was a culture shock to Alley. She began to reflect on her own life and realised that she had lived a very sheltered life; however she was willing to embrace these new experiences of her peers, although she did feel out of place and uncomfortable.

Alley's interactions with her lecturers and peers were limited. She did not make contact with her lecturers in or out of formal class time. She felt there was no need for her to make contact with them. Alley was friendly with students in her class, but she did not have many friends. She felt they were not of the same faith as her and found socialising with them difficult. Alley had no intention of going out to clubs or pubs; she regarded these activities as ethically out of her comfort zone. Her life revolved around university, home and church. Besides being a full-time student at university, Alley was engaged in Bible studies in the evenings. She found it difficult to cope with both studies at the same time. It became demanding and confusing at times. However, Alley is a motivated and goal-oriented person. Although not very confident, she is determined to complete her studies within the required time and cannot wait to make a difference to the lives of the next generation. In addition, earning an income will assist her family financially.

Alley has been faced with many challenges in her first year of study. Besides trying to adjust to the culture shock of student life on campus and the mismatch of her expectations and actual experiences of university, her family received news that her mother has cancer. Alley is trying to be strong about the news and take each day as it comes. But she says her faith in God will help her to cope with whatever happens in the future. Her mother's illness has resulted in her resigning from work. This has created a problem for Alley: her mother was paying her tuition fees and now she has to depend on her stepfather to take over the payments. She intends to either secure a student loan or bursary for the following years of her studies.

6.3.6 Katlyn

Theme: Challenges – Being a student again!

Katlyn is a 20-year-old white female student from Table View in Cape Town. She completed high school in 2010 and decided to take a gap year in 2011 before commencing university in 2012. Katlyn acknowledges her upbringing as one from a middle-class family, she acknowledges that they do have financial issues but recognises that they are not poor. Katlyn indicates that her family life is not that of a typical family. She elaborates by stating that she has six members in her unusual family: her mother, step-brother, three 'father figures' and herself. Her biological father divorced her mother when she was two years old. Katlyn remembers being raised by her mother's second husband. She regards him as one of her fathers: he made her feel loved and special. Her mother's relationship with her second father lasted ten years. Katlyn's mother's present husband is warm, loving and protective of Katlyn. She feels fortunate to have all three father figures in her life: all of whom support and care for her in special ways. Katlyn's mother works long hours and only gets home at 7:30pm every day. Although Katlyn knows that her mother cares for her and will support her in any way she can, she acknowledges that her mother did not play a major role in her schooling and university education compared with her fathers.

Katlyn enjoyed her high school experience. She regarded it as fun, but regrets that she was not very 'study orientated'. She claims that many of her peers at school regarded her as very 'talkative, silly and a party person', which she states is not who she really is. One of her intentions in going to university is to show her school friends that she does have a serious side to her and will be successful. Katlyn had a teacher at school whom she regarded as her mentor; someone that was firm, but honest and approachable. She trusted her mentor teacher and respected her opinions. Although Katlyn knew she wanted to go to university, she was unsure of what course she should study. Her parents were unable to help her make that decision: they had not been to university. It was her mentor teacher who had arranged for Katlyn to meet with a career counsellor to help her make an informed choice. The career counsellor advised her that she would be suited to a career in teaching, social work or psychology. Katlyn decided to take a gap year before enrolling at university the following year. During her gap year she worked as an au pair. Earning her 'own money' empowered Katlyn to be financially independent and self-sufficient. She got her driver's licence, bought herself a car, and was able to buy her own clothes and have money for entertainment and socialising.

When Katlyn entered university for the first time she was excited and motivated; her intention was to 'do really well, get all A symbols, make plenty of friends and have a great student social life'. Her expectations did not match up to her actual university experience. Katlyn describes her university experience as 'driving a car'. she explains: 'Initially you are all excited and very cautious and careful, making sure you follow the rules and then after a while it becomes second nature to you, it is boring and routine and you don't need to really give it much thought!' Katlyn is funding her own university studies. She has opted for a student loan instead of a bursary: she wanted the freedom to decide what she wants to do when she graduates from university and therefore not be obliged to pay back her bursary in service. She feels she needs to be successful in her studies and cannot disappoint her 'real father': he believes in her and has signed surety for her student loan. Katlyn regards him as her 'biggest supporter'.

Returning to studying after taking a gap year resulted in serious challenges in Katlyn's first year at university. Her internal conflict at not being financially independent was a difficult adjustment for her. She has to rely on her family for partial financial support. One of her fathers has opted to assist by providing Katlyn with an allowance towards her petrol bill and stationery for university. Her biological

father purchased her a laptop and cannot provide her with much financial support since he has to support her stepbrother, who is still at school. Katlyn feels her family's contribution is not sufficient: she does not have money to socialise with her friends or buy expensive clothes needed for campus. She has taken on a part-time job as an au pair that keeps her busy from 5pm to 6:30pm three times a week and full days during weekends. She feels this is affecting her academic and social life at university. Katlyn finds it challenging to balance university studies, her part-time job and socialising with her friends. This difficult situation has caused Katlyn to contemplate whether it would be a good idea to go back to her full-time job as an au pair and study part-time through correspondence. Katlyn acknowledges that she is not a person who can motivate herself to work diligently and needs the support of her peers. Katlyn is experiencing external challenges with regard to her home and personal life. She receives no support with regard to her academic work from her parents. She wishes her parents had been to university so that they could understand what it entails to be committed to succeed academically. She complains that they have a beautiful big study at home that is in a mess: her mother uses the computer in the study mainly for social media and this has priority over Katlyn's university work. Katlyn uses the study from 6pm, when she returns from her parttime job, until 7:30pm, the time her mother gets home from work. Katlyn feels her mother supports her going to university but adds that her mother does not understand the dynamics of studying at a university. Katlyn is grateful that her biological father has given her a laptop so that she can sit in her room and do her assignments and projects. Katlyn is appreciative that her father with whom she lives is caring: he will often take the time to enquire about her day and how things are going at university.

Academically, Katlyn is passing all her courses at university although she feels guilty that she is not giving herself 100% to her studies. She is aware of the rules and assessment criteria for each course and ensures that she satisfies the minimum requirements to pass the course. Katlyn claims her main intention in attending university is to earn a good salary and be financially independent. Katlyn does not make contact with her lecturers outside of formal class time, neither does she feel the need to be actively engaged in class discussions. She is mindful that her lecturers may not be aware of her or know her name, but she is aware that lecturers do seem to have their favourites with whom they regularly converse with in class.

Katlyn is grateful to her new campus friends: she feels they have motivated her to focus on her studies compared with her school friends who were not interested in school work and had a negative influence on her. She feels that her new campus friends have been academically more supportive than her parents or lecturers. Katlyn socialises with her campus friends outside university, which she finds financially challenging. She acknowledges that her friends come from 'well-off homes' and spend money on socialising which is not an issue with them. Katlyn finds it challenging to go out with them: she has to consider her financial circumstances and her commitment to her part-time job. She explains that even going out to dinner with her friends or for a quick bite during campus hours puts a strain on her budget. Her experiences with her new friends have made her reflect on her own life. She wishes that her parents had a university degree that would have allowed them to get good jobs and earn respectable incomes like her friends' parents. This she feels would have made her life much easier on campus. She is having second thoughts about her present boyfriend, which was not an issue before she came to university. Watching her friends with their boyfriends on campus made her envious: her boyfriend is older than she is. He has a fulltime job and cannot afford to study at university. Although her boyfriend treats her well, she feels she needs to have a boyfriend who is more ambitious and at university. Although Katlyn is confident she will be successful in her first year of study, she still feels that her home, personal and financial circumstances prevent her from having the full university experience. Katlyn is determined to complete her degree and earn a decent salary that will give her a good life. She recalls her childhood days of not having a lot of money and vows never to struggle the way her mother did when they were growing up.

6.3.7 Lisa

Theme: Positioning – Getting ahead at university.

Lisa is a 19-year-old white student from Cape Town who, immediately after leaving school, enrolled at university. Lisa has five members in her family. She is from a middle-class family and both her parents are professionals. Her twin sister is

studying at another university in Cape Town and her brother is in a private school. Lisa and her sister attended a 'good' school; her parents wanted them to get the best education and moved them to a school in the southern suburbs of Cape Town, a move Lisa does not regret. Lisa believes that attending a school that is regarded as one of the best does make a difference to one's academic ability, especially when preparing for university life. Lisa received all A-symbols in her Grade 12 examinations which qualified her for a bursary to study teacher education. Becoming a teacher is Lisa's passion which she says many of her relatives and school teachers cannot understand since she is so 'smart'. She could get into any other profession and attend any university in the Western Cape, like her twin sister who is studying towards her social science degree at a 'prestigious' university in Cape Town. Lisa is happy with her choice; she acknowledges we need 'smart' people to be teachers since they play an important role in advancing our society and country.

The present tertiary institution that Lisa attends was not her first choice of university. She, like many other members in her family, was concerned about attending a university of technology instead of a traditional university. She completed her research and consulted many teachers, friends and ex-students of the university of technology, all of whom informed her that her present institution was the best in the Western Cape for studying teacher education. Lisa enjoyed her first week of orientation and made a few friends. She admits that she was quite surprised to see so many white students; it was not what she expected. She was surprised by the course content. She expected it to be more challenging and difficult. Lisa found the first-year course easy and says she obtains A symbols without even working hard. She remembers doing an assignment the night before and thinking she would get an average mark, but got an A. This frustrated her since she felt she was not really being challenged sufficiently and could not understand how her peers were struggling with the course. The only stimulation was in practice teaching. She says this activity kept her on her toes and was interesting, inspiring and tiring at the same time. She has a new-found respect for all the teachers in the country, especially those in schools that are under resourced. Lisa is a self-driven, highly motivated and competitive person. She is aware of university life, the social and academic activities, and the requirements needed to be successful. She is focused and would like to gain the full benefit of being a university student. Lisa has thus opted to move out of her

family home to an apartment close to the university. Lisa's mother, however, was not happy about this decision. Lisa says her bursary covers all of her expenses and she does not need to share a flat with anyone or worry about money. Her sister also has an apartment close to her university. She proudly admits that both her parents have given her and her sister their full support with regard to university studies. Lisa acknowledges that having parents who have been to university is an advantage: they know about study commitments and how important a good education is for their future. Her mother calls her in the evenings to find out about her day and query about her academic work but, 'not to tell me to work hard or be focused'.

Lisa says her mother has never had to check up on schoolwork, assignments or projects: it is just something she and her siblings had picked up on their own from watching her mother studying for her part-time postgraduate degree while working and taking care of them as children. Lisa has good relations with her lecturers and students on her course. She has three close friends and their friendship came about because of their similar work ethic and dedication towards their academic studies. She says all of them are from good schools, are hardworking, committed and they all obtain good results at university. Lisa confesses that these are just her friends on campus and she really has no time to socialise with them off campus. Lisa adds that she would not mind working with other students in her class if she was required to. But if she was given a choice, she would chose people like herself since her priority is to excel in her studies.

Lisa is serious about her academic work and gets frustrated when her time on campus is not maximised to its full potential. For example, they have long breaks between classes and sometimes have to wait two hours before the next class. In addition, Lisa complains that the lack of academic resources and space to work hampers her study programme. Most of the time the printing lines are too long; the printers are not working; there is no available computer to work with; and no space in the library. This situation bothers Lisa. She explains if she was at home she could get a lot more work done: there she has her own printer, computer and internet access. Lisa gets frustrated with the lack of preparedness among her peers. She cannot understand why her lecturer has to explain what she believes are basic concepts and terms that her peers should have learned at school.

Lisa has built good relations with all her lecturers and addresses them by their first names. Lisa says she makes sure she talks a lot in class; asks a lot of questions and contributes to class discussions. She feels you have to be noticed by the lecturers and therefore will be taken more seriously when it comes to assignments and projects. She feels it is a good thing when the lecturer knows you personally. Lisa passed all her courses with distinctions; however, she did not feel challenged and dropped out of the course at the end of her first year.

6.3.8 Tasha

Theme: Challenges – Moving away from home.

Tasha is a 19-year-old, white female student from Hermanus who, immediately after completing school, enrolled at university. Tasha is in the upper middle-class financial category. Her father is a qualified engineer and her mother worked as a legal secretary. There are four members in Tasha's family: her parents, younger brother and herself. Tasha talks fondly of her home town because that is where her family, friends and memories are. She enjoyed her schooling experience and feels that her high school experience has equipped her to handle university studies. Tasha is not the first in her family to attend university. This helped when she was deciding to further her studies. She acknowledged that her parents sat down with her and had a discussion about her choice of career and supported her decision to enrol at a university to pursue a career in teaching.

The biggest challenge for Tasha was moving away from home. She lives with her grandparents in Constantia. She found the adjustment to her new environment difficult. She explained: 'It is fine ... but we have arguments all the time and it's ... uh it's the generation and ... they didn't grow up with me and ... yeah that's been a big change yeah!' Her second challenge was making new friends in Cape Town; she described this process as difficult because most friendships were formed during high school and none of her friends from high school attends this university. Tasha does not have any close friends at university and goes home to Hermanus every weekend. She socialises with her friends from back home.

Tasha was quite disappointed with her new university environment: it was not what she had expected. She visited other universities and her present one did not match up to her expectations. She complained about the lack of services offered by the institution. She was disappointed that there were no student activities and no 'social life' of the kind she had seen at other universities. She complained that the campus was small in size and overcrowded. It frustrated her that there was no cafeteria on campus, which meant that she had to get up early to pack her own lunch or snack.

In the first few weeks on campus Tasha was excited with her new experience, but everything quickly diminished and become old and boring. She regards herself as a diligent student who always gives of her best. She expected the same kind of commitment from her peers. She was disappointed with the students in her course: she found them to be academically below average, as she states: 'They're not as ... I won't say as intelligent as eh, that's rude, they are not as intellectual as I thought they'd be.' This was certainly something that she did not expect of university students. Tasha has a few like-minded friends on campus: they support one another and work on group projects and class tasks together. Tasha achieved distinctions in all her courses and felt disappointed that her course was not challenging. She was contemplating changing her course the following year. She hoped to transfer to the intermediate and senior phase programme: she assumes that course will be more challenging for her. Tasha is a confident student and believes there is no need for her to make contact with her lecturers outside of class time.

Tasha's study is financed by her parents and therefore she can focus her full attention on her studies. She has her own car that she drives to campus every day and does not need to depend on friends or public transport. She looks forward to going home to her friends and family every weekend.

6.3.9 Synopsis of participants

Table 6:1 on page 188 provides a synopsis of the key characteristics of the participants. The table highlights the broad themes and sub-themes that emerged from the interviews. These themes and sub-themes are further explored and discussed in detail in Section 6.5 of this chapter.

Participants' Age & Race	Student Grouping	Socio- economic Status	1 st Generation Yes/No	Broad Themes	Sub-themes
Joyce (22) African	र इ		Yes	Determination : I am here to get a degree	 Broken family Social ills Culture and
Thabo (24) African	Mature students	Working-class	Yes	Resilience : Need to be financially stable	tradition Racial stereotyping Family influence
Letti (27) African	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~		Yes	Determination: Self- improvement	 Economic hardships Connectedness
Nadia (20) Coloured	Gap students	Jass	No	Fitting in : This is where I am supposed to be	 to family Parents' educational aspirations
Alley (21) Coloured			Yes	Out-of-habitus experience: Culture shock	 Sacrifice Need to be someone – identity
Katlyn (20) White			Yes	Challenges : Being a student again	 Religious influence Social connections
Lisa (19) White		Middle-class	No	Positioning : Getting ahead at university	 School experience Coping
Tasha (19) White			No	Challenges : Moving away from home	 mechanism Dialogue with lecturers Peer support Job security Upward mobility Better life

Table 6:1 Synopsis of participants with emergent themes and sub-themes

6.4 Sort-card activity: Factors that positively contributed to first-year experience

The sort-card activity took place during the one-on-one interview session. Participants were given ten picture cards and were requested to sort them by arranging the cards from the factor that made the most significant contribution to the least contribution in influencing their first-year university experience. After ranking the factors, participants discussed the influence or contribution of each factor as related to their first-year experience. The ten picture themes that were outlined in the sort-card activity included:

- family support
- financial support
- lecturer support
- community support
- academic learning support
- new campus friends
- school experience
- need to succeed
- self-confidence
- an open card on which to write any other influence not mentioned

The sort-card activity during the one-on-one interview allowed participants to take the lead in the interview. This activity allowed participants time to reflect on the cards and contemplate on the order before discussing the influence of each factor on their university life. The sort-card activity provided rich data on the students' perceptions and influences on their first-year university experiences.

In the sort-card activity, participants were free to use all or selected cards from the pack. Table 6:2 on page 190 indicates the ranking and the use of the sort cards by the participants.

	Participants' ranking of sort-card activity							
Grouping of participants	Mature students			Gap students			School leavers	
Categories	Joyce (22)	Letti (27)	Thabo (24)	Nadia (20)	Katlyn (20)	Alley (21)	Lisa (19)	Tasha (19)
Academic support	1 st	-	-	8 th	-	8 th	6 th	-
School experience	2 nd	-	8 th	1 st	6 th	4 th	3 rd	2 nd
Lecturer support	3 rd	7 th	6 th	6 th	7 th	6 th	5^{th}	5 th
Need to succeed	4 th	2 nd	4 th	5 th	1 st	3 rd	1 st	1 st
Self-confidence	5 th	6 th	1 st	3 rd	3 rd	5^{th}	4 th	6 th
Financial	6 th	4 th	9 th	4 th	5 th	9 th	7 th	7 th
Family support	7 th	1 st	2 nd	2 nd	4 th	1 st	2 nd	3 rd
Community	8 th	5 nd	3 rd	9 th	8 th	2 nd	-	4 th
Making friends	-	3 rd	5 th	7 th	2 nd	7 th	8 th	-

Table 6:2 Participants ranking of factors influencing first-year experience – Sort-card activity

None of the participants chose to use the open card provided in the pack. Therefore this card was not used in the ranking process. Other sort cards that were not identified as factors influencing the first-year experience for some of the participants included: the 'academic support' card which was omitted by 50% of the participants, while the 'making friends' card was omitted by 25% of the participants. A further 12.5% omitted the 'school experience' and 'community' card. The highest percentage of the participants (50%) that omitted the academic support card dismissed it with either no explanation or as having no bearing on their first year of study.

Of the four participants who did give the academic support card a ranking, three participants gave it a low position, placing it in the bottom three positions; two participants ranked it in 8th place; while one put it in 6th place. All the participants that used the card gave the following explanations for their low scoring of the card:

- *it's not for me, I don't need it. (Nadia)*
- I don't have the time to utilise the service. (Alley)
- Nice to know it's there, I used it once only and it helped, but I really don't need it. (Lisa)

Joyce was the only participant who ranked the academic support card as number one in her sort-card activity. She is a returning student to the institution and therefore is familiar with accessing the service and understands the importance of such a service in enhancing her academic performance.

Joyce provides the following explanation:

- Assisted me with referencing my work.
- The writing centre helped me with my academic essay writing.
- To better my marks I ensured that I made full use of the academic services and peer tutorsthey are there to help you.

The other sort cards omitted were negligible with only two participants not using the making friends card and only one participant discarding the following two cards: school experience and community.

A significant finding from the sort-card activity related to the financial support card. The activity revealed that one of the challenges hindering participants from experiencing a successful first year was primarily due to their financial circumstances. Two of the participants had ranked financial support in 9th position, while four of the participants ranked it in 5th, 6th and 7th place. The remaining two participants ranked it in 4th place.

Six of the eight participants acknowledged that their circumstances were created by economic hardship and resulted in financial support's being a challenge in their first year, which negatively impacted on their studies. Many of the participants were involved in part-time employment to try to fund their tertiary education. Some participants were actively trying to source loans or bursaries either to supplement their first-year tuition fees or secure funding for the following academic year. The following were some of the remarks of the participants with regard to financial support in their first-year experience:

Letti:

- This bursary thing is hanging over my head, I just wish I could get it as it would really help; at least it could pay part of my res fees.
- I need to work weekends to try and pay part of my fees and buy food. It is really difficult to balance work and studies.

Thabo:

- I have received no financial support from my family or the university; I have to work night shifts to try and pay my fees and also send some money home to my family.
- If I had the financial support from my family I would have done much better in maybe some of the work; it's no excuse, I could do better if I had this support.

Nadia

• Financial aid has really helped me; however it has created an added pressure because now I have to make sure I get good results to maintain the bursary.

Two of the participants that ranked the financial support card in 9th position explained that their parents were funding their studies and therefore regarded the paying of fees as 'taken for granted' and not a factor that was either enabling or hindering their first-year university experience.

Table 6:3 on the following page (page 193) indicates the frequency and the top three ranking positions (1st, 2nd and 3rd) of the sort cards. The table highlights the factors that participants indicated as making the most significant contribution to their first-year experience.

Factors ranked according to the top three positions of the	Frequency of card use (N=8) in the	Number (N=8) ranked factor in first three positions			Frequency of card use (%) in the
sort card	top three positions	1 st	2 nd	3 rd	sort-card activity
Family support	6	2	3	1	100
Need to succeed	5	3	1	1	87.5
School experience	4	1	2	1	87.5
Self-confidence	3	1	-	2	100
Community	2	-	1	1	87.5
Making friends	2	-	1	1	75
Lecturer support	1	-	-	1	100
Academic support	1	1	-	-	62.5
*Financial support	-	-	-	-	100

Table 6:3 Identifying the top three factors influencing first-year experiencefrom the sort-card activity

* The financial support card did not feature in the top 3 positions

Eight of the nine sort cards were ranked in the top three positions during the sortcard activity. The financial support card was the only card that was not ranked in the top three positions, as indicated in Table 6:3 above.

All the factors that were ranked in the top three by the majority of participants were factors external to the university: family support, need to succeed, and school experience. The highest ranking was accorded to the family support card with 75% of the participants ranking it in the top three, indicating that their families had positively influenced their first-year experience. The second and third highest rankings were the need to succeed and school experience, with a percentage of 62.5% and 50% respectively. The three factors that ranked the lowest in the top three categories were related to factors within the institution: making friends (25%), lecturer support (12.5%), and academic support (12.5%).

Some of the remarks from the participants which reflected that their family had positively influenced their first-year experience related to the following support: financial, emotional, and academic support received from their parents, siblings or extended families. For example:

Thabo: My extended family is very supportive, my aunt's sisters and my aunt's children play a part in saying that this is good for you, keep doing what you are doing. They ask me about what is happening on campus and stuff like that so I call them my family, they care about me.

Lisa: My twin sister is studying Bachelor of Social Science at the University of Cape Town and my mum just completed her Master's in Psychology – cum laude. So we have academic discussions all the time. We are able to discuss our assignments and help each other if needed.

For the majority of participants, the 'need to succeed' seems to have developed from their economic, social and life circumstances, and a willingness to improve their own circumstances to help their families. For example:

Thabo: For me the need to succeed ... is just the need to take care of my family and putting it onto the map as well, I need to succeed for them.

Letti: I want to make a success of everything, whatever it takes I want to be successful and be able to help my mom, my dad, my family and support myself if I'm done.

Other participants' need to succeed was purely for personal reasons:

Lisa: I need to succeed, my sister and I were very competitive even though it was never mentioned, I have to make sure I succeed.

Katlyn: The degree, the course that I've chosen shouldn't be overlooked and I wanted to succeed for myself and to prove to other people that I'm not stupid and I'm doing well and you know I can do anything.

However, the outcome for all the participants from the sort-card activity was evidently either to obtain a better life; be someone; have a stable job and/or as a stepping stone to further their qualifications.

6.4.1 Synopsis of sort-card activity

Table 6:4 below provides a synopsis of the mostly highly ranked sort card selected by the participants. It provides a clear indication of the factors that were most beneficial in supporting a positive first-year experience. The table classifies the factors into two categories: factors external to the institution and factors within the institution. The most beneficial factors that contributed to a positive first-year experience were factors external to the university. The external factors related to the participants' family circumstances and the environment in which they were raised, which had either supported or given the participants the motivation and drive to succeed in their first year of study: need to succeed; family support; school experience; self-confidence. The only sort card that referred to a factor within the institution that was ranked first was the academic support card.

Ranking of No. 1 Factor	% selected card as No. 1	Category	Theme/sub-themes that emerged		
Need to succeed	37.5	External	Economic circumstances		
Family support	25	institutional factors	 Emotional & moral family support 		
School experience	12.5		Teacher influenceSocial connections		
Self-confidence	12.5		DeterminationSchool experience		
Academic support	12.5	Internal institutional factor	Commitment to succeed		

Table 6:4 Factor ranked as the No. 1 sort card by participants

The sort-card activity contributed in reaffirming and strengthening themes and subthemes that emerged from the one-on-one interviews and focus-group interview, which is further elaborated on in the next section.

6.5 Themes and sub-themes that emerged from the interviews

This section of Chapter 6 focuses mainly on the last two research questions. The aim of this section is to explore, understand and interpret how certain pre-entry academic and non-academic factors identified by the participants have influenced their first-year experience and academic performance. The conceptual framework outlined in Chapter 2 of this thesis is used as a lens to focus the gaze on the following constructs: students' prior dispositions and experiences taking into account Bourdieu's concepts of habitus, field, and capital; and students' integration into the field of higher education, that is, the transfer of their dispositions and experiences in the way they socially and academically integrate into their new higher education environment. These are relational structures which will determine the influence of the participants' circumstances on their first-year experience and academic performance. Owing to the complexity of the factors influencing the participants' firstyear experience and the interconnectedness of the influences, this section is presented thematically. Emergent themes and sub-themes were derived from reading and re-reading the transcripts of participants' one-on-one interviews and the focus-group interview. The themes and sub-themes are thus data driven and closely connected to the conceptual framework to provide a more holistic and comprehensive understanding of the phenomenon. Table 6:5 on the following page (page 197) highlights the 6 key themes and 15 sub-themes that emerged from the qualitative data.

Six key themes	Sub-themes
Determination	Family influenceBroken familiesTaking charge of one's own destiny
Resilience	Economic hardshipsSocial ills
Fitting in	School influenceParents' educational aspirations
Out-of-habitus experience	Religious influenceRacial stereotypingCulture and tradition
Positioning	Dialogue with lecturersChoosing peers for group workMaking friends
Challenges	Balancing work, study and family commitmentsDealing with diversity

Table 6:5 Key themes and sub-themes

In order to foreground the students' first-year experience and to acknowledge the importance of their voices in this study, direct quotations from the participants were used in presenting the findings. The following sub-sections highlight the key theme and then employ the sub-themes to further explain and give meaning to each of the six key themes.

6.5.1 Main theme 1: Determination

Bourdieu (1993) acknowledges individual histories to be extremely important in understanding the concept of habitus, which he defines as a complex interplay between past and present circumstances and a place where cultural capital is acquired. The data revealed that there were key events in the participants' past circumstances that had created a sense of determination among them. The participants were determined to overcome the challenges of their present circumstances and expressed a desire to attend university and assume responsibility for their own success. All participants spoke of determination as a key factor to their first-year experience and academic success. The driving force that shaped participants' determination varied between working-class and middle-class participants. The following sub-themes appropriately articulate these contexts: family influence, broken families, and taking charge of one's own destiny.

6.5.1.1 Family influence

Participants from middle-class families displayed a positive attitude towards university and were determined to succeed. Two of the participants were from stable homes, where both parents had attended university and were employed in skilled jobs. This environment contributed to the participants' awareness of the value of success and preparedness for hard work. Participants gained first-hand knowledge of such behaviour from their parents.

Tasha: ... because of my upbringing, having hardworking parents, who have been to university and have good jobs, it's forced me to be like that. I am very focused and determined to be just as successful.

Lisa: You know the fact that my mum used to get up at 5 o'clock when she was doing her Master's and then she still got a cum laude, so like now we have that in us ... my sister and I are also very focused. My mum always says, 'Well I managed to get a cum laude with like three kids so you can do better than me!' This makes me want to work harder; I am competitive and determined to get top results.

On the contrary, participants from working-class families did not have first-hand experience or knowledge from their parents with regard to university studies. Their strong determination to succeed at university was based on their family struggles as indicated in the following quotations:

Letti: I knew there was [sic] times that my mom said, 'I don't have money for your fees.' Then I told myself I'm just gonna pass every year to show them I'm not gonna waste their money on me and they feel like yeah they paid for it and I messed up. So I am determined to work hard; I'm not smart, I promise I'm not smart but I must work hard to succeed in my studies.

Katlyn: I have to think of what's going to earn me money because my mom, she really struggled, we never had a lot growing up so I didn't want to struggle like that. I want to be financially independent and being successful in my studies will help me achieve this goal.

6.5.1.2 Broken families

Six of the eight participants had varying degrees of broken family experiences including: single-parent upbringing from a very young age; death of parent/s at a young age; and living with extended families and step-families. The participants' awareness of growing up in a broken family environment and the socio-economic struggles of their parents made them more determined to improve on their own circumstances.

Joyce grew up surrounded by female members only in her family and this influenced her upbringing. Her mother was a single parent who struggled financially to support her and her siblings. The loss of Joyce's mother at a young age and her extended family circumstances made her determined to be independent and self-reliant:

I have never met my father, he lives somewhere. I grew up living with my grandmother while my mom worked in the city. My grandmother died and I moved to live with my mom, who died a year later ... so my aunt took us in as she didn't want us living with my mom's boyfriend. She wanted us with family ... we moved to Cape Town to live with her. It was tough, my aunt and I, we didn't get along! I needed to take care of me! I needed to be independent!

Thabo's determination to succeed was different when compared with Joyce's experience. Thabo's determination to succeed was primarily to improve his economic status so that he could take care of his extended family. He grew up in an all-female environment and felt the need to provide for them:

Thabo: I lost my mother when I was little. I never knew my father, so my mother ... I think she died of HIV and uhm we are only two, there's only two of us. Our aunts took over and took care of us. We had three aunts, my sister was with a much older one, and me, I was with the little one; it was difficult, we moved around a lot ... I need to do this for them and my little daughter, they are depending on me.

Nadia's determination was driven by her mother's circumstances. Nadia grew up in a single-parent environment. Her mother was a central figure in her life who kept her focused and motivated her to be someone and change her life circumstances:

Nadia: My mother, brother, sister and I live together now. So I think my mum ... because of all the years of struggling and all of that drama she had in her life, like she also kind of played a big role in like me being at university studying and wanting me to study. She would always like say, 'If you don't study and succeed you are going to end up like where we were,' and like that, so I'm like no! I'm not going to, so I'm like I'm going to do this, study hard and be a success! Be someone.

Katlyn's determination to succeed in her studies was not self-motivated. Her 'unusual' family situation, with three male figures in her life, was her main motivation. They had all supported her financially and emotionally and she wanted to make them proud of her and not let them down:

Katlyn: I kind of have three father figures, it's my biological dad and then it's my stepdad who has shown what a man should treat a woman like, which is amazing and there is my present dad, I just call him my uncle. So it's three totally different people but it's kind of awkward for me but they all contribute to my studies and want me to be successful and I don't want to let them down, especially my biological dad.

6.5.1.3 Taking charge of one's own destiny

The majority of participants interviewed articulated that their determination to succeed at university was primarily driven by the need for them to change their life circumstances; this was motivated by their family struggles and socio-economic circumstances. Many of the participants regarded university as a way of improving their lives and taking charge of their destinies.

Thabo: I wouldn't say like completely I'm enjoying myself at university; we came here to work, we want this degree, that's what we want, that is what I want. I want it so bad so I will do whatever it takes to make sure that I get the work done. I want to prove to myself that I can do this, even with all of the challenges that I have, but it's just that the need to succeed goes into supporting my family and putting them onto the map as well.

Joyce wanted to change her circumstances, especially for her son. She didn't want her son to experience the upbringing that she had:

Joyce: I don't have a family, let me just put it like that. I decided I want to study, I'm gonna quit work because it's not the life I want for me and I just said to myself 'No! You need to change your life, you need to go back to study.' I wanted to do something better for me and my son to have a better life.

Letti felt the same way as Joyce; however she wanted to further her qualification to improve her socio-economic status and take care of her parents:

Letti: So um, yeah then last year I quit, I like just didn't quit, I resigned from my job as a Grade R teacher and decided I want to come study fulltime to get my teaching degree. I wanted to improve on my qualifications and I want to make a success of everything, whatever it is I want to be successful and be able to help my mom, my dad, my family and support myself.

The next two participants were determined to change their destiny for themselves:

Katlyn: The degree, the course that I've chosen shouldn't be overlooked and I want to succeed for myself and to prove to other people that I'm not stupid and I am doing well and you know I can do anything.

Alley: Uhm ... I feel that I need to become someone who is something and I need to make a difference in someone else's life and not only mine.

6.5.1.4 Summary: determination

The three sub-themes: family influence, broken families and taking charge of one's destiny helped to articulate and deepen the understanding of the participants' experiences that had led to their determination to succeed in their studies. The majority of the experiences conveyed by the participants' involved family struggles that were either explicitly or implicitly expressed in the interviews.

6.5.2 Main theme 2: Resilience

From the qualitative data analysed, there was strong evidence of resilience and the ability to cope with adversity by the participants. According to Kobasa (1979), there are three elements essential to resilience: challenge, commitment, and personal control. She explains that resilient people view difficult circumstances as a challenge

and not a stumbling block; they are committed to their goals and work towards them; and finally resilient people take control of their personal lives, they feel confident and empowered and spend their energy on situations they can control. Such qualities of resilience and coping with adversity were evident among the participants and are further described in the following two sub-themes: economic hardships and social ills.

6.5.2.1 Economic hardships

Evidence of economic hardships was a strong factor identified mainly among the working-class participants, having faced disadvantaged and difficult challenges in their upbringing, was influential in their decision to succeed at university. These participants viewed these challenges as 'making the best of a bad situation'. Evidence of commitment and taking control of their personal circumstances and challenges is revealed in the following quotations:

Thabo reflects on his economic hardships while growing up and his ability to thrive in such challenging circumstances:

Thabo: *I grew up very poor: at school, I would run out of shoes. I made a choice to sell cigarettes so that I could buy myself a uniform: like shirts, pants, tie and everything. At school, sitting in a crowd, they couldn't tell this boy comes from this kind of family because my own business was yah good, it gave me money to look like I wasn't poor and also my stepfather helped me a lot, he would come from PE and maybe buy me 20 packs of Stuyvesant cigarettes and then say, 'You can sell this to keep pocket money,' and stuff like that. They were very helpful in terms of that.*

Joyce's experience provides an example of her commitment to her family and ability to take control of her challenging circumstances:

Joyce: My mom moved to the city with her boyfriend to find a good job. My grandmother took care of us and the other grandchildren; you know being without a parent at a young age you have to grow up so quickly so it's like you know ... I ... I had to grow up, sometimes I felt like I wanted to be a child, but I couldn't ... I had to be there for my sister and brother; I had to take responsibility for their safety.

202
Letti's economic hardships forced her to take control of her personal circumstances and get a part-time job to pay for her studies:

Letti: I'm working every weekend now to pay like for my food. So every weekend I work on a wine farm in Stellenbosch. So every Friday I take the taxi home and I work the weekend and then my dad brings me back Sunday night because the hours are long and there is no taxi so late into the city. I take my university work with me and then when it's quiet and when there is [sic] no customers I would take my bag and quickly do something and then when a customer comes I will leave it aside; however now with the new manager... she is not very nice. I won't be able to do that. So I must make another plan.

6.5.2.2 Social ills

Participants from both middle-class and working-class families were faced with social issues that were beyond their control. Participants were exposed to social issues within their immediate and extended families from a very young age, such as drug and alcohol abuse, domestic violence, teenage pregnancy, and the tragic death of family members. Participants expressed that these incidents had made them more determined and gave them the courage to take control and deal with tough situations in their own lives. Some of the social ills experienced by the participants are noted below:

Joyce:

I started high school and then I got pregnant and my aunty wasn't happy so I had to move out of the house.

My parents separated because my father was an alcoholic and he was very abusive to my mother so my mother said: 'No this is enough'.

Lisa:

We adopted this baby, he was abandoned. He was found in a toilet in Delft and our family felt we needed to adopt him ... so that is the other addition to our family.

My dad got retrenched like and we almost lost our house. Uhm ... yes, we've been through a lot!

Thabo:

I grew up with uncles who were ... they smoked weed/dagga and stuff like that. We hated Fridays after work ... they would make the house so corrupt and chasing people, every member of the family would run at night looking for places to crash and sleep and it was very very hectic.

I felt like an outsider in my house because I was a child whose biological father I didn't know and then sometimes my aunts would sit and talk and it felt as if I didn't belong there.

I lost my mother when I was little. I never knew my father, so my mother I think died of HIV and uhm we are only two, there's only two of us.

Letti:

So I attended a coloured school, a typical farm primary school and I must say the children that were there were very racial [racist] so it was the 'k' word but I went through it and so yeah.

When I started high school in 1999, yeah that's when she [sister] died. She was shot and we never found the people who did it. That was a bad year for me but I managed.

Katlyn:

Yeah, my dad's dad like hits my granny so, and my dad use to hit my mom. So that's why my mom divorced him when I was two. So my mom didn't want me growing up thinking that's the way a man treats a woman because I think that's a pathetic excuse from a man.

Some of the participants were able to use their experience with social ills to better their own circumstances. Dealing with these adversities had made them stronger and more resilient: Thabo: Luckily for me, I'm coming from a family I was not so like uhm how would I put this eh not so good upbringing uhm suffering economically and ja financial stuff that's why these challenges I'm facing at the campus here I feel that I am overcoming them bit by bit; because of my difficult life, it helps.

Joyce: I know why I'm here, ... so I have come here to study because I want to be that, I want to do something with my life, something better for me and my son so basically I just said to myself ...

Alley: My religious uhm background has given me the discipline to cope with any challenging situation.

Lisa: Uh ... I just said to myself; oh well this is what you've gotten yourself into so deal with it.

Letti: Then I told myself I'm just gonna pass every year to show them I'm not gonna waste their money on me ...

6.5.2.3 Summary: resilience

The main theme of resilience revealed that participants, in spite of their difficult upbringing, were positive and determined to change their circumstances. Their challenging experiences, due to economic hardships, and their first-hand experience with social problems motivated participants to take control of their personal circumstances. They were committed to make a positive change and to use their challenging socio-economic circumstances as a catalyst.

6.5.3 Main theme 3: Fitting in

For some of the participants, the transition into university was an easy progression. They felt comfortable and fitted in effortlessly into academic work with peers and the university environment. From the data analysed, it was evident that participants' prior school experience and parents' educational aspirations had influenced their expectation of university, thus creating a match between their expectations and the actual university experience. Participants felt familiar with their new university environment. The responses of participants are explored further under the following two sub-themes: school experience and parents' educational aspirations.

6.5.3.1 School experience

Of the eight participants, Thabo was the only participant who had attended a rural school with exclusively African students at his school. Two participants had attended private schools: Joyce attended a special school for mathematics and science achievers while Alley attended a Christian-based school run by nuns. The five remaining participants attended ex-Model C schools: these are schools that were classified as 'whites only' schools during the apartheid era in South Africa and at present are still regarded as privileged schools, that are well resourced and have mostly qualified teachers. Learners in ex-Model C schools pay high school fees, thus making entrance into such schools restricted and selective. Participants from both working-class and middle-class families who attended this type of school felt grateful to their parents for sending them to such schools, as expressed by the following participants:

Letti is from a working-class family, her parents are farm workers: she felt privileged to be in an ex-Model C school. She understands and fully appreciates the sacrifice her parents made to send her to such a well-resourced school:

Letti: I feel my family tries really had to support me but financially they are not able to really support me but I can see they want me to succeed because they made sacrifices and yeah they put me in a school which was very good; it was a Model C school and it was expensive and yeah they wanted me to get the best education even before I came here.

Lisa, however, is from a middle-class family, her family understands the importance of a good education and in this regard they had done their homework and selected the best school for their children:

Lisa: Eh, without my family support I would not have had the great school experience because I would have been sent to a bad school. My mom said, 'There is no way, you're going to a bad school', so we moved to the southern suburbs so my sister and I could attend the best school and get the best education.

Other participants described positive school experiences, supportive of an easy transition into university. They spoke about being exposed to diversity, the discipline

at school, awareness of university expectations, and exposure to English as the medium of instruction. Experiences of these participants explained their easy adjustment to a new university environment.

Joyce: I went to a very good school and I was exposed to white teachers, I even had a white lady teaching me isiXhosa. I was mainly with other races and this helped me as I don't have a problem with mixing at university ... Even with English I don't have a problem; everything that we did at school was English first language so yeah that gave me an advantage because we did all sorts of things in that school ... I am very confident and they groomed me very well to be the person I am today in terms of academic and socially.

Letti expressed a similar experience:

Letti: I must say I've been to the best high school ever. It was an all-girls high school and it was white. We were like five blacks in the whole school which made it easy for me at university. They prepared us well for university even not just the education or the quality of the education, but the teachers and like the vice-principal, they inspired me to be who I am today. And yeah I must say my mom and dad worked really hard to put me in that school.

Katlyn spoke about teacher support:

Katlyn: In my high school the teachers cared about you. I think I was very grateful to be in a good school. I had my favourite teacher. She was very supportive but she was like supportive in a gentle and firm way so she motivated me to go to university, I kind of grew to respect her more, and didn't want to let her down.

Alley articulated how her school environment had influenced her behaviour:

Alley: I think my high school actually did a very good job preparing us for university, because we were a very disciplined school, because it was a Christian school so it was like the nuns were like really strict and set in their ways and because of that discipline and what they taught us it helped at university. So at university it's easy for me to have that discipline and actually do the work and not be distracted by the normal things.

The following two participants voiced their opinion on the school influence on preparing them for university:

Tasha: I think my school did a very good job at sending us out there equipped for whatever lay ahead.

Nadia: My school was really focused on good results and university; I remember in my matric year the teachers would always like tell us like work at university is worse than school. They would say it is hard work and we needed to brush up and whatever ... and now I really understand what they meant. Like it's like so much work, it's double the amount of work than matric and I think our teachers prepared us well for this expectation.

The participants' 'good' school experience had a positive influence on their finding their feet and fitting in academically and socially. The following descriptions explain the participants' experience of fitting-in:

Joyce: In my class I talk to everyone, I love to talk and I can mix easily, so I can fit into any group or conversation.

Nadia: The work is not difficult for me, it is easy and I am passing, not with good results but that is just me ... I know I can do better but I am just lazy to put in that effort so I really have no excuse. I think I need to manage my time better, maybe cut down on socialising a bit.

6.5.3.2 Parents' educational aspirations

Three of the eight participants were non-first-generation students which meant that they either had one or both parents who had obtained a university qualification. From the data it was apparent that parents of the non-first-generation participants valued education. Many of the participants had intense conversations with their parents about their academic work and progress. Their parents were involved in the decision-making process about choice of course and university.

Nadia's mother, who is a teacher, was able to provide her with professional, academic and emotional support:

Nadia: My mom is a teacher; she studied the same course as me, even at the same campus, but it had a different name. She is always there to support me at all times; she is there to motivate and inspire me and yeah, like I can discuss my academic work with her; she even helps with my assignments.

Tasha was able to gain advice from her parents with regard to course selection and choice of university:

Tasha: My parents went to university, my dad's an engineer and my mom a legal secretary and they wanted me to do something at university. And uhm, well I've always loved teaching so my parents and I, we sat down and spoke about it and they thought this would be a good direction for me.

Lisa's dedication and commitment to her studies was influenced by her parents' educational aspirations:

Lisa: My parents have been to university; you know, even my granny wanted to do teaching but her mom forced her to be a secretary! My mom is a psychologist. Our whole family is like very focused and academic driven. My sister also got all distinctions in matric so that also helps me because it's like ... you feel like maybe I should be doing this. But definitely the hardworking side is from like what my mom, she achieved ... then you kind of like, maybe I should also do that I also need to achieve you know.

6.5.3.3 Summary: fitting in

The two sub-themes of school experience and parents' educational aspirations illustrated the participants' level of preparedness for university. These aspects reflected their realistic expectations of university life. The value placed upon education by their parents and the participants' privileged school experience helped to ensure an easy transition to university. The participants fitted into their new environment with confidence.

6.5.4 Main theme 4: out-of-habitus experience

When habitus enters a new unfamiliar field, it can create a level of discomfort. From the qualitative data it was evident that some of the participants, in their new higher education field, encountered an out-of-habitus experience. Their expectations of university life, shaped by their past experiences, were different from the reality. This was noticeable in the following sub-themes: religious influence, racial stereotyping, and culture and tradition.

6.5.4.1 Religious influence

Three of the participants spoke about the influence of religion in their lives. Their religious influence has shaped whom they are; how they behave and how they perceive others.

One of the participants' serious commitment to her faith resulted in her experiencing a culture shock. She immediately felt a sense of not belonging:

Alley: My religion has sheltered me; when I came here, and during orientation, they were talking about like the whole clinic and going for HIV testing and that they fall pregnant and stuff. I was a bit shocked. I was, like, do people do that? Because I think I'm just like still very young, I act younger than my age or I still have that mindset that I'm still a child and that I'm growing up a certain way and there is a certain standard and a way that you live your life ... a certain order ... so it was a bit like, oh gosh, people do that! Like people in your first year fall pregnant, oh pregnant, oh okay.

Although Alley's initial reaction was one of shock, she immediately began to reflect on these issues and on herself as an individual. She seemed to be positive and open about these new experiences with regard to other students:

I think I didn't expect the big age groups but I thought oh okay, you've got a child, okay. Oh you're just out of school so I didn't expect that. I thought they would be around my age or younger but you're all this age and this age ... and then some of them were like mommies already. I'm like oh okay, okay. I didn't know you were in the first-year group but cool. So that was a bit interesting for me that I wasn't really like a normal 20-year-old or 25-year-old. So I'm like okay. I still act like I'm 15, but it's okay.

The other two participants spoke about the positive effect that religion has on their person lives:

Thabo: I am a church goer; I go to church every Sunday. Although some of my peers may look at me strangely when I say that I am not afraid to say I go to church, even though I am a man. It has made me who I am ... respectful of others and honest.

Lisa: I am the only one in my family that goes to church and who believes in a higher power. My family ... not so much, although my mom followers the Buddhist teachings. So this is good; each one can believe in what they want, this has made me open to other people's religions but my behaviour is guided by my Christian faith.

6.5.4.2 Culture and traditions

Having participants from different cultures, traditions and backgrounds highlighted how participants engaged with these new experiences. It was apparent that some of the participants struggled with their out-of-habitus experience and were constantly assessing and re-assessing their behaviour.

As evident in Thabo's remarks:

My culture has influenced me a lot, especially in my academic stuff because in our culture it is mostly based on respect and how you should respect the elders and stuff like that so ... it's difficult because I am put in this position whereby I can respect the lecturer who says something even though it doesn't mean the lecturer is right but I need to respect that person as an elder ... so what can I do? I can't just talk back like I see other students do. It's tough!

However Joyce takes a different attitude towards her culture:

I think culture is a personal thing for me. I mean yeah, our grandparents and our parents groomed us to be the people to be able to respect other people and how we relate to other people. I think it becomes a personal thing as we grow up because we tend to see people in a different way, we are exposed to different races, different cultures, how people see themselves; you know for me personally because as I grew up as a teenager I had to learn things on my own.

Tasha doubts the culture that she was brought up with:

I was brought up with a culture that taught me to be kind to people and to keep ... you know, negative things to myself and just deal with them myself ... which I don't know if that's a good or a bad thing but yeah.

Nadia was able to use her cultural experience to her advantage:

My culture has taught me to be um, loyal to others, always be truthful and honest to people. Um, it actually also helps me like here at campus in the university life because it's how you make friends also, you get friends and you earn their trust and you will always have people around you that will be there to help you if you are struggling with your work.

6.5.4.3 Racial stereotyping

Some of the participants' stereotypical expectations of their peers were influenced by their past experiences and perceptions which influenced the way they interacted with students from different cultures and racial groups. This was highlighted by the following comments from some of the participants:

Joyce's perceptions of African students' attitude towards academic work:

When I do an assignment, I don't mix with my own people [African], ... because there is always an excuse when you have to do work, so I say no, let me move to people I know ... if you have my people doing the assignment it will be a problem because there is always a problem at home, there is this, there is that.

But Joyce also adds:

You know, for me it makes me feel sad ... I look at the education that the other students get from the township schools, it's like the standard is just not right, I know they're going to struggle at university.

Joyce's advice to African students:

I told them [African peers] when you come to university, you come for a reason. I am here to study, not worry about other people's marks, you need to put yourself with

people who know they are doing something positive; people that can help you achieve your goal. You are not here to make friends, friends are a bonus; focus on your marks, you are here for something, focus on that.

Joyce's views on white students:

I was surprised mostly to see white people because I ... I always thought that like white people would go for UCT, UWC, those type of campuses, you know what I mean, because I didn't think most of them would do education because most of them get good marks because they come from good schools, like I know someone who passed with 5 distinctions and 2 Bs. I mean she could have done anything but she chose education not because her marks are not good but because she wanted to do education. You know what I mean.

Thabo also had a similar perception of white students:

I see white students getting Funza bursary and it's like what! I always thought that white people are ... they have money. They are not like us [black people]. Seriously I always thought white children ... from the day that they are born money is there for them you know, to go to university. That's the thing because you know you always say why am I always working, whereas the white children just get all the money and everything they want. But they are different from us, they don't have to struggle with academic life.

Lisa's perceptions of working with students from other race groups:

I know and I get along with them [black students]. It doesn't bug me at all. It's not like I'm not gonna ... you know if I have to work with them then fine, like I'm not gonna say no ways. But when we get to choose our groups I choose the girls that are from the same background as me, we always work well together. I would never be able to work with someone who is lazy or never hands in work on time, like I'm always punctual. So ... otherwise, no I'm not like that at all. It's not an issue. Not at all. I love being able to talk to everyone and all the younger ... because you learn from them you know. Like sometimes the experience that you get from them is so nice, yeah.

Tasha's perceptions of students in her class:

The students in my class, for me, they're not as, I won't say as intelligent as ... maybe, that's rude. Maybe I should say, they are not as intellectual as I thought they'd be.

6.5.4.4 Summary: out-of-habitus experience

This main theme emerged from the participants' expressing their emotions and experiences because of their struggles with trying to fit in with their new university environment. The majority of their out-of-habitus experiences related to three sub-themes: religious influence, culture and tradition, and racial stereotypes. All of the sub-themes dealt with issues of diversity. Although it did create a bit of discomfort to the participants, it led to an awareness of diversity among the student population and their new environment.

6.5.5 Main theme 5: Positioning oneself to succeed

This main theme emerged from the participants' reflecting on their social and academic integration at university. It was evident from the data that some participants made a conscious effect to position themselves in situations that contributed positively towards their academic success. The following sub-themes were identified: dialogue with lecturers; choosing peers for group work; and making friends.

6.5.5.1 Dialogue with lecturers

Three of the eight participants felt comfortable with having dialogues with their lecturers in and out of class time:

Joyce: I am very comfortable because I know why I am here ... I don't have problems with speaking to the lecturers ... if I don't know something I'll ask them and if I don't understand I am not afraid to ask the lecturers to explain, they are here to help.

Lisa: If I have a problem or don't understand anything, I go straight to my lecturers. I am not afraid to approach them, definitely not because I feel they are the easiest people to go to, to ask for help and guidance. Nadia was not as comfortable as Joyce and Lisa with interacting with her lecturers, but did engage with them when it was necessary:

Like if I have a query or whatever, like if I have a problem I would go and ask the lecturer or I wouldn't just like sit or wait for ... I would go and ... But if they would ask me a question I would give an answer but I will only speak when I'm spoken to, like that.

The importance of being noticed by the lecturers was acknowledged by Joyce and Lisa:

Joyce: All my lecturers know me, I am very talkative and that's a good thing – then they can see I am serious about my work. I am not here to fool around. I know why I am here and they can see it, especially when they mark my work.

Lisa: I know all my lecturers, yeah. I think when you talk a lot in class and when you contribute to a discussion then I think you can get to know them well and they get to know you well. They know your standard of work and they know what to expect from you. Because I think if you're very quiet and don't attend lectures a lot then I don't think it's any good for you but being noticed, it is important.

The other five participants did acknowledge the importance of engaging with the lecturers, but were also cautious of being considered as 'being stupid' or 'disturbing the class'. Some second-language students didn't feel comfortable with speaking in English, phrasing their questions or engaging in class discussions. They were concerned that their lecturers might perceive them as not understanding their work, or alternatively, they would just not be noticed by their lecturers.

Letti explains her concerns:

You don't want to bother your lecturers with asking too many questions and looking stupid. And like yeah I try to ask my peers now to help me. I don't want to raise my hand the whole time in class. Yeah, so I just keep quiet and when I really don't understand and then afterwards I would go to my friend or whoever, just explain this, help me.

Thabo voices his concern with class interactions:

I would love to be more involved with my lecturers but I am not comfortable asking the questions in class so I would just ask my friends.

Tasha expressed her feelings with regard to not being engaged in class discussions or with her lecturers:

I don't need to engage with them, there is no need to disturb the class but if I have a problem I will approach them out of class. They are all great you know. They are all very open and if I have any problems they are more than happy to assist which is nice.

Katlyn doesn't see the need to be engaged with her lecturers although she is aware of their actions and engagement with certain students in the class:

I don't like asking questions ... and I don't think that my lecturers really know me. I think they've kind of picked their pets and the top achievers and that's it. So you can go through the lesson being unnoticed... like there is a group that do like super well and I like them, they are really nice people, honestly they are but I sat at their table that one time in maths and I got 69 and they all got in the 90s and they like joked with the lecturer asking for a re-write! Like really!

Alley remarked on her interaction with her lecturers and her perception of their engagement with students:

I don't talk in class or engage with my lecturers ... Uhm ... not really but if I need my notes and stuff, if I'm not there then I will go to the lecturer to get it. Uh ... I know some of them know us. It depends on how we interact in class but I think most of them would know the students that talk a lot. I think they know more of the problematic girls than the ones that are actually quiet or the ones they feel have an attitude.

6.5.5.2 Choosing peers for group work

Participants were strategic when choosing peers for group activities. It was evident that they preferred peers who were similar to them with regard to work ethics, attitude towards academic work and similar prior schooling experience. This behaviour was apparent in the following participants' voices:

Katlyn: I am very careful when I choose peers to work with and therefore the friends that I have made have really contributed a lot to my academic work, because in high school my friends weren't very um, ambitious, they just wanted 40% or 45% and now all the people I have surrounded myself with are very ambitious ... they aim high and I mean I didn't get anything below 65%.

Lisa: Yeah. We [peers for group work] just kind of ... like similar; I knew some of these students and then I asked them if we could stay together because we all work hard, all four of us, we all have the same standard. We're like, this is what we want, so when it comes to group projects, we always work together and then we know we are fine.

Thabo: I would like to be in my comfort zone as I've said before. So they [black peers] want to be with people who understand them, who understand them better and not only about school work and all of that, that's what I noticed, so usually I get pulled into this group although I would also like to experience being in other groups with different races, but also working with my race group is comfortable.

Alley: I used to have a group of friends but then I just felt like they weren't very good for me. I like being with people that can make a difference to my life, I can learn from them and be friends with them as well because that's just how I can get better and improve.

Joyce: When I do an assignment, I don't mix with my own people [African], ... because there is always an excuse when you have to do work, ... so I say no, let me move to people I know that are like me and we have the same attitude towards our work.

6.5.5.3 Making friends

It was evident that participants' choice of friends at university was based on the same premise of choosing peers for group work. Most of the participants' intentions for building friendships were based on achieving their academic goals.

Participants wanted to be around friends who could motivate and support them at university as indicated in the following remarks:

Katlyn: I was bullied when I was younger and it's horrible not to have any friends. It's the most humiliating feeling in the world. But now at university I have amazing friends. I think these are friends to keep for life. Because we have so much in common. They helped me a lot and are very ambitious with their marks and that's helped me a lot because I spent a lot of my time with them.

Letti: I love my friends. They're always there when I need them. They are 'like don't worry you are going to be fine'. I'm a stress ball, I stress a lot. So it's like 'no you stress for nothing'. I'm like 'yeah I know'. They're know you're gonna be fine so they're very helpful, very motivating you know. They're there for you when you need them.

6.5.5.4 Summary: positioning oneself to succeed

It was apparent from the evidence that all participants were motivated to succeed in their studies. They had positioned themselves differently according to their prior school experience or background circumstances, to take full advantage of their own potential and the potential of others who could assist them to reach their academic goals. The three related sub-themes, dialogue with lecturers, choosing peers for group work, and making friends, describe how participants were able to position themselves to advance their own academic interests.

6.5.6 Main theme 6: Challenges

The findings under this section illuminate the many challenges that participants encountered on a daily basis in their first year at university. The data revealed how participants reflected and reacted to challenges internal and external to the university which had an influence on their academic studies. The sub-themes that informed the main theme were: balancing work, study, and family commitments; and dealing with diversity.

6.5.6.1 Balancing work, study and family commitments

All participants interviewed had highlighted challenges with regard to either balancing work and study or study and family commitments, or all three aspects: work, study and family commitments.

Joyce regards herself as lucky: she received a bursary and did not have to worry about work. Her only focus was her studies and her son. She seems to have worked out a convenient way to balance all her commitments:

Most weekends I work at home, during the week after campus; my focus is on my son. I have to make sure he does his homework. I have to do his laundry in-between and also cook in-between. I know if I cook in the week then weekends I can focus on my studies.

However, she still considers her situation a challenge and is constantly aware of the difficulties:

I mean I also have things that I have to do at home. I have a son and he keeps me busy all the time but I have to make time for my studies as well because that is just as important – you know what I mean.

Thabo is not as fortunate as Joyce when it comes to financial issues. He has to balance work, study and family commitments. He has a young daughter, a sister and an extended family to take care of. He finds balancing all three activities a challenge, but acknowledges that balancing all these obligations is important to achieve his goal of being financially stable and thereby obtaining a good life for himself and his family:

So ja I'm working, studying and supporting my family again, it's been quite a ride, not an easy one, it's quite daunting at times when you sleep on the weekends, you pass out. Ja because it's been like yoh! the week, is nearly to the end.

Thabo's part-time job is demanding:

Yes I'm still working, I'm working night shift. Joh its quite difficult, it's not easy, because this workload at university it's quite tough, eh sometimes you lack that [sic] levels of concentration, you didn't sleep, you were working nights and at times, not

most of the times, there's this thing where the lesson is boring and you begin to doze off.

It is affecting his academic work:

This week we were submitting a handwriting file on the day when I don't sleep, so it was rush rush to meet the due date and then I got a low mark for that. Had I done it a week before and had time to type it and do well with it and check spellings and everything, I would have done better, much better. Ja so this working is also playing a part. I wouldn't say, I don't want to call it an excuse, because I came in here knowing that it's gonna be like that.

Thabo had to live up to his family's expectations of sending money back home:

Uhm because I grew up there, I think there is that expectation, even though they are not saying that to me, I think there are those expectations of getting money from me. And I try my best to move away from those thoughts, because they're gonna disturb me, in terms of my studies, but I need to send money home.

Katlyn is experiencing problems trying to balance her part-time work and study commitments:

One day I literally got home, I went straight from university, I went ... it was a Friday so I came home, we had a test. I worked from 5:00 till 6:15 and from 6:30 till 12:00 and then the next day from about 9:00 till the whole day and I was exhausted. I was like I can't breathe, but I still need to complete my university work.

Katlyn finds it challenging to share the study at home with the family:

Yeah because I will get home on weekday nights about 6:30 and then I will like have about half an hour to use the study and I try and cram everything into that study session and then as soon as she [mom] gets home I'm like okay, I already know she's gonna come in and say I need to go like on the computer now.

Letti's financial circumstances resulted in her experiencing a challenge in balancing work and study:

......I just have to... I just have to do my work during the week because on weekends I work on a wine farm.

6.5.6.2 Dealing with diversity

The sub-theme, dealing with diversity, articulates the participants' feelings, beliefs and perceptions that were informed by their background environment and prior schooling experiences. From the data it was apparent that the participants felt challenged when dealing with diversity with regard to race, culture, religion, age, educational background, and socio-economic status. An awareness of such diversity posed some serious challenges for the participants in understanding and responding to their peers, lecturers and new university environment.

Thabo talked about his dilemma about interacting with his lecturers; although he is mindful of his culture and the respect due to elders, he understands the importance of having a dialogue with his lecturers:

I'm younger than them [lecturers] so I give them that respect, because our culture is more about respect. Eh I wouldn't say it's the best one, I don't know about that one, but I know when it comes to comparing it and putting culture into it and where it comes from ... eh a lecturer is a lecturer who is older than you; if you don't get a chance to ask this in a class and stuff like that, you'll get the chance maybe next time but for now I don't know if I am comfortable with it.

Thabo adds how this has an influence on his academic work:

My culture is influencing me a lot according to my academic stuff also that has put me in a position whereby I can respect the lecturer who says something even though it doesn't mean the lecturer is right, but I need to respect that person as an elder.

Participants' awareness of diversity was evident in the way they interacted with and perceived their peers.

Joyce's perceptions of diversity and interacting with students from other race groups were more positive and embracing due to her prior school experience:

I just mix with everyone, I talk to everyone. I don't have a problem with that ...

Thabo's attempt at interacting with students from other race groups can be regarded as challenging:

Sometimes you try to fit in, we try to fit in with white people or maybe coloured people 'cause we are African; we try to do that and sometimes it's hard, because eh you get to be in a conversation. Then they [white students] chat and then the chat is about their stuff, while we were chatting on something maybe serious and then they change it and turn to each other and chat and then you feel like an outsider, honestly we feel like an outsider, that's my honest opinion.

This was a challenge for Lisa. She was willing to work with diverse students with regard to academic work; yet she was aware of their diverse educational backgrounds and was sympathetic to their circumstances:

If I have to work with them [African students] it doesn't bug me at all. It's not like I'm not gonna ... you know if I have to work with them then fine, like I'm not gonna say, no ways.

Lisa adds:

I think some of them [African students] struggle. I really do think so ... Shame I feel sorry for them.

Katlyn feels sympathetic to students who are not English first-language speakers and is concerned about how they are able to cope with the situation:

I sometimes feel sorry for some of the Xhosa students because, in say, Professional Studies, we get a thick pile of notes with heavy in-depth English and I don't know how they cope ... I sometimes don't even know what it is and I've been speaking English my whole life so I feel very sorry for them.

Letti's prior experience with diverse populations makes her feel comfortable with the diverse student population as she explains she has been exposed to a diverse population during her upbringing:

I'm Sotho but I grew up in Stellenbosch which exposed me to a lot of people from different races. So I'm even comfortable speaking Afrikaans and isiXhosa. I wasn't raised like in rural areas so ... I was surrounded by all kinds of people, yeah so I don't know where I belong because I'm happy with everybody.

Thabo's lack of interaction with other race groups prior to entering university created a different experience:

Comparing my school experience with the other students [white students], there's quite a major difference, as I've said before, because we come from different backgrounds and schools; we have different personalities and different ways of seeing things and we go about solving issues differently, so ja, those expectations of diversity were just my personal expectations, the ones I made for myself when I thought of a university as being a diverse one; but true diversity is not experienced here, I think that is what I would say politically that is what is lacking.

Some of the participants acknowledged that their intentions of mixing with students from other race groups and cultures were borne out of curiosity to learn more from the diverse student population as remarked by Lisa and Thabo:

Lisa: I love being able to talk to everyone and all the younger [black students] ... because you learn from them you know. Like sometimes the experience that you get from them is so nice, yeah.

Thabo: Eh I want to be amongst other race groups, for instance we were doing a presentation about Islam religion, I found that I wanted to be in another group, a mixed one because I wanted to be more exposed to other people's like culture and the way they think and get challenged in terms of my English and all of that but then my friends wanted me to go with them.

Evident from the data collected was how participants' who interacted with students from other race groups were perceived by their friends and peers.

Joyce adds that students of her own race group have an issue with her interacting with students from other race groups:

My people, the black people, sometimes I feel they have this attitude towards me because I interact with other races most of the time.

Thabo had a similar experience with peers from his race group:

Sometimes my friends see me talking to the white students and will call me this word 'coconut' being white inside and black outside. Yes they would call me that [laughing], and the others will pass by and say 'coconut'.

However, Thabo is willing to embrace these new experiences and take full advantage of getting to know the diverse range of students:

It's nice when I see Xhosa people mixing up with white. There was one mixed group working together in class and I like that, I love that so much. So sometimes it's not right to eh separate yourself to exclude ourselves from that because at the end of the day we are all students, we can learn from one another.

6.5.6.3 Summary: challenges

The main theme in this section highlighted the range of challenges experienced by participants and how they were influenced by their background circumstances and experiences. The other main themes highlighted varying experiences or challenges faced by participants. This section specifically illuminates their challenges in terms of two sub-themes: balancing work, study and family commitments; and dealing with diversity. The findings emphasised that participants' life experiences were a positive force in enabling them to deal with challenging situations inside and outside the university environment.

6.6 Conclusion

This chapter presented the findings of the qualitative data which emanated from oneon-one interviews, the sort-card activity and focus-group interview. A total of eight participants were involved in the one-on-one interviews and six of the eight participants formed part of the focus-group interview. The findings from the sort-card activity revealed that the factors that were most beneficial in supporting a positive first-year experience were external to the university: family support, school experience, self-confidence, and need to succeed. The sort card findings revealed that the main factor hindering a positive first-year experience by participants was their financial plight.

The following main themes, which were data driven, emerged from the one-on-one and focus-group interviews: *determination; self-reliance; fitting in; out-of-habitus experience; positioning oneself to succeed; and challenges.* Sub-themes were identified and provided a deeper and more comprehensive understanding of preentry factors influencing students' first-year experience. Direct quotations were used in this chapter, as they foregrounded the students' voices. Their experiences formed the heart of this study. The students' voices provided a rich description of their firstyear experience and factors influencing these experiences. Overall, participants expressed how their life experiences, which can be viewed as their habitus, had influenced their commitment to improve their socio-economic status, obtain job security and a better life either for their families and/or themselves. This commitment made them more focused and diligent towards their academic performance. Although all these participants had the same intentions, they were underscored by different life circumstances and thus influenced the way they reacted to situations.

The next chapter is the discussion chapter. The key findings from both the quantitative and qualitative data will be discussed by employing the conceptual framework developed to guide this study. The key findings are framed around Bourdieu's key concepts of habitus, capital, and field; and Tinto's concepts of social and academic integration. The conclusions of this study discuss the contributions of this study to students' first-year experience in higher education and the pre-entry academic and non-academic factors influencing these experiences

CHAPTER 7 DISCUSSION OF FINDINGS

7.1 Introduction

The purpose of this chapter is to examine the pre-entry academic and non-academic factors influencing students' first-year experience and academic performance. These factors emerged from the quantitative and qualitative data. The participants' voices are central to the overall research process; hence I have foregrounded the qualitative data as the primary source of information for this discussion chapter. The quantitative data provided an overview and a backdrop to this study and is included in the discussion where appropriate. The conceptual framework, together with relevant literature reviewed in Chapter 3, guided the interpretation and integration of the key findings discussed in this chapter. The conceptual framework developed and explained in Chapter 2 is grounded in key concepts from Bourdieu's conceptual tools (habitus, capital, and field), together with two key concepts from Tinto's integration model (social and academic integration). These concepts have been utilised as analytical tools in this study to comprehend how students' past experiences influence their present and future actions. The conceptual framework provided a foundation from which to view and understand the totality of how students' habitus and capital resources influence their first-year university experience. The framework offered a context to understand how certain key pre-entry factors challenge or assist students' social and academic university integration.

The six key themes and 15 sub-themes that emanated from the data in Chapters 5 and 6 are discussed in this chapter. However, owing to the wide range of sub-themes, and to offer clarity in this discussion chapter, Table 7.1 on the next page provides a structure for the discussion of the findings. I outline how the themes and sub-themes are further categorised, grouped and reduced to two data sets for discussion in Table 7:1 (page 227).

Six key themes	Sub-themes of the findings	Categories for discussion	Data set
Determination	 Family influence Broken families Taking charge of one's own destiny 	Family support Life circumstances	Pre-entry academic & non-academic factors influencing FYE and academic performance
Resilience	Economic hardshipsSocial ills	 Economic status Prior school experience 	
Fitting in	 School influence Parents' educational aspirations 	Diversity in student population	
Out-of-habitus experience	Religious influenceRacial stereotypingCulture and tradition		
Positioning	 Dialogue with lecturers Choosing peers for group work Making friends 	Academic integration Social integration	Adjusting to FYE and academic
Challenges	 Balancing work, study and family commitments Dealing with diversity 	A sense of belonging	performance

Table 7:1 Structure for discussion of findings

The 15 sub-themes (see second column of Table 7.1), are grouped into eight categories (as indicated in the third column of Table 7.1) and listed below:

- Diversity in student population
- Prior school experience
- Economic status
- Family support
- Life circumstance

- Academic integration
- Social integration
- Sense of belonging

These eight categories were further grouped into two larger sets of data (column 4 of Table 7:1 on page 227) and are listed below:

- Pre-entry academic and non-academic factors influencing first-year experience and academic performance.
- Adjusting to the first-year experience and academic performance.

It is important to note that although the findings are discussed under these two broad data sets, the themes and sub-themes under the eight categories are not isolated concepts, but are interconnected and integrated in the discussion. The two data sets frame the focus of this research study and address the following main research question:

How do pre-entry academic and non-academic factors influence teacher education students' first-year experience and academic performance?

7.2 Pre-entry academic and non-academic factors influencing first-year experience and academic performance

In order to capture the participants' first-year university experience and academic performance fully, it was essential to investigate their past experiences. Participants' backgrounds provided a deeper understanding and context to view their multi-layered first-year university experience. The findings in this study identified the following pre-entry factors as influential in participants' first-year experience and academic performance: student diversity, prior academic experience, economic status, family support, and life circumstances. In identifying these significant pre-entry factors, the following key themes emerged: *determination, resilience,* and *challenges,* which are discussed in the following sections in Section 7.2.

7.2.1 Diversity in student population

The diverse nature of students with regard to age, gender, race and socio-economic status, could either have a positive or negative influence on students' university

experience, which could contribute to their either staying or dropping out of university (Fraser & Killen, 2003). The quantitative data provides evidence that the profile of a 'typical' student has shifted. The findings revealed that there was a wide range of ages of participants entering university as first-year students: 19 to 45 years. Consistent with many other studies on changing profiles of first-year students (Tinto, 1993; James, 2002b; Terezini & Reason, 2005; Krause, 2006), the findings from this study reveal that first-year student profiles have become more diverse and inclusive. Close to half of the first-year student population are mature students: 40% within the 20–29-year age group; 6.2% in the 30–39 age group and 0.5% over 40 years of age. The study's sample reveals that the ratio of male to female students entering university has also changed, indicating a ratio of 1:3 for male: female participants. It was found that there are more students entering higher education from fulltime employment than directly from school. Students entering first-year university can be classified into three groups of students: school-leavers, gap students and mature students. The study indicates that 26.2% of the participants were school-leavers entering university directly from school, while 31.8% were gap students and 42% were mature students, either returning directly from fulltime employment or a long period of absence from studying or working. Also evident was a shift in the racial composition of students entering higher education, towards better representation of the racial composition of the country. The first-year intake constituted 82.6% black students and 17.4% white. According to Swail (2004), race can be a factor related to retention at institutions where the diversity of the student body is not met with the same diversity in faculty and institutional leadership. McInnis and James (1995) warn that many universities and academics are not adequately aware of the increasing diversity of the student population or how this may impact on their institution and/or staff. This study intends to address this gap by providing nuances of diverse students' first-year experience and academic performance and uncovering the factors influencing their integration into their new university environment.

The findings on student pathways to university reveal that a large percentage of participants (34.4%) were employed in fulltime jobs and 39% were unemployed prior to entering university. An additional finding on student diversity indicates that for the majority of mature students entering tertiary education, their decision to further their studies was not an impulsive or taken-for-granted one. For many of them, this

decision emanated from their disadvantaged social and economic circumstances. Participants in this study provided the following reasons for their intentions to go to university: secure job, good salary, better life for themselves and their families/extended families, and improved social status 'to be someone'. The research of Tones et al. (2009) obtained similar findings and they elaborate that many of the students entering university from fulltime employment are mature students, predominantly from low socio-economic backgrounds. They add that these students have more family and home obligations that could impede their studies.

An interesting finding is that the majority of the participants entering university are first-generation students: 83.6% and 87.7% of the participants' male and female parents respectively do not possess a tertiary qualification. Studies reviewed reveal that a lack of parental tertiary education deprives first-generation students of the crucial cultural capital associated with education, resulting in certain students being academically advantaged and others disadvantaged at university (Bourdieu, 1984; Pascarella et al., 2004; Reay, 2004). Bourdieu (1984) describes the cultural capital, transmitted tacitly from parental education, to be knowledge, language, values, experiences and ways of doing things that belong to the dominant social group. As a result of parental educational influences, first-generation students, as Noyes (2007-2008:56) states: 'did not enter the university on equal terms' and may struggle with first-year university transition. The findings on this topic are elaborated on further in this chapter under the family support section.

7.2.1.1 Dealing with diversity

The qualitative data reveals that diversity with regard to prior academic experience, culture, and socio-economic background, created a level of discomfort with fitting into the new university environment for some of the participants, mainly, first-generation and mature students. According to Bourdieu (1990), habitus shapes individuals' present and future practices which are structured by past and present circumstances, such as family upbringing, culture and educational experiences. Habitus gives individuals the confidence and feeling of comfort to operate in a particular field. Many of the first-generation participants and those from disadvantaged schools and backgrounds felt an out-of-habitus experience at university. Their expectations of university life, shaped by their past experiences,

were different from their actual first-year experience. Many of the first-generation participants struggled in their new environment and were constantly assessing and re-assessing their behaviour to try to fit in:

Sometimes you try to fit in, we [black students] try to fit in with white people or maybe coloured people cause we black we try to do that and sometimes it's hard, because eh you get to be in a conversation, then they [white students] chat and then the chat is about their stuff, while we were chatting on something maybe serious and then they change it and turn to each other and chat and then you feel like an outsider, honestly we feel like an outsider, that's my honest opinion.

Bourdieu (1993:88) describes this behaviour as the 'power of adaptation'.

The findings reveal that in confronting issues with diversity, participants expressed feelings, beliefs and perceptions that were informed by their background and prior schooling experiences. For example, one participant expressed his perceptions of white students:

I see white students getting Funza bursary and it's like what! I always thought that white people are ... they have money! They are not like us [black people]. Seriously, I always thought white children, from the day that they are born, money is there for them you know, to go to university.

Many of the participants felt 'out of their comfort zone' when dealing with issues related to diversity in race, culture, religion, age, educational background, and socioeconomic status. In addition, an awareness of student diversity created difficult challenges for some of the participants, particularly those from homogeneous schools and communities that limited the participants' exposure to diversity in race, culture and religion. This finding is evident in the following quotation by one of the participants:

I've been doing everything in my first language [isiXhosa] at school. It was a bit of a problem to now do everything in English, because I didn't do that. I just came from a black school.

The participants found it difficult to adjust to their new diverse environment and struggled to integrate and interact with their peers, lecturers and the university culture. The findings also reveal that some participants' stereotypical expectations and cultural beliefs, influenced by their habitus, prejudiced the way they interacted with academic staff and students from different cultures and racial groups. One of the participants found it difficult to interact with lecturers because of his cultural upbringing:

it's difficult because I am put in this position whereby I can respect the lecturer who says something even though it doesn't mean the lecturer is right but I need to respect that person as an elder so what can I do? I can't just talk back like I see other students do, it's tough!

An interesting finding that emerged was that although the diverse experience among some of the participants had brought them a degree of discomfort, it did lead to an awareness of diversity among the student population and a curiosity and preparedness to embrace their new environment which was motivated by their need to succeed in their first year of study. This finding was consistent with that of Tieu and Pancer's (2009) study of first-year undergraduates' involvement with and adjustment to university.

7.2.2 Prior school experience

From the total sample, 79% of the participants attended urban schools while 21% attended rural schools. A significant number of participants (51.7%) attended feepaying schools and 40.4% no-fee schools. Many of the participants who attended fee-paying schools studied at ex-Model C schools. Participants that attended these well-resourced schools exhibited an effortless and natural transition into university life. These participants felt comfortable and fitted easily into their new academic environment with peers, academic staff and the university culture. The following quotation provides evidence of the advantages of being in a good school and the level of preparedness for university:

I went to a very good school and I was exposed to white teachers; I even had a white lady teaching me isiXhosa. I was mainly with other races and this helped me as I don't have a problem with mixing at university. Bourdieu argues that schools in which students traditionally made the transition to university transmit cultural capital to their learners as a necessary pre-requisite for transition to university (Scanlon et al., 2007). This study's data corroborates Bourdieu's statement and affirms that the transition to university was an easy progression for participants who came from well-resourced schools. The findings indicate that these participants' advantaged school experience had prepared them for university, thus creating an alignment between their expectations of university life and their actual university experience. Participants from both working-class and middle-class families who attended ex-Model C schools felt grateful to their parents for sending them to such schools and felt obliged to succeed in their academic studies:

they [her parents] made sacrifices and yeah they put me in a school which was very good, it was a Model C school and it was expensive and yeah they wanted me to get the best education.

These participants were able to see the benefits of their privileged school education and experiences in their first-year university transition. Some of the salient points that these participants from well-resourced schools identified as supportive of an easy transition into university were being exposed to student diversity, teacher support, stringent school discipline, being made aware of university expectations, acquiring academic writing skills, and exposure to English as a medium of instruction.

Evidence from the data indicates that participants who entered university from disadvantaged educational backgrounds lacked the dominant cultural capital needed and were relatively disadvantaged with regard to university expectations and academic performance. For many of these participants, it was their first experience of being in a heterogeneous academic environment with students from diverse racial, cultural and language backgrounds. This created a challenge for these participants as they struggled to fit in. Although many of these students initially struggled to adjust to their new environment, their determination to succeed academically was driven by the need to change their life circumstances. They showed a high level of determination, motivation and self-resilience that influenced their commitment to their

studies and their academic performance. This finding is evident in the following quotation by one of the participants:

I wouldn't say like completely I'm enjoying myself at university, we [black students] came here to work, we want this degree, that's what we want, that is what I want. I want it so bad so I will do whatever it takes to make sure that I get the work done.

Research on personal factors such as motivation, personal interest, confidence, determination and self-esteem indicates that such variables can positively influence academic performance (Kim et al., 2010). My findings are supported by the study of Guiffrida et al. (2013), which uses the concept of Self-Determination Theory (SDT) to reveal that students who attend university to fulfil an intrinsic motivation (personal goal) are positively influenced with regard to their level of academic success and intention to persist. Studies completed on student satisfaction reveal that students' decisions to withdraw from campus are often linked to their lack of motivation and adjustment to the campus environment (Nes et al., 2009).

7.2.3 Economic status

Both the quantitative and qualitative data confirm the lack of financial assistance as a major factor impacting on students' first-year experience and academic performance. Bourdieu (1986) suggests that the most powerful form of capital is economic capital as all forms of capital are inherently linked to economic capital. He adds that students at higher education institutions who are endowed with economic capital resources are at an advantage – they are likely to be more competitive and to succeed. Economic, cultural and social capital is bestowed to children by their parents, thus access to such resources could advantage some students while disadvantaging others. The findings in this study indicate that access to economic capital, determined by the participants' parental qualifications which were aligned with the types of employment they were engaged in, influenced the amount of financial support that parents and 22.2% of female parents were employed in professional occupations. Many parents' lack of tertiary education and their subsequent type of employment had a negative impact on the participants' financial

support. A total of 40.8% of the participants received financial support from their parents, while 53.1% sourced external funding (10.1% received student loans and 43% obtained student bursaries) and 6.1% paid for their own university studies. A similar finding was reported in Noyes' (2007-2008) study which indicates that the financial barriers that accompany the lack of parental tertiary education makes first-generation students' lives more challenging, which may impact on their academic success and university persistence.

Access to economic resources primarily by working-class participants created a barrier to their university studies which resulted in many of them seeking alternative means to fund their studies. Six of the eight participants acknowledged that their financial circumstances were created by their families' economic hardships which made focusing on academic studies a challenge in their first year at university. The findings reveal that many of the participants were compelled to take on part-time employment to fund their tertiary education. Some participants were actively trying to source loans or bursaries, either to supplement their first-year tuition fees or secure funding for the following academic year. These circumstances of trying to balance part-time work and studies, and seek funding for university studies, resulted in many of the participants not being able to give their full attention to their academic studies:

I'm working night shift. Joh! It's quite difficult, it's not easy, because this workload at university, it's quite tough, eh sometimes you lack that [sic] levels of concentration, you didn't sleep,..... you were working nights and at times, not most of the times, there's this thing where the lesson is boring and you begin to doze off.

The data affirms that these participants were unable to enjoy the full benefits of being a university student. Taniguchi and Kaufman's (2005) study indicates that these challenging circumstances of trying to balance study, work and home commitments could create a barrier to students successfully completing their studies. Swail's (2004) study adds that fulltime students working while attending university, paying tuition through loans or grants, and being financially dependent on family, are all factors that could influence academic success and undergraduate retention.

Evidence of economic hardship was strongly identified in this study, mainly among the working-class participants. Findings reveal that participants from disadvantaged communities faced difficult challenges in their upbringing. In most cases, workingclass participants viewed difficult challenges as 'making the best of a bad situation'. However, Cabrera et al. (1992) found that providing financial assistance to disadvantaged students early in the academic year created an equal playing field among the students that were financially advantaged and those that were not.

7.2.4 Family support

From the data it was evident that all the African participants regarded their extended family, aunts, uncles and grandparents, as part of their immediate family. Two of the three African participants were raised by their aunts and grandmothers who played a pivotal role in their upbringing. The discussion in this section includes extended families as part of family support. Findings from the sort-card activity reveal that the highest ranking was accorded to the family support card, with 75% of the participants ranking it in the top three positions, indicating that their families had positively influenced their first-year experience. The cultural capital within family networks influences the kinds of family support received by first-generation and non-first-generation participants. However it was evident that the value of the support received from all the participants' family networks was significant to their first-year university experience.

The nature of family support offered to the non-first-generation participants was both tangible and intangible. Non-first-generation participants noted the following kinds of tangible support: financial, informational, advisory, academic support, and physical assistance. The intangible support included encouragement, motivation and personal support. The findings affirm that parents of non-first-generation participants valued education. Many of the non-first-generation participants had intense conversations with their parents about their academic work and progress:

My twin sister is studying Bachelor of Social Science at the University of Cape Town and my mum just completed her Master's in Psychology – cum laude. So we have academic discussions all the time. The parents of the non-first-generation participants were also involved in the decision-making process about choice of course and university. Non-first-generation participants' parental influence enabled these participants to gain first-hand knowledge of the value of success and preparedness for hard work. These participants displayed a positive attitude towards university and were determined to succeed. Longden's (2004) study supports these findings and highlights that non-first-generation students were able to fit in and thrive in their new environment, without any challenges, due to the tacit knowledge and cultural capital received from their parents.

In stark contrast, first-generation participants from working-class families did not imbibe first-hand experience or knowledge from their families with regard to university studies. They mainly received intangible support from their families, namely, spiritual, emotional, motivational and personal encouragement:

My extended family is very supportive, my aunt's sisters and my aunt's children play a part in saying that this is good for you, keep doing what you are doing.

Dennis et al. (2005) state that despite the lack of first-hand university experience, families can provide encouragement and emotional support which could influence first-generation students' decision to remain and succeed at university. The first-generation participants in this study displayed a positive attitude and were determined to succeed at university. Their purpose for wanting to succeed was based on their family struggles and social ills. They wanted a better life for themselves and their families:

I want to make a success of everything; whatever it takes I want to be successful and be able to help my mom, my dad, my family and support myself if I'm done.

Six of the eight participants had varying degrees of broken family experiences, including, single-parent upbringing from a young age; death of parent/s at a young age; and living with extended families and step-families. These participants had personal experiences of growing up in a broken family environment. The socio-

economic struggles of these participants made them more determined to improve their own circumstances and make their families proud of their achievements:

I have to think of what's going to earn me money because my mom she really struggled, we never had a lot growing up so I didn't want to struggle like that. I want to be financially independent and being successful in my studies will help me achieve this goal.

This finding is also supported in Cabrera's (2014) study on factors influencing firstgeneration minority students' preparedness to attend higher education. In Cabrera's study, attending university was a stepping stone for first-generation minority students and their families. They considered themselves role models for other first-generation minority students. As expressed by one of the participants:

I want to prove to myself that I can do this, even with all of the challenges that I have, but it's just that the need to succeed goes into supporting my family and putting them onto the map as well.

A significant finding from the data is that irrespective of parental educational attainment, both tangible and intangible family support influenced all participants' first-year experience, educational goals and goal commitments (Stage & Rushin, 1993; Martin et al., 1999). Bourdieu (1977:495) emphasised that first-generation students' commitment to succeed and reach their educational goals placed them in an advantageous position equal to the students from the upper sectors of society that understand the need for their children to increase their cultural capital.

7.2.5 Life circumstances

Results from the one-on-one interviews affirm that adversities from participants' past experiences contributed to their wanting to succeed in achieving their educational goals. Two key themes that emerged from the data addressing family struggles and socio-economic circumstances were participants' determination to improve their personal circumstances and their strong resilience against economic hardships and social issues. Findings from this study indicate that the majority of participants articulated determination as a strong factor influencing their need to succeed at university, which was primarily driven by the need to change their life circumstances.
Many of the participants regarded university as a way of improving their lives and taking charge of their destiny:

I decided I want to study, I'm gonna quit work because it's not the life I want for me and I just said to myself, 'No! you need to change your life, you need to go back to study.' I wanted to do something better for me and my son to have a better life!

Both middle-class and working-class participants were faced with social issues beyond their control: drug and alcohol abuse, domestic violence, teenage pregnancy, and the tragic deaths of family members. These social ills contributed to some participants being motivated and determined to succeed at university. In addition, the findings indicate that economic hardships emerged as a strong factor that was influential in working-class participants reinforcing their commitment to their educational goals. The findings of this study suggest that these incidents made the participants more determined and gave them the courage to take control and deal with tough situations in their own personal lives. This finding was evident in the following two quotations:

... these challenges I'm facing at the campus here I feel that I am overcoming them bit by bit; because of my difficult life, it helps.

and

I told myself I'm just gonna pass every year to show them [her parents] I'm not gonna waste their money on me.

Evidence of working-class participants' resilience was acknowledged in their reflection on family hardships. These participants displayed characteristics of confidence, having high expectations, self-esteem and coping techniques to balance work, study and family commitments. The findings from this study affirm previous findings by researchers such as Yorke and Longden (2004), Cleyle and Philpott (2012), and Toni and Olivier (2004) who realised that factors such as family circumstances, past experiences and economic forces were decisive for students to be committed to their educational goals and university studies.

The findings concede that participants were committed to make a positive change by using their challenging socio-economic circumstances as a catalyst. Toni and Olivier's (2004) study supports this conclusion, when they acknowledge that students' past experiences form the root of their beliefs, sense of security, confidence, self-esteem and identity, all of which contribute to their educational goal achievement and academic success.

7.3 Adjusting to first-year experience and academic performance

It was evident in this study that participants' habitus influenced their perceptions, actions and interactions in their new higher education environment. Habitus, according to Bourdieu (1984), refers to one's values, expectations and life-styles gained through everyday experiences and individual disposition. For many of the participants in this study, the key themes of motivation, determination and selfresilience which resulted from their habitus, influenced their social and academic integration and their sense of belonging. According to Tinto (1975, 1993), adjustment to first-year university and academic performance occurs when a student successfully integrates into the institution academically and socially. Tinto maintains that integration is influenced by pre-entry characteristics and goal commitment, interactions with peers and faculty, and out-of-classroom factors (Jensen, 2011). For Bourdieu (1984) integration and a sense of belonging are informed by one's habitus and past experiences. Bourdieu explains that habitus operates in two key ways: it is structured by one's circumstances and it is structuring structure, as it shapes one's present and future practices (Bourdieu, 1984:170). Paulsen and St John (2002) advise that when studying diverse student groups, it is imperative that the students' situated circumstances be taken into account. These authors elaborate situated circumstances to encompass the students' prior educational and background experiences that are shaped by their values, limited resources and capital. The participants' situated contexts helped to explain how the participants' habitus shaped their first-year experience and academic performance. The themes of *determination*, motivation and self-resilience helped to understand why the participants choose to behave in a particular manner while trying to navigate through their first year at university. The key themes that emerged while participants integrated into the new higher education field were *fitting-in*, out-of-habitus experience, positioning oneself to *succeed*, and *challenges*. These themes are discussed under the following three main headings: academic integration, social integration and a sense of belonging.

7.3.1 Academic integration

Participants that were non-first-generation students and also those that came from well-resourced schools were able to fit easily into the academic environment of the university. Their level of academic integration was a natural progression. The findings demonstrate that these participants understood the importance of being connected to their lecturers, support staff and learning environment. Bourdieu and Passeron (1977) suggest that children who have inherited appropriate cultural capital speak the same language as their lecturers and therefore are seen to be academically better than their peers who did not have these privileges. The findings also illustrate how participants deliberately positioned themselves in situations that made them visible to their lecturers. They were aware that this could positively contribute to their academic success. For example, one of the participants stated:

All my lecturers know me, I am very talkative and that's a good thing, then they can see I am serious about my work. I am not here to fool around. I know why I am here and they can see it, especially when they mark my work.

Three of the eight participants that were non-first-generation students felt comfortable having dialogues with their lecturers in and out of class time. They were quite vocal in class discussions and did not hesitate to meet with their lecturers outside of formal class time to discuss academic issues. The findings suggest that these participants' cultural knowledge, which is valued in higher education, was acquired from their home and school environments.

Martin and Spenner's (2009) study revealed similar results to those of Ball et al. (2002), indicating that the different forms of capital, especially cultural capital which more privileged students simply inherit from their parents and prior school experience, play an important role in university educational attainment. Schmid's (2001) study argues that children whose parents are better educated make more money, have higher status jobs and tend to attain higher levels of education than children whose parents lack higher education qualifications. Terenzini et al. (1994) indicate that for non-first-generation students, university is a logical next step after

the completion of high school. They add it is an 'expected and desired stage in the passage toward personal and occupational achievement' (Terenzini et al., 1994:62). For many of the non-first-generation participants and those that attended well-resourced schools, their expectations of university were aligned to their experiences at university. They felt comfortable; the cultural capital that they inherited had placed them in an advantageous position. As stated by Lane and Taber (2012), if a student's cultural capital aligns with that which is prized in educational institutions and that of their lecturers, then that student is more likely to be academically successful.

In contrast, the findings from the first-generation participants that either attended rural or under-resourced schools indicate that they struggled to fit in to their new university environment. Bourdieu (1990) notes that these participants lack the cultural capital needed to make the transition to higher education as they are not aware of the 'rules of the game' and are therefore constantly trying to 'catch up' to their peers that have the relevant cultural capital needed for university life. The two themes that emerged from the data with regard to the first-generation participants were out-of-habitus experience and challenges experienced at university. The mismatch between the first-generation participants' past experiences and expectation of university created an out-of-habitus and challenging first-year experience. The first-generation participants' university expectations were influenced by the following factors: religion, cultural practices, language barriers, lack of parental education, and racial stereotyping. The findings reveal that although these participants were aware of the importance of engaging with their lecturers to enhance their academic success, their values, attitudes and dispositions produced by their cultural capital created a barrier to their academic integration. The findings affirm that many of the first-generation participants did not want to engage in class discussions for fear of being labelled as 'being stupid' or 'disturbing the class'. Some of the English second-language participants did not feel comfortable with communicating in English in class: they were afraid of 'phrasing their questions' incorrectly. According to Bourdieu (1993), language is an embodied cultural capital that originates in the family. Therefore English second- and third-language participants who do not possess the correct linguistic capital dominating the institution have a distinct disadvantage compared with their peers that are English

first-language speakers. Bourdieu explains that linguistic capital is central to educational success as it is closely linked to other forms of capital. However, Bourdieu and Passeron (1990) advise that the notion of habitus helps explain English second-language students' acceptance of the status quo and internalising their inferiority by subscribing to the importance of English competency in higher education. As one of the participants expressed:

I would love to be more involved with my lecturers but I am not comfortable asking the questions in class so I would just ask my friends.

The first-generation participants were concerned that their lecturers might perceive them as not understanding their work or alternatively as not being bright. These participants were also cognisant that not being vocal in class could result in their being unnoticed by their lecturers which could impact on their academic success. Some of the first-generation participants' culture created a barrier to their engagement with their lecturers, for example, they regarded their lecturers as 'elders' who needed to be respected and not questioned. As a result of this situation, these participants found it difficult to engage with their lecturers in and out of formal class time.

The findings from this study concur with the findings from Orsuwan and Cole's (2007) study on Hawaiian students, which reveals that low-income and low parental educational background of students may cause students to experience difficulties as they navigate through the university system which, in turn, could cause them to be less satisfied with their educational experience. A significant finding in this study that is not consistent with many studies of first-generation student persistence is that many of the participants used their challenging circumstances to commit to their academic success and goal achievement. In spite of their low cultural, social and economic capital, inherited from their habitus, they were determined and motivated to persist in their studies. This finding is supported by other studies: Nes et al. (2009), Toni and Olivier (2004), and Nora and Cabrera (1996).

Nes et al. (2009) found that dispositional and academic optimism were associated with better motivation and adjustment to university which reduced dropout rates. They concluded that with increased self-efficacy and a belief in a positive outcome, students can succeed in the academic world. Toni and Olivier's (2004) study adds that students' life goals, which are ignited by specific characteristics and emotions from their past experiences, influence their motivation and commitment to their studies which contribute to academic success. In Nora and Cabrera's (1996) study on minority students, they reveal that even though these students experience negative and challenging experiences at university, it is the support and encouragement from their families that give them the motivation to persist in their studies.

Other studies that highlight challenges faced by first-generation students' persistence include those of Engle (2007), Noyes (2007-2008) and Pascarella et al. (2004). These studies indicate that first-generation students are more likely to experience problems with higher education adjustment and academic and social integration than traditional students and therefore may be more likely to drop out of university.

7.3.2 Social integration

All participants' levels of social integration were influenced by two factors: their commitment to their educational goals and their economic constraints. The findings acknowledge that all participants had positioned themselves strategically to take full advantage of their own potential and the potential of others who could assist them to reach their academic goals. The findings concede that participants used the same criteria for choosing peers for group work and making friends on campus. Participants preferred peers and friends who were similar to them with regard to work ethics, attitude towards academic work and prior schooling experience:

We [peers for group work] just kind of ... like similar, I knew some of these students and then I asked them if we could stay together because we all work hard, all four of us, we all have the same standard. We're like, this is what we want, so when it comes to group projects, we always work together and then we know we are fine.

Participants wanted to be around friends who could motivate and support them at university. The findings admit that covertly students preferred being with peers from their own culture, race and age group. Kuh and Love (2000) found that students who made cultural connections through social groups that reflected their culture of origin were more likely to persist in higher education. This finding was evident in this study; although it advantaged some of the participants from well-resourced backgrounds, it created a challenge for others. Some participants were faced with a dilemma as they also wanted the benefit of being with peers from other cultural and racial groups:

I would like to be in my comfort zone as I've said before. So they [black peers] want to be with people who understand them, who understand them better and not only about school work and all of that, that's what I noticed, so usually I get pulled into this group although I would also like to experience being in other groups with different races, but also working with my race group is comfortable.

Economic capital was another important factor that influenced participants' level of social integration. Their financial constraints resulted in a lack of funding to support their university studies and accommodation, which compelled them to take on part-time employment. The findings consequently reveal that the majority of the participants' low level of social integration was influenced by their commitment to their part-time jobs, families and financial constraints. Rubin's (2012) meta-analysis of 35 studies found that socio-economic status is related to social integration among students in higher education; he adds that working-class students are less integrated at university than middle-class students. An additional factor influencing low levels of social integration of participants was the large percentage of participants' being mature students. The mature students' commitment to their family obligations took priority over social integration at university.

A significant finding of this study is that participants' levels of academic integration were higher than their levels of social integration. As academic success and completion were the ultimate goals for participants, they placed greater emphasis on academic integration than social integration. The findings reveal that participants' decisions to socially integrate with peers and friends at university were purely to advance their own academic goals. This result was more evident among first-generation and mature students. However, in research done by Kuh et al. (2007), they claim that, with regard to increasing student persistence, social integration is essential for non-traditional older students. Other studies undertaken on student

success and retention examine social and academic integration as a cohesive whole. These studies indicate that students' being socially and academically integrated into the university domain can lead to increased academic success, persistence and student satisfaction (Tinto, 1995; McKenzie & Schweitzer, 2001; Astin, 1993). For Terenzini and Pascarella (1978), students' academic and social integration at university is a more significant predictor of university performance and attrition than academic performance. It was evident that participants that possessed the relevant cultural and social capital needed for higher education were better able to make connections with the relevant social networks that benefited them with regard to their academic success than participants that did not have this privilege.

Tinto (1995) suggests that a high level of academic integration can compensate for a lower level of social integration; this was evident in this study's findings, since participants sacrificed their level of social integration for academic integration. Tinto adds that the greater the level of social integration, the greater the level of subsequent commitment to the university. By the same token, the greater the level of academic integration support Tinto's graduation. This study's findings on social and academic integration support Tinto's statement. It was evident from the data that all participants were more committed to their first-year studies than socially integrated. In this regard, the finding suggests that participants were more committed to their educational goals than to their university commitment. The next section discusses this in more detail as it alludes to participants' lack of a sense of belonging to the university.

7.3.3 A sense of belonging

According to Thomas (2012:12), feeling a sense of belonging can be defined as 'students' subjective feelings of relatedness or connectedness'. Literature on the first-year student transition, reviewed in Chapter 3, reveals that for students to successfully make the transition to university life, they need to develop a sense of belonging and connectedness with their new peer groups and the wider academic community. The findings from this study indicate that some participants lacked a sense of belonging or connectedness with the university. The participants' lack of connectedness to the wider academic community was mainly due to their financial

constraints and family obligations. The data generated revealed that many of the participants were engaged in part-time employment which resulted in their spending less time at university; they therefore missed opportunities to socialise with peers.

Another important finding indicates that many participants did not make close friendships at university and expressed a greater level of commitment to their academic studies than socialising at university. These findings are supported by a study completed by McInnis, James and Hartley (2000) on students' level of engagement. Results from their study of a five-year trend on students' level of engagement with university life, show a decrease in student engagement due to part-time work. Their study also indicates that such students spend less time on campus; they do not have close friends, and generally keep to themselves.

Mature participants spent less time on campus because of family obligations. These participants expressed a level of commitment to their studies, family and work. They were faced with many challenges and tough decisions about commitment and sacrifice while trying to balance work and study, work and family obligations, or all three. This created a sense of alienation with and disconnectedness to the university community. Other studies on mature students that concur with this study's finding discovered that mature students often experienced a lack of belonging and fit with traditional university culture (Johnson & Watson, 2004; Kahu, 2014). Wyatt's (2011) study found that mature students may have less desire and time for non-academic activities at university.

First-generation students, mature students and students from working-class families were unable to position themselves to take full advantage of the opportunities available at university or gain the full experience of being a university student, mainly owing to part-time work commitments and family obligations. Increasingly, researchers have argued that higher education institutions have a responsibility to develop a sense of belonging among students which could be acquired through sustained programmes of small-scale initiatives that meet wide-ranging transition needs, such as supportive peer-to-peer relations, encouraging positive interactions between staff and students, and integrating curricular and co-curricular activities (Kift et al., 2010; Thomas, 2012). Pym et al. (2011) suggest that when providing support to first-year students, especially in the South African higher education context,

universities need to take into account students' range of diverse histories, experiences and contexts to develop a supportive community that can promote a sense of belonging at university. Researchers have consistently affirmed that for universities to increase students' commitment to higher education institutions, they need to take a broader view of the student experience and include interactions outside the confines of the university.

7.4 Concluding remarks

The discussion of the findings in this chapter was grouped into the following eight categories that emerged from the themes and sub-themes presented in Chapters 5 and 6: student diversity, prior school experience, economic status, family support, life circumstances, academic integration, social integration, and a sense of belonging. These categories were further grouped into two data sets: pre-entry academic and non-academic factors influencing the first-year experience and academic performance; and adjusting to the first-year experience and academic performance. I now provide a succinct overview of the main threads of the findings drawn from the discussion.

7.4.1 Pre-entry academic and non-academic factors influencing first-year experience and academic performance

Participants' cultural, economic, and social capital, together with their habitus, influenced the way they navigated through their first year at university. It was noted that the pre-entry factors influencing students' habitus and capital which impacted on their academic and social integration and academic performance were: student diversity, prior school experience, economic status, family support, and life circumstances. The data confirms that the first-year student population is extremely diverse and complex, with varying levels of capital resources and habitus. This study identified the following groupings of students: first-generation, non-first-generation, working class, middle class, school-leavers, gap students, and mature students. The data reveals that with the diversity in student population, a typical first-year experience was difficult to identify. However, this study confirms that students' university experience was informed by their past experiences, life circumstances and socio-economic backgrounds.

Life circumstances were a major contributory factor to students attending university. The findings reveal that for non-first-generation participants and middle-class participants, attending university was the next logical step and they had no problems fitting into the university environment. On the other hand, first-generation participants, working-class participants and mature students struggled to adjust to university life. However, their high levels of motivation and determination to change their present circumstances were the incentives driving them to succeed and achieve their academic goals. This in turn influenced their social and academic integration at university.

Economic constraints and family hardships were the primary intention for firstgeneration, mature students and working-class participants to enrol at university. Going to university was not a natural progression for these participants, but an action bred out of necessity. In fact for these particular participants, going to university was breaking family tradition. The motivating force for them to succeed academically was influenced by their parents' struggles and economic adversities. These participants appreciate how privileged they are to have the educational opportunities that their parents lacked. In addition, these participants' educational goals were also driven by their need to change their economic status. The impetus to succeed at university, for these participants, was a means to make money and obtain a stable job with a steady income.

Another important finding was the influence of family support. Participants received varying levels of tangible and intangible support from parents, siblings and extended families. A distinct pattern emerged with the kinds of support offered by families to non-first-generation and first-generation participants. Non-first-generation students regarded their parents and siblings as family. Because of these parents having first-hand knowledge of university life and being financially stable, they were able to offer tangible and intangible support. Not only did these parents assist with the choice of institution and offer advisory, emotional, informational and moral support, but they were also able to financially support their children with university fees and expenses. The data affirms a contrasting pattern with first-generation participants. For many of these participants, family included their aunts, cousins, and grandparents. Many of the first-generation students received intangible support from their extended families:

emotional, religious, motivational and reinforcement. These participants felt proud as they were not only representing their families at university, but their communities. This type of support negated any negative experiences and challenges they had at university as they felt motivated and determined to change their family circumstances.

7.4.2 Adjusting to the first-year experience and academic performance

The participants' level of academic and social integration was guided by their habitus. These predispositions interacted with the university environment to either assist or challenge the participants' first-year experience and academic performance. Their habitus created a sense of belonging to the university. Non-first-generation participants found it much easier to integrate academically into their new university environment than first-generation participants. Overall there was a low level of social integration reported among all participants. However, first-generation participants experienced a lower level of social integration than non-first-generation participants. The findings affirm that participants did not form strong on-campus friendships with their peer groups. Most interaction with regard to social support at university, initiated by participants, was predominantly driven by academic matters. The participants shared educational goals and created on-campus friendships that were helpful to them academically. Participants were strategic in the formation of their friendships, as they tended to gravitate towards peers that shared their own educational goals; came from similar prior schooling experience; or were motivated and had the same work ethics as they had. The results indicate that this behaviour benefited the nonfirst-generation participants in their academic work. With regard to first-generation participants, these friendships provided moral and emotional support that spearheaded their motivation and determination to succeed academically.

The data confirms that some participants lacked a sense of belonging to the university. This finding was affirmed by the participants' low level of social integration, which according to Tinto, brings about a lower level of commitment to the institution. Most participants spent little time at university, outside of formal classes. Participants' financial constraints formed the main factor contributing to their lack of connectedness to the university. The findings concur that participants' limited financial resources and lack of financial assistance compelled them to take up part-

time employment to cover the cost of tuition fees, living expenses and family obligations. This influenced participants' opportunities to socially integrate at university. The findings bring to light that most of the participants saw the university as a 'vehicle' to get to their 'destination'. Achieving their academic goal was their main focus. Toman's (2010) study also reveals that the habitus of mature students and working-class students is associated with the university mainly as a place of learning in which the social aspects are secondary.

It was clear from the findings that although all the participants had similar university experiences and educational goals, their socio-economic backgrounds and life experiences influenced the way they responded to these experiences. Some students felt like 'a fish in water' at university: a privilege gained from their advantaged socio-economic backgrounds and life experiences. However, students from disadvantaged backgrounds felt like 'a fish out of water' as they struggled to fit in. However, the challenges that the latter participants faced at university made them more determined to work hard and succeed.

The next chapter concludes this study. It provides an overview of the study, summarises and concludes the thesis, and provides recommendations based on the findings.

CHAPTER 8 CONCLUDING REMARKS

8.1 Introduction

This study is being concluded at an important juncture in South Africa's development trajectory. The importance of the higher education system in providing a foundation for sustainable development is documented in many of our country's policies. In the most recent State of the Nation address, the President of the country re-affirmed the latter. In the forthcoming period, government will substantively commit to the overhaul of the higher education system. This is evident in terms of the substantive investment of over R600 million a year, earmarked by the Minister of the Department of Higher Education and Training (DHET), to bring about improvements in targeted areas of need in the higher education institutions (Nzimande, 2014a:6). Among the identified areas of need, the 'first-year experience' is currently one of the initiatives aimed at assisting students with the transition from school to higher education (Nzimande, 2014b:7). Programmes such as the first-year experience underscore that the low throughput and high dropout rates continue to be one of the most pressing challenges facing higher education institutions.

Given the foregoing, the initial problem statement which underpinned this study is still relevant, and probably even more so, in the five years since the commencement of this study. The need to ensure a more efficient throughput of first-year students to graduation and into the world of practice is paramount. At the heart of this objective is the question which framed this research study: *How do pre-entry academic and non-academic factors influence teacher education students' first-year experience and academic performance?*

My rationale for undertaking such a study was to highlight the importance of students' past histories and socio-economic and academic backgrounds for higher education institutions and planners of first-year initiatives. If higher education institutions intend to address the challenge of student retention and success, specifically in the first year of study, they need to have a holistic and realistic view of who their students are and where they come from.

8.2 A synopsis of the research journey

In Chapter 1, I outlined the crux of the problem. The shift from an elite to a mass education system to redress past inequalities has brought about a host of new challenges to higher education institutions: high attrition, low throughput, and low graduation rates. This has created an imperative for these institutions to appreciate and adjust to such diversity, particularly among first-year students. Research has recognised the first-year experience as a crucial period with regard to student persistence, academic performance, retention, and success.

This study has been undertaken in response to the need for universities to understand and adapt to the more entrenched diversity of the student body which is one of the characteristics of a post-democratic era. Recent studies have indicated the need to better understand student backgrounds and their influence on the first-year experience and academic performance. Nelson et al. (2011) emphasises the importance of examining the impact of student diversity on the first-year experience and persistence. Further, McInnis (2003) notes that more studies are needed on how student diversity adds value to the first-year experience. Reay (2012) adds that much of the research and literature on higher education student experience and retention has overlooked the influence of students' backgrounds and characteristics. This study addressed this gap in the literature by contributing empirical research to the first-year experience field by means of examining what, how and why certain preentry academic and non-academic factors influence students' first-year experience and academic performance.

Through the journey of this investigation, my focus was not exclusively to provide an in-depth exploration of pre-entry factors influencing students' first-year experience and academic performance, but more importantly to bring about an awareness of students' habitus and social, economic and cultural capital to higher education institutions and first-year curriculum planners so that effective first-year programmes and initiatives could be planned to create an inclusive first-year experience. The main research question and sub-questions in this study provided answers with regard to the nature of the first-year students entering university. The research further addressed the complex and dynamic interplay of first-year students' habitus and capital that influenced their social and academic integration at university.

In order to unpack the problem and obtain a deeper and holistic understanding of the investigation, I developed a conceptual framework to frame the research questions and to analyse and interpret the findings of the study. The framework took into consideration the significance of diversity in understanding the first-year experience. It thus employed concepts from two sociological models: Tinto's integration model (social and academic integration) and Bourdieu's theoretical tools of capital, habitus, and field, to understand how students' past experiences influenced their commitment to their university studies which allowed them to socially and academically integrate into their new environment. The model, as shown in Chapter 2 page 50, served as a conceptual map where the various themes, sub-themes and categories were expanded on.

A review of the literature in Chapter 3 produced a robust set of evidence that supported the need for this study. It was evident from the literature review that much of the research completed on student success and persistence focused mainly on educational experiences within the university environment. Lane and Taber (2012), recommend observing the roles played by social class, capital and habitus in students' educational experiences and academic success, which is addressed in the findings and discussion chapters of this study. The literature review introduced many theories that highlight students' first-year experience from different perspectives. Although many of these theories take into account students' pre-entry characteristics, they do so as control variables rather than key factors 'investigated as variables whose effects are important to understand' (Walpole, 2007:8).

This study's research design and methodology outlined in Chapter 4 provided the grounds and justification for the study's being designed within the qualitative research paradigm. It employed a case study strategy to collect both quantitative and qualitative data to address the research questions of this study. Included in this chapter were the strategies undertaken to ensure the study's credibility, transferability and dependability.

The quantitative findings chapter of this study provided a detailed overview of who the first-year students are who enter teacher education programmes at this university. This chapter specifically explored student diversity under the following headings: student demographics, socio-economic background, prior schooling, and student expectations of university experience. The qualitative findings were presented in Chapter 5. The first section provided a biographical description of each of the eight participants. The second section highlighted the outcomes from the sort-card activity and the final section presented the qualitative findings from the one-on-one and focus-group interviews, under themes and sub-themes that emerged. The six themes and 15 sub-themes presented in Table 6.5 included the following: *determination, self-reliance, fitting-in, out-of-habitus experience, positioning oneself to succeed, and challenges.* The sub-themes identified provided a deeper, more comprehensive understanding of pre-entry factors influencing students' first-year experience.

8.3 Research outcomes

8.3.1 Theoretical contribution

The conceptual framework developed in Chapter 2 guided the interpretation and integration of the key findings of this study. It provided a lens for a critical analysis of the extant literature on the first-year experience and academic performance.

The use of Bourdieu's three concepts (habitus, capital, and field) revealed the realities of students' situations. The utilisation of the concepts provided a more detailed and coherent account of how the environments in which students were raised, as well as their habitus and capital resources, shaped their attitudes and interpretation of their first-year experience. A key contribution of using Bourdieu's concepts revealed that participants who were economically and socially disadvantaged possessed highly valued forms of cultural capital which assisted them to be committed to their educational goals and persist in their challenging first-year experience. In addition, the findings indicated that for many of the mature students, attending university was not their initial aim. They decided to attend university because of their job expectations, economic hardships and life circumstances. In addition, many of the working-class and first-generation students displayed a high level of determination, commitment and resilience to achieve their academic goals. For these participants, achieving academic success was seen as a solution to eradicate their economic hardships and improve their family status. In addition, the findings reveal that students from privileged backgrounds and non-first-generation students were equipped to thrive and succeed in their first year at university. They were able to use the economic, social and cultural resources gained from their privileged backgrounds to position themselves more favourably and benefit academically and socially from their new environment.

The use of Tinto's concepts of social and academic integration highlighted a significant contribution. The findings indicated that, while Tinto's model of student integration demonstrates that academic and social integration operates concurrently at varying levels of students' first-year experience, in contrast, the participants of this study viewed the academic and social domains at the university as successive. Students' academic and social integration was influenced by their limited economic capital which resulted in their prioritising their concerns. For example, many of the first-generation and working-class participants regarded their financial concerns as a priority. Once they were able to deal with these concerns, only then did they shift their focus to their academic concerns. Socialising on campus was only considered by participants once they felt confident and secure that the other two concerns had been achieved. For many of the participants, their educational goal was viewed as their primary focus. This level of prioritising between social and academic integration was also evident among many of the mature students. Owing to their lack of capital resources, they focused exclusively on their academic success to the exclusion of campus social life. For these participants, their personal and family commitments took priority over socialising on campus. Another important finding with regard to social and academic integration was the manner in which participants created strategic relationships with peers. This was visible in the way they had formed bonds with peers for class group activities and out-of-class support. The main intention of their actions was to advance their educational goals rather than their social needs, as stated by one of the participants: 'Primarily we are here to study, not make friends.'

This research study has provided an example of how an integrated conceptual framework of Bourdieu's concepts and Tinto's integration model can be applied in today's university settings to bring about a nuanced perspective to investigate and understand the first-year experience. The development of a conceptual framework modelling the forms of and the relations between students' pre-entry factors and

university social and academic integration into a microcosm of the higher education field represents a contribution to the body of knowledge informing the understanding of students' first-year experience.

8.3.2 Contribution to practice

This study's findings could be extended to other universities that are committed to provide an inclusive first-year experience, as its application could influence students' first-year persistence and academic success. Analysis of the collected data and findings of this study could benefit first-year curriculum planners, student support services, first-year experience initiatives and orientation programmes. Readers should be cautioned, however, not to over-generalise the findings as they pertain to a specific university setting.

Much of the literature on first-generation students acknowledges that these students are less prepared for higher education and are more likely to drop out in their first year of study. This research study has identified that, in spite of these expected outcomes, participants from disadvantaged backgrounds, with diverse educational experience, enter university with high levels of resilience, determination and motivation to succeed. My recommendation is that higher education institutions need to recognise these traits in first-generation students and incorporate them in planning initiatives for first-year student success and retention. In addition, this study has highlighted that students predominantly make use of their own inner strengths and limited capital resources to navigate their first year at university. The results further indicate that participants' level of engagement with regard to accessing academic support is limited. In this respect, higher education institutions need to be proactive in the kinds of support offered to students. The support offered needs to be inclusive and acknowledge the diversity of students' past experiences. This type of support also needs to be embedded in the mainstream first-year programme delivery.

Another important contribution to practice identified by this study was that the firstyear experience for many of the participants was adversely influenced by their financial constraints. This resulted in many of the students being engaged in parttime employment which had an impact on their level of social integration into and subsequent level of commitment to the institution. Studies by Hillman (2005) and Tinto (1995) acknowledge that for students to be successful at university, they need to be socially and academically integrated into the university domain. The findings from this study identified that students' social and academic integration was dependent on their financial security. A recommendation is that if higher education institutions want their students to enjoy the full benefits of integration into the social and academic domain of university life, these institutions need to play an active role in ensuring that financial provision is made in first-year programme implementation and policies that facilitate this. For example, universities could assist students with financial aid prior to their entering university. Student loans and bursaries could be confirmed prior to first-year registration, which would mitigate the burden and stress of trying to secure financial aid while also trying to cope with first-year studies and other commitments such as finding a part-time job and personal issues like accommodation. The findings of this study indicate that a large number of students are involved in part-time employment and have out-of-campus obligations. In this regard first-year programme planners need to consider this important factor and be flexible when planning the first-year timetable.

This study's research findings identified that students' level of academic integration was higher for non-first generation students than first-generation students. Academic staff teaching on first-year programmes need to be aware of this factor and find creative, meaningful ways of making contact with these students and increasing their level of academic integration in a non-threatening and approachable manner. These students need to feel comfortable with addressing their lecturers and being involved in class discussions. Universities need to find proactive ways to develop student – lecturer interactions to improve student success and retention.

My research highlighted that all participants' levels of social integration were notably lower than their academic integration; this was supported by the fact that all these students felt more committed to their own educational goals than to the goals of the institution. Many studies have implied that social integration is related to better academic performance, greater persistence, and retention (Tinto, 1975; Pascarella & Terenzini, 2005). In this regard, it is recommended that higher education institutions assume the responsibility to ensure that students socially integrate into the university environment. Universities need to provide space and opportunities for students to socialise on campus. They need to offer support to enhance students' social networking skills – something that is not only needed for students at university, but also in the professional environment. In addition, encouraging social networking at universities would allow students with limited social and cultural capital the opportunity to engage with students who have the capital needed for success in higher education. Research has shown that students are spending less time on campus and few or no close friendships are formed at university.

Finally, this research does not only have implications for broad educational practice, but more specifically for the policies and educational standards within specific departments and programmes concerned with first-year experience initiatives. Valuing diversity or acknowledging and addressing the challenges to facilitate an inclusive first-year experience is pivotal to academic success and retention.

8.4 Critique of the research and future research opportunities

As the journey through this investigation neared conclusion, I was able to reflect on the research process and identify some limitations in my study, which otherwise could have enhanced the research findings of this study. As with similar studies of this nature, I experienced the normal limitations of a postgraduate student engaged in part-time studies, such as time constraints and access to resources. More importantly, and in retrospect, the findings could have been enriched by multiple research sites. The study was confined to a single first-year programme in a single institution. A richer texture to the data could have been obtained if the sample had included more than one programme as well as more than one institution.

Secondly, from a methodological perspective, the use of reflective diaries to collect evidence may have yielded deeper insight into students' attitudes, thoughts and actions. The reflective diaries would have captured accounts of the participants' firstyear experiences that they perceived as somewhat significant and meaningful to them. It would have provided a more holistic understanding of students' experiences impossible to capture in a single one-on-one interview. Data from the reflective diaries would have provided greater depth and clarity, and further enriched the findings of this study. Analysing the data from participants' diaries would have provided me with a sound understanding of students' fears, concerns and the role players influencing their learning environment and academic performance. In addition it would have also strengthened the validity of the study by providing another element for triangulation.

8.4.1 Future research

I suggest the following key areas for future research to advance the knowledge and understanding of the first-year experience of university students:

- An investigation into the extent to which the conceptual framework developed in this research is applicable to other higher education settings and programmes.
- Longitudinal research into the first-year experience to provide a broader perspective of the pre-entry factors influencing students' university experience and academic success.
- An examination of how specific first-year student cohorts, for example first-generation or mature students' social, economic and cultural capital could be utilised to enhance social integration.
- An investigation into the manner in which students' habitus and capital resources enable them to successfully navigate the first year of study.

8.5 A personal reflection of what this journey meant to me as the researcher

I was a first-generation undergraduate university student. I grew up in an extended family environment, living with parents, grandparents, aunts, uncles, cousins and siblings all under one roof. Growing up during the apartheid era in South Africa afforded me a school experience that had limited resources, but dedicated teachers and a homogeneous group of students based on race. My habitus was quite similar to many of the non-white, first-generation participants of this study.

I was fortunate to have had the economic resources to attend university. My parents' lack of senior high school and tertiary education made them determined to provide better opportunities for their children. I was the first of my siblings to attend university. My limited social and cultural resources created a real challenge in transitioning from school to university.

My own first-year university experience served as an invaluable resource in conducting this study. It made it easier for me to relate to, and understand, issues raised by the participants from the diverse and also disadvantaged backgrounds. Sharing my own experiences with the participants made them feel comfortable to share their experiences. In addition, my context of being a first-generation student, and my present position as an academic, provided a common ground for me to probe deeply and gain clarity on issues pertaining to this study.

In my reflection of my entrée into the higher education field, concomitant with my disadvantaged context and not possessing the dominant culture of the university, I felt marginalised within the higher education field. At times I found it quite daunting and difficult to adjust to the dominant culture of the institution. But, my experiences in the field resulted in my habitus 'structuring the structured', which directed me to the focus of my doctoral study. In retrospect, I think my research reflects my desire to enable and legitimise marginal field positions which are brought about by certain habitus and capital resources not recognised by the dominant culture within higher education institutions.

8.6 Conclusion

The aim of this study was to investigate who the first-year students are who enter teacher education programmes and what pre-entry factors influence their first-year experience and academic performance. The purpose was to address the research problem: challenges of high student dropout, low throughput, and poor graduation rates experienced by higher education institutions. The use of a conceptual framework developed to analyse and discuss the actions and interactions of participants' past experiences on their social and academic integration into their new university environment assisted in answering the research question. The results revealed that students' first-year university experience was dynamic and in continuous flux, structured and restructured by their past experiences is a shifting target that evolves and changes with time and culture. Therefore studies on students' first-year experience require research and practice to continuously re-evaluate relevance and points of possible enhancement.

Finally, the findings of this study provide insight into how unequal social, economic and educational backgrounds, created by an unjust apartheid system in South Africa (pre-1994), have created challenging circumstances for students entering higher education institutions. It is imperative that these institutions move away from viewing today's first-year students from a deficit model. They should, instead, capitalise on the qualities of determination, optimism, enthusiasm and openness to learning to create an inclusive first-year experience.

Most importantly, this investigation has provided evidence that the improvement of success, especially at the first-year level, requires recognition and acknowledgement of the after effects of the apartheid era upon the majority of disadvantaged students in South Africa, and by doing so will reduce dropout rates, improve throughput and graduation rates, and inevitably have a positive net effect on the South African human capital skills base.

References

Abramson, M. & Jones, P. 2003. Tinto's model of retention revisited: A case study from a 'new' British university. In Thomas, L.,. Cooper, M. & Quinn, J. (eds). *Improving completion rates among disadvantaged students*. Stoke-on-Trent, Trentham Books: 133-148.

ACT (American College Test). 2004. U.S. colleges falling short on helping students stay in college. ACT News Release, December 13. http://www.act.org/news/releawses/2004/2012-2013.html [15 November 2011].

ACT (American College Test). 2007. Issues in college success: The role of nonacademic factors in college readiness and success. Iowa City, IA. <u>http://www.act.org/research/policymakers/pdf/nonacademic factors.pdf</u> [12 September 2012].

Alhojailan, M.L. 2012. Thematic analysis: a critical review of its process and evaluation. *West East Journal of Social Sciences*, 1(1):8-18.

Asmar, C. 2004. Cultural difference in Western universities: intercultural and internationalized responses to a changing world. In Rust, C. (ed.). *Improving student learning: diversity and inclusivity*. Oxford: Oxford Centre for Staff and Learning Development: 9-20.

Astin, A.W. 1975. *Preventing students from dropping out.* San Fransisco, CA: Jossey-Bass.

Astin, A.W. 1985. Achieving educational excellence. San Francisco, CA: Jossey-Bass.

Astin, A.W. 1993. What matters in college? Four critical years revisited. San Francisco, CA: Jossey-Bass.

Atkins, L. & Wallace, S. 2012. Qualitative research in education. London: Sage.

Atkinson, W. 2011. From sociological fictions to social fictions. *British Journal of Sociology of Education*, 32(3):331-347, May.

Attinasi, L.C. 1989. Getting in: Mexican Americans' perceptions of university attendance and the implications for freshman year persistence. *Journal of Higher Education*, 60(3):247-277, May–June.

Australian Government. 2009. *Transforming Australia's higher education system*. Canberra: Dept of Education, Employment and Workplace Relations. http://www.go8.edu.au/__documents/go8-policy-analysis/2010/accountability-forguality-agenda/6_referencesattachments.pdf [3] October 2013].

Babbie, F. & Mouton, J. 2001. *The practice of social research*. Cape Town: Oxford University Press.

Badat, S. 2010. *The challenges of transformation in higher education and training institutions in South Africa*. Midrand: Development Bank of Southern Africa.

Ball, S.J., Davies, J., David, M. & Reay, D. 2002. 'Classification' and 'judgement': social class and the 'cognitive structures' of choice of higher education. *British Journal of Sociology of Education*, 23(1):51-72.

Bean, J.P. 1980. Dropouts and turnover: the synthesis and test of a causal model of student attrition. *Research in Higher Education*, 12(2):155-187.

Bean, J. P. 1990. Why students leave: insights from research. In Hossler, D. & Bean, J.P. (eds.). *The strategic management of college enrollments*. San Francisco, CA: Jossey-Bass: 147-169.

Bean, J.P. & Eaton, S.B. 2000. A psychological model of college student retention. In Braxton, J.M. (ed.). *Reworking the student departure puzzle: new theory and research on college student retention.* Nashville, TN: Vanderbilt University Press: 48-61.

Bean, J.P. & Metzner, B.S. 1985. A conceptual model of nontraditional undergraduate student attrition. *Review of Educational Research*, 55(4):485-540, Winter.

Bell, J. 2005. *Doing your research project: a guide for first-time researchers in education, health and social science.* 4th ed. Maidenhead: Open University Press.

Benham, M.K.P. 2006. A challenge to Native Hawaiian and Pacific Islander scholars: what the research literatures teaches us about our work. *Race Ethnicity and Education*, 9(1):29-50, March.

Berger, J.B. 2000. Optimizing capital, social reproduction, and undergraduate persistence: a sociological perspective. In Braxton, J.M. (ed.). *Reworking the student departure puzzle: new theory and research on college student retention.* Nashville, TN: Vanderbilt University Press: 95-124.

Berger, J.B. & Milem, J.F. 2000. Organizational behavior in higher education and student outcomes. In Smart, J.C. & Tierney, W.G. (eds). *Higher education: handbook of theory and research, Vol. 15.* New York, NY: Agathon Press: 268-338.

Bertram, C., Appleton, S., Muthukrishna, N. & Wedekind, V. 2006. The career plans of newly qualified South African teachers. *South African Journal of Education*, 26(1):1-13.

Bessoondyal, H. 2005. Gender and other factors impacting on mathematics achievement at the secondary level in Mauritius. Unpublished DEd thesis, Curtin University of Technology, Perth, Western Australia.

Bitzer, E. 2005. First-year students' perceptions of generic skills competence and academic performance: a case study at one university. *South African Journal of Higher Education*, 19(3):172-187.

Bloor, M., Frankland, J., Thomas, M. & Robson, K. 2001. Focus groups in social research. London: Sage.

Bogdan, R.C. & Biklen, S.K. 2007. *Qualitative research for education: an introduction to theory and methods*. 5th ed. Boston, MA: Pearson Allyn & Bacon.

Bourdieu, P. 1971. Systems of education and systems of thought. In Young, M.F.D. (ed.). *Knowledge and control: new directions for the sociology of education.* London: Collier-Macmillan: 189-207.

Bourdieu, P. 1973. Cultural reproduction and social reproduction. In Brown, R. (ed.). *Knowledge, education and cultural change: papers in the sociology of education.* London: Tavistock: 71-112.

Bourdieu, P. 1977. Cultural reproduction and social reproduction. In Karabel, J. & Halsey, A.H. (eds). *Power and ideology in education.* New York, NY: Oxford University Press: 487-511.

Bourdieu, P. 1984. *Distinction: a social critique of the judgement of taste.* Cambridge, MA: Harvard University Press.

Bourdieu, P. 1986. The forms of capital. In Richardson, J.E. (ed.). *Handbook of theory and research for the sociology of education.* Westport, CT: Greenwood Press: 241-258.

Bourdieu, P. 1990. The logic of practice. Stanford, CA: Stanford University Press.

Bourdieu, P. 1993. *The field of cultural production: essays on art and literature.* Cambridge: Polity Press.

Bourdieu, P. 1998. *Practical reason: on the theory of action*. Stanford, CA: Stanford University Press.

Bourdieu, P. & Passeron, J.C. 1977. *Reproduction in education, society and culture*. London: Sage.

Bourdieu, P. & Passeron, J.C. 1990. *Reproduction in education, society and culture*. 2nd ed. London: Sage.

Bourdieu, P. & Wacquant, L.J.D. 1992. *An invitation to reflexive sociology*. Cambridge: Polity Press.

Braun, V. & Clarke, V. 2006. Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3(2):77-101.

Braxton, J.M. & Hirschy, A.S. 2005. Theoretical developments in the study of college student departure. In Seidman, A. (ed.). *College student retention: formula for student success.* Westport, CT: Praeger.

Braxton, J.M., Hirschy, A.S. & McClendon, S.A. 2004. Understanding and reducing college student departure. San Francisco, CA: Jossy-Bass.

Briggs A.R.J., Clark, J. & Hall, I. 2012. Building bridges: understanding student transition to university. *Quality in Higher Education*, 18(1):3-21.

Brunsden, V., Davies, M., Shevlin, M. & Bracken, M. 2000. Why do HE students drop out? A test of Tinto's model. *Journal of Further and Higher Education*, 24(3):301-310.

Bryman, A. 1988. Quantity and quality in social research. London: Unwin Hyman.

Bryman, A. 2008. Social research methods. 3rd ed. Oxford: Oxford University Press.

Bunting, I. 2006. The higher education landscape under apartheid. In Cloete, N., Fehnel, R., Gibbon, T., Maassen, P. & Moja, T. (eds). *Transformation in higher education: global pressures and local realities.* Amsterdam: Springer: 35-52.

Cabrera, A.F., Nora, A. & Castaneda, M.B. 1992. The role of finances in the persistence process: a structural model. *Research in Higher Education*, 33(5):571-593.

Cabrera, A.S.P. 2014. First generation minority students: understanding the influential factors that contributed to their preparation and decision to pursue higher education. *Portland State University McNair Scholars Research Online Journal,* 8(1), Article 2. DOI: 10.15760/mcnair.2014.1 http://pdxscholar.library.pdx.edu/mcnair/vol8/iss1/2 [8 April 2015].

Cameron, R. 2011. Mixed methods in business and management: a call to the 'first-generation'. *Journal of Management & Organisation*, 17(2):245-267, March.

Case, J., Marshall, D. & Grayson, D. 2013. Mind the gap: science and engineering education at the secondary-tertiary interface. *South African Journal of Science*, 109(7/8):1-5, July-August.

CHE (Council on Higher Education). 2009. *The state of higher education in South Africa.* Pretoria: Council on Higher Education.

CHE (Council on Higher Education). 2013. A proposal for undergraduate curriculum reform in South Africa: the case for a flexible curriculum structure: Report of the Task Team on Undergraduate Curriculum Structure. Pretoria: Council on Higher Education.

Chetty, R. 2012. Theoretical reflections on critical literacy: an engagement with power and marginalisation. Inaugural professorial address, Cape Peninsula University of Technology, Cape Town, 15 March.

Cleyle, S. & Philpott, D. 2012. Developing an effective first year experience for students with academic challenges: a proposal for a pilot program. St. John's, Newfoundland: Memorial University of Newfoundland.

Cohen, L., Manion, L. & Morrison, K. 2001. *Research methods in education*. 5th ed. London: RoutledgeFalmer.

Cohen, L., Manion, L. & Morrison, K. 2008. *Research methods in education*. 6th ed. London: Routledge.

CPUT (Cape Peninsula University of Technology). 2014. IMIS (Institutional Management Information System). https://connect.cput.ac.za/ [15 July 2014].

Creswell, J.W. 2003. *Research design: qualitative, quantitative, and mixed methods approaches.* 2nd ed. Thousand Oaks, CA: Sage.

Creswell, J.W. 2009. *Research design: qualitative, quantitative, and mixed methods approaches*. 3rd ed. Los Angeles, CA: Sage.

Creswell, J.W. & Plano Clark, V.L. 2007. *Designing and conducting mixed methods research.* Thousand Oaks, CA: Sage.

Crotty, M. 1998. *The foundations of social research: meaning and perspective in the research process.* St Leonards, NSW: Allen & Unwin.

Crouch, L. & Perry, H. 2003. Educators. In *HSRC human resources development review 2003: education, employment and skills in South Africa*. Cape Town: HSRC Press: 477-499.

Curtis, B. & Curtis, C. 2011. *Social research: a practical introduction.* Los Angeles, CA: Sage.

Czerniewicz, L. & Brown, C. 2013. The habitus of digital 'strangers' in higher education. *British Journal of Educational Technology*, 44(1):45-53, January.

Davey, G. 2009. Using Bourdieu's concept of habitus to explore narratives of transition. *European Educational Research Journal*, 8(2):276-284, June.

DBE (Department of Basic Education). 2011. Annual report: 2010/11. Pretoria: DBE.

DBE (Department of Basic Education). 2012. *Education statistics in South Africa:* 2010. Pretoria: DBE.

DBE & DHET (Department of Basic Education & Department of Higher Education and Training). 2011. *Integrated strategic planning framework for teacher education and development in South Africa: 2011–2025.* Pretoria: DBE; DHET.

Dennis, J.M., Phinney, J.S. & Chuateco, L.I. 2005. The role of motivation, parental support, and peer support in the academic success of ethnic minority first-generation college students. *Journal of College Student Development*, 46(3):223-236, May–June.

Devas, A. 2011. Widening participation and the media student experience, *Higher Education*, 62(6):815-828.

Dey, E.L. & Astin, A.W. 1989. *Predicting college student retention: comparative national data from the 1985 freshman class*. Los Angeles, CA: Higher Education Research Institute, Graduate School of Education, UCLA.

DHET (Department of Higher Education & Training). 2010. Report on the Stakeholder Summit on Higher Education Transformation held at the Cape Peninsula University of Technology, Bellville, 22–23 April 2010. http://www.cepd.org.za/files/Report_Stakeholder_Summit_on_HE_Transformation_2 010.pdf [9 September 2011].

DHET (Department of Higher Education & Training). 2012a. *Green Paper for post*school education and training. Pretoria: DHET.

DHET (Department of Higher Education & Training). 2012b. *Guidelines: Organising framework for occupations (OFO) 2012.* Pretoria. DHET.

DHET (Department of Higher Education & Training). 2013. *Statistics on post-school education and training in South Africa: 2011.* Pretoria: DHET.

DiMaggio, P. 1979. Review essay: On Pierre Bourdieu. *American Journal of Sociology*, 84(6):1460-1474.

DiMaggio, P. & Mohr, J. 1985. Cultural capital, educational attainment, and marital selection. *American Journal of Sociology*, 90(6):1231-1261, May.

DiRamio, D.C. & Jarvis, K. 2011. Transition 2.0: Using Tinto's model to understand student veterans' persistence. In DiRamio, D.C. & Jarvis, K. *Veterans in higher education: when Johnny and Jane come marching to campus.* San Francisco, CA: Wiley Jossey-Bass: 35-49.

DoE (Department of Education). 1997. *Education White Paper 3: a programme for the transformation of higher education.* Pretoria: DoE.

DoE (Department of Education). 2005. *Student enrolment planning in public higher education.* Pretoria: DoE.

DoE (Department of Education). 2007. *The national policy framework for teacher education and development in South Africa*. Pretoria: DoE.

DoE (Department of Education). 2009. *Educator profile report: 2004–2007*. Pretoria: Directorate: Education Planning, Provisioning and Monitoring.

Driscoll, D.L., Appiah-Yeboah, A., Salib, P. & Rupert, D.J. 2007. Merging qualitative and quantitative data in mixed methods research: how to and why not. *Ecological and Environmental Anthropology*, 3(1):19-28.

Dumais, S.A. 2002. Cultural capital, gender, and school success: the role of habitus. *Sociology of Education*, 75(1):44-68, January.

Dumais, S.A. 2006. Early childhood cultural capital, parental habitus, and teachers' perceptions. *Poetics*, 34(1):83-107.

Dumais, S.A. & Ward, A. 2010. Cultural capital and first-generation college success. *Poetics*, 38(3):245–265.

Duquette, C. 2000. Experiences at university: perceptions of students with disabilities. *Canadian Journal of Higher Education*, 30(2):123-142.

Engle, J. 2007. Postsecondary access and success for first-generation college students. *American Academic*, 3(1):25-48.

Engstrom, C.M. 2008. Curricular learning communities and unprepared students: how faculty can provide a foundation for success. *New Directions for Teaching and Learning*, (115):5-19, Fall.

Essop, A. 2008. The incorporation of colleges of education into higher education: a brief synopsis. Pretoria: DoE.

Finch, H. & Lewis, J. 2004. Focus groups. In Ritchie, J. & Lewis, J. (eds). *Qualitative research practice: a guide for social science students and researchers*. London: Sage.

Fjortoft, N.F. 1995. College student employment: opportunity or deterrent? Paper presented at the annual meeting of the American Educational Research Association, San Francisco, CA, 18–22 April.

Flick, U. 2006. An introduction to qualitative research. 3rd ed. London: Sage.

Frank, A.W. 2002. "How can they act like that?" Clinicians and patients as characters in each other's stories. *Hastings Center Report*, 32(6):14–22,November–December.

Fraser, W.J. & Killen, R. 2003. Factors influencing academic success or failure of first-year and senior university students: do education students and lecturers perceive things differently? *South African Journal of Education*, 23(4):254-263.

Gaddis, S.M. 2013. The influence of habitus in the relationship between cultural capital and academic achievement. *Social Science Research*, 42(1):1-13, January.

Geyser, H.C. & Wolhuter, C.C. 2001. Career paths of education graduates and the implications thereof for higher education. *South African Journal of Higher Education*, 15(2):93-97.

Gibb, A. 2005 *Towards the entrepreneurial university: entrepreneurship education as a lever for change.* Birmingham: National Council for Graduate Entrepreneurship.

Gordon, A. 2009. *Restructuring teacher education*. Johannesburg: Centre for Education Policy Development.

Gregory, M. 2012. The first-year experience at the University of Botswana. In Leibowitz, B., Van der Merwe, A &. Van Schalkwyk, S. (eds). *Focus on first-year success: perspectives emerging from South Africa and beyond*. Revised ed. Stellenbosch: Sun Media: 99-111.

Grenfell, M. 2004. Pierre Bourdieu: agent provocateur. London: Continuum.

Grenfell, M. & James. D. (eds.). 1998. *Bourdieu and education: acts of practical theory*. London: Falmer.

Guiffrida, D.A. 2006. Toward a cultural advancement of Tinto's theory. *Review of Higher Education*, 29(4):451-472, Summer.

Guiffrida, D.A., Lynch, M.F., Wall, A.F. & and Abel, D.S. 2013. Do reasons for attending college affect academic outcomes? A test of a motivational model from a self-determination theory perspective. *Journal of College Student Development*, 54(2):121-139, March–April.

Hancock, D.R. & Algozzine, B. 2006. *Doing case study research.* New York, NY: Teachers College Press.

Harris, B.A. 2006. The importance of creating a "sense of community." *Journal of College Student Retention: Research, Theory & Practice*, 8(1):83-105, August.

Harrison, A. 2002. Case study research. In Partington, D. (ed.). *Essential skills for management research*. London: Sage: 158-181.

Harvey, L. 2008. Analytical quality glossary. http://www.qualityresearchinternational.com/glossary/ [2 July 2012].

Harvey, L., Drew, S. & Smith, M. 2006. *The first-year experience: a review of literature for the Higher Education Academy*. York: Higher Education Academy.

Henning, E., Van Rensburg, W. & Smit, B. 2004. *Finding your way in qualitative research*. Pretoria: Van Schaik.

Hillier, J. & Rooksby, E. (eds). 2002. *Habitus: a sense of place*. Aldershot: Ashgate.

Hillman, K. 2005. The first year experience: the transition from secondary school to university and TAFE in Australia. Melbourne: ACER.

Holliday, A. 2007. *Doing and writing qualitative research.* 2nd ed. London: Sage.

Hooper, C. 2005. Afterword: The first year experience fifty years ago. In Krause, K.L., Hartley, R., James, R. & McInnis, C. (eds). *The first year experience in*

Australian universities: findings from a decade of national studies. Canberra: Australian Government Department of Education, Science and Training: 91-96.

Horstmanshof, L. & Zimitat, C. 2007. Future time orientation predicts academic engagement among first year university students. *British Journal of Educational Psychology*, 77(3):703-718, September.

Huysamen, G.K. 2000. The differential validity of matriculation and university performance as predictors of post-first-year performance. *South African Journal of Higher Education*, 14(1):146-151.

Imenda, S. 2014. Is there a conceptual difference between theoretical and conceptual frameworks? *Journal of Social Sciences*, 38(2):185-195.

Inman, P. & Pascarella, E.T. 1998. The impact of college residence on the development of critical thinking skills in college freshmen. *Journal of College Student Development*, 39(6):557-568, November–December.

Jabareen, Y. 2009. Building a conceptual framework: philosophy, definitions, and procedure. *International Journal of Qualitative Methods*, 8(4):49-62.

Jackson, P. 2008. Pierre Bourdieu, the 'cultural turn' and the practice of international history. *Review of International Studies*, 34(1):155-181, January.

Jaffee, D., Carle, A.C., Phillips, R. & Paltoo, L. 2008. Intended and unintended consequences of first-year learning communities: an initial investigation. *Journal of The First-Year Experience & Students in Transition*, 20(1):53-70.

James, D. 2011. Beginning with Bourdieu in educational research. <u>http://www.bera.ac.uk/resources/beginning-bourdieu-educational-research</u> [2 June 2012].

James, R. 2002a. Students' changing expectations of higher education and the consequences of mismatches with the reality. In Coaldrake, O.P. & Stedman, L. (eds). *Responding to student expectations.* Paris: OECD: 71-73.

James, R. 2002b. Socioeconomic background and higher education participation: an analysis of school students' aspirations and expectations. Canberra: Commonwealth of Australia Department of Education, Science and Training.

James, R., Baldwin, G. & McInnis C., 1999. *Which university? The factors influencing the choices of prospective undergraduates*. Canberra: Department of Education, Training and Youth Affairs.

Jansen, E. & Suhre, C. 2011. Preparedness, first-year experiences and outcomes. A comparison between students in domestic and international degree programmes in a Dutch university. In Krause, K.L., Buckridge, M., Grimmer, C. and Purbrick-Illek, S. (eds.). Research and Development in Higher Education: Reshaping Higher Education: Proceedings of the 34th HERDSA Annual Conference, Gold Coast, Queensland, Australia, 4–7 July. Milperra, NSW: HERDSA: 174-187.

Jansen, E.P.W.A. & Van der Meer, J. 2012. Ready for university? A cross-national study of students' perceived preparedness for university. *Australian Educational Researcher*, 39(1):1-16.

Jansen, J. 2012. National Senior Certificate results belie conceptual and skill limitations of school-leavers. *South African Journal of Science*, 108(3/4), Art. #1165, 2 pages. http://dx.doi.org/10.4102/sajs.v108i3/4.1165

Jarvis, P., Holford, J. & Griffin, C. 2004. *The theory and practice of learning.* 2nd ed. London: RoutledgeFalmer.

Jawitz, J. 2009. Learning in the academic workplace: the harmonization of the collective and the individual habitus. *Studies in Higher Education*, 34(6):601-614.

Jensen, U. 2011. Factors influencing student retention in higher education. Summary of influential factors in degree attainment and persistence to career or further education for at-risk/high educational need students. Honolulu, HI: Kamehameha Schools Research & Evaluation Division.

Johnson, G. & Watson, G. 2004. 'Oh Gawd, how am I going to fit into this?': Producing [mature] first-year student identity. *Language and Education*, 18(6):474-487.

Johnson, R.B. & Onwuegbuzie, A.J. 2004. Mixed methods research: a research paradigm whose time has come. *Educational Researchers*, 33(7):14-26, October.

Jones, L. (ed.). 2001. *Retaining African Americans in higher education: challenging paradigms for retaining students, faculty, & administrators.* Sterling, VA: Stylus.

Kahu, E. 2014. Increasing the emotional engagement of first year mature-aged distance students: interest and belonging. *International Journal of the First Year in Higher Education*, 5(2):45-55, August.

Kantanis, S.T. 2000. The role of program cohesion in effecting a smooth transition from secondary school to university. *Journal of Institutional Research*, 9(1):100-110.

Kerlinger, F.N. 1973. *Foundations of behavioral research*. 2nd ed. New York, NY: Holt, Rinehart and Winston.

Kift, S. & Nelson, K. 2005. Beyond curriculum reform: embedding the transition experience. In Brew, A. & Asmar, C. (eds). *Proceedings of the 28th HERDSA 2005 Conference: Higher Education in a Changing World, Sydney, Australia.* Milperra, NSW: 225-235.

Kift, S., Nelson, K. & Clarke, J. 2010. Transition pedagogy: a third generation approach to FYE – A case study of policy and practice for the higher education sector. *International Journal of the First Year in Higher Education*, 1(1):1-20, July.

Kim, E., Newton, F.B., Downey, R.G. & Benton, S.L. 2010. Personal factors impacting college student success: constructing college learning effectiveness inventory (CLEI). *College Student Journal*, 44(1):112-125, March.

Kingston, P.W. 2001. The unfulfilled promise of cultural capital theory. *Sociology of Education*, 74(Special Issue):88-99.

Kloot, B. 2009. Exploring the value of Bourdieu's framework in the context of institutional change. *Studies in Higher Education*, 34(4):469-481.

Kobasa, S.C. 1979. Stressful life events, personality and health: an inquiry into hardiness. *Journal of Personality and Social Psychology*, 37(1):1-11, January.

Kothari, C.R. 2006. *Research methodology: methods and techniques.* 2nd ed. New Delhi: New Age.

Krause, K.L. 2006. Student diversity in the first year: challenges and rewards. Keynote address delivered at the 4th Annual Conference on Teaching and Learning in Higher Education: The Challenge of Diversity – Teaching, Support and Student Learning, Galway 8–9 June 2006.

Krause, K.L., Hartley, R., James, R. & McInnis, C. (eds). 2005. *The first year experience in Australian universities: findings from a decade of national studies.* Canberra: Australian Government Department of Education, Science and Training.

Krueger, R.A. & Casey, M.A. 2000. *Focus groups: a practical guide for applied research*. 3rd ed. Thousand Oaks, CA: Sage.

Kuh, G.D. 2001. Assessing what really matters to student learning: inside the National Survey of Student Engagement. *Change*, 33(3):10-17, 66, May–June.

Kuh, G.D. & Love, P.G. 2000. A cultural perspective on student departure. In Braxton, J.M. (ed.). *Reworking the student departure puzzle: new theory and research on college student retention.* Nashville, TN: Vanderbilt University Press: 127-156.

Kuh, G.D., Cruce, T.M., Shoup, R., Kinzie, J. & Gonyea, R.M. 2008. Unmasking the effects of student engagement on first-year college grades and persistence. *Journal of Higher Education*, 79(5):540-563, September–October.

Kuh, G.D., Kinzie, J., Buckley, J.A., Bridges, B.K. & Hayek, J.C. 2007. Piecing together the student success puzzle: research, propositions and recommendations. San Francisco, CA: Jossey-Bass. ASHE Higher Education Report Series, Vol. 32, No. 5.

Kuh, G.D., Kinzie, J., Schuh, J.H. & Whitt, E.J. 2005. Assessing conditions to enhance educational effectiveness: the inventory for student engagement and success. San Francisco, CA: Jossey-Bass.

Kvale, S. 1996. InterViews – an introduction to qualitative research interviewing. Thousand Oaks, CA: Sage.

Lane, L. & Taber, N. 2012. Experiences with cultural capital in education: exploring the educational life stories of first-generation post-secondary students. *Canadian Journal for the Study of Adult Education*, 24(2):1-15, April.

Lareau A. & Weininger, E.B. 2003. Cultural capital in educational research: a critical assessment. *Theory and Society*, 32(5/6):567-606, December.

Lawrence, J. 2003. The 'deficit-discourse' shift: university teachers and their role in helping first year students persevere and succeed in the new university culture. Paper presented at the 6th Pacific Rim First Year in Higher Education Conference 2002: Changing Agendas, Christchurch, New Zealand, 8–10 July, 2002. http://ultibase.rmit.edu.au/Articles/march03/lawrence1.pdf [4 March 2011].

Leathwood, C. & O'Connell, B. 2003. 'It's a struggle': the construction of the 'new student' in higher education. *Journal of Education Policy*, 18(6):597-615, November–December.

Lehmann, W. 2007. 'I just didn't feel like I fit in': the role of habitus in university dropout decisions. *Canadian Journal of Higher Education*, 37(2):89-110.

Leibowitz, B., Van der Merwe, A &. Van Schalkwyk, S. (eds). 2012. *Focus on firstyear success: perspectives emerging from South Africa and beyond*. Revised ed. Stellenbosch: Sun Media.

Levy, S. & Earl, C. 2012. *Student voices in transition: the experiences of pathways students.* Pretoria: Van Schaik.

Lewis, J. & Ritchie, J. 2003. Generalising from qualitative research. In Ritchie, J. & Lewis, J. (eds). *Qualitative research practice: a guide for social science students and researchers*. London: Sage: 264-286.

Liamputtong, P. & Ezzy, D. 2005. Focus groups. In Liamputtong, P. & Ezzy, D. (eds). *Qualitative research methods*. 2nd ed. Melbourne: Oxford University Press: 75-99.

Liehr, P. & Smith, M.J. 1999. Middle range theory: spinning research and practice to create knowledge for the new millennium. *Advances in Nursing Science*, 21(4):81-91, June.

Light, R.J. 2001. *Making the most of college: students speak their minds.* Cambridge, MA: Harvard University Press.

Lincoln, Y.S. & Guba, E.C. 1985. Naturalistic inquiry. Beverly Hills, CA: Sage.

Longden, B. 2004, Interpreting student early departure from higher education through the lens of cultural capital. *Tertiary Education and Management*, 10(2):121-138, June.
Longwell-Grice, R. & Longwell-Grice, H. 2008. Testing Tinto: how do retention theories work for first-generation, working-class students? *Journal of College Student Retention: Research, Theory & Practice*, 9(4):407-420.

Lotkowski, V.A., Robbins, S.B. & Noeth, R.J. 2004. *The role of academic and non-academic factors in improving college retention.* ACT Policy Report. <u>www.atc.org/research/policy/index.htlm</u> [9 April 2015].

Mackenzie, N. & Knipe, S. 2006. Research dilemmas: paradigms, methods and methodology. *Issues In Educational Research*, 16(2):193-205.

Mackie, S. 1998. Jumping the hurdles. Paper presented at the Higher Education Close Up 1 Conference, Preston, 6–8 July 1998.

Malefo, V. 2000. Psycho-social factors and academic performance among African women students at a predominantly white university in South Africa. *South African Journal of Psychology*, 30(4):40-45.

Mansfield, P.M., Pinto, M.B., Parente, D.H. & Wortman, T.I. 2009. College students and academic performance: a case of taking control. *NASPA Journal*, 46(3):502-518.

Martin Jr., W.E., Swartz-Kulstad, J.L. & Madison, M. 1999. Psychosocial factors that predict the college adjustment of first-year undergraduate students: implications for college counselors. *Journal of College Counseling*, 2(2):121-133, Fall.

Martin, N.D. & Spenner, K.I. 2009. Capital conversion and accumulation: a social portrait of legacies at an elite university. *Research in Higher Education*, 50(7):623-648.

Mason, J. 2002. *Qualitative researching.* 2nd ed. London: Sage.

Maton, K. 2008. Habitus. In Grenfell, M. (ed.). *Pierre Bourdieu: key concepts*. Durham: Acumen: 49-66.

McInnis, C. 2001. Researching the first year experience: where to from here? *Higher Education Research & Development*, 20(2):105-114.

McInnis, C. 2003. New realities of the student experience: how should universities respond? Keynote address to the 25th Annual Forum of the European Association for Institutional Research, Limerick, Ireland, 24–27 August 2003.

McInnis, C. & James, R. 1995. *First year on campus: diversity in the initial experiences of Australian undergraduates*. Canberra: Australian Government Publishing Service.

McInnis, C., Hartley, R., Polesol, J. & Teese, R. 2000. *Non-completion in vocational education and training and higher education: a literature review commissioned by the Department of Education, Training and Youth Affairs.* Canberra: AusInfo.

McInnis, C., James, R. & Hartley, R. 2000. *Trends in the first year experience in Australian universities*. Canberra: Department of Education, Training and Youth Affairs.

McKenzie, K. & Schweitzer, R. 2001. Who succeeds at university? Factors predicting academic performance in first year Australian university students. *Higher Education Research & Development*, 20(1):21-33.

McKenzie, K., Gow, K. & Schweitzer, R. 2004. Exploring first-year academic achievement through structural equation modelling. *Higher Education Research & Development*, 23(1):95-112, February.

McMillan, J.H. & Schumacher, S. 2001. *Research in education: a conceptual introduction*. 5th ed. New York, NY: Longman.

Macnaghten, P. & Myers, G. 2004. Focus groups. In Seale, C., Gobo, G., Gubrium, J.F. & Silverman, D. (eds). *Qualitative research practice.* Thousand Oaks, CA: Sage: 65-79.

Melguizo, T. 2011. A review of the theories developed to describe the process of college persistence and attainment. In Smart, J.C. & Paulsen, M.B. (eds.). *Higher education: handbook of theory and research, Vol. 26.* Dordrecht: Springer Science+Business Media: 396-424.

Merriam, S.B. 1998. *Case study research in education: a qualitative approach*. 2nd ed. San Francisco, CA: Jossey-Bass.

Merriam, S.B. 2009. *Qualitative research: a guide to design and implementation*. San Francisco, CA: Jossey-Bass.

Mertens, D.M. 2005 Research and evaluation in education and psychology: integrating diversity with quantitative, qualitative and mixed methods. 2nd ed. Thousand Oaks, CA: Sage.

Mertens, D.M. 2010. Philosophy in mixed methods teaching: the transformative paradigm as illustration. *International Journal of Multiple Research Approaches*, 4(1): 9-18, April.

Metz, G.W. 2002. Challenges and changes to Tinto's Persistence Theory. Paper presented at the Annual Meeting of the Mid-Western Educational Research Association, Columbus, OH, 16–19 October 2002.

Miles, M.B. & Huberman, A.M. 1994. *Qualitative data analysis: a sourcebook of new methods*. Beverly Hills, CA: Sage.

Miller, J.V. 2013. A model of first-year student academic achievement at four-year universities. Unpublished Microsoft Word manuscript.

Mitrushina, M., Boone, K.B., Razani, J. & D'Elia, L.F. 2005. *Handbook of normative data for neuropsychological assessment.* 2nd ed. Oxford: Oxford University Press.

Molina Azorín, J. & Cameron, R. 2010. The application of mixed methods in organisational research: a literature review. *Electronic Journal of Business Research Methods*, 8(2):95-105.

Morgan, D.L. 1988. Focus groups as qualitative research. Newbury Park, CA: Sage.

Morgan, D.L. 2007. Paradigms lost and pragmatism regained: methodological implications of combining qualitative and quantitative methods. *Journal of Mixed Methods Research*, 1(1):48-76, January.

Morsi, R., Smith, P.C. & DeLoatch, S.J. 2007. Student success seminars: a school level freshman intervention program. In *Proceedings of 37th Annual Frontiers in Education Conference – Global Engineering: Knowledge Without Borders, Opportunities Without Passports, Milwaukee, WI, 10–13 October 2007*: F4A-11-F4A-15. doi:10.1109/FIE.2007.4418178

Mounts, N.S., Valentiner, D.P., Anderson, K.L. & Boswell, M.K. 2006. Shyness, sociability, and parental support for the college transition: relation to adolescents' adjustment. *Journal of Youth and Adolescence*, 35(1):68-77, February.

Mouton, J. 1996. Understanding social research. Pretoria: Van Schaik.

Mullendore, R.H. & Banahan, L.A. 2005. Designing orientation programs. In Upcraft, M.L., Gardner, J.N. & Barefoot, B.O. (eds). 2005. *Challenging and supporting the first-year student: a handbook for improving the first year of college*. San Francisco, CA: Jossey-Bass: 391-409.

NAO (National Audit Office). 2007. *Staying the course: the retention of students in higher education*. London: Stationery Office.

Nash, R. 1990. Bourdieu on education and social and cultural reproduction. *British Journal of Sociology of Education*, 11(4):431-447.

Nash, R. 1999. Bourdieu, 'habitus' and educational research: is it all worth the candle? *British Journal of Sociology of Education*, 20(2):176-187, June.

Nash, R. 2002. A realist framework for the sociology of education: thinking with Bourdieu. *Educational Philosophy and Theory*, 34(3):273-288, August.

Nelson, K.J., Clarke, J.A., Kift, S.M. & Creagh, T.A. 2011. Trends in policies, programs and practices in the Australasian first year experience literature 2000–2010. Brisbane: Queensland University of Technology.

Nes, L.S., Evans, D.R. & Segerstrom, S.C. 2009. Optimism and college retention: mediation by motivation, performance, and adjustment. *Journal of Applied Psychology*, 39(8):1887-1912, August.

Nora, A. & Cabrera, A.F. 1996. The role of perceptions of prejudice and discrimination on the adjustment of minority students to college. *Journal of Higher Education*, 67(2):119-148, March–April.

Nora, A., Cabrera, A., Hagedorn, L.S. & Pascarella, E. 1996. Differential impacts of academic and social experiences on college-related behavioral outcomes across different ethnic and gender groups at four-year institutions. *Research in Higher Education*, 37(4):427-451, August.

Noyes, J. 2007–2008. Book review of *First-generation entry into higher education*, by Liz Thomas and Josey Quinn. *Learning and Teaching in Higher Education*, (3):113-116.

NPC (National Planning Commission). 2012. *National Development Planning 2030: Our future – make it work.* Pretoria: National Planning Commission.

Nurmuliani, N., Zowghi, D. & Williams, S.P. 2004. Using card sorting technique to classify requirements change. *Proceedings of the 12th IEEE International Requirements Engineering Conference, Kyoto, Japan, 6–11 September 2004.* Kyoto: IEEE: 240-248.

Nutt, D. & Calderon, D. (eds). 2009. *International perspectives on the first-year experience in higher education.* Columbia, SC: National Resource Center for the First-Year Experience & Students in Transition, University of South Carolina.

Nzimande, B.E. 2014a. Budget speech, National Assembly, 2014. Pretoria: DHET.

Nzimande, B.E. 2014b. Keynote address. Seventh Annual International South African Technology Network (SATN) Conference, Umhlali, KwaZulu-Natal, 14–16 October 2014.

OECD (Organisation for Economic Co-operation and Development). 2008. *Reviews of national policies for education: South Africa 2008.* Paris: OECD Publications.

Orsuwan, M. & Cole, D. 2007. The moderating effects of race/ethnicity on the experience of Asian American and Pacific Islander community college students. *Asian American Policy Review*,16:61-85.

Österåker, M. 2001. To put your cards on the table –collection of data through silent interviews. *Management Decision*, 39(7) 578-582.

Ovink, S.M. & Veazey, B.D. 2010. More than 'getting us through:' a case study in cultural capital enrichment of underrepresented minority undergraduates. *Research in Higher Education*, 52(4):370-394, June.

Pargetter, R. 2000, Transition: from a school perspective. *Journal of Institutional Research*, 9(1):1-6.

Pascarella, E.T. & Terenzini, P.T. 1978. Student–faculty informal relationships and freshman year educational outcomes. *Journal of Educational Research*, 71(4):183-189.

Pascarella, E.T. & Terenzini, P.T. 1979. Student–faculty informal contact and college persistence: a further investigation. *Journal of Educational Research*, 72(4):214-218.

Pascarella, E.T. & Terenzini, P.T. 1980. Predicting freshman persistence and voluntary dropout decisions from a theoretical model. *Journal of Higher Education*, 51:60-75, January–February.

Pascarella, E.T. & Terenzini, P.T. 1983. Predicting voluntary freshman year persistence/withdrawal behavior in a residential university: a path analytic validation of Tinto's model. *Journal of Educational Psychology*, 75(2):215-226, April.

Pascarella, E.T. & Terenzini, P.T. 1991. *How college affects students: findings and insights from twenty years of research*. San Francisco, CA: Jossey-Bass.

Pascarella, E.T. & Terenzini, P.T. 2005. *How college affects students: a third decade of research.* San Francisco, CA: Jossey-Bass.

Pascarella, E.T., Pierson, C.T., Wolniak, G.C. & Terenzini, P.T. 2004. First generation college students: additional evidence on college experiences and outcomes. *Journal of Higher Education*, 75(3):249-284, May–June.

Pather, S. 2013. Social and academic integration of first-year at-risk students in a mathematics intervention programme. *Proceedings of the Canadian International Conference on Education (CICE), University of Toronto, Canada, 27 – 29 June 2013.* Toronto: CICE: 68-73.

Patton, M.Q. 1987. *How to use qualitative methods in evaluation*. Newbury Park, CA: Sage.

Patton, M.Q. 2002. *Qualitative research and evaluation methods*. 3rd ed. Thousand Oaks, CA: Sage.

Paulsen, B.M. & St. John, P.E. 2002. Social class and college costs: examining the financial nexus between college choice and persistence. *Journal of Higher Education*, 73(2):189-236, March–April.

Perna, L.W. & Titus, M. 2005. The relationship between parental involvement as social capital and college enrollment: an examination of racial/ethnic group differences. *Journal of Higher Education*, 76(5):485-518, September–October.

Perry, C. & Allard, A. 2003. Making the connections: transition experiences for firstyear education students. *Journal of Educational Enquiry*, 4(2): 74–89.

Pitsoe, V.J. 2013. Teacher attrition in South Africa: trends, challenges and prospects. *Journal of Social Sciences*, 36(3):309-318.

Pitzalis, M., Sulis, I. & Porcu, M. 2008. *Differences of Cultural Capital among Students in Transition to University. Some First Survey Evidences.* Working Paper

CRENoS 2008-05. Cagliari: Centre for North South Economic Research, University of Cagliari and Sassari, Sardinia.

Polit, D.F. & Beck, C.T. 2004. *Nursing research: principles and methods*. 7th ed. Philadelphia, PA: Lippincott Williams & Wilkins.

Pym, J.M., Goodman, S.L. & Patsika, N. 2011. Does belonging matter? Exploring the role of social connectedness as a critical factor in students' transition to higher education. *Pins: Psychology in Society*, 42:35-50.

Reason, R.D. 2009. An examination of persistence research through the lens of a comprehensive conceptual framework. *Journal of College Student Development*, 50 (6):659-682, November–December.

Reason, R.D., Terenzini, P.T. & Domingo, R.J. 2006. First things first: developing academic competence in the first year of college. *Research in Higher Education*, 47(2):149-175, March.

Reay, D. 1995. 'They employ cleaners to do that': habitus in the primary classroom. *British Journal of Sociology of Education*, 16(3):353-371.

Reay, D. 2004. It's all becoming a habitus': beyond the habitual use of habitus. *British Journal of Sociology of Education*, 25(4):431-444, September.

Reay, D. 2010. From the theory of practice to the practice of theory: working with Bourdieu in research in higher education choice. In Silva, E.B. & Warde, A. (eds). *Cultural analysis and Bourdieu's legacy: settling accounts and developing alternatives*. London: Routledge: 75-86.

Reay, D. 2012. Researching class in higher education. <u>www.bera.ac.uk/.../Researching%20Class%20in%20Higher%20Educucation.pdf</u> [14 July 2012].

Reay, D., Crozier, G. & Clayton, J. 2010. 'Fitting in' or 'standing out': working-class students in UK higher education. *British Educational Research Journal*, 36(1):107-124, February.

Reay, D., Hollingworth, S., Williams, K., Crozier, G., Jamieson, F., James, D. & Beedell, P. 2007. A darker shade of pale? Whiteness, the middle classes and multiethnic inner city schooling. *Sociology*, 41(6):1041-1060, December.

Rendón, L.I., Jalomo, R.E. & Nora, A. 2000. Theoretical considerations in the study of minority retention. In Braxton, J.M. (ed.). *Reworking the student departure puzzle: new theory and research on college student retention.* Nashville, TN: Vanderbilt University Press: 127-156.

Rezgui, Y. & Miles, J. 2010. Exploring the potential of SME alliances in the construction sector. *Journal of Construction Engineering and Management*, 136(5): 558-567, May.

Rhodes, C.P. & Nevill, A. 2004. Academic and social integration in higher education: a survey of satisfaction and dissatisfaction within a first year education studies cohort at a new university. *Journal of Further and Higher Education*, 28(2):179-193, May.

Rugg, G. & McGeorge, P. 2005. The sorting techniques: a tutorial paper on card sorts, picture sorts and item sorts. *Expert Systems*, 22(3):94-107, July.

Russell, M. 2008. Leveraging student engagement with assessments: collecting intelligence to support teaching, student progress and retention. In Crosling, G.M., Thomas, L. & Heagney, M. (eds). *Improving student retention in higher education: the role of teaching and learning*. London: Routledge: 108-116.

Rubin, M. 2012. Social class differences in social integration among students in higher education: a meta-analysis and recommendations for future research. *Journal of Diversity in Higher Education*, 5(1):22-38, March.

Ryan, G.W. & Bernard, H.R. 2003. Techniques to identify themes. *Field Methods*, 15(1):85-109, February.

Sahu, P.K. 2013. *Research methodology: a guide for researchers in agricultural science, social science and other related fields.* New Delhi: Springer.

Samodien, L. 2008. Teacher exodus. *IOL News*, August 8, 2008 at 12:05pm. <u>http://www.iol.co.za/news/south-africa/teacher-exodus-1.411686#.U4hE9XKSyoM</u> [4 May 2012].

Saunders, M.N.K. & Thornhill, A. 2004. Trust and mistrust in organizations: an exploration using an organizational justice framework. *European Journal of Work and Organizational Psychology*, 13:493-515.

Sayer, R.A. 2005. *The moral significance of class.* Cambridge: Cambridge University Press.

Scanlon, L., Rowling, L. & Weber, Z., 2007. 'You don't have like an identity ... you are just lost in a crowd': forming a student identity in the first-year transition to university. *Journal of Youth Studies*, 10(2):223-241, May.

Schmid, L.C. 2001. Educational achievement, language-minority students, and the new second generation. *Sociology of Education*, 74 (special issue):71-87.

Schneider, M. 2010. *Finishing the first lap: the cost of first-year student attrition in America's four-year colleges and universities*. Washington, DC: American Institutes for Research.

Scott, I. 2012. First-year experience as terrain of failure or platform for development? Critical choices for higher education. In Leibowitz, B., Van der Merwe, A &. Van Schalkwyk, S. (eds). *Focus on first-year success: perspectives emerging from South Africa and beyond*. Revised ed. Stellenbosch: Sun Media: 17-35.

Scott, I., Yeld, N. & Hendry J. 2007. *A case for improving teaching and learning in South African higher education*. Pretoria: Council on Higher Education.

Sedlacek, W.E. 2004. Why we should use noncognitive variables with graduate and professional students. *The Advisor: The Journal of the National Association of Advisors for the Health Professions*, 24(2):32-39.

Sedlacek, W.E. 2011. Using noncognitive variables in assessing readiness for higher education. *Readings on Equal Education*, 25:187-205.

Seekings, J. 2003. Social stratification and inequality in South Africa at the end of apartheid. Centre for Social Science Research, University of Cape Town.

Shields, P.M. 2003. A pragmatic teaching philosophy. *Journal of Public Affairs Education*, 9(1):7-12.

Shisana, O., Peltzer, K., Zungu-Dirwayi, N. & Louw, J. (eds). 2005. *The health of our educators : a focus on HIV/AIDS in South African public schools*. Cape Town: HSRC Press.

Shupp, M.R. 2005. New student orientation and a book discussion group: strategies and justifications. <u>http://www.newfoundations.com/PolHeader.html</u> [3 June 2011].

Silverman, D. 2011. *Qualitative research: issues of theory, method and practice.* 3rd ed. London: Sage.

Slonimsky, L. & Shalem, Y. 2005. Pedagogic responsiveness for academic depth. In Griesel, H. (ed.). *Curriculum responsiveness: case studies in higher education*. Pretoria: South African Universities Vice-Chancellors Association: 81-101.

Smit, R. 2012. Towards a clearer understanding of student disadvantage in higher education: problematising deficit thinking. *Higher Education Research & Development*, 31(3):369-380, June.

Smith, L. 2011. Towards a transition pedagogy: a case study of a regional Australian university's approach to enhancing the first year experience. Paper presented at the 14th Pacific Rim First Year in Higher Education (FYHE) Conference: Design for Student Success, Fremantle, Australia, 28 June–1 July 2011. http://www.fyhe.com.au/past_papers/papers11/FYHE-2011/content/pdf/15D.pdf [27 February 2012].

Smyth, R. 2004. Exploring the usefulness of a conceptual framework as a research tool: a researcher's reflections. *Issues in Educational Research*, 14(2):167-180.

South Africa. 1996. *Constitution of the Republic of South Africa, Act No. 108 of 1996.* Pretoria: Government Printer.

Spady, W.G. 1970. Dropouts from higher education: an interdisciplinary review and synthesis. *Interchange*, 1(1):64-85, April.

Stage, F.K. & Anya, G. 1996. A transformational view of college student research. In Stage, F.K., Anya, G., Bean, J., Hossler, D. & Kuh, G. (eds). *College students: the evolving nature of research.* Boston, MA: Simon & Schuster: 12-22.

Stage, F.K. & Hossler, D. 2000. Where is the student? Linking student behaviors, college choice, and college persistence. In Braxton, J.M. (ed.). *Reworking the student departure puzzle: new theory and research on college student retention.* Nashville, TN: Vanderbilt University Press: 170-194.

Stage, F.K. & Rushin, P.W. 1993. A combined model of student predisposition to college and persistence in college. *Journal of College Student Development*, 34(4): 276-282, July.

Stake, R.E. 1995. The art of case study research. Thousand Oaks, CA: Sage.

Statistics South Africa. 2012. Census 2011 statistical release – P0301.4. Pretoria: Statistics South Africa. <u>http://www.statssa.gov.za/publications/P03014/P030142011.pdf</u> [10 April 2015].

Steenkamp, L.P., Baard, R.S. & Frick, B.L. 2009. Factors influencing success in firstyear accounting at a South African university: a comparison between lecturers' assumptions and students' perceptions. *South African Journal of Accounting Research*, 23(1):113-140.

Steyn, H.J. 2006. The numbers (game) in teacher education. Presentation at the Deans' Forum, Deans of Faculties of Education in South Africa.

Struwig, F.W. & Stead, G.B. 2001. *Planning, designing and reporting research.* Cape Town: Pearson Education South Africa.

Stuart, M. 2006. 'My friends made all the difference': getting into and succeeding at university for first-generation entrants. *Journal of Access Policy and Practice*, 3(2): 162-184, August.

Sullivan, A. 2001. Cultural capital and educational attainment. *Sociology*, 35(4):893-912, November.

Sullivan, A. 2002. Bourdieu and education: how useful is Bourdieu's theory for researchers? *Netherlands' Journal of Social Sciences*, 38(2):144-166.

Swail, W.S. 2004. *The art of student retention: a handbook for practioners and administrators*. Austin, TX: Educational Policy Institute.

Swartz, D. 1997. *Culture & power: the sociology of Pierre Bourdieu*. Chicago, IL: University of Chicago Press.

Taniguchi, H. & Kaufman, G. 2005. Degree completion among nontraditional college students. *Social Science Quarterly*, 86(4):912-927, December.

Tashakkori, A. & Teddlie, C. 1998. *Mixed methodology: combining qualitative and quantitative approaches.* Thousand Oaks, CA: Sage.

Tashakkori, A. & Teddlie, C. (eds). 2003. *Handbook of mixed methods in social & behavioral research.* Thousand Oaks, CA: Sage.

Tashakkori, A. & Teddlie, C. 2010. *Sage handbook of mixed methods in social & behavioral research.* 2nd ed. Los Angeles, CA: Sage.

Taylor, J. & Bedford, T. 2004. Staff perceptions of factors related to non-completion in higher education. *Studies in Higher Education*, 29(3):375-394.

Terenzini, P.T. 1993. The transition to college: diverse students, diverse stories. Paper presented at the 46th Annual Forum of the Association for Institutional Research (AIR), Chicago, IL, May 1993.

Terenzini, P.T. & Pascarella, E.T. 1978. The relation of students' pre-college characteristics and freshman year experience to voluntary attrition. *Research in Higher Education*, 9(4):347-366.

Terenzini, P.T. & Reason, R.D. 2005. Parsing the first year of college: rethinking the effects of college on students. Paper presented at the Annual Meeting of the Association for the Study of Higher Education (ASHE), Philadelphia, PA, 17–19 November. http://www.ed.psu.edu/educ/parsing-project/.pdf%20documents/ASHE05ptt.pdf [12 April 2012].

Terenzini, P.T., Cabrera, A. & Bernal, E.M. 2001. *Swimming against the tide: the poor in American higher education.* New York, NY: College Entrance Examination Board.

Terenzini, P.T., Rendon, L.I., Upcraft, M. L., Millar, S.B., Allison, K.W., Gregg, P.L. & Jalomo, R. 1994. The transition to college: diverse students, diverse stories. *Research in Higher Education*, 35(1):57-73.

Thomas, L. 2002. Student retention in higher education: the role of institutional habitus. *Journal of Educational Policy*, 17(4):423-442.

Thomas, L. 2012. Building student engagement and belonging in higher education at a time of change. Final Report from the What Works? Student Retention & Success programme.

https://www.heacademy.ac.uk/sites/default/files/What_works_final_report.pdf [10 April 2015].

Thomas, L. & Yorke, M. 2003. Improving the retention of students from lower socioeconomic groups. *Journal of Higher Education Policy and Management*, 25(1):63-74, May.

Tierney, W.G. 1992. An anthropological analysis of student participation in college. *Journal of Higher Education*, 63(6):603-618, November–December.

Tierney, W.G. 2000. Power, identity and the dilemma of college student departure. In Braxton, J.M. (ed.). *Reworking the student departure puzzle: new theory and research on college student retention.* Nashville, TN: Vanderbilt University Press: 213-234.

Tieu, T.T. & Pancer, S.M. 2009. Cocurricular involvement and first-year students' transition to university: quality vs. quantity of involvement. *Journal of the First-Year Experience & Students in Transition*, 21(1):43-63, January.

Tinto, V. 1975. Dropout from higher education: a theoretical synthesis of recent research. *Review of Educational Research*, 45(1):89-125, Winter.

Tinto, V. 1982. Limits of theory and practice in student attrition. *Journal of Higher Education*, 53(6):687-700, November–December.

Tinto, V. 1988. Stages of student departure: reflections on the longitudinal character of student leaving. *Journal of Higher Education*, 59(4):438-455, July.

Tinto, V. 1993. *Leaving college: rethinking the causes and cures of student attrition*. 2nd ed. Chicago, IL: University of Chicago Press.

Tinto, V. 1995. Educational communities and student success in the first year of university. Paper presented at the Monash University Conference on the Transition from Secondary School to University, Melbourne, 29 November–1 December 1995.

Tinto, V. 2007. Research and practice of student retention: what's next? *Journal of College Student Retention: Research, Theory & Practice*, 8(1):1-19.

Toman, N. 2010. Diverse students? Diverse identities? Can conceptions of habitus and identity facilitate an understanding of learner experience within contemporary higher education? Paper presented at the Academic Identities for the 21st Centurty, Conference, Glasgow, 16–18 June 2010.

Tones, M.J., Fraser, J., Elder, R.L. & White, K.M. 2009. Supporting mature-aged students from a low socioeconomic background. *Higher Education*, 58(4):505-529, October.

Toni, M.N. & Olivier, M.A.J. 2004. Academic identities of black female first-year students. *South African Journal of Education*, 24(3):194-199.

Travers, M.H. 2006. Qualitative interviewing methods. In Walter, M. (ed.). Social Research methods: an Australian perspective. Melbourne: Oxford University Press: 83-112.

Trow, M. 2006. Reflections on the transition from elite to mass to universal access: forms and phases of higher education in modern societies since WWII. In Forest, J. & Altbach, P.J.J.F. (eds). *International handbook of higher education.* Dordrecht: Springer: 243-280.

Upcraft, M.L., Gardner, J.N. & Barefoot, B.O. (eds). *Challenging and supporting the first-year student: a handbook for improving the first year of college*. San Francisco, CA: Jossey-Bass.

Van den Berg, M.N. & Hofman, W.H.A. 2005. Student success in university education: a multi-measurement study into the impact of student and faculty factors on study progress. *Higher Education*, 50(3):413-446, October.

Van der Meer, J., Scott, S. & Neha, T. 2010. Retention of first-year Māori students at university. *Mai Review* 2(1): 2-14.

Van Gennep, A. 1960. The rites of passage. London: Routledge & Kegan Paul.

Van Schalkwyk, S., Leibowitz, B. & Van der Merwe, A. 2012. Introduction: Perspectives on the first-year experience. In Leibowitz, B., Van der Merwe, A &. Van Schalkwyk, S. (eds). *Focus on first-year success: perspectives emerging from South Africa and beyond*. Revised ed. Stellenbosch: Sun Media: 3-13.

Wacquant, L.J.D. 1989. Towards a reflexive sociology: a workshop with Pierre Bourdieu. *Sociological Theory*, 7(1):26-63, Spring.

Wacquant, L.J.D. 1998. Pierre Bourdieu. In Stones, R. (ed.) Key sociological thinkers. Basingstoke: Palgrave Macmillan: 215-229.

Wacquant, L.J.D. 2005. (ed.). *Pierre Bourdieu and democratic politics: the mystery of ministry.* Cambridge: Polity Press.

Wagner, A. 2000. Students at the centre of the education paradigm: changing patterns of demand. In *OECD Thematic Review of the First Years of Tertiary Education : Australia Seminar, Sydney, 20–21 April 1998.* Canberra: Higher Education Division, Department of Education, Training and Youth Affairs: 57-69.

Wahr, F., Gray, K. & Radloff, A. 2009. Improving student transition by working with academics' conceptions of the student experience: academic development for organisational change. In Wozniak, H. & Bartoluzzi, S. (eds). *The Student Experience: Proceedings of the 32nd HERDSA Annual Conference, Darwin, Australia, 6–9 July 2009.* Milperra, NSW: HERDSA: 434-443.

Walpole, M. 2007. *Economically and educationally challenged students in higher education: access to outcomes.* San Francisco, CA: Jossey-Bass.

Warde, A. 2004. Practice and field: revising Bourdieusian concepts. Centre for Research on Innovation and Competition (CRIC), University of Manchester.

Watson, J., Nind, M., Humphris, D. & Borthwick, A. 2009. Strange new world: applying a Bourdieuian lens to understanding early student experiences in higher education. *British Journal of Sociology of Education*, 30(6):665-681.

Watson, J.C. & Flamez, B. 2014. *Counseling assessment and evaluation: fundamentals of applied practice*. Thousand Oaks, CA: Sage.

Webb, J., Schirato, T. & Danaher, G.R. 2002. *Understanding Bourdieu*. Crows Nest, NSW: Allen & Unwin.

Weidman, J. 1989. Undergraduate socialization: a conceptual approach. In Smart, J.C. (ed.). *Higher education: handbook of theory and research, Vol. 5.* New York, NY: Agathon Press: 289-322.

Welch, T. & Gultig, J. 2002. *Teacher education: looking in the mirror to plan the future*. 10th anniversay edition. Johannesburg: South African Institute for Distance Education.

Wellington, J.J. 2000. *Educational research: contemporary issues and practical approaches.* London: Continuum.

Wells, R. 2009. Social and cultural capital, race and ethnicity, and college student retention. *Journal of College Student Retention: Research, Theory & Practice*, 10(2): 103-128.

Wetzels, R., Matzke, D., Lee, M.D., Rouder, J.N., Iverson, G.J. & Wagenmakers, E-J. 2011. Statistical evidence in experimental psychology: an empirical comparison using 855 t tests. *Perspectives on Psychological Science*, 6(3):291-298.

Whaley, A.L. & Longoria, R.A. 2009. Preparing card sort data for multidimensional scaling analysis in social psychological research: a methodological approach. *Journal of Social Psychology*, 149(1):105-115, February.

Wheeldon, J. & Åhlberg, M.K. 2012.Visualizing social science research: maps, methods and meaning. Thousand Oaks, CA: Sage.

Wildhagen, T. 2010. Capitalizing on culture: how cultural capital shapes educational experiences and outcomes. *Sociology Compass*, 4(7):519-531, July.

Wiersma, W. & Jurs, S.G. 2005. *Research methods in education: an introduction*. 8th ed. Boston, MA: Pearson Allyn & Bacon.

Wilcox, P., Winn, S. & Fyvie-Gauld, M. 2005. 'It was nothing to do with the university, it was just the people': the role of social support in the first-year experience of higher education. *Studies in Higher Education*, 30(6):707-722.

Wolhuter, C.C. 2006. Teacher training in South Africa: past, present and future. *Education Research and Perspectives*, 33(2):124-139.

Wolhuter, C.C. 2010. South Africa: worldwide educational reform programme telescoped into an instant time space. In Wolhuter, C.C & Herman, H.D. (eds). *Educational reform in Southern Africa: prospects for the new millennium*. Potchefstroom: CC Wolhuter: 6-7.

Wolhuter, C.C., Van der Walt, H., Potgieter, F., Meyer, L. & Mamiala, T. 2012. What inspires South African student teachers for their future profession? *South African Journal of Education*, 32(2):178-190.

Woollacott, L. & Henning, L. 2004. Dealing with under-preparedness in engineering entrants: a perspective from Wits. *CREE for Engineering Educators*, 8(1):3-8.

Woosley, S.A. & Shepler, D.K. 2011. Understanding the early integration experiences of first-generation college students. *College Student Journal*, 45(4): 700-714, December.

Wyatt, L.G. 2011. Nontraditional student engagement: increasing adult student success and retention. *Journal of Continuing Higher Education*, 59(1):10-20.

Yanto, H., Mula, J.M. & Kavanagh, M.H. 2011. Developing student's [*sic*] accounting competencies using Astin's I-E-O model: an identification of key educational inputs based on Indonesian student perspectives. In *Proceedings of 2nd RMIT Accounting Educators' Conference, 14 Nov 2011, Melbourne, Australia.* <u>http://eprints.usq.edu.au/20077/2/Yanto_Mula_Kavanagh_RMIT_2011_PV.pdf</u> [3 November 2013].

Yazedjian, A., Purswell, K.E., Sevin, T. & Toews, M.L. 2007. Adjusting to the first year of college: students' perceptions of the importance of parental, peer, and institutional support. *Journal of the First-Year Experience & Students in Transition*, 19(2):29-46.

Yeld, N. 2013. Time to stop blaming schools. *Mail and Guardian,* 4 October: 5.

Yin, R.K. 1984. Case study research: design and methods. Beverly Hills, CA: Sage.

Yin, R.K. 1994. *Case study research: design and methods*. 2nd ed. Thousand Oaks, CA: Sage.

Yin, R.K. 2003. *Case study research: design and methods.* 3rd ed. Thousand Oaks, CA: Sage.

Yin, R.K. 2009. *Case study research: design and methods*. 4th ed. Los Angeles, CA: Sage.

Yin, R.K. 2011. *Qualitative research from start to finish*. New York, NY: Guilford Press.

Yorke, M. 1999. Leaving early: undergraduate non-completion in higher education. London: Falmer Press.

Yorke, M. 2000. The quality of the student experience: what can institutions learn from data relating to non-completion. *Quality in Higher Education*, 6(1):61-75.

Yorke, M. & Longden, B. 2004. Retention and student success in higher education. Maidenhead: Society for Research into Higher Education; Open University Press.

Yorke, M. & Longden, B. 2008. *The first-year experience of higher education in the UK: final report.* York: Higher Education Academy.

Zainal, Z. 2007. Case study as a research method. Jurnal Kemanusiaan, 9:1-6, June.

Zimitat, C. 2003. The impact of full-time employment and family commitments on the first year experience of full-time students. Paper presented at the 7th Annual Pacific Rim First Year in Higher Education Conference, Brisbane, Australia, 9–11 July. 10 pp. <u>http://www.griffith.edu.au/ data/assets/pdf_file/0020/52328/zim_nzare2004.pdf</u>. [10 November 2013].

APPENDIX A: Pre-entry questionnaire



FACULTY OF EDUCATION

MOWBRAY CAMPUS

P O Box 652, CAPE TOWN, 8000

Pre-University Student Characteristics and Expectation Questionnaire

Dear BEd Students

Congratulations and welcome to the Faculty of Education at Cape Peninsula University of Technology (CPUT), Mowbray Campus. This questionnaire forms part of a DEd project.

You are requested to please complete the attached questionnaire (N.B. the completion of the questionnaire is optional). Once completed you can return the completed questionnaire in the self-addressed envelope provided in your acceptance pack. Your acceptance letter could also be returned in the same envelope.

This survey contributes towards a doctoral research project that aims to assist the Faculty to understand who their first-year students are and thus to improve the first- year experience of students entering the BEd programme at CPUT.

Please take note that your responses are <u>confidential</u> and will only be used to produce summarised results that will be used in my doctoral study. The questions below are divided into three areas:

- A. Student background information
- B. Pre-university academic background
- C. Expectations of first year at university

Thank you very much for your time and contribution to my research project titled: An investigation of teacher education students' pre-entry factors influencing first-year experience and academic performance.

We look forward to welcoming you to CPUT.

Yours faithfully

Sue Pather Academic Development Lecturer pathersu@cput.ac.za

Pre-University Student Characteristics and Expectation Questionnaire

Instructions:

- Please respond to all questions honestly and to the best of your knowledge.
- The responses require your opinion or perception regarding various issues. Therefore there are **no right or wrong** responses.
- Write your responses in the space provided.
- In cases where options are provided, indicate the appropriate response with a cross (X).
- Please answer <u>all</u> questions.

A. Background Information

1.	Name & Surname:
2.	Age:
3.	Gender: Male Female
4.	Race:
	African Coloured Indian White
5.	What is your home language?

6.	Did your	parents	complete	a degree?
----	----------	---------	----------	-----------

6.1	Father
	Yes No
6.2	Mother
	Yes No
7.	What is/was your parents' occupation?
7.1	Father
7.2	Mother
8.	Are you the first in your immediate family to attend university?
	Yes No
9.	How will you fund your university studies?
	Parents Yourself Loan Bursary
	Other: State
10.	In which residential area will you be staying while studying?
11.	How will you travel to campus?
	Bus Taxi Train Walk Private Vehicle
	Other: State
В.	Pre-University Academic Background
12.	What is your present occupation?
	Student Working Unemployed If you are currently working, describe your occupation:

13. Which BEd Programme will you be registering for?

Further Education & Training (FET): Grade 10–12	
General Education & Training (GET): Foundation Phase, Grade R to 3	
General Education & Training (GET): Intermediate & Senior Phase	

- 14. In what year did you complete or will complete Grade 12? _____
- 15. Name of school/institution where you completed your Grade 12 or are presently studying:
- 15.1 Where is the school/institution located?

Province: _____

Residential Area:	
-------------------	--

16. Subjects completed at Grade 12 level and symbols (if presently in school indicate your June 2011 examination subjects and results).

Subjects	Symbols

17. Do you have any other qualifications?

Yes	No	

If yes, state them:

18. Do you feel that you are prepared for university life?

	Yes No
	Write down at least two reasons for your answer above.
19.	Have you used a computer?
	Yes No
	If your answer is No to question 19, explain why you have not?

If your answer is <u>Yes</u> to question 19, answer the following questions:

19.1 How often do you use the computer?

Every day	
3 – 4 times a week	
Once a week	
Once a month	
A few times a year	

19.2 Where do you use the computer?

19.3 What do you use the computer for? You may mark an "X" next to more than one option.

Facebook	Browsing the internet	
Playing games	Career choices/jobs	
Email	Download music/movies	
Blog	Using Office applications e.g. Word, Spreadsheets	
Other: State		

19.4 List your **top 3** uses of the computer from the options in 19.3 above.

_

C. Career Choice

20.	Was the BEd degree your first choice of study?	Yes	No
20.1	If no, to the above question, what was your first choice of study?		
21.	Did your family support your choice of study?	Yes	No
21.1	How did or didn't they support you? Explain.		
22.	Did your friends support your choice of study?	Yes	No
22.1	How did or didn't they support you? Explain.		

23. Did you seek career advice before choosing this course?	
---	--

23.1 If yes, from whom? Explai	23.1	If yes,	from	whom?	Explair
--------------------------------	------	---------	------	-------	---------

24.	Did you consider choosing a different university before settling for this institution?	Yes	No
24.1	If yes, what was your choice of institution?		
25.	How did you hear about the teaching degree?		
26.	Do you have family in the teaching profession?	Yes	No
20.	Do you have family in the teaching profession?	165	NO
	Explain relationship:		

27. Are you aware that all your subjects will be taught in English at the Yes No Mowbray campus?

D. Expectation of first-year university experience

The following statements describe a number of possible expectations which you may have of your first-year experience in the BEd course.

Indicate your response with a cross (x). You may choose only one option per statement.

Think about 2012, and your first year of study in the BEd programme at CPUT.

With regard to your expectations of your first-year experience:

INDICATE THE EXTENT TO WHICH YOU AGREE OR DISAGREE WITH THE FOLLOWING STATEMENTS

		Strongly Agree	Agree	Disagree	Strongly disagree
28.	I expect to join organisations/clubs on campus, e.g. sports club, student union, etc.				
29.	I will make many new friends.				
30.	The course of study will not be difficult.				
31.	Librarians will be important people to help me with finding information for my assignments and projects.				
32.	I will make friends from different racial groups.				
33.	My lecturer will expect me to write well-structured academic essays.				
34.	I will attend many social functions, e.g. sports day, student braai, fund-raising events, etc. in my first year.				
35.	I will be in contact with campus friends and peers using one or more of the following social networks: Facebook, Email, SMS, Mxit.				
36.	Going out to schools for teaching practice will be easy and I do not expect to encounter any difficulties.				
37.	I will not be involved in academic discussions with my peers outside of formal lectures.				
38.	Joining social clubs/organisations at university will distract me from academic work.				

		Strongly Agree	Agree	Disagree	Strongly disagree
39.	I expect that group work will be an important feature of lectures.				
40.	My lecturers will expect me to know how to correctly reference my assignments and projects.				
41.	I expect to be in contact with my lecturers using any of the following social networks : Facebook, Email, SMS, Mxit.				
42.	The library is not a place where I expect to spend a lot of time.				
43.	My first-year university work load will be easier than school.				
44.	I expect to spend a lot of time preparing for lectures.				
45.	My lecturers will not expect me to attend every single lecture.				
46.	It will be more difficult to make friends at university than it was at school or work.				
47.	My lecturer will expect me to take part in class discussions and debates.				
48.	During my first year of studies there will only be a few assignments and projects.				
49.	Lecturers will not know me and other students by their first name.				
50.	It will be a challenge for me to participate in group work during class lessons.				
51.	Lecturers will make themselves available outside of the formal lecture time to assist and advise me.				
52.	I am looking forward to contributing to lively classroom discussions and debates.				
53.	I will seek academic support from my friends whenever I need help with my academic work.				
54.	Making friends from different racial groups will be difficult for me.				
55.	I am aware of academic support services, e.g. writing centre, peer tutors and tutorials that are available on campus to support students that need help.				

				0	
		Strongly Agree	Agree	Disagree	Strongly disagree
56.	I will seek academic help from senior students assigned as peer tutors in the various subjects if I have difficulties with my courses.				
57.	I am shy and will not personally introduce myself to my lecturers outside of class.				
58.	I expect the lecturers to prepare/equip me before I go to schools for my first teaching practice session.				
59.	I am shy and will not contribute to class discussions and debates.				
60.	Having conversations with my class peers outside the classroom is easy for me.				
61.	I expect my first-year courses to be difficult.				
62.	During my first year of studies there will only be a few assignments and projects.				
63.	Having a conversation with my lecturer outside of formal class will be easy for me.				
64.	I expect that my lecturers will provide regular feedback to me on class work, assignments and tests.				
65.	My friends can easily persuade me to bunk lectures.				
66.	My lecturers will expect me to be able to use the computer for assignments and projects.				
67.	I expect to write fewer formal tests in my first year of study compared with school.				

Thank you for taking the time to complete this survey.

Sue Pather & Rajendra Chetty

Place this survey together with your acceptance letter in the self-addressed envelope and return to the university.

If there are any other expectations that you have about your first-year experience at CPUT Mowbray campus that are not covered in this questionnaire, please state them:

If you would like to be a participant in this research prodetails below and I will be in contact with you once you next year.	
Name:	
Cell number:	
Email address:	
Thanks	
Sue Pather	
pathersu@cput.ac.za	

First-Year Experience: Student Interview Schedule

Research Question: How do pre-entry academic and non-academic factors influence TE students' first-year educational experience and academic performance?

1. FYE: Perception	What are you enjoying about your first-year experience?What is the worst thing about being a first-year student?	
2. Academic Integration	 What is the worst thing about being a hist-year student? What are your reasons for studying this course (goal)? Are you enjoying your studies (intrinsic goal)? Do you think it is worth studying (extrinsic goal)? What difficulties are you experiencing with the course (method)? How does the language of teaching and learning (English) impact on your learning (method)? How do you think you are performing in your courses? Do you get regular feedback? Have you encountered any academic challenges? How did you cope with these? 	
3. Social Integration	 Do you feel comfortable being a student at this campus? Do you think it is important to know your lecturer and peers (goal)? Have you made contact with your lecturers outside of class time/how often? Do you have a lot of friends on campus (intrinsic goal)? Do you think being in contact with lecturers and peers is important (extrinsic goal)? How do you feel about collaborative/group work? Have you encountered any challenges with students? How did you cope with these? 	
4. First-Year Expectations	 Is the campus life what you expected it to be? Are the students on campus what you expected them to be like? What was that expectation? Is the course you are studying what you expected to encounter? How did you cope? 	
5. Academic Background	How did you enjoy your high school experience?What was your academic performance like?	
6. Family Background	 Tell me something about your family and home life? Are you the first in your family to go to university? What are your parents' occupations? What kind of support do you receive from your family? 	

		What do your family think about your studying?	
What challenges have you and your family encountered with your studying?			
7.	Support Structure	 How do you think the following people have supported you? Immediate family Extended family Friends off campus Collaboration/group work with peers New campus friends Academic Development: tutorials/workshops – student services Lecturers Administration staff: library, admin office, bursary 	
8.	Sort-Card Activity	 Students were given a pack of 10 cards with titles and pictures. The cards were in no particular order. Students had to study the cards and then arrange them according to preference of the most influential factor contributing towards their academic studies to the least influential. Students were given a chance to arrange the cards and talk about their experiences. 	

Consent Form for DEd Research

Participant Consent Form

DEd Research Project – Pre-entry academic and non-academic factors influencing teacher education students' first-year academic experience and academic performance.

I, ______ have read an explanation of the purpose of the research project and have volunteered to be a participant in the above research project. I am aware that the researcher (Mrs Sue Pather) will be conducting two interviews with me. I am also aware of and agree to the interview's being recorded. Finally, I have been assured that my interview and reflective journal in which I will be recording all my first-year academic experiences will be used for research purposes only. My confidentiality will be respected also. I hereby give my permission for any information I provide during my interviews and in my journal to be used in the DEd research study.

Signature of participant

Date

Signature of researcher

Date

APPENDIX D: Ethical clearance

Department Name: General Education & Training

Research Ethical Clearance

Compliance Form

All postgraduate students and researchers are required to complete this form before commencing with research or submitting applications to the **University Research Fund (URF).** All URF applicants are requested to please submit this form together with all the relevant URF application forms to their Faculty Research Committee (FRC) for approval.

(Where applicable mark relevant boxes with an X)

Project Title: DEd Research

Factors influencing BEd students' first-year experience and academic performance.

Applica	Supervisor (if applicable):	
Surname	Pather	Chetty
First Name	Subethra	Rajendra
Title	Mrs	Prof
Office Telephone	021 680 1586	021 680 1532
Cell	084 6898584	
Email	pathersu@cput.ac.za	chettyr@cput.ac.za
Undergraduate		
Postgraduate		
Staff	Х	

Rese	earch Checklist:	Yes	No
1.	Does the study involve participants who are unable to give informed consent? Examples include children, people with learning disabilities, or your own students. Animals?		Х
2.	Will the study require the co-operation of a gatekeeper for initial access to the groups or individuals to be recruited? Examples include students at school, members of self-help groups, residents of nursing homes — anyone who is under the legal care of another.		х
3.	Will it be necessary for participants to participate in the study without their knowledge and consent at the time — for example, covert observation of people in non-public places?		Х
4.	Will the study with the research subject involve discussion of sensitive topics? Examples would include questions on sexual activity or drug use.		Х
5.	Will the study involve invasive, intrusive, or potentially harmful procedures of any kind (e.g. drugs, placebos or other substances to be administered to the study participants)?		Х
6.	Will the study involve prolonged or repetitive testing on sentient subjects?		Х
7.	Will financial inducements (other than reasonable expenses and compensation for time) be offered to participants?		х
8.	Does your research involve environmental studies which could be contentious or use materials or processes that could damage the environment? Particularly the outcome of your research?		Х

If you have answered '*No*' to <u>all</u> questions, submit the completed and signed form to your FRC together with the completed URF application forms.

If you have answered 'Yes'...

If you have answered '**Yes**' to <u>one or more</u> questions, kindly attach a report describing more fully how you plan to deal with the ethical issues raised by your research. It does not mean that you cannot do the research, only that your proposal will need to be approved by the Research Ethics Committee. You will need to submit your plans for addressing the ethical issues raised by your proposal to your faculty's Research Ethics Committee.

Declaration

As Researcher/Applicant I acknowledge that it is my responsibility to follow Ethical Standards and good research practice.

I declare that I am not aware of any potential conflicts of interest, other than those declared on THIS form, which may influence the ethical conduct of this study.

Signatures:

Researcher/Applicant	Date	
Supervisor/Senior (if applicable)	Date	
Faculty Research Ethics Committee of	comments:	
Approved	Referred back	
Chairperson	Date	











APPENDIX F: Focus-group interview



Slide 1



Slide 2

Introduction Activity

Introduce yourself to the class.

- What is your name?
- Where do you come from?
- Do you have a part-time job?
- What is your hobby (you do in your spare time)?
- What do you consider to be your strengths with regard to being a university students?





Slide 4

Activity 3: Mind map



In your group discuss what qualities you have brought with you to university/higher education.

- Once you have all had a chance to reflect and discuss, you need to draw a mind map with the groups qualities.
- Your mind map should have in the middle: Qualities I bring with me to HE.
- Arrows moving outwards should have the following topics: experience, education, skills, & personal characteristics/personality.

Slide 5



Slide 6





Do you have any closing remarks, comments or suggestions before we conclude

Thank you for participating in this focus group

interview Danki, Siyabonga



Slide 8

APPENDIX G: Job category grouping

The eight major job groups are:

- 1. Managers
- 2. Professionals
- 3. Technicians and Associate Professionals
- 4. Clerical Support Workers
- 5. Service and Sales Workers
- 6. Skilled Agricultural, Forestry, Fishery, Craft and Related Trades Workers
- 7. Plant and Machine Operators and Assemblers
- 8. Elementary Occupations

Figure 1 below provides an indicative comparison between the skill level of occupational groups with the National Qualifications Framework as well as with the entry, intermediate and advanced levels referred to in the National Skills Development Strategy.

A mapping between ISCO skill levels and levels of education in ISCED 97 is provided in Table 1 below.

ISCO – 08 Skill Level	ISCED-97 groups				
4	 6 – Second stage of tertiary education (leading to an advanced research qualification) 5a – First stage of tertiary edicatopm. 1st degree (medium duration) 				
3	5b –First stage of tertiary education (short or medium duration)				
2	 4 – Post-secondary, non-tertiary education 3 – Upper secondary level of education 2 – Lower secondary level of education 				
1	1 – Primary level of education				

NSDA	NQF LevelsKILL Level	OFO Major Groups				
HIGH	10 ↑ 4 2 Profession 7			1 Managers		
ENTRY INTERMEDIATE	6	3	3 Technicians and Associate Profesionals			
	5		4 Clerical Support Workers	5 Service and Sales Workes	6 Skilled Agricultural, Forestry, Fishery, Craft & Related Trades Workers	7 Plant and Machine Operators and Assemblers
	4	2				
	3					
	2	1		{	Dccupation	
	1	•		Elementary		